UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

DATE OF REPORT (DATE OF EARLIEST EVENT REPORTED)
July 1, 2008 (June 25, 2008)

MOBILE MINI, INC.

(Exact name of registrant as specified in its charter)

Delaware1-1280486-0748362(State or other jurisdiction of incorporation)(Commission File Number)(IRS Employer Identification No.)7420 South Kyrene Road, Suite 101, Tempe, Arizona85283

dui Kyrene Koad, Suite 101, Tempe, Arizon

(Zip Code)

(Address of principal executive offices)

Registrant's telephone number, including area code: **(480) 894-6311**Former name or former address, if changed since last report: **Not Applicable**

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

TABLE OF CONTENTS

<u>Item 1.01 Entry into a Material Definitive Agreement.</u>

<u>Item 1.02 Termination of a Material Definitive Agreement.</u>

<u>Item 2.01 Completion of Acquisition or Disposition of Assets.</u>

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

Item 3.02 Unregistered Sales of Equity Securities.

<u>Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers;</u>

Compensatory Arrangements of Certain Officers.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

Item 8.01 Other Information.

<u>Item 9.01. Financial Statements and Exhibits.</u>

SIGNATURES

EXHIBIT INDEX

EX-3.1

EX-3.2

EX-4.1

EX-4.3

EX-10.1

EX-10.2

EX-10.3

EX-99.1

Item 1.01 Entry into a Material Definitive Agreement.

Mobile Mini, Inc. ("Mobile Mini") entered into five agreements relating to Mobile Mini's consummation of its merger with MSG WC Holdings Corp and two of its subsidiaries (collectively, "Mobile Storage Group"). These five agreements are: the Escrow Agreement, the Stockholders Agreement, the Credit Agreement, the MSG Supplemental Indenture and the Mobile Mini Supplemental Indenture (each as defined below).

Escrow Agreement

On June 27, 2008, Mobile Mini, Welsh, Carson, Anderson & Stowe X, L.P. ("WCAS") and Wells Fargo Bank, N.A. entered into an Escrow Agreement (the "Escrow Agreement"), pursuant to the terms of which, the Mobile Storage Group stockholders placed into escrow \$15,000,000 of the merger consideration in cash and shares of newly issued Mobile Mini convertible redeemable participating preferred stock (the "Preferred Stock"), valued at \$18.00 per share, to satisfy any post-closing adjustments of the merger consideration and to secure their indemnification obligations under the Agreement and Plan of Merger, dated February 22, 2008, by and among Mobile Mini, Cactus Merger Sub, Inc., a wholly-owned subsidiary of Mobile Mini, MSG WC Holdings Corp., the indirect parent of Mobile Storage Group, Inc. and Mobile Services Group, Inc., and WCAS, as representative of the stockholders of MSG WC Holdings Corp. (the "Merger Agreement"). At all times while such shares of Preferred Stock are held in escrow, the stockholders of Mobile Storage Group have the right to (i) exercise any voting rights with respect to the escrowed shares of Preferred Stock and (ii) receive all products and proceeds of any of the escrowed shares of Preferred Stock, including all dividends, whether in the form of cash, stock or any other form, and any other rights and other property which the stockholders of Mobile Storage Group are, from time to time, entitled to receive in respect of, or in exchange for, any or all of the escrowed shares.

The foregoing description of the Escrow Agreement is qualified in its entirety by the Escrow Agreement, a copy of which is attached as Exhibit 10.1 hereto and is incorporated by reference herein.

Stockholders Agreement

On June 27, 2008, WCAS, WCAS Capital Partners IV., L.P. and WCAS Management Corporation entered into a Stockholders Agreement (the "Stockholders Agreement"), which primarily governs the shares of Preferred Stock issued to the stockholders of Mobile Storage Group as partial consideration in the merger with Mobile Storage Group. The Stockholders Agreement principally covers: Transfer of Equity Securities, Registration Rights and the Board of Directors of Mobile Mini, as described below.

Transfer of Equity Securities

Under the terms of the Stockholders Agreement, Mobile Storage Group stockholders party to the stockholders agreement are restricted from transferring, voluntarily or involuntarily, directly or indirectly, in any manner, any equity or debt securities of Mobile Mini, in whole or in part, or any other right or interest therein, or entering into any transaction which results in the economic equivalent of a transfer to any person except pursuant to a permitted transfer (i) following such stockholder's death to a permitted transferee, (ii) to its affiliates subject to certain restrictions and (iii) on or after June 27, 2009, provided, that private sales of common stock of Mobile Mini or sales of preferred stock to any single transferee shall not exceed 3% of the fully diluted common stock of Mobile Mini subject to certain restrictions. In addition, in the event of such permitted transfers, the aggregate number of permitted transferees in connection with transfers by any single stockholder party to the stockholders agreement is limited and subject to certain restrictions.

Subject to certain exceptions, the Stockholders Agreement provides that, at all times during which WCAS has the right to nominate a director, WCAS or any of its controlled affiliates are restricted from selling any securities other than during any period when the directors and officers of WCAS and its subsidiaries are not prohibited from selling securities pursuant to the written policies and procedures of Mobile Mini governing transfers of securities by such officers and directors.

Subject to certain exceptions, the Stockholders Agreement also provides that, until the date on which WCAS no longer hold equity securities constituting 5% or more of the outstanding shares of common stock of Mobile Mini, WCAS will be restricted from directly or indirectly (i) acquiring securities of Mobile Mini, (ii) participating in any merger or other extraordinary transaction involving Mobile Mini, (iii) soliciting proxies with respect to Mobile Mini, (iv) forming a "group" with respect to Mobile Mini, (v) participating in any financing for the purchase of any securities or assets of Mobile Mini that are not equity and debt securities, (vi) acting to seek

control or direct the Board of Directors, stockholders, policies or affairs of Mobile Mini, (vii) taking any action intended to require Mobile Mini to make a public announcement of the foregoing, (viii) making public announcements with respect to any of the foregoing and (ix) proposing to Mobile Mini that it amend or waive any term of the foregoing standstill provisions.

Notwithstanding the foregoing, in the event WCAS has not acquired up to 2.0 million shares of Mobile Mini common stock under the waiver of standstill provision of the Merger Agreement, WCAS is permitted to purchase up to 2.0 million shares of Mobile Mini common stock.

Registration Rights

Shelf Registration Statement. The Stockholders Agreement provides that Mobile Mini must use all commercially reasonable efforts to file a shelf registration statement under the U.S. Securities Act of 1933, as amended (the "Securities Act"), on or about April 27, 2009 covering all of the shares of Mobile Mini common stock issuable upon conversion of the Preferred Stock and shares of Mobile Mini common stock acquired pursuant to the Merger Agreement (the "registrable securities") then held by the Mobile Storage Group stockholders party to the Stockholders Agreement on Form S-3 to enable the resale of such registrable securities after June 27, 2009.

Required Registrations. The Stockholders Agreement also provides that at any time after the date, if any, that (x) Mobile Mini is not permitted to file or maintain a Form S-3 in connection with the shelf registration in accordance with the terms of the Stockholders Agreement, or (y) the shelf registration expired in accordance with the terms of the Stockholders Agreement and not all registrable securities registered in such shelf registration have been sold, the holders of registrable securities representing at least a majority of the outstanding registrable securities have the right to request Mobile Mini to effect a registration under the Securities Act of registrable securities held by such stockholders. Mobile Mini shall not be required to comply with more than 1 such request during any 6 month period and shall only be obligated to comply with 4 such requests in total.

Incidental Registration. If, at any time after June 27, 2009, Mobile Mini proposes to register any of its securities under the Securities Act for sale to the public, any Mobile Storage Group stockholder party to the Stockholders Agreement has the right at each such time to include registrable securities held by it that are not otherwise covered by the shelf registration statement or a required registration statement in such registration statement.

The registration rights granted in the Stockholders Agreement are subject to customary restrictions such as blackout periods and limitations on the number of shares to be included in any underwritten offering imposed by the managing underwriter. In addition, the Stockholders Agreement contains other limitations on the timing and ability of the holders of registrable securities to exercise demands.

Board of Directors of Mobile Mini

As long as WCAS continues to hold, in the aggregate, at least 2,000,000 shares of Preferred Stock or common stock issued upon conversion or exchange of the Preferred Stock, WCAS has the right to designate a director whose term ends in 2011, a director whose term ends in 2009, and an observer from and after January 1, 2010 on the Board of Directors of Mobile Mini subject to the limitations set forth on the Stockholders Agreement.

The foregoing description of the Stockholders Agreement is qualified in its entirety by the Stockholders Agreement, a copy of which is attached as Exhibit 10.2 hereto and is incorporated by reference herein.

Credit Agreement

On June 27, 2008, Mobile Mini and certain of its subsidiaries entered into an ABL Credit Agreement (the "Credit Agreement") with Deutsche Bank AG New York Branch ("Deutsche Bank") and the other lenders party thereto, with Deutsche Bank as administrative agent. The Credit Agreement provides for a five-year, \$900 million first lien senior secured revolving credit facility, a portion of which is for borrowing in US Dollars (the "US Subfacility) and a portion of which is available for borrowing in British Pounds Sterling and Euros (the "UK Subfacility"). The US Subfacility is subject, among other things, to the terms of a borrowing base calculated as a

discount to the value of certain pledged United States and Canadian collateral; the UK Subfacility is subject to a similar borrowing base which includes pledged UK collateral. Both borrowing bases are subject to certain reserves and caps customary for financings of this type. If at any time the aggregate amounts outstanding under either the US Subfacility or UK Subfacility exceed the respective borrowing base then in effect, a prepayment of an amount sufficient to eliminate such excess is required to be made. The terms of the Credit Agreement provide that Mobile Mini is required to prepay loans with net cash proceeds from sales of debt or equity securities or asset sales, including casualty proceeds. Mobile Mini has the right to prepay loans under the Credit Agreement in whole or in part at any time. All amounts borrowed under the Credit Agreement must be repaid on or before June 27, 2013. The Credit Agreement also provides for the issuance of irrevocable standby letters of credit by domestic lenders in amounts totaling up to \$50 million, and by UK-based lenders in amounts totaling up to \$10 million.

Loans made under the US Subfacility will bear interest at a rate equal to, at Mobile Mini's option, either (a) the London Interbank Operating Rate ("LIBOR") plus an applicable margin ("LIBOR Loans") or (b) the prime rate plus an applicable margin ("Base Rate Loans"). With some exceptions, Mobile Mini may freely convert LIBOR Loans to Base Rate Loans and vice versa. Loans made under the UK Subfacility denominated in Pounds Sterling will bear interest at a rate equal to the LIBOR plus an applicable margin; loans denominated in Euros will bear interest at a rate equal to the Euro Inter-Bank Offered Rate ("EURIBOR") plus an applicable margin. The initial applicable margin will be 1.00% with respect to Base Rate Loans and 2.50% with respect to LIBOR Loans, and for loans under the UK Subfacility the initial applicable margin will be 2.50%. The applicable margins will be readjusted quarterly based upon Mobile Mini's leverage ratio.

In addition to paying interest on outstanding principal under the Credit Agreement, Mobile Mini will be required to pay an unused line fee to the lenders under the revolving credit facilities in respect of the unutilized commitments thereunder. The unused line fee rate is .375% per annum if the unused amount is less than or equal to 50% of the aggregate commitments or .25% per annum if the unused amount is greater than 50%. Mobile Mini will also pay customary letter of credit fees.

Ongoing extensions of credit under the Credit Agreement are subject to customary conditions, including sufficient availability under the borrowing bases. The Credit Agreement also contains covenants that require Mobile Mini to, among other things, periodically furnish financial and other information to the various lenders. The Credit Agreement contains customary negative covenants applicable to Mobile Mini and its subsidiaries, including negative covenants that restrict the ability of such entities to, among other things, (i) make capital expenditures in excess of defined limits, (ii) allow certain liens to attach to Mobile Mini or subsidiary assets, (iii) repurchase or pay dividends or make certain other restricted payments on capital stock and certain other securities, or prepay certain indebtedness, (iv) incur additional indebtedness or engage in certain other types of financing transactions, and (v) make acquisitions or other investments.

The US Subfacility is guaranteed by Mobile Mini and all of its US subsidiaries, including its new subsidiaries as a result of the merger with Mobile Storage Group. Mobile Mini, its US subsidiaries, and its UK subsidiaries are guarantors of the UK Subfacility. The US Subfacility will be secured by a first priority lien on substantially all assets of the US borrowers and US guarantors. The UK Subfacility will be secured by a first priority lien on substantially all of the assets of Mobile Mini and its US and UK subsidiaries.

The Credit Agreement also includes other covenants, representations, warranties, indemnities, and events of default that are customary for facilities of this type, including events of default relating to a change of control of Mobile Mini.

The foregoing description of the Credit Agreement is qualified in its entirety by the Credit Agreement, a copy of which is attached as Exhibit 10.3 hereto and is incorporated by reference herein.

MSG Supplemental Indenture

In connection with the Merger (as defined below), Mobile Mini entered into a Supplemental Indenture, dated as of June 27, 2008 (the "MSG Supplemental Indenture"), with Mobile Mini of Ohio LLC, a Delaware limited liability company, Mobile Mini, LLC, a California limited liability company, Mobile Mini, LLC, a Delaware limited liability company, Mobile Mini I, Inc., an Arizona corporation, A Royal Wolf Portable Storage, Inc., a

California corporation, Temporary Mobile Storage, Inc., a California corporation, Delivery Design Systems, Inc., an Arizona corporation, Mobile Mini Texas Limited Partnership, LLP, a Texas limited liability partnership (collectively, the "New MSG Guarantors"), A Better Mobile Storage Company, a California corporation, and Mobile Storage Group (Texas), LP, a Texas limited partnership (the "Existing MSG Guarantors"), Mobile Storage Group, Inc., a Delaware corporation, and Wells Fargo Bank, N.A., as trustee ("Wells Fargo"), pursuant to which Mobile Mini became an "Issuer" for all purposes under the MSG Indenture. Mobile Storage Group, Inc. and Mobile Services Group, Inc., a Delaware corporation (the "Original Issuers"), the Existing MSG Guarantors and Wells Fargo previously entered into an Indenture (the "MSG Indenture"), dated as of August 1, 2006, pursuant to which the Original Issuers issued \$200,000,000 in aggregate principal amount of 9³/4% Senior Notes due 2014 (the "MSG Notes"). A copy of the MSG Indenture is attached as Exhibit 4.2 hereto and is incorporated by reference herein, and a copy of the MSG Supplemental Indenture is attached as Exhibit 4.1 hereto and is incorporated by reference herein. The MSG Indenture includes covenants, indemnities and events of default that are customary for indentures of this type, including restrictions on the incurrence of additional debt, sales of assets and payment of dividends. The foregoing description of the MSG Indenture is qualified in its entirety by the MSG Indenture.

Mobile Mini Supplemental Indenture

In connection with the Merger, Mobile Mini entered into a Supplemental Indenture, dated as of June 27, 2008 (the "Mobile Mini Supplemental Indenture"), with Mobile Storage Group, Inc., a Delaware corporation, A Better Mobile Storage Company, a California corporation, and Mobile Storage Group (Texas), LP, a Texas limited partnership (the "New Mobile Mini Guarantors"), the guarantors (the "Existing Mobile Mini Guarantors") party to the Mobile Mini Indenture (as defined below) and Law Debenture Trust Company of New York, as trustee ("LDTC"), pursuant to which the New Mobile Mini Guarantors became "Guarantors" for all purposes under the Mobile Mini Indenture. Mobile Mini, the Existing Mobile Mini Guarantors and LDTC previously entered into an Indenture (the "Mobile Mini Indenture"), dated as of May 7, 2007, pursuant to which Mobile Mini issued \$150,000,000 in aggregate principal amount of its 67/8% Senior Notes due 2015 (the "Mobile Mini Notes"). A copy of the Mobile Mini Supplemental Indenture is attached as Exhibit 4.3 hereto and is incorporated by reference herein.

Item 1.02 Termination of a Material Definitive Agreement.

In connection with entering the Credit Agreement, on June 27, 2008, Mobile Mini repaid in full all obligations and liabilities owing under, and terminated, its \$425 million credit facility under that certain Second Amended and Restated Loan and Security Agreement, dated as of February 17, 2006, as amended as of May 7, 2007, with Mobile Mini UK Limited and Deutsche Bank AG, New York Branch, as Agent for the lending group (the "**Prior Mobile Mini Facility**").

Item 2.01 Completion of Acquisition or Disposition of Assets.

On June 27, 2008, a wholly-owned subsidiary of Mobile Mini merged with and into MSG WC Holdings Corp. (the "Merger"), and immediately thereafter, each of MSG WC Holdings Corp. and two of its subsidiaries merged with and into Mobile Mini. Mobile Mini acquired all outstanding shares of Mobile Storage Group. Upon closing, Mobile Mini assumed Mobile Storage Group's outstanding indebtedness and paid cash totaling approximately \$562 million and issued 8,555,556 shares of Preferred Stock with a liquidation preference of \$154.0 million, subject to certain post-closing adjustments. On June 27, 2008, Mobile Mini issued a press release announcing, among other things, the completion of the Merger, a copy of which is attached as Exhibit 99.1 hereto and is incorporated by reference herein.

The Merger was effected pursuant to the Merger Agreement Mobile Mini stockholders approved the Merger at a special meeting of stockholders in Phoenix, Arizona on June 26, 2008.

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

On June 27, 2008, Mobile Mini entered into the Credit Agreement. The description of the Credit Agreement set forth under Item 1.01 is hereby incorporated by reference into this Item 2.03 as if fully set forth

herein.

Pursuant to the Merger Agreement, on June 27, 2008, Mobile Mini assumed Mobile Storage Group's outstanding indebtedness and paid cash totaling approximately \$562 million. The assumed debt includes the \$200.0 million in aggregate principal amount of MSG Notes issued pursuant to the MSG Indenture, which remain outstanding. The MSG Indenture includes covenants, indemnities and events of default that are customary for indentures of this type, including restrictions on the incurrence of additional debt, sales of assets and payment of dividends. Mobile Mini used a portion of the proceeds from an initial borrowing under the Credit Agreement to refinance its borrowings under the Prior Mobile Mini Facility and the borrowings of Mobile Storage Group under its prior revolving credit facility, which was also terminated on June 27, 2008.

Item 3.02 Unregistered Sales of Equity Securities.

On June 27, 2008, Mobile Mini completed the sale of 8,555,556 shares of Mobile Mini Preferred Stock to Mobile Storage Group's stockholders, in a transaction exempt from registration pursuant to Section 4(2) of the Securities Act. The shares formed a portion of the consideration used to acquire Mobile Storage Group. Mobile Mini assumed Mobile Storage Group's outstanding indebtedness and paid cash totaling approximately \$562 million and issued 8,555,556 shares of Mobile Mini Preferred Stock to acquire all of the outstanding shares of capital stock of Mobile Storage Group.

The Preferred Stock has a liquidation preference of \$154 million and will vote together with common stock as a single class. It will rank senior to the common stock only with respect to a distribution upon the occurrence of the bankruptcy, liquidation, dissolution or winding up of Mobile Mini. Mobile Mini will redeem all of the outstanding Preferred Stock if (i) Mobile Mini enters into a binding agreement in respect of a sale of Mobile Mini or (ii) at any time after the tenth anniversary of the closing date, the holders of a majority of the then outstanding shares of the Preferred Stock being issued to WCAS exercise their right at such time to require Mobile Mini to redeem such Preferred Stock then outstanding. The Preferred Stock will be initially convertible into 8,555,556 shares of Mobile Mini's common stock, representing a conversion price of \$18.00 per Mobile Mini share. The Preferred Stock will be mandatorily convertible into Mobile Mini common stock if, after the first anniversary of the issuance of the Preferred Stock, Mobile Mini's common stock trades above \$23.00 per share for a period of 30 consecutive days. The Preferred Stock will not have any cash or payment-in-kind dividends (unless and until a dividend is paid with respect to the common stock, in which case dividends will be paid on an equal basis with the common stock, on an as-converted basis) and will not impose any covenants upon Mobile Mini.

On June 26, 2008, Mobile Mini filed such certificate of designation for the Preferred Stock with the Secretary of State of the State of Delaware. The foregoing description of the certificate of designation for the Preferred Stock is qualified in its entirety by the certificate of designation for the Preferred Stock, a copy of which is attached as Exhibit 3.1 hereto and is incorporated by reference herein.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

(b) Resignation of Director.

On June 25, 2008, Mr. Ronald J. Marusiak resigned as a director of Mobile Mini effective June 25, 2008. Mr. Marusiak had also served Mobile Mini as a member of the Nominating and Corporate Governance Committee, the Audit Committee and the Compensation Committee. There were no disagreements between Mobile Mini and Mr. Marusiak that led to his resignation. On June 27, 2008, Mobile Mini issued a press release announcing, among other things, the resignation of Mr. Marusiak as a director of Mobile Mini, a copy of which is attached as Exhibit 99.1 hereto and is incorporated by reference herein.

(d) Appointment of Directors.

On June 27, 2008, Mr. Sanjay Swani and Mr. Michael E. Donovan were appointed, and on June 25, 2008, Mr. Frederick G. McNamee was appointed to the Board of Directors with terms expiring in 2011, 2009 and 2010, respectively. Messrs. Swani and

Donovan were appointed, effective upon the consummation of the mergers under the Merger Agreement on June 27, 2008 and pursuant to the right of WCAS under the Stockholders Agreement to designate two directors to the Board of Directors of Mobile Mini upon consummation of the merger of Mobile Mini with Mobile Storage Group. Mr. McNamee was appointed to fill the vacancy on the Board of Directors of Mobile Mini created by Mr. Marusiak's resignation, and was appointed to serve on the Audit Committee, the Compensation Committee, and the Nominating and Corporate Governance Committee. Other Board committee assignments have not been determined as of the date hereof. On June 27, 2008, Mobile Mini issued a press release announcing, among other things, the appointment of Messrs. Swani, Donovan and McNamee as directors of Mobile Mini, a copy of which is attached as Exhibit 99.1 hereto and is incorporated by reference herein.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

At the special meeting of stockholders held on June 26, 2008, Mobile Mini's stockholders approved two amendments to our Amended and Restated Certificate of Incorporation (the "Certificate of Incorporation"). Consequently, the Board of Directors of Mobile Mini amended Article IV of the Certificate of Incorporation in its entirety to (i) increase the number of authorized shares of preferred stock, par value \$0.01 per share, from 5,000,000 shares to 20,000,000 shares and (ii) authorize the designation of a series of preferred stock as Series A Convertible Redeemable Participating Preferred Stock. These amendments were effective on June 26, 2008. As a result, we filed the amendments with the Secretary of State of the State of Delaware on June 26, 2008.

A copy of the certificate of amendment of the Certificate of Incorporation is attached as Exhibit 3.2 hereto and is incorporated by reference herein.

The information set forth in Item 3.02 above is incorporated by reference as if fully set forth herein.

Item 8.01 Other Information.

On June 27, 2008, Mobile Mini issued a press release announcing the completion of the Merger and the appointment of Messrs. Swani, Donovan and McNamee as directors of Mobile Mini. This press release is attached as Exhibit 99.1 hereto and is incorporated by reference herein.

Item 9.01. Financial Statements and Exhibits.

(a) Financial Statements of Business Acquired.

Not applicable. The financial statements of MSG WC Holdings Corp. required pursuant to Rule 3-05 of Regulation S-X were previously filed with Mobile Mini's definitive proxy statement, filed with the SEC on June 4, 2008 (the "**Proxy Statement**") and pursuant to General Instruction B.3 of Form 8-K are not required to be reported herein.

(b) Pro Forma Financial Information.

Not applicable. The pro forma financial statements required by Article 11 of Regulation S-X were previously filed under "Unaudited Pro Forma Condensed Combined Financial Statements" included in the Proxy Statement and pursuant to General Instruction B.3 of Form 8-K are not required to be reported herein.

- (d) Exhibits.
- 3.1 Certificate of Designation of Mobile Mini Series A Convertible Redeemable Participating Preferred Stock
- 3.2 Certificate of Amendment of the Amended and Restated Certificate of Incorporation of Mobile Mini
- 4.1 Supplemental Indenture, dated as of June 27, 2008, among Mobile Mini, Inc., Mobile Mini of Ohio LLC, Mobile Mini, LLC, Mobile Mini, LLC, Mobile Mini I, Inc., A Royal Wolf Portable Storage, Inc., Temporary Mobile Storage, Inc., Delivery Design Systems, Inc., Mobile Mini Texas Limited Partnership, LLP, Mobile Storage Group, Inc., the guarantors

10.3

- party to the MSG Indenture and Wells Fargo Bank, N.A., as trustee. 4.2 Indenture, dated as of August 1, 2006, by and among Mobile Services Group, Inc., Mobile Storage Group, Inc., the subsidiary guarantors named therein and Wells Fargo Bank, N.A., as trustee (incorporated by reference to Exhibit 4.1 to the Mobile Storage Group, Inc.'s Form S-4 filed on September 18, 2007). 4.3 Supplemental Indenture, dated as of June 27, 2008, among Mobile Mini, Inc., Mobile Storage Group, Inc., A Better Mobile Storage Company, Mobile Storage Group (Texas), LP, the guarantors party to the Mobile Mini Indenture and Law Debenture Trust Company of New York, as trustee. 10.1 Escrow Agreement dated as of June 27, 2008, between Mobile Mini, Welsh, Carson, Anderson & Stowe X, L.P. and Wells Fargo Bank, N.A. Stockholders Agreement, dated as of June 27, 2008, between Mobile Mini and the certain stockholders. 10.2 Credit Agreement, dated June 27, 2008, between Mobile Mini, Deutsche Bank AG New York Branch and the other lenders.
- 99.1 Press Release of Mobile Mini dated June 27, 2008.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this Report to be signed on its behalf by the undersigned thereunto duly authorized.

MOBILE MINI, INC.

By: / s / Lawrence Trachtenberg
Lawrence Trachtenberg

Executive Vice President and Chief Financial Officer

Dated: July 1, 2008

EXHIBIT INDEX

- 3.1 Certificate of Designation of Mobile Mini Series A Convertible Redeemable Participating Preferred Stock.
- 3.2 Certificate of Amendment of the Amended and Restated Certificate of Incorporation of Mobile Mini.
- 4.1 Supplemental Indenture, dated as of June 27, 2008, among Mobile Mini, Inc., Mobile Mini of Ohio LLC, Mobile Mini, LLC, Mobile Mini, LLC, Mobile Mini I, Inc., A Royal Wolf Portable Storage, Inc., Temporary Mobile Storage, Inc., Delivery Design Systems, Inc., Mobile Mini Texas Limited Partnership, LLP, Mobile Storage Group, Inc., the guarantors party to the MSG Indenture and Wells Fargo Bank, N.A., as trustee.
- 4.2 Indenture, dated as of August 1, 2006, by and among Mobile Services Group, Inc., Mobile Storage Group, Inc., the subsidiary guarantors named therein and Wells Fargo Bank, N.A., as trustee (incorporated by reference to Exhibit 4.1 to the Mobile Storage Group, Inc.'s Form S-4 filed on September 18, 2007).
- 4.3 Supplemental Indenture, dated as of June 27, 2008, among Mobile Mini, Inc., Mobile Storage Group, Inc., A Better Mobile Storage Company, Mobile Storage Group (Texas), LP, the guarantors party to the Mobile Mini Indenture and Law Debenture Trust Company of New York, as trustee.
- 10.1 Escrow Agreement dated as of June 27, 2008, between Mobile Mini, Welsh, Carson, Anderson & Stowe X, L.P. and Wells Fargo Bank, N.A.
- 10.2 Stockholders Agreement, dated as of June 27, 2008, between Mobile Mini and the certain stockholders.
- 10.3 Credit Agreement, dated June 27, 2008, between Mobile Mini, Deutsche Bank AG New York Branch and the other lenders.
- 99.1 Press Release of Mobile Mini dated June 27, 2008.

CERTIFICATE OF DESIGNATION OF MOBILE MINI, INC.

Pursuant to Section 151 of the General Corporation Law of the State of Delaware

SERIES A CONVERTIBLE REDEEMABLE PARTICIPATING PREFERRED STOCK

MOBILE MINI, INC., a Delaware corporation (the "<u>Corporation</u>"), hereby certifies that the following resolution has been duly adopted by the Board of Directors of the Corporation (the "<u>Board</u>"):

WHEREAS, pursuant to the amended and restated certificate of incorporation of the Corporation (the "<u>Amended and Restated Certificate of Incorporation</u>"), the Corporation is authorized to issue up to 20,000,000 shares of preferred stock, par value \$.01 per share (the "<u>Preferred Stock</u>"), with such designations, preferences, rights and qualifications, limitations or restrictions as may be determined by the Board from time to time; and

RESOLVED, that pursuant to the authority expressly granted to and vested in the Board by the provisions of the Amended and Restated Certificate of Incorporation, there hereby is created, out of the 20,000,000 shares of Preferred Stock authorized by Article IV of the Amended and Restated Certificate of Incorporation, a series of the Preferred Stock consisting of 8,555,556 shares, which series shall have the following powers, designations, preferences and relative, participating, optional and other special rights, and the following qualifications, limitations and restrictions:

- 1. <u>Designation</u>. This series of Preferred Stock shall be designated as the "Series A Convertible Redeemable Participating Preferred Stock".
- 2. <u>Authorization</u>. The Corporation shall have the authority to issue 8,555,556 shares of the Series A Convertible Redeemable Participating Preferred Stock, par value \$.01 per share, of the Corporation (the "Series A Preferred Stock").
- 3. <u>Rank</u>. The Series A Preferred Stock shall, with respect to dividend rights, rights upon a Liquidity Event, rights to any other distributions or payments with respect to capital stock, voting rights and all other rights and preferences, rank junior to each other class or series of capital stock of the Corporation (including, without limitation, any class or series of capital stock created after the date hereof which by its terms ranks senior to the Series A Preferred Stock), other than (i) in all respects all classes or series of common stock existing or created after the date hereof, (ii) with respect to any other class or series of capital stock (including, without limitation, each class or series of capital stock created after the date hereof) which by its terms ranks junior to the Series A Preferred Stock, it shall rank senior to each such series of capital stock, (iii) with respect to dividends and voting rights, it shall rank *pari passu* with the Common Stock, and (iv) with respect to a distribution upon the occurrence of a Liquidity Event it shall rank senior to the Common Stock.

4. Dividends.

- (a) Holders of record of shares of Series A Preferred Stock shall be entitled to receive when, as and if declared by the Board, out of funds legally available therefor, cash dividends payable on shares of the Series A Preferred Stock at the same time as cash dividends are paid on shares of Common Stock (and in the same manner and amount as if such shares of the Series A Preferred Stock had been converted at the Conversion Rate then in effect into shares of Common Stock at the time of such cash dividends). All dividends described in this Section 4(a) shall be payable on the date determined by the Board (a "<u>Dividend Payment Date</u>"). Such dividends shall be paid to the holders of record of shares of Series A Preferred Stock at the close of business on the date specified by the Board at the time such dividend is declared; <u>provided</u>, that such date shall not be more than sixty (60) days nor less than ten (10) days prior to the respective Dividend Payment Date.
- (b) All dividends paid with respect to the shares of Series A Preferred Stock pursuant to this Section 4 shall be paid *pro rata* to the holders of Series A Preferred Stock.
- (c) The shares of Series A Preferred Stock are subject to the same restrictions on the payment of dividends that are applicable to the shares of Common Stock, as set forth in the Senior Credit Agreement.

5. Preference Upon a Liquidity Event.

- (a) Upon the occurrence of a Liquidity Event, each holder of shares of Series A Preferred Stock then outstanding shall be entitled to be paid out of the assets of the Corporation legally available for distribution to its stockholders, whether such assets are stated capital or surplus of any nature, a per share amount (such amount being herein referred to as the "Series A Liquidation Preference") on such date equal to the greater of (i) the Original Liquidation Preference plus all declared and unpaid dividends per each share of Series A Preferred Stock and (ii) the amount such holder would be entitled to receive if each share of Series A Preferred Stock held by it were converted at the Conversion Rate then in effect into shares of Common Stock immediately prior to the occurrence of such Liquidity Event, before any payment is made to the holders of Common Stock or any other class or series of capital stock of the Corporation ranking on liquidation junior to the Series A Preferred Stock in connection with such Liquidity Event. If, upon the occurrence of any such Liquidity Event, the remaining assets of the Corporation legally available for distribution to its stockholders shall be insufficient to pay such holders the aggregate Series A Liquidation Preference then the entire remaining assets and funds of the Corporation legally available for distribution to the holders of the Series A Preferred Stock shall be allocated among, and distributed to, such holders *pro rata* based on the number of shares of Series A Preferred Stock then held by each such holder.
- (b) The amount distributed to the holders of the Series A Preferred Stock in connection with a transaction referred to in this Section 5 shall be the cash or the fair market value of the property, rights or other securities distributed to such holders by the Corporation or acquiring Person, as the case may be. The fair market value of property, rights or other securities shall (i) in the case of publicly traded securities, be equal to the value as calculated in accordance with the definitive agreement(s) giving effect to the Liquidity Event and, if the agreement(s) do not prescribe a means for calculating such value the value shall be equal to the volume weighted

average of the securities on their principal exchange over the thirty (30) trading days immediately preceding the effective date of the Liquidity Event, or (ii) in all other cases, be determined by and in the good faith discretion of the Board; provided, however, that (x) if the Majority Holders (the "Disputing Party.") dispute in writing the Board's determination of fair market value within ten (10) days of the Board's determination of fair market value and the Disputing Party and the Board are unable to reach agreement as to the fair market value within twenty (20) days after the date the Board receives the Disputing Party's written dispute of the Board's determination of fair market value, the Corporation and the Disputing Party shall seek an independent appraisal of such fair market value by an independent appraiser experienced in valuing the property, rights or other securities in question and mutually agreeable to the Corporation and the Disputing Party, and the determination of such appraiser shall be final and binding upon the Corporation and the Disputing Party and (y) in connection with such appraisal, the cost and expense of such appraisal shall be borne 50% by the Corporation and 50% by the Disputing Party. Payment of all amounts (including, without limitation the amounts payable pursuant to Section 5(a) hereunder) required to be paid to the holders of the Series A Preferred Stock shall constitute a redemption in full of such Series A Preferred Stock and after the payment of such amounts, all such shares of Series A Preferred Stock shall cease to be outstanding for all purposes.

(c) The amounts set forth above and throughout this Section 5 shall be subject to equitable adjustment whenever there shall occur a stock dividend, stock split or combination involving the Series A Preferred Stock.

6. Voting.

- (a) Holders of record of shares of Series A Preferred Stock shall be entitled to vote, voting together with the holders of Common Stock as a single class, on all matters on which the holders of Common Stock are entitled to vote as if such shares of Series A Preferred Stock had been converted to Common Stock at the time of such vote at the Conversion Rate then in effect as set forth in Section 7(a).
- (b) In addition to the rights set forth in Section 6(a) hereof, without the affirmative vote of the Majority Holders, the Corporation may not amend this Certificate of Designation.
 - 7. Optional Conversion. The holders of the Series A Preferred Stock shall have conversion rights as follows (the "Conversion Rights"):
- (a) <u>Right to Convert</u>. Each share of the Series A Preferred Stock shall be convertible, at the option of the holder thereof, at any time and from time to time, and without the payment of additional consideration by the holder thereof, into such number of fully paid and nonassessable shares of Common Stock as is determined by dividing the Original Liquidation Preference by the Conversion Price (as defined below) in effect at the time of conversion. The "<u>Conversion Price</u>" shall initially be \$18.00 and shall be subject to adjustment from time to time as described below.
- (b) <u>Fractional Shares</u>. No fractional shares of Common Stock shall be issued upon conversion of Series A Preferred Stock. For purposes of determining the number of

fractional shares of Common Stock that would be issued upon conversion of Series A Preferred Stock, the following calculation shall be performed: the aggregate number of shares of Series A Preferred Stock that a holder proposes to convert into shares of Common Stock shall be multiplied by the Conversion Rate then in effect. In lieu of issuing any fractional shares to which the holder would otherwise be entitled based on the result of the foregoing calculation, the Corporation shall pay cash equal to such fraction multiplied by the then effective Conversion Price.

(c) Mechanics of Conversion.

- (i) In order for a holder of Series A Preferred Stock to convert shares of Series A Preferred Stock into shares of Common Stock, such holder shall surrender the certificate or certificates for such shares of Series A Preferred Stock at the office of the transfer agent for Series A Preferred Stock (or at the principal office of the Corporation if the Corporation serves as its own transfer agent), together with written notice that such holder elects to convert all or any number of the shares of Series A Preferred Stock represented by such certificates. Such notice shall state such holder's name or the names of the nominees in which such holder wishes the certificate or certificates for shares of Common Stock to be issued. If required by the Corporation, certificates surrendered for conversion shall be endorsed or accompanied by a written instrument or instruments of transfer, in form reasonably satisfactory to the Corporation, duly executed by the registered holder or his, her or its attorney duly authorized in writing. The date of receipt of such certificates and notice by the transfer agent (or by the Corporation if the Corporation serves as its own transfer agent) shall be the conversion date ("Conversion Date"). The Corporation shall, as soon as practicable after the Conversion Date, issue and deliver to such holder of Series A Preferred Stock, or to his, her or its nominees, a certificate or certificates for the number of shares of Common Stock to which such holder shall be entitled, together with cash in lieu of any fraction of a share.
- (ii) The Corporation shall at all times when Series A Preferred Stock shall be outstanding, reserve and keep available out of its authorized but unissued Common Stock, for the purpose of effecting the conversion of Series A Preferred Stock, such number of its duly authorized shares of Common Stock as shall from time to time be sufficient to effect the conversion of all outstanding shares of Series A Preferred Stock.
- (iii) Upon any such conversion, no adjustment to the applicable Conversion Price shall be made for any declared but unpaid dividends on Series A Preferred Stock surrendered for conversion or on the Common Stock delivered upon conversion.
- (iv) All shares of Series A Preferred Stock which shall have been surrendered for conversion as herein provided shall no longer be deemed to be outstanding and all rights with respect to such shares shall immediately cease and terminate on the date that such holder of Series A Preferred Stock becomes the record holder of the shares of Common Stock received upon conversion, except only the right of the holders thereof to receive shares of Common Stock in exchange therefor and the payment of any dividends declared but unpaid thereon. Any shares of Series A Preferred Stock so converted shall be retired and cancelled and shall not be reissued, and the

Corporation (without the need for stockholder action) may from time to time take such appropriate action as may be necessary to reduce the authorized number of shares of Series A Preferred Stock accordingly.

- (v) The Corporation shall pay any and all issue and other taxes that may be payable in respect of any issuance or delivery of shares of Common Stock upon conversion of shares of Series A Preferred Stock pursuant to this Section 7. The Corporation shall not, however, be required to pay any tax which may be payable in respect of any transfer involved in the issuance and delivery of shares of Common Stock in a name other than that in which the shares of Series A Preferred Stock so converted were registered.
- (d) <u>Adjustment for Stock Splits and Combinations</u>. If the Corporation shall at any time or from time to time after the Original Issue Date effect a subdivision of the outstanding Common Stock, the applicable Conversion Price then in effect immediately before that subdivision shall be proportionately decreased. If the Corporation shall at any time or from time to time after the Original Issue Date combine the outstanding shares of Common Stock, the applicable Conversion Price then in effect immediately before the combination shall be proportionately increased. Any adjustment under this paragraph shall become effective at the close of business on the date the subdivision or combination becomes effective.
- (e) <u>Adjustment for Certain Dividends and Distributions</u>. In the event the Corporation at any time, or from time to time after the Original Issue Date shall make or issue, or fix a record date for the determination of holders of Common Stock entitled to receive, a dividend or other distribution payable in additional shares of Common Stock, then and in each such event the applicable Conversion Price then in effect immediately before such event shall be decreased as of the time of such issuance or, in the event such a record date shall have been fixed, as of the close of business on such record date, to the amount obtained by multiplying the applicable Conversion Price then in effect by a fraction:
 - (i) the numerator of which shall be the total number of shares of Common Stock issued and outstanding immediately prior to the time of such issuance or the close of business on such record date (calculated assuming conversion of issued and outstanding shares of Series A Preferred Stock, and the exercise, exchange or conversion of all then outstanding options, warrants, or exchange or subscription rights), and
 - (ii) the denominator of which shall be the total number of shares of Common Stock issued and outstanding immediately prior to the time of such issuance or the close of business on such record date (calculated assuming conversion of issued and outstanding shares of Series A Preferred Stock, and the exercise, exchange or conversion of all then outstanding options, warrants, or exchange or subscription rights) <u>plus</u> the number of shares of Common Stock issuable in payment of such dividend or distribution; <u>provided</u>, that if such record date shall have been fixed and such dividend is not fully paid or if such distribution is not fully made on the date fixed therefor, the applicable Conversion Price shall be recomputed accordingly as of the close of business on such record date and thereafter and the applicable Conversion Price shall be adjusted pursuant to this paragraph as of the time of actual payment of such dividends or distributions; and <u>provided</u>, <u>further</u>, that no such adjustment shall be made if, at the election of the

Corporation, the holders of shares of Series A Preferred Stock simultaneously receive a dividend or other distribution of shares of Common Stock in a number equal to the number of shares of Common Stock as they would have received if all outstanding shares of Series A Preferred Stock had been converted into Common Stock on the date of such event.

- (f) <u>Adjustments for Other Dividends and Distributions</u>. In the event the Corporation at any time or from time to time after the Original Issue Date shall make or issue, or fix a record date for the determination of holders of shares of Common Stock entitled to receive, a dividend or other distribution payable in securities of the Corporation other than shares of Common Stock, then and in each such event provision shall be made so that the holders of shares of Series A Preferred Stock shall receive upon conversion thereof in addition to the number of shares of Common Stock receivable thereupon, the amount of securities of the Corporation that they would have received had the Series A Preferred Stock been converted into Common Stock on the date of such event and had they thereafter, during the period from the date of such event to and including the conversion date, retained such securities receivable by them as aforesaid during such period, giving application to all adjustments called for during such period under this paragraph with respect to the rights of the holders of Series A Preferred Stock; <u>provided</u>, that no such adjustment shall be made if, at the election of the Corporation, the holders of shares of Series A Preferred Stock simultaneously receive a dividend or other distribution of such securities in an amount equal to the amount of such securities as they would have received if all outstanding shares of Series A Preferred Stock had been converted into Common Stock on the date of such event.
- (g) <u>Adjustment for Reclassification</u>, <u>Exchange</u>, <u>or Substitution</u>. If the shares of Common Stock issuable upon the conversion of shares of Series A Preferred Stock shall be changed into the same or a different number of shares of any class or classes of stock, whether by capital reorganization, reclassification, or otherwise (other than a subdivision or combination of shares or stock dividend provided for above), then and in each such event the holder of each such share of Series A Preferred Stock shall have the right thereafter to convert such share into the kind and amount of shares of stock and other securities and property receivable, upon such reorganization, reclassification, or other change, by holders of the number of shares of Common Stock into which such shares of Series A Preferred Stock might have been converted immediately prior to such reorganization, reclassification, or change, all subject to further adjustment as provided herein.
- (h) <u>Certificate as to Adjustments</u>. Upon the occurrence of each adjustment of the applicable Conversion Price pursuant to this Section 7, the Corporation, at its expense, shall compute such adjustment or readjustment in accordance with the terms hereof and furnish to each holder of shares of Series A Preferred Stock a statement setting forth such adjustment or readjustment and showing in detail the facts upon which such adjustment or readjustment is based. The Corporation shall, upon the written request at any time of any holder of Series A Preferred Stock, furnish or cause to be furnished to such holder a similar statement setting forth (i) such adjustments and readjustments, (ii) the applicable Conversion Price then in effect, and (iii) the number of shares of Common Stock and the amount, if any, of other property which then would be received upon the conversion of such Series A Preferred Stock.

(i) Notice of Record Date. In the event:

- (i) that the Corporation declares a dividend (or any other distribution) on its shares of Common Stock payable in shares of Common Stock or other securities of the Corporation;
 - (ii) that the Corporation subdivides or combines its outstanding shares of Common Stock;
- (iii) of any reclassification of the shares of Common Stock of the Corporation (other than a subdivision or combination of its outstanding shares of Common Stock or a stock dividend or stock distribution thereon);
 - (iv) of the occurrence of a Liquidity Event or the consummation of a Sale of the Company; or
- (v) that the Corporation receives a Holder Optional Redemption written election or a Holder Corporation Sale Optional Redemption written election in accordance with Section 8(a) or 8(c).

then the Corporation shall use its commercially reasonable efforts to cause to be filed at its principal office or at the office of the transfer agent of the Series A Preferred Stock, and shall cause to be mailed to the holders of Series A Preferred Stock at their last addresses as shown on the records of the Corporation or such transfer agent, at least ten (10) days prior to the date specified in (A) below or thirty (30) days before the date specified in (B) below, a notice stating

- (A) the record date of such dividend, distribution, subdivision or combination, or, if a record is not to be taken, the date as of which the holders of Common Stock of record to be entitled to such dividend, distribution, subdivision or combination are to be determined or the date upon which the Holder Optional Redemption is expected to be consummated; or
- (B) the date on which such Liquidity Event is expected to occur or the date on which such Sale of the Company is expected to be consummated, and the date as of which it is expected that holders of Common Stock of record shall be entitled to exchange their shares of Common Stock for securities or other property deliverable upon such Liquidity Event or Sale of the Company; <u>provided</u>, that the Corporation's failure to provide any notice required under this Section 7(j) after using commercially reasonable efforts shall not be deemed a default, breach or violation of this Section 7(j).

8. Holder Optional Redemptions.

(a) At any time after the tenth (10th) anniversary of the Original Issue Date, at the written election of the Majority Holders the Corporation shall redeem (the "<u>Holder Optional Redemption</u>") all of the shares of Series A Preferred Stock then outstanding on the date specified in the Majority Holders' written election (which date shall not be earlier than thirty (30) days

after the date of such written election or later than ninety (90) days after the date of such written election) (such date, the "Holder Optional Redemption Date. If the funds of the Corporation legally available for the redemption of shares of Series A Preferred Stock shall be insufficient to permit the payment of the amounts due to such holders on the Holder Optional Redemption Date, then the holders of Series A Preferred Stock shall share in any legally available funds *pro rata* based on the number of shares of Series A Preferred Stock held by each such holder. During the Default Period, (i) the Corporation shall use commercially reasonable efforts to obtain the funds and/or make funds legally available as necessary to make the remaining payments required under this Section 8, (ii) the number of directors on the Board shall be increased by one (1) and the Majority Holders shall have the right to appoint the one (1) additional director (with such director holding office until the expiration of the Default Period) and (iii) the amount of the remaining payments required under this Section 8 shall accrue interest at a rate of ten percent (10%) per annum, compounded quarterly, until the remaining payments are paid in full. As soon as practicable after the Corporation has funds legally available therefor, the Corporation shall make the remaining payments required under this Section 8.

(b) The closing of the Corporation's redemption pursuant to Section 8(a) shall take place at 10:00 a.m. Central Standard time on the Holder Optional Redemption Date at the Corporation's principal executive office or place of business. At the closing the Corporation shall pay to each holder of Series A Preferred Stock, against the Corporation's receipt from such holder of the certificate or certificates representing the shares of such Series A Preferred Stock then held by such holder, an amount equal to the aggregate payment due pursuant to Section 8(a) for all such shares, by wire transfer of immediately available funds, or if such holder shall not have specified wire transfer instructions to the Corporation prior to the closing, by check made payable to the order of such holder; provided, that if any certificate representing the shares of Series A Preferred Stock has been lost, stolen or destroyed, such holder will execute and deliver to the Corporation an affidavit of loss in connection with such lost, stolen or destroyed certificate(s), in a form reasonably acceptable to the Corporation. If so required by the Corporation, certificates surrendered for redemption shall be endorsed or accompanied by written instrument or instruments of transfer, in form reasonably satisfactory to the Corporation, duly executed by the registered holder or by his, her or its attorney duly authorized in writing.

(c) If the Corporation enters into a binding agreement in respect of a Sale of the Company (x) on or prior to June 27, 2008, (y) on or prior to the date that is ninety (90) days after the Original Issue Date and the negotiations in connection with such Sale of the Company commenced prior to the Original Issue Date, or (z) after the date which is ninety (90) days after the Original Issue Date, and in any such case the per-share purchase price of the Common Stock in connection with such Sale of the Company is less than \$23.00 per share (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences), then at the written election of the Majority Holders made within ten (10) days of the receipt of the Corporation's notice pursuant to Section 7(ij)(B) the Corporation shall redeem (the "Holder Corporation Sale Optional Redemption") all of the shares of Series A Preferred Stock then outstanding at a price per share equal to (i) in the case if clauses (x) or (y) above, \$21.60 per share (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences), and (ii) in the case of clause (z) above, the Original Liquidation Preference plus declared but unpaid dividends (such price, the "Holder Corporation Sale

Optional Redemption Price"). If any of the outstanding shares of Common Stock would receive any proceeds in connection with such Sale of the Company, then the aggregate Holder Corporation Sale Optional Redemption Price shall be paid to the holders of Series A Preferred Stock prior to any payment to the holders of Common Stock in connection with such Sale of the Company; provided, that if the assets of the Corporation legally available for distribution shall be insufficient to pay such holders the aggregate Holder Corporation Sale Optional Redemption Price, then (i) the entire assets and funds of the Corporation legally available for distribution to the holders of the Series A Preferred Stock, shall be allocated among, and distributed to, such holders pro rata based on the number of shares of Series A Preferred Stock then held by each such holder and (ii) the acquiring entity in such Sale of the Company shall pay to the holders of Series A Preferred Stock the aggregate Holder Corporation Sale Optional Redemption Price that remains outstanding after the payment contemplated by the foregoing clause (i) has been made; or (B) if the Holder Corporation Sale Optional Redemption is prohibited by the application of Section 8(e), then the acquiring entity in such Sale of the Company shall pay to the holders of Series A Preferred Stock the aggregate Holder Corporation Sale Optional Redemption Price pursuant to this Section 8(c) shall constitute a redemption in full of such Series A Preferred Stock shall cease to be outstanding for all purposes.

(d) The closing of the Holder Corporation Sale Optional Redemption shall take place at the Corporation's principal executive office or place of business on the date the Sale of the Company is consummated. At the closing the Corporation (or, if applicable, the acquiring entity in such Sale of the Company) shall pay to each holder of Series A Preferred Stock, against the Corporation's receipt from such holder of the certificate or certificates representing the shares of such Series A Preferred Stock then held by such holder, an amount equal to the aggregate payment due pursuant to Section 8(c) for all such shares, by wire transfer of immediately available funds, or if such holder shall not have specified wire transfer instructions to the Corporation prior to the closing, by check made payable to the order of such holder; provided, that if any certificate representing the Series A Preferred Stock has been lost, stolen or destroyed, such holder will execute and deliver to the Corporation an affidavit of loss in connection with such lost, stolen or destroyed certificate(s), in a form reasonably acceptable to the Corporation. If so required by the Corporation, certificates surrendered for redemption shall be endorsed or accompanied by written instrument or instruments of transfer, in form satisfactory to the Corporation, duly executed by the registered holder or by his, her or its attorney duly authorized in writing.

(e) Notwithstanding any of the foregoing, the Corporation shall not redeem any shares of Series A Preferred Stock until after such time as both (i) all obligations under the Senior Credit Agreement shall have been satisfied in full in cash (other than contingent reimbursement obligations and contingent indemnification obligations for which no claim has been made) and (ii) the revolving loan commitments under the Senior Credit Agreement have been terminated.

9. Corporation Optional Redemption.

(a) Subject to the holders Conversion Rights, in the event the Majority Holders did not exercise their Holder Corporation Sale Optional Redemption right, if applicable, the Corporation may redeem (the "Corporation Optional Redemption") all of the shares of Series A Preferred Stock then outstanding simultaneously with the consummation of a Sale of the Company in which the per-share purchase price of the Common Stock in connection with such Sale of the Company is less than \$23.00 per share (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences) for an amount per share equal to the Original Liquidation Preference plus declared but unpaid dividends (the "Corporation Optional Redemption Price"). If any of the outstanding shares of Common Stock would receive any proceeds in connection with such Sale of the Company, then the aggregate Corporation Optional Redemption Price shall be paid to the holders of Series A Preferred Stock prior to any payment to the holders of Common Stock in connection with such Sale of the Company; provided, that if the assets of the Corporation legally available for distribution shall be insufficient to pay such holders the aggregate Corporation Optional Redemption Price, then (i) the entire assets and funds of the Corporation legally available for distribution to the holders of the Series A Preferred Stock, shall be allocated among, and distributed to, such holders pro rata based on the number of shares of Series A Preferred Stock then held by each such holder and (ii) the acquiring entity in such Sale of the Company shall pay to the holders of Series A Preferred Stock the aggregate Corporation Optional Redemption Price that remains outstanding after the payment contemplated by the foregoing clause (i) has been made; or (B) if the Corporation Optional Redemption is prohibited by the application of Section 9(c), then the acquiring entity in such Sale of the Company shall pay to the holders of Series A Preferred Stock the aggregate Corporation Optional Redemption Price that remains outstanding. Payment of the Corporation Optional Redemption Price pursuant to this Section 9(a) shall constitute a redemption in full of such Series A Preferred Stock and after the payment in full of such amounts, all such shares of Series A Preferred Stock shall cease to be outstanding for all purposes.

(b) The closing of the Corporation Optional Redemption shall take place at the Corporation's principal executive office or place of business on the date the Sale of the Company is consummated. At the closing the Corporation (or, if applicable, the acquiring entity in such Sale of the Company) shall pay to each holder of Series A Preferred Stock, against the Corporation's receipt from such holder of the certificate or certificates representing the shares of such Series A Preferred Stock then held by such holder, an amount equal to the aggregate payment due pursuant to this Section 9 for all such shares, by wire transfer of immediately available funds, or if such holder shall not have specified wire transfer instructions to the Corporation prior to the closing, by check made payable to the order of such holder; provided, that if any certificate representing the Series A Preferred Stock has been lost, stolen or destroyed, such holder will execute and deliver to the Corporation an affidavit of loss in connection with such lost, stolen or destroyed certificate(s), in a form reasonably acceptable to the Corporation. If so required by the Corporation, certificates surrendered for redemption shall be endorsed or accompanied by written instrument or instruments of transfer, in form satisfactory to the Corporation, duly executed by the registered holder or by his, her or its attorney duly authorized in writing.

(c) Notwithstanding any of the foregoing, the Corporation shall not redeem any shares of Series A Preferred Stock until after such time as both (i) all obligations under the Senior Credit Agreement shall have been satisfied in full in cash (other than contingent reimbursement obligations and contingent indemnification obligations for which no claim has been made) and (ii) the revolving loan commitments under the Senior Credit Agreement have been terminated.

10. Mandatory Conversion.

(a) If (x) at any time after the one (1) year anniversary of the Original Issue Date, the Market Price of the Common Stock exceeds \$23.00 per share (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences), then effective as of the close of business on the last trading day of the thirty (30) day period in which the Market Price so exceeds \$23.00 per share (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences), or (y) the Corporation enters into a binding agreement in respect of a Sale of the Company in which the per-share purchase price of the Common Stock in connection with such Sale of the Company is less than \$23.00 per share (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences) and the Majority Holders did not exercise their Holder Corporation Sale Optional Redemption right, if applicable, or the Corporation does not exercise its Corporation Optional Redemption right, then all outstanding shares of Series A Preferred Stock shall automatically be converted into shares of Common Stock at the then effective Conversion Rate as of the close of business on the last day of such thirty day period or the day immediately preceding the consummation of such Sale of the Company, as the case may be (either of such date, the "Mandatory Conversion Date"). The Corporation shall file a Certificate of Elimination with the Secretary of State of the State of Delaware terminating this Certificate of Designation as soon as practicable after the Mandatory Conversion Date.

(b) All holders of record of shares of Series A Preferred Stock shall be given written notice of the applicable Mandatory Conversion Date and the place designated for mandatory conversion of all such shares of Series A Preferred Stock pursuant to this Section 10. Such notice shall be given (x) in the case of Section 10 (a) (x), within five (5) business days after the occurrence of the Market Price exceeding \$23.00 per share (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences), or (y) in the case of Section 10 (a) (y), thirty (30) days prior to the consummation of such Sale of the Company. Such notice shall be sent by overnight courier or first class or registered mail, postage prepaid, to each record holder of applicable Series A Preferred Stock at such holder's address last shown on the records of the transfer agent for Series A Preferred Stock (or the records of the Corporation, if it serves as its own transfer agent). Upon receipt of such notice, each holder of the applicable shares of Series A Preferred Stock shall surrender his, her or its certificate or certificates for all such shares to the Corporation at the place designated in such notice, and shall thereafter receive certificates for the number of shares of Common Stock to which such holder is entitled pursuant to this Section 10; provided, that if any certificate representing the Series A Preferred Stock has been lost, stolen or destroyed, such holder will execute and deliver to the Corporation an affidavit of loss in connection with such lost, stolen or destroyed certificate(s), in a form reasonably acceptable to the Corporation. Upon such holder of Series A Preferred Stock becoming the record holder of the shares of Common Stock received upon conversion, all rights

with respect to Series A Preferred Stock so converted will terminate, except only the rights of the holders thereof, upon surrender of their certificate or certificates therefore (or, if applicable, upon delivery to the Corporation of the affidavit of loss), to receive certificates for the number of shares of Common Stock into which such Series A Preferred Stock has been converted, and payment of any declared but unpaid dividends thereon, if any. If so required by the Corporation, certificates surrendered for conversion shall be endorsed or accompanied by written instrument or instruments of transfer, in form satisfactory to the Corporation, duly executed by the registered holder or by his, her or its attorney duly authorized in writing. As soon as practicable after the applicable Mandatory Conversion Date and the surrender of the certificate or certificates for the shares of Series A Preferred Stock, the Corporation shall cause to be issued and delivered to such holder, or on his, her or its written order, a certificate or certificates for the number of full shares of Common Stock issuable on such conversion in accordance with the provisions hereof and cash as provided in Section 7(b) in respect of any fraction of a share of Common Stock otherwise issuable upon such conversion.

- (c) Such converted Series A Preferred Stock may not be reissued, and the Corporation may (but shall not be obligated to) thereafter take such appropriate action (without the need for stockholder action) as may be necessary to reduce the authorized number of shares of Common Stock by the number of shares of Series A Preferred Stock so converted.
- (d) Any Series A Preferred Stock converted pursuant to this Section 10 will be cancelled and will not under any circumstances be reissued, sold or transferred and the Corporation may (but shall not be obligated to) from time to time take such appropriate action as may be necessary to reduce the authorized Preferred Stock accordingly.
- 11. <u>Restrictive Legends</u>. Each certificate representing shares of Series A Preferred Stock (and shares of Common Stock into which such shares of Series A Preferred Stock are converted) shall be stamped or otherwise imprinted with legends in substantially the following form:

"The sale, transfer, hypothecation, negotiation, pledge, assignment, encumbrance or other disposition of this share certificate and the shares of Preferred Stock represented hereby are restricted by and are subject to all of the terms, conditions and provisions of that certain Preferred Stockholders Agreement, dated June 27, 2008, as amended from time to time, by and between the Company and the investors party thereto, which agreement is on file at the principal office of the Company."

"The securities represented by this certificate have not been registered under the Securities Act of 1933, as amended, or pursuant to any state securities laws. The securities have been acquired for investment and may not be sold or transferred except in compliance with the registration requirements of the Securities Act of 1933, as amended, and applicable state securities laws or pursuant to an exemption therefrom."

Upon the written request of any holder of Series A Preferred Stock, (i) the Corporation shall remove the portion of such legend that relates to the Stockholders Agreement from the certificates evidencing any shares of Series A Preferred Stock (or shares of Common Stock into which such shares of Series A Preferred Stock are converted) which cease to be bound by the Stockholders Agreement and (ii) the Corporation shall remove the Securities Act portion of such

legend from such certificate or certificates (<u>provided</u>, that such shares are eligible (as reasonably determined by the Board) for sale pursuant to Rule 144 (or any similar rule or rules then in effect) under the Securities Act).

- 12. Exclusivity. Except as expressly set forth herein and in the Stockholders Agreement, the holders of Series A Preferred Stock shall have no rights other than those provided by applicable law.
- 13. <u>Amendments</u>. The terms, conditions, rights and preferences contained in this Certificate of Designation may be amended, modified, waived, amended and restated or replaced in its entirety upon the approval of the Board with the consent of the Majority Holders.
- 14. <u>Definitions; Gender</u>. (a) As used in this Certificate of Designation, and unless the context requires a different meaning, the following terms have the meanings indicated:
 - "Amended and Restated Certificate of Incorporation" shall have the meaning set forth in the recitals.
 - "Board" shall have the meaning set forth in the preamble.
 - "Certificate of Designation" shall have the meaning set forth in the recitals.
 - "Common Stock" shall mean, collectively, the common stock, par value \$.01, of the Corporation.
 - "Conversion Date" shall have the meaning set forth in Section 7(c)(i).
 - "Conversion Price" shall have the meaning set forth in Section 7(a).
- "Conversion Rate" means the rate at which shares of Series A Preferred Stock may be converted into shares of Common Stock, as determined by dividing the Original Liquidation Preference by the Conversion Price in effect at the time of conversion.
 - "Conversion Rights" shall have the meaning set forth in Section 7.
 - "Corporation" shall have the meaning set forth in the preamble.
 - "Corporation Optional Redemption" shall have the meaning set forth in Section 9.
 - "Corporation Optional Redemption Price" shall have the meaning set forth in Section 9.
- "<u>Default Period</u>" means the period beginning on the date that the Corporation fails to make any required payment to the holders of Series A Preferred Stock pursuant to Section 8(a) and ending on the date that the amounts required to be paid by the Corporation to the holders of Series A Preferred Stock pursuant to Section 8(a) have been paid in full.
 - "Disputing Party" shall have the meaning set forth in Section 5(b).

- "Dividend Payment Date" shall have the meaning set forth in Section 4(a).
- "Holder Corporation Sale Optional Redemption" shall have the meaning set forth in Section 8(c).
- "Holder Corporation Sale Optional Redemption Price" shall have the meaning set forth in Section 8(c).
- "Holder Optional Redemption" shall have the meaning set forth in Section 8(a).
- "Holder Optional Redemption Date" shall have the meaning set forth in Section 8(a).
- "<u>Liquidity Event</u>" shall mean a voluntary or involuntary filing of bankruptcy of the Corporation, any liquidation, dissolution or winding up of the Corporation but shall not include a Sale of the Company.
- "Majority Holders" means the holders of Series A Preferred Stock representing at least a majority of the then outstanding shares of the Series A Preferred Stock, voting as a single class.
 - "Mandatory Conversion Date" shall have the meaning set forth in Section 10(a).
- "Market Price" means the average of the closing prices of the Common Stock on the NASDAQ reporting system or on the principal exchange on which the Common Stock is traded (as reported in the Wall Street Journal) over a period of thirty (30) days consisting of the day as of which "Market Price" is being determined and the twenty nine (29) consecutive trading days prior to such day (including trading days occurring prior to the one year anniversary of the Original Issue Date); provided, that if the Common Stock is not traded on any exchange or the over-the-counter market, then "Market Price" shall be determined in good faith by the Board (provided that the dispute mechanism set forth in Section 5(b) shall apply if the Majority Holders dispute the Board's determination of "Market Price").
 - "Original Issue Date" shall be June 27, 2008.
- "Original Liquidation Preference" shall mean \$18.00 per share of the Series A Preferred Stock (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences).
- "Person" shall mean any individual, corporation, limited partnership, general partnership, limited liability company, joint stock company, joint venture, association, company, trust, or any governmental or political subdivision or any agency, department or instrumentality thereof.
 - "Preferred Stock" shall have the meaning set forth in the recitals.
 - "Sale of the Company" shall mean
- (i) any consolidation or merger of the Corporation or a subsidiary of the Corporation in which the shares of Common Stock are converted into cash, securities or other

property other than any consolidation or merger of the Corporation in which holders of the Corporation's capital stock immediately prior to the consolidation or merger own greater than 50% of the voting stock and voting power of the surviving corporation immediately after the consolidation or merger;

- (ii) any sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all, or substantially all, of the assets of the Corporation and its subsidiaries other than any sale, lease, exchange or other transfer to an entity where the Corporation owns, directly or indirectly, greater than 50% of the outstanding voting securities of such entity after the transfer or series of transfers as the case may be; or
- (iii) any Person has become the beneficial owner (within the meaning of Rule 13d-3 promulgated under the Securities Exchange Act of 1934, as amended) of shares of the capital stock of the Corporation representing greater than 50% of the outstanding voting power of the Corporation.

"Securities Act" means the Securities Act of 1933, as amended.

"Senior Credit Agreement" means that certain Senior Credit Agreement, by and among the Corporation, Deutsche Bank AG New York Branch, as administrative agent, the lenders party from time to time thereto, and the agents named therein providing for up to \$1.0 billion of revolving credit borrowings, including any related notes, guarantees, collateral documents, instruments and agreements executed in connection therewith, and in each case as amended, restated, modified, renewed, refunded or extended in whole or in part from time to time under the same or any other agent, lender or group of lenders.

"Stockholders Agreement" means that certain Stockholders Agreement, dated as of June 27, 2008 by and the Corporation and the holders of Series A Preferred Stock party thereto, or the same may be amended, modified or waived from time to time.

"subsidiary" shall mean, with respect to any Person, any corporation, association or other business entity of which more than fifty percent (50%) of the total voting power of shares of capital stock or other equity interests entitled (without regard to occurrence of any contingency) to vote in the election of directors or other managing authority thereof is at the time owned or controlled, directly or indirectly, by such Person or its subsidiaries.

(b) Words expressed in the masculine shall include the feminine and neuter gender and vice versa.

IN WITNESS WHEREOF the foregoing Certificate of Designation has been duly executed on behalf of the Corporation by the undersigned on this 26th day of June, 2008.

MOBILE MINI, INC.

By: /s/ STEVEN BUNGER
Title: STEVEN BUNGER
Name: PRESIDENT

[Signature Page for Certificate of Designation]

CERTIFICATE OF AMENDMENT OF THE

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

of

MOBILE MINI, INC.

MOBILE MINI, INC., a corporation organized and existing under the General Corporation Law of the State of Delaware (the "Corporation") does hereby certify:

FIRST: That at a meeting of the Board of Directors of the Corporation a resolution was duly adopted setting forth a proposed amendment to the Amended and Restated Certificate of Incorporation of the Corporation, declaring said amendment to be advisable and calling a meeting of the stockholders of the Corporation for consideration thereof. The resolution setting forth the proposed amendments is as follows:

RESOLVED, that Article FOURTH of the Amended and Restated Certificate of Incorporation is amended in its entirety to read as follows:

"FOURTH: The total number of shares of all classes of stock which the Corporation shall have authority to issue is one hundred fifteen million (115,000,000) of which ninety five million (95,000,000) shares shall be common stock of the par value of one cent (\$0.01) per share and twenty million (20,000,000) shares shall be preferred stock with the par value of one cent (\$0.01) per share."

SECOND: That thereafter, pursuant to a resolution of the Board of Directors, a special meeting of the stockholders of the Corporation was duly called and held, upon notice in accordance with Section 222 of the General Corporation Law of the State of Delaware, at which meeting the necessary number of shares as required by statute were voted in favor of the amendment.

THIRD: That said amendment was duly adopted in accordance with the provisions of Section 242 of the General Corporation Law of the State of Delaware.

* * *

The Corporation has caused this Certificate to be executed by Steven G. Bunger, its President, and by Lawrence Trachtenberg, its Secretary, as of June 26, 2008.

MOBILE MINI, INC.

By: /s/ STEVEN G. BUNGER

Steven G. Bunger, President

ATTEST:

/s/ LAWRENCE TRACHTENBERG

Lawrence Trachtenberg, Secretary

[Signature Page to Certificate of Amendment]

SUPPLEMENTAL INDENTURE

THIS SUPPLEMENTAL INDENTURE (this "Supplemental Indenture"), dated as of June 27, 2008, among Mobile Mini, Inc., a Delaware corporation ("Parent"), Mobile Mini of Ohio LLC, a Delaware limited liability company, Mobile Mini, LLC, a California limited liability company, Mobile Mini, LLC, a Delaware limited liability company, Mobile Mini I, Inc., an Arizona corporation, A Royal Wolf Portable Storage, Inc., a California corporation, Temporary Mobile Storage, Inc., a California corporation, Delivery Design Systems, Inc., an Arizona corporation, Mobile Mini Texas Limited Partnership, LLP, a Texas limited liability partnership (each a "New Guarantor") and collectively, the "New Guarantors"), Mobile Storage Group, Inc., a Delaware corporation (the "Company"), the Guarantors party to the Indenture (as defined below) (the "Existing Guarantors") and Wells Fargo Bank, N.A., as Trustee.

WITNESSETH:

WHEREAS, the Company, Mobile Services Group, Inc., a Delaware corporation ("<u>Mobile Services</u>" and, together with the Company, the "<u>Original Issuers</u>"), the Existing Guarantors and the Trustee entered into an indenture (the "<u>Indenture</u>"), dated as of August 1, 2006, pursuant to which the Original Issuers issued \$200,000,000 in principal amount of 9³/4% Senior Notes due 2014 (the "Notes");

WHEREAS, on the date hereof, pursuant to that certain Agreement and Plan of Merger, dated as of February 22, 2008 (the "Merger Agreement"), among Parent, Cactus Merger Sub, Inc., a Delaware corporation and a direct wholly-owned subsidiary of Parent, MSG WC Holdings Corp., a Delaware corporation, and Welsh, Carson, Anderson & Stowe X, L.P., a Delaware limited partnership, the Merger and Subsequent Mergers (each as defined in the Merger Agreement) were consummated (collectively, the "Reorganization Transactions");

WHEREAS, as a result of the Reorganization Transactions, Mobile Services was merged with and into Parent, and the separate corporate existence of Mobile Services ceased, and Parent continued as the surviving corporation under the laws of the State of Delaware;

WHEREAS, Sections 5.1 and 5.2 of the Indenture require that as a result of the Reorganization Transactions, Parent execute and deliver to the Trustee this Supplemental Indenture pursuant to which Parent shall become a party to the Indenture and an "Issuer" for all purposes under the Notes, the Indenture and the Registration Rights Agreement and liable for all obligations of an "Issuer" thereunder;

WHEREAS, Section 4.16 of the Indenture provides that under certain circumstances new Restricted Subsidiaries are required to execute and deliver to the Trustee a supplemental indenture pursuant to which such new Restricted Subsidiary shall unconditionally guarantee all of Parent's and the Company's Obligations under the Notes and the Indenture pursuant to a Guarantee on the terms and conditions set forth in the Indenture;

WHEREAS, Section 9.1 of the Indenture provides that the Original Issuers and the Trustee may amend or supplement the Indenture in order to: (i) provide for the assumption of the Obligations of the Original Issuers or any Existing Guarantor to Holders of Notes in the case of a merger or consolidation or sale of all or substantially all of their assets and (ii) provide for

additional Guarantors, in each case without the consent of the Holders of the Notes and in accordance with the terms of the Indenture; and

WHEREAS, all acts and things prescribed by the Indenture, by law and by the certificate of incorporation and the bylaws (or comparable constituent documents) of the parties hereto necessary to make this Supplemental Indenture a valid instrument legally binding on each of the parties hereto, in accordance with its terms, have been duly done and performed.

NOW THEREFORE, to comply with the provisions of the Indenture, and in consideration of the foregoing, the parties hereto mutually covenant and agree for the equal and ratable benefit of the Holders of the Notes as follows:

ARTICLE 1

Section 1.01. This Supplemental Indenture is supplemental to the Indenture and does and shall be deemed to form a part of, and shall be construed in connection with and as part of, the Indenture for any and all purposes.

Section 1.02. This Supplemental Indenture shall become effective immediately upon its execution and delivery by each of the parties hereto.

ARTICLE 2

Section 2.01. Parent hereby becomes a party to and bound by all of the terms, conditions and other provisions of the Indenture with all attendant rights, duties and obligations stated therein, and hereby becomes an "Issuer" for all purposes under the Notes and the Indenture and liable for all Obligations of an "Issuer" thereunder, with the same force and effect as if Parent was originally named as an "Issuer" in the Indenture and as if Parent executed the Indenture on the date thereof.

Section 2.02. Parent hereby becomes a party to and bound by all of the terms, conditions and other provisions of the Registration Rights Agreement with all attendant rights, duties and obligations stated therein, and hereby becomes an "Issuer" for all purposes under the Registration Rights Agreement and liable for all obligations of an "Issuer" thereunder, with the same force and effect as if Parent was originally named as an "Issuer" in the Registration Rights Agreement and as if Parent executed the Registration Rights Agreement on the date thereof.

Section 2.03. For the avoidance of doubt, nothing herein shall or shall be deemed to modify or otherwise affect the obligations of the Company or the Existing Guarantors under the Notes, the Indenture or the Registration Rights Agreement, and the Company shall remain liable, jointly and severally with Parent, for all obligations of an "Issuer" thereunder.

ARTICLE 3

Section 3.01. Each of the New Guarantors hereby agrees to be bound by the terms, conditions and other provisions of the Indenture with all attendant rights, duties and obligations stated therein, on a joint and several basis with the parties hereto and thereto, with the same force

and effect as if originally named as a Guarantor therein and as if such party executed the Indenture on the date thereof.

ARTICLE 4

Section 4.01. Except as specifically modified herein, the Indenture, the Notes and the Registration Rights Agreement are in all respects ratified and confirmed (mutatis mutandis) and shall remain in full force and effect in accordance with their terms.

Section 4.02. All capitalized terms used but not defined herein shall have the same respective meanings ascribed to them in the Indenture.

Section 4.03. Except as otherwise expressly provided herein, no duties, responsibilities or liabilities are assumed, or shall be construed to be assumed, by the Trustee by reason of this Supplemental Indenture. This Supplemental Indenture is executed and accepted by the Trustee subject to all of the terms and conditions set forth in the Indenture with the same force and effect as if those terms and conditions were repeated at length herein and made applicable to the Trustee with respect hereto.

Section 4.04. THIS SUPPLEMENTAL INDENTURE SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

Section 4.05. The parties may sign any number of copies of this Supplemental Indenture. Each signed copy shall be an original, but all of them together represent the same agreement.

Section 4.06. The headings herein are inserted for convenience of reference only and are not intended to be part of, or to affect the meaning or interpretation of, this Supplemental Indenture.

Section 4.07. The recitals hereto are statements only of Parent, the Original Issuers, the Existing Guarantors and the New Guarantors and shall not be considered statements of or attributable to the Trustee.

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have caused this Supplemental Indenture to be duly executed and attested, all as of the date first above written.

PARENT:

MOBILE MINI, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

COMPANY:

MOBILE STORAGE GROUP, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

EXISTING GUARANTORS:

A BETTER MOBILE STORAGE COMPANY

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE STORAGE GROUP (TEXAS), LP

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

[Signature Page to Supplemental Indenture to Mobile Storage Group, Inc. Indenture]

NEW GUARANTORS:

MOBILE MINI OF OHIO LLC

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE MINI, LLC (CA)

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE MINI, LLC (DE)

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE MINI I, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

[Signature Page to Supplemental Indenture to Mobile Storage Group, Inc. Indenture]

A ROYAL WOLF PORTABLE STORAGE, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

TEMPORARY MOBILE STORAGE, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

DELIVERY DESIGN SYSTEMS, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE MINI TEXAS LIMITED PARTNERSHIP, LLP

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

[Signature Page to Supplemental Indenture to Mobile Storage Group, Inc. Indenture]

TRUSTEE:

WELLS FARGO BANK, N.A., as Trustee

By: /s/ Jane Y. Schweiger

Name: Jane Y. Schweiger Title: Vice President

[Signature Page to Supplemental Indenture to Mobile Storage Group, Inc. Indenture]

SUPPLEMENTAL INDENTURE

THIS SUPPLEMENTAL INDENTURE (this "<u>Supplemental Indenture</u>"), dated as of June 27, 2008, among Mobile Mini, Inc., a Delaware corporation (the "<u>Issuer</u>"), Mobile Storage Group, Inc., a Delaware corporation, A Better Mobile Storage Company, a California corporation, Mobile Storage Group (Texas), LP, a Texas limited partnership (each a "<u>New Guarantor</u>" and collectively, the "<u>New Guarantors</u>"), the Guarantors party to the Indenture (as defined below) (the "<u>Existing Guarantors</u>") and Law Debenture Trust Company of New York, as Trustee.

WITNESSETH:

WHEREAS, the Issuer, the Existing Guarantors and the Trustee entered into an Indenture (the "<u>Indenture</u>"), dated as of May 7, 2007, pursuant to which the Issuer issued \$150,000,000 in principal amount of 67/8% Senior Notes due 2015 (the "Notes");

WHEREAS, Section 4.16 of the Indenture provides that under certain circumstances new Restricted Subsidiaries are required to execute and deliver to the Trustee a supplemental indenture pursuant to which such Restricted Subsidiary shall unconditionally guarantee all of the Issuer's Obligations under the Notes and the Indenture pursuant to a Guarantee on the terms and conditions set forth in the Indenture;

WHEREAS, Section 9.l(d) of the Indenture provides that the Issuer, the Guarantors and the Trustee may amend or supplement the Indenture in order to make any change that would provide any additional rights or benefits to the Holders of the Notes or that does not adversely affect the rights thereunder of any Holder of the Notes in any material respect, without the consent of the Holders of the Notes; and

WHEREAS, all acts and things prescribed by the Indenture, by law and by the certificate of incorporation and the bylaws (or comparable constituent documents) of the parties hereto necessary to make this Supplemental Indenture a valid instrument legally binding on each of the parties hereto, in accordance with its terms, have been duly done and performed;

NOW THEREFORE, to comply with the provisions of the Indenture, and in consideration of the foregoing, the parties hereto mutually covenant and agree for the equal and ratable benefit of the Holders of the Notes as follows:

ARTICLE 1

Section 1.01. This Supplemental Indenture is supplemental to the Indenture and does and shall be deemed to form a part of, and shall be construed in connection with and as part of, the Indenture for any and all purposes.

Section 1.02. This Supplemental Indenture shall become effective immediately upon its execution and delivery by each of the parties hereto.

ARTICLE 2

Section 2.01. Each of the New Guarantors hereby agrees to be bound by the terms, conditions and other provisions of the Indenture with all attendant rights, duties and obligations stated therein, on a joint and several basis with the parties hereto and thereto, with the same force and effect as if originally named as a Guarantor therein and as if such party executed the Indenture on the date thereof.

ARTICLE 3

Section 3.01. Except as specifically modified herein, the Indenture and the Notes are in all respects ratified and confirmed (mutatis mutandis) and shall remain in full force and effect in accordance with their terms.

Section 3.02. All capitalized terms used but not defined herein shall have the same respective meanings ascribed to them in the Indenture.

Section 3.03. Except as otherwise expressly provided herein, no duties, responsibilities or liabilities are assumed, or shall be construed to be assumed, by the Trustee by reason of this Supplemental Indenture. This Supplemental Indenture is executed and accepted by the Trustee subject to all of the terms and conditions set forth in the Indenture with the same force and effect as if those terms and conditions were repeated at length herein and made applicable to the Trustee with respect hereto.

Section 3.04. THIS SUPPLEMENTAL INDENTURE SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

Section 3.05. The parties may sign any number of copies of this Supplemental Indenture. Each signed copy shall be an original, but all of them together represent the same agreement.

Section 3.06. The headings herein are inserted for convenience of reference only and are not intended to be part of, or to affect the meaning or interpretation of, this Supplemental Indenture.

Section 3.07. The recitals hereto are statements only of the Issuer, the Existing Guarantors and the New Guarantors and shall not be considered statements of or attributable to the Trustee.

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have caused this Supplemental Indenture to be duly executed and attested, all as of the date first above written.

ISSUER:

MOBILE MINI, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

EXISTING GUARANTORS:

MOBILE MINI OF OHIO LLC

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE MINI, LLC (CA)

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE MINI, LLC (DE)

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE MINI I, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

A ROYAL WOLF PORTABLE STORAGE, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

TEMPORARY MOBILE STORAGE, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

DELIVERY DESIGN SYSTEMS, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE MINI TEXAS LIMITED PARTNERSHIP, LLP

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

NEW GUARANTORS:

MOBILE STORAGE GROUP, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

A BETTER MOBILE STORAGE COMPANY

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

MOBILE STORAGE GROUP (TEXAS), LP

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President and
Chief Financial Officer

TRUSTEE:

LAW DEBENTURE TRUST COMPANY OF NEW YORK, as Trustee

By: /s/ James D. Heaney

Name: James D. Heaney
Title: Vice President

ESCROW AGREEMENT

This Escrow Agreement (this "<u>Escrow Agreement</u>"), dated as of June , 2008, is entered into by and among Mobile Mini, Inc., a Delaware corporation ("<u>Parent</u>"), Welsh, Carson, Anderson & Stowe X, L.P., a Delaware limited partnership, as Target Stockholder Representative (the "<u>Stockholder Representative</u>", together with Parent, the "<u>Other Parties</u>"), and Wells Fargo Bank, N.A., as escrow agent (the "<u>Escrow Agent</u>").

WITNESSETH:

WHEREAS, reference is hereby made to that certain Agreement and Plan of Merger dated February 22, 2008 (the "Merger Agreement"), entered into by and among Parent, MSG WC Holdings Corp., a Delaware corporation ("Target"), and the Stockholder Representative, solely for the purposes of representing the Target Stockholders in accordance with the Merger Agreement, that provides for, among other transactions, the merger of Target with and into the Parent, with the Parent continuing as the surviving company, as more fully set forth in the Merger Agreement. Capitalized terms used but not otherwise defined herein shall have respective the meanings given to such terms in the Merger Agreement, a copy of which is attached hereto as Exhibit C;

WHEREAS, Section 3.3 of the Merger Agreement provides that the Merger Consideration is subject to certain post-Closing adjustments;

WHEREAS, Article XI of the Merger Agreement provides that the Target Stockholders are obligated to indemnify Parent Indemnitee under certain circumstances;

WHEREAS, pursuant to Section 3.3(d) of the Merger Agreement, (i) an amount of \$2,811,659.79 in cash (the "Escrow Amount", and together with all interest and other amounts from time to time held by the Escrow Agent under this Escrow Agreement, the "Escrow Funds")) and (ii) 677,130.01 shares of Series A Convertible Redeemable Participating Preferred Stock, par value \$0.01 per share, of Parent (the "Shares", and together with all other shares issued on, in respect of, upon conversion of, or in substitution of the Shares according to Section 4.4, the "Escrowed Shares")) are being delivered to the Escrow Agent on the date of this Escrow Agreement;

WHEREAS, the Escrow Funds and the Escrowed Shares, collectively, are referred to herein as the "Escrowed Property"; and

WHEREAS, the Other Parties have requested the Escrow Agent to act in the capacity of escrow agent under this Escrow Agreement, and the Escrow Agent, subject to the terms and conditions hereof, has agreed so to do.

NOW, THEREFORE, in consideration of the premises and mutual covenants and agreements contained herein, the parties hereto hereby agree as follows:

ARTICLE I RECEIPT AND RETENTION OF THE ESCROW PROPERTY

Section 1.1 The Other Parties each hereby appoints the Escrow Agent to act as agent and custodian for the Escrowed Property for their respective benefit pursuant to the terms of this Escrow Agreement, and the Escrow Agent hereby accepts such appointment pursuant to such terms.

Section 1.2 The Escrow Agent hereby acknowledges receipt of the Escrowed Property.

Section 1.3 The Escrowed Property shall be retained, managed and disbursed by the Escrow Agent strictly in accordance with the terms and conditions of this Escrow Agreement. The Escrow Agent shall not transfer, assign, loan, convert, reinvest, or otherwise dispose of any Escrowed Property but shall hold the Escrowed Property in accordance with this Escrow Agreement until the disbursement thereof pursuant to Article IV or Section 5.6.

ARTICLE II INVESTMENT

Section 2.1 At the written direction of the Stockholder Representative, pursuant to the attached Exhibit D, the Escrow Agent will invest (and reinvest) the Escrow Funds in one or more of: (i) direct obligations of and obligations fully guaranteed by the United States of America, or any agency thereof, the principal and interest of which are guaranteed by the United States of America or its agencies; (ii) participation under a revolving repurchase agreement maintained by the Escrow Agent with other entities relative to an agreement for the sale and repurchase of obligations listed in item (i) above; (iii) any time deposit which is fully insured by the Federal Deposit Insurance Corporation; (iv) commercial paper notes which, at the time of investment, are rated in one of the two highest credit ratings by Moody's Investors Service, Inc. and/or Standard & Poor's Corporation; (v) certificates of deposit of any bank organized under the laws of the United States; or (vi) any money market fund (including money market funds for which the Escrow Agent serves in an advisory capacity and/or other money market funds with which the Escrow Agent has an existing relationship), the assets of which are any of those obligations listed in items (i) through (vi) above (collectively, the "Permitted Investments").

Section 2.2 The Escrow Agent is hereby authorized to execute the purchase and sale of Permitted Investments as directed by the Stockholder Representative, in writing, through the facilities of its own trading or capital markets operations or those of any affiliated entity. In the event that the Escrow Agent does not receive investment instructions to invest the Escrow Funds, the Escrow Agent shall invest such funds in direct obligations of and obligations fully guaranteed by the United States of America, or any agency thereof, the principal and interest of which are guaranteed by the United States of America or its agencies. The Escrow Agent can liquidate any investment in order to comply with disbursement instructions without any liability for any resulting loss. Any loss incurred from an investment will be borne by the Escrow Funds. The Stockholder Representative and the Parent acknowledge that the Escrow Agent is not providing investment supervision, recommendations or advice.

Section 2.3 All income earned with respect to any Permitted Investment shall be delivered from time to time, but at least quarterly, to the Stockholder Representative for distribution to the Target Stockholders.

ARTICLE III TAXES

Section 3.1 The Other Parties agree that, for United States federal, state, local and other tax purposes, all taxable interest, dividends and other income, if any, attributable to the Escrowed Property or any other amount held in escrow by the Escrow Agent pursuant to this Escrow Agreement shall be allocable to the Target Stockholders.

Section 3.2 The Escrow Agent shall report to the United States Internal Revenue Service (the "<u>IRS</u>") and to each Target Stockholder, as of each calendar year-end, all income, if any, attributable to the Escrowed Property or any other amount held in escrow by the Escrow Agent pursuant to this Escrow Agreement, as and to the extent required under the provisions of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>"), on IRS Form 1099 or other applicable form.

Section 3.3 The Escrow Agent shall make payments of income earned on the Escrow Funds as provided for in this Escrow Agreement. Prior to closing, each such payee shall provide to the Escrow Agent an IRS Form W-9, for tax identification number certification, or an applicable Form W-8, for certification that such payee is not a "United States person" within the meaning of the Code, as appropriate and such other forms and documents as the Escrow Agent may request. The Other Parties and the Target Stockholders understand that if such tax reporting documentation is not provided and certified to the Escrow Agent, the Escrow Agent may be required by the Code, and the Regulations promulgated thereunder, to withhold a portion of any interest or other income earned on the investment of the Escrowed Property.

Section 3.4 The Other Parties intend that the Target Stockholders shall be treated as the owners of the Escrowed Shares for all United States federal, state and local tax purposes (except to the extent that any such Escrowed Shares are disbursed to Parent pursuant to <u>Article IV</u>), and none of the parties shall take any actions or positions that are inconsistent with such treatment.

ARTICLE IV DISBURSEMENT OF THE ESCROW PROPERTY; SUBSTITUTION

Section 4.1 The Escrow Agent shall make disbursements of the Escrowed Property upon receipt of and in accordance with (a) a joint written instruction ("<u>Joint Instruction</u>") signed by authorized signers of both Parent and the Stockholder Representative evidenced in Exhibits B-1 and B-2, respectively, directing delivery of all or a portion of the Escrowed Property, or (b) an order from a court of competent jurisdiction that orders such disbursement, together with an opinion of counsel to the effect that such order is final and not subject to further appeal (collectively, the "<u>Court Order</u>").

Section 4.2 Notwithstanding anything herein to the contrary:

- (a) To the extent the Escrowed Property has not been fully disbursed on or prior to the date that is twelve (12) months from the date of this Escrow Agreement (the "Escrow Release Date"), the Escrow Agent shall disburse within two (2) Business Days after such date to the Stockholder Representative the remaining Escrowed Property minus the Retained Property. The Escrow Agent shall retain the Retained Property and shall only release the Retained Property in accordance with Section 4.1 above.
 - (b) For purposes of this Article IV, the following terms shall have the following meanings:

"Retained Property" means the amount of cash and/or the number Shares set forth in a certificate of the Chief Financial Officer of Parent delivered to the Escrow Agent and the Target Shareholder Representative at least two (2) Business Days prior to the Escrow Release Date, which certificate shall set forth Parent's good faith estimate of the aggregate amount of all pending indemnification claims of Parent under the Merger Agreement as of the date of such certificate. Such certificate shall contain Parent's good faith calculation of the Fair Market Value (as defined in the Merger Agreement) of the Shares as of the date of such certificate.

Section 4.3 Whenever a partial release of the Escrowed Shares is anticipated, Parent shall, upon request by the Escrow Agent, promptly cause the Parent's Secretary to reissue the stock certificates of the Escrowed Shares in the number of certificates necessary so as to enable such partial release.

ARTICLE V PROVISIONS RELATING TO THE ESCROW AGENT

Section 5.1 The Escrow Agent shall have no duties or responsibilities whatsoever with respect to the Escrowed Property except as are specifically set forth herein. The Escrow Agent shall neither be responsible for or under, nor chargeable with knowledge of the terms and conditions of, any other agreement, instrument or document in connection herewith, including without limitation the Merger Agreement. The Escrow Agent may conclusively rely upon, and shall be fully protected from all liability, loss, cost, damage or expense in acting or omitting to act pursuant to any written notice, instrument, request, consent, certificate, document, letter, opinion, order, resolution or other writing hereunder that is in a form and manner consistent with the requirements set forth in this Escrow Agreement without being required to determine the authenticity of such document, the correctness of any fact stated therein, the propriety of the service thereof or the capacity, identity or authority of any party purporting to sign or deliver such document. The Escrow Agent shall have no responsibility for the contents of any such writing contemplated herein and may rely without any liability upon the contents thereof. Concurrently with the execution of this Escrow Agreement, the Other Parties shall deliver to the Escrow Agent authorized signers' forms in the form of Exhibit B-1 and Exhibit B-2 to this Escrow Agreement.

The Escrow Agent shall not be liable for any action taken or omitted by it in good faith and reasonably believed by it to be authorized hereby or within the rights or powers conferred upon it

hereunder, nor for action taken or omitted by it in good faith, and in accordance with advice of counsel (which counsel may be of the Escrow Agent's own choosing), and shall not be liable for any mistake of fact or error of judgment or for any acts or omissions of any kind except for its own willful misconduct or gross negligence. The Parties, jointly and severally, shall indemnify, defend and hold harmless the Escrow Agent from and against any and all loss, liability, cost, damage and expense, including, without limitation, attorneys' fees and expenses or other professional fees and expenses which the Escrow Agent may suffer or incur by reason of any action, claim or proceeding brought against the Escrow Agent, arising out of or relating in any way to this Escrow Agreement or any transaction to which this Escrow Agreement relates, unless such loss, liability, cost, damage or expense shall have been finally adjudicated to have been directly caused by the willful misconduct or gross negligence of the Escrow Agent. The provisions of this Section 5.1 shall survive the resignation or removal of the Escrow Agent and the termination of this Escrow Agreement.

THE ESCROW AGENT SHALL NOT BE LIABLE, DIRECTLY OR INDIRECTLY, FOR ANY (I) DAMAGES, LOSSES OR EXPENSES ARISING OUT OF THE SERVICES PROVIDED HEREUNDER, OTHER THAN DAMAGES, LOSSES OR EXPENSES WHICH HAVE BEEN FINALLY ADJUDICATED TO HAVE DIRECTLY RESULTED FROM THE ESCROW AGENT'S GROSS NEGLIGENCE OR WILLFUL MISCONDUCT, OR (II) SPECIAL, INDIRECT OR CONSEQUENTIAL DAMAGES OR LOSSES OF ANY KIND WHATSOEVER (INCLUDING WITHOUT LIMITATION LOST PROFITS), EVEN IF THE ESCROW AGENT HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH LOSSES OR DAMAGES AND REGARDLESS OF THE FORM OF ACTION.

Section 5.2 The Escrow Agent may resign as such following the giving of sixty (60) days' prior written notice to each of the Other Parties. Similarly, the Escrow Agent may be removed and replaced following the giving of sixty (60) days' prior written notice to the Escrow Agent jointly by the Other Parties. In either event, the duties of the Escrow Agent shall terminate sixty (60) days after the date of such notice (or at such earlier date as may be mutually agreeable), except for its obligations to hold and deliver the Escrowed Property to the successor Escrow Agent; and the Escrow Agent shall then deliver the balance of the Escrowed Property then in its possession to such a successor Escrow Agent as shall be appointed by the Other Parties as evidenced by Joint Instructions delivered to the Escrow Agent. Upon acknowledgment by any successor Escrow Agent of the receipt of the then remaining Escrowed Property (or the deposit or delivery of the Escrowed Property by the Escrow Agent pursuant to Section 5.4 below), the then acting Escrow Agent shall be fully released and relieved of all duties, responsibilities and obligations under this Escrow Agreement.

Section 5.3 The Escrow Agent shall be under no duty to institute or defend any arbitration or legal proceeding with respect to the Escrowed Property or under this Escrow Agreement and none of the costs or expenses of any such proceeding shall be borne by the Escrow Agent. The costs and expenses of any such proceeding shall be borne as decided by the arbitrators or court and shall be direct obligations of the Other Parties, as the case may be, and shall not be satisfied in any way by the Escrowed Property.

Section 5.4 Should any controversy arise involving the parties hereto or any of them or any other person with respect to this Escrow Agreement or the Escrowed

Property, or should a substitute escrow agent fail to be designated as provided in Section 5.2 hereof, or if the Escrow Agent should be in doubt as to what action to take, the Escrow Agent shall have the right, but not the obligation, either to (a) withhold delivery of the Escrowed Property until the controversy is resolved, the conflicting demands are withdrawn or its doubt is resolved, (b) deposit all Escrowed Property into the registry of any court of competent jurisdiction and notify the Other Parties of such deposit, and thereupon the Escrow Agent shall be discharged from all further duties and responsibilities as Escrow Agent under this Escrow Agreement or (c) file an interpleader action in any court of competent jurisdiction, and upon the filing thereof and deposit of all Escrowed Property into the registry of such court, the Escrow Agent shall be relieved of all liability as to the Escrowed Property and shall be entitled to recover from the losing party of such action reasonable attorneys' fees, expenses and other costs incurred in commencing and maintaining any such interpleader action.

ARTICLE VI NOTICES

Section 6.1 All notices, consents, requests, demands, waivers and other communications required or permitted to be given under this Escrow Agreement shall be in writing and shall be deemed to have been duly given if delivered in person or mailed, certified or registered mail with postage prepaid, or sent by reputable overnight courier or facsimile (upon confirmation of receipt), as follows:

(a) If to the Escrow Agent:

Wells Fargo Bank, N.A. Corporate Municipal & Escrow Solutions MAC S4101-22E 100 West Washington Street, 22nd Floor Phoenix, AZ 85003 Phone: 602-378-2305

(b) If to Parent:

Mobile Mini, Inc. 7420 South Kyrene Road, Suite 101 Tempe, AZ 85283 Fax: (480) 894-6433 Attn: Larry Trachtenberg

Facsimile: 602-378-2333

with a copy (which shall not constitute notice) to:

White & Case LLP 1155 Avenue of the Americas New York, NY 10036 Fax: (212) 354-8113

Attn: John M. Reiss, Esq.

Daniel M. Latham, Esq.

(c) If to the Stockholder Representative:

Welsh, Carson, Anderson & Stowe X, L.P. 320 Park Avenue Suite 2500

New York, NY 10022 Fax: (212) 893-9575 Attn: Sanjay Swani Michael Donovan

with a copy (which shall not constitute notice) to:

Kirkland & Ellis LLP 153 East 53rd Street New York, NY 10022 Fax: (212) 446-6460

Attn: Michael Movsovich, Esq.

or to such other person or address as any party shall specify by notice in writing in accordance with this <u>Section 6.1</u> to each of the other parties. All such notices, requests, demands, waivers and communications shall be deemed to have been received on the date of delivery unless if mailed, in which case on the third (3rd) Business Day after the mailing thereof, except for a notice of a change of address, which shall be effective only upon receipt thereof.

ARTICLE VII COMPENSATION; EXPENSES

Section 7.1 The Escrow Agent shall be entitled to payment of customary fees and expenses for all services rendered by it hereunder in accordance with Exhibit A attached hereto (as such schedule may be amended from time to time), payable on the Closing Date. After the Closing Date, the Escrow Agent shall be entitled to annual fees in accordance with Exhibit A. All fees and expenses owed to the Escrow Agent pursuant to this Article VII shall be paid 50% by Parent and 50% by the Stockholder Representative.

ARTICLE VIII TERM

Section 8.1 This Escrow Agreement shall terminate upon the disbursement, in accordance with <u>Article IV</u> or <u>Section 5.6</u> hereof, of the Escrowed Property in full; <u>provided</u>, <u>however</u>, that the rights of the Escrow Agent and the obligations of the other parties hereto under <u>Article VII</u> and <u>Article VII</u> shall survive the termination thereof and the resignation or removal of the Escrow Agent.

ARTICLE IX DESIGNEES FOR INSTRUCTIONS

Section 9.1 An Other Party may, by notice to the Escrow Agent and the other Other Party, designate one or more persons who will execute notices and from whom the Escrow Agent may take instructions hereunder. Such designations may be changed from time to time upon written notice to the Escrow Agent from such Other Party. The Escrow Agent will be entitled to rely conclusively on any notices or instructions from any person so designated by an Other Party.

ARTICLE X MISCELLANEOUS

Section 10.1 No amendment, modification or waiver in respect of this Escrow Agreement shall be effective unless it shall be in writing and signed by all parties hereto.

Section 10.2 This Escrow Agreement and the rights and obligations hereunder shall not be assignable or transferable by a party hereto without the prior written consent of the other parties hereto; provided, however, that any banking association or corporation into which the Escrow Agent may be merged, converted or with which the Escrow Agent may be consolidated, or any corporation resulting from any merger, conversion or consolidation to which the Escrow Agent shall be a party, or any banking association or corporation to which all or substantially all of the corporate trust business of the Escrow Agent shall be transferred, shall succeed to all the Escrow Agent's rights, obligations and immunities hereunder without the execution or filing of any paper or any further act on the part of any of the parties hereto, anything herein to the contrary notwithstanding. Any attempted assignment in violation of this Section 10.2 shall be void *ab initio*.

Section 10.3 If any term, provision, covenant or restriction contained in this Escrow Agreement is held by a court of competent jurisdiction or other authority to be invalid, void, unenforceable or against its regulatory policy, the remainder of the terms, provisions, covenants and restrictions contained in this Escrow Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated, and this Escrow Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable term, provision, covenant or restriction or any portion thereof had never been contained herein.

Section 10.4 This Escrow Agreement, including the Exhibits hereto and the documents and instruments and other agreements specifically referred to herein or delivered pursuant hereto, contains the entire understanding of the parties hereto with respect to the subject matter contained herein and supersedes all prior agreements and understandings, oral and written, with respect thereto. The Escrow Agent is bound only by the terms of this Escrow Agreement.

Section 10.5 THIS ESCROW AGREEMENT AND THE LEGAL RELATIONS BETWEEN THE PARTIES HERETO SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.

THE STATE OR FEDERAL COURTS LOCATED WITHIN THE STATE OF NEW YORK SHALL HAVE EXCLUSIVE JURISDICTION OVER ANY AND ALL DISPUTES BETWEEN THE PARTIES HERETO, WHETHER IN LAW OR EQUITY, ARISING OUT OF OR RELATING TO THIS ESCROW AGREEMENT AND THE PARTIES CONSENT TO AND AGREE TO SUBMIT TO THE EXCLUSIVE JURISDICTION OF SUCH COURTS. EACH OF THE PARTIES HEREBY WAIVES AND AGREES NOT TO ASSERT IN ANY SUCH DISPUTE, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY CLAIM THAT (I) SUCH PARTY IS NOT PERSONALLY SUBJECT TO THE JURISDICTION OF SUCH COURTS, (II) SUCH PARTY AND SUCH PARTY'S PROPERTY IS IMMUNE FROM ANY LEGAL PROCESS ISSUED BY SUCH COURTS OR (III) ANY LITIGATION OR OTHER PROCEEDING COMMENCED IN SUCH COURTS IS BROUGHT IN AN INCONVENIENT FORUM. THE PARTIES HEREBY AGREE THAT MAILING OF PROCESS OR OTHER PAPERS IN CONNECTION WITH ANY SUCH ACTION OR PROCEEDING IN THE MANNER PROVIDED IN SECTION 6.1, OR IN SUCH OTHER MANNER AS MAY BE PERMITTED BY LAW, SHALL BE VALID AND SUFFICIENT SERVICE THEREOF AND HEREBY WAIVE ANY OBJECTIONS TO SERVICE ACCOMPLISHED IN THE MANNER HEREIN PROVIDED.

Section 10.6 Each of the parties to this Escrow Agreement hereby irrevocably waives all right to a trial by jury in any action, proceeding or counterclaim arising out of or relating to this Escrow Agreement or the transactions contemplated hereby.

Section 10.7 This Escrow Agreement shall be binding on and shall inure to the benefit of the parties hereto and their successors and permitted assigns. This Escrow Agreement is for the sole benefit of the parties hereto and their successors and permitted assigns and nothing herein expressed or implied shall give or be construed to give to any person, other than the parties hereto and such successors and assigns, any legal or equitable rights hereunder. Nothing herein is intended or shall be construed to give any other person (including, without limitation, any creditors of the Escrow Agent or the Other Parties) any right, remedy or claim under, in or with respect to the Escrowed Property held hereunder.

Section 10.8 This Escrow Agreement may be executed in one or more counterparts (including by facsimile or electronic means), all of which shall be considered one and the same agreement, and shall become effective when one or more such counterparts have been signed by each of the parties and delivered to the other party.

Section 10.9 Construction. In this Escrow Agreement, unless the context otherwise requires:

- (a) any reference in this Agreement to "writing" or comparable expressions includes a reference to facsimile transmission or comparable means of communication:
- (b) words expressed in the singular number shall include the plural and vice versa, words expressed in the masculine shall include the feminine and neuter gender and vice versa;
 - (c) references to Articles, Sections, Exhibits and Recitals are references to articles, sections, exhibits, schedules and recitals of this Agreement;

- (d) reference to "day" or "days" are to calendar days;
- (e) this "Escrow Agreement" or any other agreement or document shall be construed as a reference to this Agreement or, as the case may be, such other agreement or document as the same may have been, or may from time to time be, amended, varied, novated or supplemented; and
- (f) "include," "includes," and "including" are deemed to be followed by "without limitation" whether or not they are in fact followed by such words or words of similar import.

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Escrow Agreement as of the day and year first above written.

ESCROW AGENT:

Wells Fargo Bank, as Escrow Agent

By: /s/ Jeffrey B. Kassels
Name: Jeffrey B. Kassels
Title: Vice President

PARENT:

Mobile Mini, Inc.

By: /s/ Lawrence Trachtenberg
 Name: Lawrence Trachtenberg
 Title: Executive Vice President

STOCKHOLDER REPRESENTATIVE:

Welsh, Carson, Anderson & Stowe X, L.P.

By: WCAS X Associates LLC, its General Partner

By: /s/ Sanjay Swani
Name: Sanjay Swani
Title: Vice President

[SIGNATURE PAGE TO ESCROW AGREEMENT]

EXHIBIT A

Escrow Agent's Fee Schedule to act as ESCROW AGENT for the Mobile Mini, Inc. Indemnity Escrow

Acceptance Fee: \$750.00

Initial Fees as they relate to Wells Fargo Bank acting in the capacity of Escrow Agent — includes review of the Escrow Agreement; acceptance of the Escrow appointment; setting up of Escrow Account(s) and accounting records; and coordination of receipt of funds for deposit to the Escrow Account(s).

Acceptance Fee payable at time of Escrow Agreement execution.

Escrow Agent Annual Administration Fee:

\$1,500.00

For ordinary administrative services by Escrow Agent — includes daily routine account management; investment transactions; cash transaction processing (including wire and check processing); monitoring claim notices pursuant to the agreement; disbursement of funds in accordance with the agreement; and mailing of trust account statements to all applicable parties.

Tax reporting is included for up to Five (5) entities. Should additional reporting be necessary, a \$25 per reporting charge will be assessed.

This fee is payable in advance, with the first installment due at the time of Escrow Agreement execution. The Annual Fee covers a full year or any part thereof, and therefore will not be prorated or refunded in the year of early termination.

Wells Fargo's bid is based on the following assumptions:

- Number of Escrow Accounts to be established: One (1)
- Number of Deposits to Escrow Account: Not more than One (1)
- Number of Withdrawals from Escrow Fund: Not more than Ten (10)
- Term of Escrow: Not more than One (1) year
- APPOINTMENT SUBJECT TO RECEIPT OF REQUESTED DUE DILIGENCE INFORMATION AS PER THE USA PATRIOT ACT
- THIS PROPOSAL ASSUMES THAT BALANCES IN ESCROW ACCOUNT WILL BE INVESTED IN WELLS FARGO ADVANTAGE MONEY MARKET FUNDS
- · ALL FUNDS WILL BE RECEIVED FROM OR DISTRIBUTED TO A DOMESTIC OR AN APPROVED FOREIGN ENTITY
- IF THE ACCOUNT(S) DOES NOT OPEN WITHIN THREE (3) MONTHS OF THE DATE SHOWN BELOW, THIS PROPOSAL WILL BE DEEMED TO BE NULL AND VOID

Out-of Pocket Expenses: At Cost

We will charge for out-of-pocket expenses in response to specific tasks assigned by the client or provided for in the escrow agreement. Possible expenses would be, but are not limited to, express mail and messenger charges, travel expenses to attend closing or other meetings. There are no charges for indirect out-of- pocket expenses.

This fee schedule is based upon the assumptions listed above which pertain to the responsibilities and risks involved in Wells Fargo undertaking the role of Escrow Agent. These assumptions are based on information provided to us as of the date of this fee schedule. Our fee schedule is subject to review and acceptance of the final documents. Should any of the assumptions, duties or responsibilities change, we reserve the right to affirm, modify or rescind our fee schedule. Extraordinary services (services other than the ordinary administration services of Escrow Agent described above) are not included in the annual administration fee and will be billed as incurred at the rates in effect from time to time.

Submitted on: 06/09/08

EXHIBIT B-1 CERTIFICATE AS TO AUTHORIZED SIGNATURES OF PARENT

Account Name:

The specimen signatures shown below are the specimen signatures of the individuals who have been designated as Authorized Representatives of Mobile Mini, Inc. and are authorized to initiate and approve transactions of all types for the above-mentioned account on behalf of Parent.

| Name / Title | Specimen Signature | | |
|-------------------------|------------------------|--|--|
| Larry Trachtenberg | /s/ Larry Trachtenberg | | |
| Name | Signature | | |
| Chief Financial Officer | | | |
| Title | | | |
| Christopher Miner | /s/ Christopher Miner | | |
| Name General Counsel | Signature | | |
| Title | | | |
| | | | |
| | | | |

EXHIBIT B-2 CERTIFICATE AS TO AUTHORIZED SIGNATURES OF STOCKHOLDER REPRESENTATIVE

Account Name:

The specimen signatures shown below are the specimen signatures of the individuals who have been designated as the Authorized Representatives of the Stockholder Representative and are authorized to initiate and approve transactions of all types for the above-mentioned account on behalf of the Stockholder Representative.

| | Specimen Signature |
|-------------------|---------------------|
| Jonathan Rather | /s/ Jonathan Rather |
| Name | Signature |
| Managing Member | |
| Title | |
| Sanjay Swani | /s/ Sanjay Swani |
| Name | Signature |
| Managing Member | |
| Title | |
| Michael Donovan | /s/ Michael Donovan |
| Name | Signature |
| Additional Member | |
| Title | |
| Name | Signature |
| Title | |
| | |
| | |

EXHIBIT C

Agreement and Plan of Merger, dated as of February 22, 2008, among Mobile Mini, Inc., Cactus Merger Sub, Inc., MSG WC Holdings Corp. and Welsh, Carson, Anderson & Stowe X, L.P.

Incorporated by reference to Exhibit 2.1 to Mobile Mini, Inc.'s Form 8-K filed on February 28, 2008.

EXHIBIT D INVESTMENT DIRECTION

Direction for Investment of Cash Balances

Wells Fargo Advantage Funds

Direction to use Wells Fargo Advantage Funds for Cash Balances for the following account(s):

Account Name:

Account Number(s):

You are hereby directed to invest, as indicated below or as I shall direct further from time to time, all cash in the Account in the following money market portfolio of Wells Fargo Advantage Funds (the "Fund") or another permitted investment of my choice (Check One):

- o Wells Fargo Advantage Funds, 100% Treasury Money Market Fund
- ☑ Wells Fargo Advantage Funds, Government Money Market Fund
- o Wells Fargo Advantage Funds, Cash Investment Money Market Fund
- o Wells Fargo Advantage Funds, Prime Investment Money Market Fund
- o Wells Fargo Advantage Funds, Treasury Plus Money Market Fund
- o Wells Fargo Advantage Funds, Heritage Money Market Fund
- o Wells Fargo Advantage Funds, National Tax-Free Money Market Fund

I acknowledge that I have received, at my request, and reviewed the Fund's prospectus and have determined that the Fund is an appropriate investment for the Account. Each Fund's prospectus can be downloaded from the Wells Fargo website at the following link:

http://www.wellsfargo.com/funds/fmg_fund/fund_type/fundtype.jhtml?fundType=MoneyMarket&tab=literature

I understand from reading the Fund's prospectus that Wells Fargo Funds Management, LLC, ("Wells Fargo Funds Management"), a wholly-owned subsidiary of Wells Fargo & Company, provides investment advisory and other administrative services for the *Wells Fargo Advantage Funds*. Other affiliates of Wells Fargo & Company provide sub-advisory and other services for the Funds. Boston Financial Data Services serves as transfer agent for the Funds. The Funds are distributed by Wells Fargo Funds Distributor, LLC, Member NASD/SIPC, an affiliate of Wells Fargo & Company. I also understand that Wells Fargo & Company will be paid, and its bank affiliates may be paid, fees for services to the Funds and that those fees may include Processing Organization fees as described in the Fund's prospectus.

I understand that you will not exclude amounts invested in the Fund from Account assets subject to fees under the Account agreement between us.

I understand that investments in the Fund are not obligations of, or endorsed or guaranteed by, Wells Fargo Bank or its affiliates and are not insured by the Federal Deposit Insurance Corporation.

I acknowledge that I have full power to direct investments of the Account.

I understand that I may change this direction at any time and that it shall continue in effect until revoked or modified by me by written notice to you.

I understand that if I choose to communicate this investment direction solely via facsimile, then the investment direction will be understood to be enforceable and binding.

| Signature | | |
|-----------|--|--|
| Date | | |

STOCKHOLDERS AGREEMENT

Dated June 27, 2008

By and Among

MOBILE MINI, INC.

and

THE STOCKHOLDERS SIGNATORY HERETO

TABLE OF CONTENTS

| ARTICLE I CERTAI | IN DEFINITIONS | <u>Page</u> 1 |
|-------------------|-------------------------------------|-------------------------|
| Section 1.1 | Certain Definitions | 1 |
| ARTICLE II TRANS | SFER OF EQUITY SECURITIES | 7 |
| Section 2.1 | Restrictions | 7 |
| Section 2.2 | Permitted Transfers | 8 |
| Section 2.3 | Standstill | 10 |
| ARTICLE III REGIS | STRATION RIGHTS | 11 |
| Section 3.1 | Required Registrations | 11 |
| Section 3.2 | Incidental Registration | 13 |
| Section 3.3 | Registration Procedures | 14 |
| Section 3.4 | Registration Expenses | 17 |
| Section 3.5 | Indemnification; Contribution | 17 |
| Section 3.6 | Holdback Agreements | 19 |
| Section 3.7 | Availability of Information | 19 |
| Section 3.8 | Information Concerning Stockholders | 19 |
| ARTICLE IV BOAR | D OF DIRECTORS OF THE COMPANY | 20 |
| Section 4.1 | Composition | 20 |
| Section 4.2 | Vacancy and Removal | 22 |
| Section 4.3 | Board Observation Rights | 22 |
| Section 4.4 | Transfer of Preferred Stock by WCAS | 23 |
| ARTICLE V MISCE | LLANEOUS | 23 |
| Section 5.1 | Entire Agreement | 23 |
| Section 5.2 | Table of Contents; Captions | 23 |
| Section 5.3 | Counterparts | 23 |
| Section 5.4 | Notices | 24 |
| Section 5.5 | Successors and Assigns | 24 |
| Section 5.6 | Governing Law | 24 |
| Section 5.7 | Submission to Jurisdiction | 24 |
| Section 5.8 | Waiver of Jury Trial | 25 |
| Section 5.9 | Third Party Beneficiaries | 26 |
| Section 5.10 | Confidentiality | 26 |
| Section 5.11 | Expenses | 26 |
| Section 5.12 | Amendments; Waivers | 26 |
| Section 5.13 | No Strict Construction | 27 |
| Section 5.14 | Specific Performance | 27 |

i

STOCKHOLDERS AGREEMENT

STOCKHOLDERS AGREEMENT (this "<u>Agreement</u>"), dated June ____, 2008, by and among Mobile Mini, Inc., a Delaware corporation (the "<u>Company</u>") and the Persons listed on Schedule I attached hereto (each, a "<u>Stockholder</u>" and collectively, the "<u>Stockholders</u>").

WITNESSETH:

WHEREAS, on the date hereof, in connection with the consummation of the transactions contemplated by that certain Agreement and Plan of Merger (the "Merger Agreement"), dated February 22, 2008, by and among the Company, Cactus Merger Sub, Inc., a Delaware corporation and direct wholly-owned subsidiary of the Company, MSG WC Holdings Corp., a Delaware corporation ("MSG") and Target Stockholder Representative (as defined in the Merger Agreement), each Stockholder became the record and beneficial owner of that number of shares of Series A Convertible Redeemable Participating Preferred Stock of the Company, par value \$0.01 per share ("Preferred Stock"), listed opposite such Stockholder's name on Schedule I attached hereto;

WHEREAS, Welsh, Carson, Anderson & Stowe X, L.P. ("<u>WCAS</u>") has the right in accordance with Section 7.7 of the Merger Agreement to acquire up to an additional 2,000,000 shares of Common Stock;

WHEREAS, it is a condition to the consummation of the transactions contemplated by the Merger Agreement that, simultaneously with such consummation, the Company and the Stockholders enter into this Agreement; and

WHEREAS, the Company and the Stockholders each desire to enter into this Agreement to, <u>inter alia</u>, regulate and limit certain rights relating to any Securities which may be held by any of the Stockholders from time to time and to limit the sale, assignment, transfer, encumbrance or other disposition of such Securities and to provide for certain arrangements regarding the management of the Company as set forth herein.

NOW, THEREFORE, in consideration of the mutual covenants set forth herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I

CERTAIN DEFINITIONS

Section 1.1 Certain Definitions. For purposes of this Agreement, the following terms shall have the following meanings:

"Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under common control with, such Person; provided that, for the purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the

management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

- "Agreement" shall have the meaning set forth in the preamble to this Agreement.
- "Approved Sale of the Company" shall mean a Sale of the Company that has been approved by the Board and the Board has not withdrawn or modified its recommendation (to the extent it is obligated to recommend the approval of such transaction to the Company's stockholders under applicable Law) of such Sale of the Company.
 - "Board" shall mean the Board of Directors of the Company.
- "Business Day" means any day, other than a Saturday, Sunday or other day on which banks located in New York City, New York or Tempe, Arizona are authorized or required by Law to close.
- "Certificate of Designation" shall mean that certain Certificate of Designation of the Company with respect to Preferred Stock, filed with the Secretary of State of the State of Delaware on the date hereof.
 - "Common Stock" shall mean the common stock of the Company, par value \$0.01 per share.
 - "Company" shall have the meaning set forth in the preamble to this Agreement.
- "Competitive Business" shall mean any business of the type and character engaged in and competitive with that conducted by the Company from time to time (which constitutes at least 20% of its gross revenues on a consolidated basis), including, without limitation, the rental, acquisition, refurbishment, renovation, or resale of shipping containers and other similar portable storage solutions.
 - "Confidentiality Agreement" shall mean that certain Mutual Confidentiality Agreement dated as of December 14, 2007, between Target and Parent.
 - "Confidential Information" shall have the meaning set forth in the Confidentiality Agreement.
- "Controlled Affiliate" shall mean any Affiliate of a Stockholder (other than (x) Affiliates of a portfolio company who are not otherwise an Affiliate of such Stockholder and (y) a portfolio company of such Stockholder, except the following portfolio companies shall be deemed a "Controlled Affiliate": any portfolio company (i) in which such Stockholder and its other Affiliates (other than another portfolio company of such Stockholder) has the power or right to nominate at least 50% of the board of directors or other similar governing body of such portfolio company or owns at least 50% of the outstanding voting equity securities of such portfolio company, (ii) that has received confidential information concerning the Company and its Subsidiaries (provided that possession or knowledge of such confidential information by any Affiliate of such Stockholder serving on the board of directors or other similar governing body of any entity without more shall not be imputed to such entity), or (iii) has taken any action at the

direction of such Stockholder or another Affiliate of such Stockholder that is otherwise prohibited or restricted by the terms of Section 2.3 of this Agreement).

"<u>Debt Securities</u>" shall mean all bonds, debentures, notes and other instruments evidencing indebtedness for borrowed money of the Company or any of its Subsidiaries from time to time (including, without limitation, those certain 9³/4% Senior Notes of Mobile Services Group, Inc. and Mobile Storage Group, Inc. due 2014 and those certain 67/8% Senior Notes of the Company due 2015 and any participation in senior secured, second lien, asset-backed or other credit facilities of the Company or any of its Subsidiaries), and any other instruments exchangeable or convertible therefor.

- "Demand Notice" shall have the meaning set forth in Section 3.1(b) of this Agreement.
- "Demand Registration Statement" shall have the meaning set forth in Section 3.1(b) of this Agreement.
- "Demand Request" shall have the meaning set forth in Section 3.1(b) of this Agreement.
- "Equity Securities" shall mean all shares of Common Stock of the Company, all securities, directly or indirectly, convertible into or exchangeable for shares of Common Stock of the Company (including, without limitation, the Preferred Stock) and all options, warrants, and other rights to purchase or otherwise, directly or indirectly, acquire from the Company shares of Common Stock, or securities convertible into or exchangeable for shares of Common Stock, whether at the time of issuance or upon the passage of time or the occurrence of some future event.
 - "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations of the SEC promulgated thereunder.
- "GAAP" shall mean United States generally accepted accounting principles consistently applied by the Company and its Subsidiaries throughout the periods indicated.
- "Governmental Entity" shall mean any instrumentality, subdivision, court, administrative agency, commission, official or other authority of the United States or any other country or any state, province, prefect, municipality, locality or other government or political subdivision thereof, or any quasi-governmental or private body exercising any regulatory, taxing, importing or other governmental or quasi-governmental authority.
 - "Holders' Counsel" shall have the meaning set forth in the definition of "Registration Expenses".
 - "Incidental Registration" shall have the meaning set forth in Section 3.2(a) of this Agreement.
 - "Law" means any statute, law, common law, order, ordinance, rule or regulation of any Governmental Entity.

"Majority Holders" shall mean the holders of Registrable Securities representing at least a majority of the outstanding Registrable Securities from time to time.

"Merger Agreement" shall have the meaning set forth in the first recital to this Agreement.

"NASDAQ" shall mean The Nasdaq Stock Market, Inc.

"Non-Qualified Person" shall mean any Person who is (i) directly or indirectly engaged in any business which the Board determines, in good faith, to be a Competitive Business, (ii) an adverse party in any material legal proceeding or material arbitration proceeding or other significant dispute (as determined in good faith by the Board), or (iii) an Affiliate of any Person described in clauses (i) and (ii).

"Original Stockholder" shall mean each Person that is either (a) a Stockholder as of the date hereof or (b) a Permitted Transferee pursuant to a Transfer effected in accordance with clause (i), (ii) or (iii) of Section 2.2(a) of this Agreement.

"Permitted Transfer" shall have the meaning set forth in Section 2.2(a) of this Agreement.

"Permitted Transferee" shall have the meaning set forth in Section 2.2(a) of this Agreement.

"<u>Person</u>" shall mean and include an individual, a partnership, a joint venture, a corporation, a limited liability company, a limited liability partnership, a trust, an incorporated organization or any other entity or organization, including a Governmental Entity.

"Preferred Stock" shall have the meaning set forth in the first recital to this Agreement.

"Qualifying Securities" shall mean the Preferred Stock and any Equity Securities issued upon conversion or exchange for the Preferred Stock pursuant to the Certificate of Designation.

"Registrable Securities" shall mean shares of Common Stock issued upon the conversion of Preferred Stock pursuant to the Certificate of Designation and any Common Stock acquired by WCAS as permitted by Section 7.7 of the Merger Agreement to the extent such shares have not been previously registered and sold pursuant to an effective registration statement and any other shares of Common Stock that may be received in respect of any of the foregoing securities; provided, that any Registrable Securities shall cease to be Registrable Securities:

(i) when a registration statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been disposed of in accordance with such registration statement;

- (ii) when such securities shall have been distributed by the holder thereof to the public pursuant to Rule 144 under the Securities Act (or any successor provision); or
 - (iii) when such securities shall have ceased to be outstanding.
 - "Registration" shall mean the Shelf Registration, each Required Registration and each Incidental Registration.

"Registration Expenses" shall mean all expenses incident to the Company's performance of or compliance with Article III including, without limitation, all registration and filing fees, fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel in connection with blue sky qualifications of any Registrable Securities), expenses of printing certificates for any Registrable Securities in a form eligible for deposit with the Depository Trust Company, messenger and delivery expenses, internal expenses (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), and fees and disbursements of counsel for the Company and its independent certified public accountants (including the expenses of any management review, cold comfort letters or any special audits required by or incident to such performance and compliance), securities acts liability insurance (if the Company elects to obtain such insurance), the reasonable fees and expenses of any special experts retained by the Company in connection with such registration, fees and expenses of other Persons retained by the Company, the fees and expenses of one (1) counsel not to exceed \$50,000 (the "Holders' Counsel") and applicable local counsel for the holders of Registrable Securities to be included in such Registration; but not including any underwriting fees, discounts or commissions attributable to the sale of securities or fees and expenses of counsel representing the holders of Registrable Securities included in such Registration (other than the Holders' Counsel and applicable local counsel) incurred in connection with the sale of Registrable Securities.

"Required Registration" shall have the meaning set forth in Section 3.1(b) of this Agreement.

"Sale of the Company" shall mean

- (i) any consolidation or merger of the Company or a Subsidiary of the Company in which the shares of Common Stock are converted into cash, securities or other property;
- (ii) any sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all, or substantially all, of the assets of the Company and its Subsidiaries; or
- (iii) any Person has become the beneficial owner (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of shares of the capital stock of the Company representing greater than 50% of the outstanding voting power of the Company.

"SEC" shall mean, at any time, the Securities and Exchange Commission or any other federal agency at such time administering the Securities Act.

- "Securities" shall mean, collectively, the Equity Securities and the Debt Securities held by a Stockholder from time to time.
- "Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations of the SEC promulgated thereunder.
- "Selection Date" shall mean the date that is sixty (60) days prior to the date on which the Company distributes to its stockholders the proxy statement relating to each applicable annual meeting.
 - "Shelf Registration" shall have the meaning set forth in Section 3.1(a) of this Agreement.
- "Shelf Registration Lapse Date" shall mean the date, if any, that (x) the Company is not permitted to file or maintain a Form S-3 in connection with the Shelf Registration in accordance with Section 3.1(a), or (y) the Shelf Registration expired in accordance with Section 3.1(a)(i) and not all Registrable Securities registered in such Shelf Registration have been sold.
 - "Shelf Registration Statement" shall have the meaning set forth in Section 3.1(a) of this Agreement.
 - "Standstill Period" shall have the meaning set forth in Section 2.3(a) of this Agreement.
- "<u>Standstill Securities</u>" shall mean any Equity Securities of the Company or any of its Subsidiaries (in each case, other than Qualifying Securities), and Debt Securities of the Company or any of its Subsidiaries.
 - "Stockholder" shall have the meaning set forth in the preamble to this Agreement, subject to Section 2.2 hereof.
 - "Subject Stockholder" shall mean WCAS and each Permitted Transferee of WCAS pursuant to a Transfer described in clause (iii) of Section 2.2(a).
- "Subsidiary" or "Subsidiaries" shall mean, with respect to any Person, (i) any corporation more than 50% of whose stock of any class or classes having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at the time stock of any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is owned by such Person directly or indirectly through one or more Subsidiaries of such Person and (ii) any partnership, association, joint venture or other entity in which such Person directly or indirectly through one or more Subsidiaries of such Person has more than a 50% equity interest.
- "<u>Transaction Documents</u>" shall mean, collectively, (i) this Agreement, (ii) the Merger Agreement, (iii) that certain Joinder to the Merger Agreement, dated February 22, 2008, executed by WCAS Management Corporation and certain of the other Stockholders, (iv) that certain Joinder, dated June 9, 2008, executed by Douglas Waugaman, (v) that certain Escrow Agreement, dated as of the date hereof, among the Company, MSG and Wells Fargo Bank, N.A.,

as escrow agent, and (vi) each other agreement, instrument and document delivered pursuant to or in connection with any of the transactions contemplated by the documents described in clauses (i) through (v) of this definition.

- "Transfer" shall have the meaning set forth in Section 2.1(a) of this Agreement.
- "WCAS" shall have the meaning set forth in the second recital to this Agreement.
- "WCAS Directorship Term End Date" shall have the meaning set forth in Section 4.4 of this Agreement.
- "WCAS Observer" shall have the meaning set forth in Section 4.3(a) of this Agreement.
- "WCAS Permanent Director" shall have the meaning set forth in Section 4.1(a) of this Agreement.
- "WCAS Temporary Director" shall have the meaning set forth in Section 4.1(a) of this Agreement.

ARTICLE II

TRANSFER OF EQUITY SECURITIES

Section 2.1 Restrictions.

- (a) No Stockholder shall, voluntarily or involuntarily, directly or indirectly, sell, assign, donate, hypothecate, pledge, encumber, grant a security interest in or in any other manner transfer, any Securities, in whole or in part, or any other right or interest therein, or enter into any transaction which results in the economic equivalent of a transfer to any Person (each such action, a "Transfer") except pursuant to a Permitted Transfer.
- (b) From and after the date hereof, all certificates or other instruments representing Securities held by each Stockholder shall bear a legend which shall state:

"The sale, transfer, hypothecation, assignment, pledge, encumbrance or other disposition of this [note and the obligations of the issuer] [share certificate and the shares of [Preferred] [Common] Stock] represented hereby are restricted by and are subject to all of the terms, conditions and provisions of that certain Stockholders Agreement, dated as of [______], 2008, as amended from time to time, by and between the Company and the investors party thereto, which agreement is on file at the principal office of the Company."

(c) In addition to the legend required by Section 2.1(b) above, all certificates representing Equity Securities held by each Stockholder (other than Equity Securities acquired pursuant to Section 7.7 of the Merger Agreement or that have otherwise been previously registered and sold pursuant to an effective registration statement under the Securities Act) shall bear a legend which shall state:

"The securities represented by this certificate have not been registered under the Securities Act of 1933, as amended, or pursuant to any state securities laws. The securities have been acquired for investment and may not be sold or transferred except in compliance with the registration requirements of the Securities Act of 1933, as amended, and applicable state securities laws or pursuant to an exemption therefrom."

- (d) Any attempt to transfer any Security which is not in accordance with this Agreement shall be null and void and the Company agrees that it will not cause, permit or give any effect to any Transfer of any Securities to be made on its books and records unless such Transfer is permitted by this Agreement and has been made in accordance with the terms hereof.
- (e) Each Stockholder agrees that it will not effect any Transfer of Securities unless such Transfer is a Permitted Transfer and is made (i) pursuant to an effective registration statement under the Securities Act or pursuant to an exemption from the registration requirements of the Securities Act or pursuant to Rule 144 or Rule 144A promulgated under the Securities Act and (ii) in accordance with all applicable Laws (including, without limitation, all securities laws).

Section 2.2 Permitted Transfers.

- (a) Notwithstanding anything to the contrary contained herein and subject to Sections 2.2(b) and 2.2(c), a Stockholder may at any time effect any of the following Transfers (each a "Permitted Transfere", and each transferee of such Stockholder in respect of such Transfer, a "Permitted Transferee"):
 - (i) any Transfer of any or all Securities held by a Stockholder who is a natural Person following such Stockholder's death by will or intestacy to such Stockholder's legal representative, heir or legatee;
 - (ii) any Transfer of any or all Securities held by a Stockholder who is a natural Person as a gift or gifts during such Stockholder's lifetime to such Stockholder's spouse, children, grandchildren or a trust or other legal entity for the exclusive benefit of such Stockholder or any one or more of the foregoing;
 - (iii) any Transfer of any or all Securities held by a Stockholder to any Affiliate of such Stockholder; <u>provided</u> that such Affiliate is a Person who is not a Non-Qualified Person; <u>provided</u>, further that any such Affiliate shall Transfer such Securities to the Stockholder from whom the Securities were originally received or acquired within five (5) calendar days after ceasing to be an Affiliate of such Stockholder;
 - (iv) any Transfer, occurring on or after the first (1st) anniversary of the date hereof, of any or all Common Stock or Debt Securities or other securities, other than the Preferred Stock held by a Stockholder provided that if any such Transfer is a private sale and not in connection with an underwritten offering or a block trade to a financial intermediary or other public sale (including any Rule 144 sale in a brokered transaction) then such sale may not be made if (A) the transferee (or any group (as defined in Rule 13(d)(3) of the Exchange Act) of transferees) would purchase in that transaction, (or

pursuant to all transactions with that Stockholder or group (as defined in Rule 13(d)(3) of the Exchange Act) of Stockholders) more than three percent (3%) of the fully diluted Common Stock (as of the relevant time of determination), which will be provided promptly upon the written request of any Stockholder to the Company's Secretary or Chief Financial Officer determined either by reference to the Company's most recently filed Form 10-K or Form 10-Q or upon information provided by the Company, or (B) the transferee with respect to such Transfer is a Non-Qualified Person; and

- (v) any Transfer, occurring on or after the first (1st) anniversary of the date hereof, of the Preferred Stock if (A) the transferee with respect to such Transfer is not a Non-Qualified Person, (B) the transferee (or any group (as defined in Rule 13(d)(3) of the Exchange Act) of transferees) would purchase in that transaction, (or pursuant to all transactions with that Stockholder or group (as defined in Rule 13(d)(3) of the Exchange Act) of Stockholders) more than three percent (3%) of the fully diluted Common Stock (as of the relevant time of determination), which will be provided promptly upon the written request of any Stockholder to the Company's Secretary or Chief Financial Officer determined either by reference to the Company's most recently filed Form 10-K or Form 10-Q or upon information provided by the Company, and (C) the aggregate number of Permitted Transferees in connection with all Transfers of Equity Securities by (i) any single Stockholder (other than WCAS, Lehman or Calsters) pursuant to this clause (v) does not exceed one (1) person; and (ii) with respect to WCAS, Lehman and Calsters pursuant to this clause (v) does not exceed five (5), two (2) and two (2) Persons, respectively; provided, that prior to effecting any Transfer pursuant to this clause (v), the transferee(s) with respect to a Transfer pursuant to this clause (v) shall agree in writing not to Transfer any of the Preferred Stock to a Non-Qualified Person pursuant to a private sale.
- (b) In any Transfer referred to above in clauses (i), (ii) or (iii) of Section 2.2(a), the Permitted Transferee shall agree in writing to be bound by all of the provisions of this Agreement, shall execute and deliver to the Company a counterpart to this Agreement, and shall hold all such Securities as a "Stockholder" hereunder as if such Permitted Transferee was an original signatory hereto and shall be deemed to be a party to this Agreement. In addition, each Permitted Transferee (A) pursuant to a Transfer by a Stockholder other than WCAS referred to in clause (i), (ii) or (iii) of Section 2.2(a) shall hold all Equity Securities an "Original Stockholder" hereunder, and (B) pursuant to a Transfer by WCAS referred to in clause (iii) of Section 2.2(a) shall hold all Equity Securities as a "Subject Stockholder" and an "Original Stockholder" hereunder.
- (c) Notwithstanding anything to the contrary contained in this Agreement, prior to the WCAS Directorship Term End Date, at all times during the Company's customary black-out periods (i.e., relating to the public release of quarterly or annual financial information) neither WCAS nor any of its Controlled Affiliates shall sell any Securities other than during any period when the directors and officers of the Company and its Subsidiaries are not prohibited from selling Securities pursuant to the written policies and procedures of the Company governing transfers of Securities by such officers and directors during such ordinary black-out periods as may be in effect from time to time; <u>provided</u>, <u>however</u>, that (i) the foregoing limitation shall not apply to WCAS or any of its Controlled Affiliates if the black-out period imposed by

the Company results from the occurrence of an extraordinary event affecting the Company or any of its Subsidiaries, and (ii) if circumstances exist such that the Company would have the right to take the actions specified in Section 3.1(e) of this Agreement, then (notwithstanding anything to the contrary set forth in clause (i)) the foregoing prohibition may be imposed by the Company under the circumstances described in Section 3.1(e) of this Agreement, for the periods and subject to the limitations set forth in such Section.

Section 2.3 Standstill.

For the period (the "<u>Standstill Period</u>") commencing on the date hereof and ending on the date on which the Subject Stockholders, in the aggregate, no longer hold Equity Securities constituting (or representing upon the conversion thereof) five percent (5%) or more of the outstanding shares of Common Stock, no Subject Stockholder shall, and each Subject Stockholder shall cause its respective Controlled Affiliates, unless expressly agreed in writing, in advance, by the Company, directly or indirectly, in any manner whatsoever:

- (a) acquire, announce an intention to acquire, offer or propose to acquire, solicit an offer to sell or agree to acquire, or enter into any arrangement or undertaking to acquire, directly or indirectly, by purchase, or otherwise, record or direct or indirect beneficial ownership interest in any Standstill Securities or any assets (other than purchases of assets in the ordinary course of business) or other securities of the Company or any of its Subsidiaries or any direct or indirect rights, warrants or options to acquire record or direct or indirect beneficial ownership of any securities or assets of the Company or any of its Subsidiaries;
- (b) make, effect, initiate, cause or participate in any take-over bid, tender offer, exchange offer, merger, consolidation, business combination, recapitalization, restructuring, liquidation, dissolution or other extraordinary transaction involving the Company or any of its Subsidiaries;
- (c) solicit, make, effect, initiate, cause, or in any way participate in, directly or indirectly, any solicitation of proxies or consents from any holders of any securities of the Company or any of its Subsidiaries or call or seek to have called any meeting of stockholders of the Company or any of its Subsidiaries;
- (d) form, join or participate in, or otherwise encourage the formation of, any "group" (within the meaning of Section 13(d)(3) of the Exchange Act) with respect to any securities of the Company or any of its Subsidiaries that are not Standstill Securities;
- (e) arrange, facilitate, or in any way participate, directly or indirectly, in any financing for the purchase of any securities or assets of the Company or any of its Subsidiaries that are not Standstill Securities;
- (f) (i) act, directly or indirectly, to seek control or direct the board of directors, stockholders, policies or affairs of the Company or any of its Subsidiaries; (ii) solicit, propose, seek to effect or negotiate with any other Person with respect to any form of business combination transaction involving the Company or any take-over bid, tender, exchange offer, merger, consolidation, recapitalization, restructuring, liquidation, dissolution, or other extraordinary transaction involving the Company or any of its Subsidiaries; or (iii) disclose an

intent, purpose, plan or proposal with respect to the Company, or any securities or assets of the Company or any of its Subsidiaries that are not Standstill Securities:

- (g) take any action that is intended to or reasonably expected to require the Company or any of its Subsidiaries to make a public announcement regarding any of the types of matters set forth in clauses (a) through (f), inclusive, of this Section 2.3;
- (h) agree or offer to take, or encourage or propose (publicly or privately) the taking of, or announce an intention to take, or otherwise make any public announcement with respect to, any action referred to in clauses (a) through (g), inclusive, of this Section 2.3; and
- (i) request of, or propose to the Company, or any of its representatives that the Company amend or waive or consider the amendment or waiver of any provision of this Section 2.3.

Notwithstanding anything to the contrary in this Section 2.3, (i) any pooled investment vehicle managed or under the control of WCAS or any of its Affiliates that primarily invests on a passive basis in debt securities or debt instruments shall be permitted to acquire up to \$50 million in the aggregate of all classes or types of Debt Securities that are not convertible into Equity Securities or otherwise have voting features permitting the holders thereof to vote with the holders of any Equity Security on any matter, (ii) this Section 2.3 shall not apply to any indirect interest held by any Controlled Affiliate of a Person if such Controlled Affiliate is an individual or trust or other investment vehicle formed for the benefit of individuals or charitable concerns and such interest is held through another entity that is not controlled by such Controlled Affiliate (e.g. a publicly traded mutual fund, a "blind" trust or investment account or a private equity or other private investment fund with respect to which such Controlled Affiliate is not affiliated) and (iii) nothing herein shall limit in anyway the conduct of any person who is serving on the Board and who is affiliated with WCAS or any of its Controlled Affiliates from taking any actions in his or her capacity as such, nor prohibit any such person or WCAS or any of its Controlled Affiliates from taking any action in furtherance of or in order to assure that any person who may be designated by them to serve on the Board as contemplated by this Agreement is elected to serve on the Board. In addition, notwithstanding anything to the contrary in this Section 2.3, each Subject Stockholder shall be permitted to sell its Equity Securities in any Approved Sale of the Company.

ARTICLE III

REGISTRATION RIGHTS

Section 3.1 Required Registrations.

(a) <u>Shelf Registration Statement</u>. Subject to each Stockholder's compliance with Section 3.8, the Company shall use all commercially reasonable efforts to file a registration statement under the Securities Act on or about the date that is ten (10) months from the date hereof covering all of the Registrable Securities then held by the Stockholders on Form S-3 or such other available forms (the "<u>Shelf Registration</u>") and to have such Registration Statement declared effective to enable the resale of such Registrable Securities after the first (1st)

anniversary of the date hereof on a delayed or continuous basis pursuant to Rule 415 under the Securities Act (the "Shelf Registration Statement") through NASDAQ or such other market as may be the principal market on which the Registrable Securities are then quoted or listed. The Company will use all commercially reasonable efforts to cause the Shelf Registration Statement to remain continuously effective under the Securities Act until the earlier of (i) the date that is two (2) years (or such longer period as may be permitted under applicable Law) after the initial effectiveness thereof and (ii) the earliest date on which all Registrable Securities held by the Stockholders shall have either (A) been sold in accordance with this Section 3.1(a) or (B) ceased to be outstanding.

- (b) Required Registrations. If, at any time after the Shelf Registration Lapse Date, the Company shall be requested in writing, which writing shall specify the Registrable Securities to be sold and the intended method of disposition thereof (a "Demand Request"), by the Majority Holders, to effect a registration under the Securities Act of Registrable Securities held by such Stockholders (each, a "Required Registration"), then the Company shall promptly use all commercially reasonable efforts to effect such Required Registration by filing, at the Company's option, either a Form S-1 or Form S-3 registration statement (a "Demand Registration Statement"); provided the Company shall not be required to comply with more than one (1) Demand Request during any six (6) month period and shall only be obligated to comply with four (4) Demand Requests in total; provided, that if a Shelf Registration Lapse Date occurs during the period beginning on the date that the Shelf Registration contemplated by Section 3.1(a) is declared effective and ending on the second anniversary of the date thereof, then the limit for Demand Requests shall be increased by one (1). Upon receipt by the Company of a Demand Request, the Company shall deliver a written notice (a "Demand Notice") to each Stockholder who did not make such Demand Request stating that the Company intends to comply with a Demand Request and informing each such Stockholder of its right to include Registrable Securities in such Required Registration. Within ten (10) Business Days after receipt of a Demand Notice, each Stockholder shall have the right to request in writing that the Company include all or a specific portion of the Registrable Securities held by such Stockholder in such Required Registration. Notwithstanding anything to the contrary set forth herein, the Company shall be obligated to effect any one or more of such Required Registrations pursuant to a Shelf Registration Statement if the Majority Holders so request in connection with any Demand Request.
- (c) Selection of Underwriters. In the event that the Registrable Securities to be registered pursuant to a Required Registration are to be disposed of in an underwritten public offering, the underwriters of such public offering shall be one or more underwriting firms of nationally recognized standing selected by the Majority Holders and shall be reasonably acceptable to the Company. In the event the Company elects to file a Demand Registration Statement on Form S-3 and the underwriters, if any, in such public offering or the Majority Holders request that the Company provide disclosures otherwise required in connection with a Form S-1 registration statement, then the Company shall include in such Demand Registration Statement such "long form" disclosures.
- (d) Priority on Required Registrations. In the event that, in the case of any Required Registration, the managing underwriter for the public offering contemplated by Section 3.1(b) shall advise the Company in writing (with a copy to each holder of Registrable Securities

requesting sale) that, in such underwriter's opinion, the amount of securities requested to be included in such Required Registration would adversely affect the public offering and sale (including pricing) of such Registrable Securities (such writing to state the basis of such opinion and the approximate number of Registrable Securities that may be included in such public offering without such effect), the Company will include in such Required Registration the number of Registrable Securities that the Company is so advised can be sold in such public offering, in the following amounts:

- (i) <u>first</u>, all Registrable Securities requested to be sold by holders of Registrable Securities pursuant to Section 3.1(b) <u>pro rata</u> among such holders on the basis of the number of Registrable Securities owned by each such holders; and
 - (ii) second, securities proposed to be sold by the Company for its own account.
- (e) <u>Black Out Period</u>. Notwithstanding any other provision of this Agreement to the contrary, if the Board reasonably determines that the registration and distribution of Registrable Securities (i) would reasonably be expected to impede, delay or interfere with, or require premature disclosure of, any material financing, offering, acquisition, merger, corporate reorganization, segment reclassification or discontinuation of operations, or other significant transaction or any negotiations, discussions or pending proposals with respect thereto, involving the Company or any of its Subsidiaries, or (ii) would require disclosure of non-public material information, the disclosure of which would reasonably be expected to adversely affect the Company, the Company shall (x) be entitled to postpone the filing or effectiveness or suspend the effectiveness of a registration statement and/or the use of any prospectus for a period of time not to exceed sixty (60) days and (y) promptly give the Stockholders written notice of such postponement or suspension (which notice need not specify the nature of the event giving rise to such suspension); <u>provided</u>, that the Company shall not utilize the right described in Section 3.1(b) more than once in any six (6) month period and provided further that the Company may extend such period to be up to ninety (90) days in the aggregate, but if it elects to do so it shall not be permitted to impose a subsequent black out period until a time that is more than six (6) months after the end of such extended black out period.

 Notwithstanding anything to the contrary set forth herein, any application of the provisions of Section 2.2(c) of this Agreement that results in a postponement of the effectiveness of a registration statement pursuant to this Section 3.1(e) shall not be included in calculating the 60-day period or 90-day period above.

Section 3.2 Incidental Registration.

(a) Filing of Registration Statement. If, at any time after the first (1st) anniversary of the date hereof the Company proposes to register, for its own account or for the account of any other Person any of its securities (an "Incidental Registration") under the Securities Act (other than pursuant to a registration statement on Form S 4 or Form S 8 or any successor forms thereto) for sale to the public, it will at each such time give prompt written notice to all Stockholders of its intention to do so, which notice shall be given at least thirty (30) days prior to the date that a registration statement relating to such registration is proposed to be filed with the SEC. Upon the written request of any Stockholder to include Registrable Securities held by it that are not otherwise covered by the Shelf Registration Statement or a

Demand Registration Statement in such Incidental Registration statement (which request shall (i) be made within fifteen (15) days after the receipt of any such notice, and (ii) specify the Registrable Securities intended to be included by such holder), the Company will use all commercially reasonable efforts to effect the registration of all Registrable Securities that the Company has been so requested to register by such Stockholder; <u>provided</u>, <u>however</u>, that if, at any time after giving written notice of its intention to register any securities and prior to the effective date of the registration statement filed in connection with such registration, the Company shall determine for any reason to terminate such registration statement and not to register such securities, the Company may, at its election, give written notice of such determination to each such holder and, thereupon, shall be relieved of its obligation to register any Registrable Securities of such Persons in connection with such registration.

- (b) <u>Selection and Use of Underwriters</u>. Underwriters, if any, in connection with any offering pursuant to this Section 3.2 shall be selected at the sole and exclusive discretion of the Company. No Stockholder shall Transfer any Registrable Securities included in the Incidental Registration other than through the underwriter or underwriters so selected by the Company.
- (c) <u>Priority on Incidental Registrations</u>. If the managing underwriter for the offering contemplated by this Section 3.2 shall advise the Company in writing that, in such underwriter's opinion, the number of securities requested to be included in such Incidental Registration would adversely affect the offering and sale (including pricing) of such securities, the Company shall include in such Incidental Registration the number of securities that the Company is so advised can be sold in such offering, in the following amounts and order of priority:
 - (i) first, securities proposed to be sold by the Company for its own account or for the account of any other Person not a party hereto; and
 - (ii) <u>second</u>, the Registrable Securities requested to be registered by Stockholders pro rata among such Stockholders on the basis of the number of Registrable Securities owned by each such Stockholders.

Section 3.3 Registration Procedures.

The Company will use all commercially reasonable efforts to effect the Shelf Registration and Required Registration pursuant to Section 3.1 and each Incidental Registration pursuant to Section 3.2, and to cooperate with the sale of such Registrable Securities in accordance with the intended method of disposition thereof as quickly as reasonably practicable, and the Company will as expeditiously as reasonably practicable:

- (a) subject, in the case of an Incidental Registration, to the proviso to Section 3.2(a), prepare and file with the SEC the registration statement and use all commercially reasonable efforts to cause the Registration to become effective;
- (b) subject, in the case of an Incidental Registration, to the proviso to Section 3.2(a), prepare and file with the SEC such amendments and post-effective amendments to any registration statement and any prospectus used in connection therewith as may be necessary to

keep such registration statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all Registrable Securities covered by such registration statement until such time as all of such Registrable Securities have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such registration statement and cause the prospectus to be supplemented by any required prospectus supplement, and as so supplemented to be filed pursuant to Rule 424 under the Securities Act;

- (c) furnish, upon request, to each holder of Registrable Securities to be included in such Registration and the underwriter or underwriters, without charge, at least one copy of the signed registration statement and any post-effective amendment thereto, and such number of conformed copies thereof and such number of copies of the prospectus (including each preliminary prospectus and each prospectus filed under Rule 424 under the Securities Act), any amendments or supplements thereto and any documents incorporated by reference therein, as such holder or underwriter may reasonably request in order to facilitate the disposition of the Registrable Securities being sold by such holder (it being understood that the Company consents to the use of the prospectus and any amendment or supplement thereto by each holder of Registrable Securities covered by such registration statement and the underwriter or underwriters, in connection with the offering and sale of the Registrable Securities covered by the prospectus or any amendment or supplement thereto);
 - (d) notify each holder of the Registrable Securities to be included in such Registration and the underwriter or underwriters:
 - (i) of any stop order or other order suspending the effectiveness of any registration statement, issued or threatened by the SEC in connection therewith, and take all commercially reasonable actions required to prevent the entry of such stop order or to remove it or obtain withdrawal of it at the earliest possible moment if entered;
 - (ii) when such registration statement or any prospectus used in connection therewith, or any amendment or supplement thereto, has been filed and, with respect to such registration statement or any post-effective amendment thereto, when the same has become effective;
 - (iii) of any written request by the SEC for amendments or supplements to such registration statement or prospectus; and
 - (iv) of the receipt by the Company of any notification with respect to the suspension of the qualification of any Registrable Securities for sale under the applicable securities or blue sky laws of any jurisdiction;
- (e) if requested by the managing underwriter or underwriters, promptly incorporate in a prospectus supplement or post-effective amendment such information relating to such underwriting as the managing underwriter or underwriters reasonably request to be included therein; and make all required filings of such prospectus supplement or post-effective amendment as soon as practicable after being notified of the matters incorporated in such prospectus supplement or post-effective amendment; provided, however, that the Company shall

not be required to take any action pursuant to this Section 3.3(e) that would, in the opinion of counsel to the Company, violate applicable Law;

- (f) on or prior to the date on which a Registration is declared effective, use all commercially reasonable efforts to register or qualify, and cooperate with the holders of Registrable Securities to be included in such Registration, the underwriter or underwriters, if any, and their counsel, in connection with the registration or qualification of the Registrable Securities covered by such Registration for offer and sale under the securities or "blue sky" laws of each state and other jurisdiction of the United States as any such holder or underwriter reasonably requests in writing; use all commercially reasonable efforts to keep each such registration or qualification effective, including through new filings, or amendments or renewals, during the period such registration statement is required to be kept effective; and do any and all other acts or things reasonably necessary or advisable to enable the disposition of the Registrable Securities in all such jurisdictions reasonably requested to be covered by such Registration; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it is not then so qualified or to take any action which would subject it to general service of process in any such jurisdiction where it is not then so subject;
- (g) in connection with any sale pursuant to a Registration, cooperate with the holders of Registrable Securities to be included in such Registration and the managing underwriter or underwriters, to facilitate the timely preparation and delivery of certificates (not bearing any restrictive legends including, without limitation, those set forth in Section 2.1) representing securities to be sold under such Registration, and enable such securities to be in such denominations and registered in such names as the managing underwriter or underwriters, if any, or such holders may request;
- (h) use all commercially reasonable efforts to cause the Registrable Securities to be registered with or approved by such other governmental agencies or authorities within the United States and having jurisdiction over the Company or any Subsidiary as may be necessary to enable the seller or sellers thereof or the underwriter or underwriters, as applicable, to consummate the disposition of such securities;
 - (i) use all commercially reasonable efforts to obtain such legal opinions and auditors' consents as may be required by applicable Law;
- (j) otherwise comply with all applicable rules and regulations of the SEC, and make generally available to its security holders (as contemplated by Section 11(a) under the Securities Act) an earnings statement satisfying the provisions of Rule 158 under the Securities Act no later than ninety (90) days after the end of the twelve (12) month period beginning with the first month of the Company's first fiscal quarter commencing after the effective date of the registration statement, which statement shall cover said twelve (12) month period; and
- (k) use all commercially reasonable efforts to cause its senior executive officers to participate in "road shows" at the request of the underwriters in connection with a Required Registration; <u>provided</u>, that such senior executive officers shall not be required to participate in "road shows" for more than two (2) Required Registrations.

Section 3.4 Registration Expenses.

The Company will pay all Registration Expenses in connection with each registration of Registrable Securities, including, without limitation, any such registration not effected by the Company. WCAS shall promptly reimburse the Company for any incremental Registration Expenses incurred by the Company in connection with the registration of any shares of Common Stock acquired by WCAS in accordance with Section 7.7 of the Merger Agreement as reasonably agreed by the Company and WCAS.

Section 3.5 Indemnification; Contribution,

(a) The Company shall indemnify, to the fullest extent permitted by applicable Law, each holder of Registrable Securities, its officers, directors, partners, employees and agents, if any, and each Person, if any, who controls such holder within the meaning of Section 15 of the Securities Act, against all losses, claims, damages, liabilities (or proceedings in respect thereof) and expenses (under the Securities Act or common law or otherwise), joint or several, resulting from any violation by the Company of the provisions of the Securities Act or any untrue statement or alleged untrue statement of a material fact contained in any registration statement or prospectus (and as amended or supplemented if amended or supplemented) or any preliminary prospectus or caused by any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein (in the case of any prospectus, in light of the circumstances under which they were made) not misleading, except to the extent that such losses, claims, damages, liabilities (or proceedings in respect thereof) or expenses are caused by any untrue statement or alleged untrue statement contained in or by any omission or alleged omission from information concerning any holder of Registrable Securities furnished in writing to the Company by such holder expressly for use therein. No action or failure to act on the part of the underwriters (whether or not such underwriter is an Affiliate of any holder of Registrable Securities) shall affect the obligations of the Company to indemnify any holder of Registrable Securities or any other Person pursuant to the preceding sentence. In connection with any underwritten offering pursuant to Section 3.2, the Company agrees to enter into an underwriting agreement in customary form with the applicable underwriters, and the Company agrees to indemnify such underwriters, their officers, directors, employees and agents, if any, and each Person, if any, who controls such underwriters within the meaning of Section 15 of the Securities Act to the same extent as herein before provided with respect to the indemnification of the holders of Registrable Securities; provided that the Company shall not be required to indemnify any such underwriter, or any officer, director or employee of such underwriter or any Person who controls such underwriter within the meaning of Section 15 of the Securities Act, to the extent that the loss, claim, damage, liability (or proceedings in respect thereof) or expense for which indemnification is claimed results from such underwriter's failure to send or give a copy of an amended or supplemented final prospectus to the Person asserting an untrue statement or alleged untrue statement or omission or alleged omission at or prior to the written confirmation of the sale of Registrable Securities to such Person if such statement or omission was corrected in such amended or supplemented final prospectus prior to such written confirmation and the underwriter was provided with such amended or supplemented final prospectus.

- (b) In connection with any registration statement in connection with an offering in which a holder of Registrable Securities is participating, each such holder, severally and not jointly, shall indemnify, to the fullest extent permitted by applicable Law, the Company, each underwriter and their respective officers, directors, employees and agents, if any, and each Person, if any, who controls the Company or such underwriter within the meaning of Section 15 of the Securities Act, against any losses, claims, damages, liabilities (or proceedings in respect thereof) and expenses resulting from any untrue statement or alleged untrue statement of a material fact, or any omission or alleged omission of a material fact required to be stated in the registration statement or prospectus or preliminary prospectus or any amendment thereof or supplement thereto or necessary to make the statements therein (in the case of any prospectus, in light of the circumstances under which they were made) not misleading, but only to the extent that such untrue statement is contained in or such omission is from information so concerning a holder furnished in writing by such holder expressly for use therein; provided that such holder's obligations hereunder shall be limited to an amount equal to the net proceeds to such holder of the Registrable Securities sold pursuant to such registration statement.
- (c) Any Person entitled to indemnification under the provisions of this Section 3.5 shall (i) give prompt notice to the indemnifying party of any claim with respect to which it seeks indemnification and (ii) permit such indemnifying party to assume the defense of such claim, with counsel reasonably satisfactory to the indemnified party; and if such defense is so assumed, such indemnifying party shall not enter into any settlement without the consent of the indemnified party if such settlement attributes liability to the indemnified party and such indemnifying party shall not be subject to any liability for any settlement made without its consent (which shall not be unreasonably withheld); and any underwriting agreement entered into with respect to any registration statement provided for under this Article III shall so provide. In the event an indemnifying party shall elect not to assume the defense of a claim, such indemnifying party shall not be obligated to pay the fees and expenses of more than one counsel or firm of counsel for all parties indemnified by such indemnifying party in respect of such claim.
- (d) If for any reason the foregoing indemnity is unavailable, then the indemnifying party shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities or expenses (i) in such proportion as is appropriate to reflect the relative benefits received by the indemnifying party on the one hand and the indemnified party on the other or (ii) if the allocation provided by clause (i) above is not permitted by applicable Law or provides a lesser sum to the indemnified party than the amount hereinafter calculated, in such proportion as is appropriate to reflect not only the relative benefits received by the indemnifying party on the one hand and the indemnified party on the other but also the relative fault of the indemnifying party and the indemnified party as well as any other relevant equitable considerations. Notwithstanding the foregoing, no holder of Registrable Securities shall be required to contribute any amount in excess of the amount such holder would have been required to pay to an indemnified party if the indemnity under Section 3.5(b) were available. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation. The obligation of any Person to contribute pursuant to this Section 3.5 shall be several and not joint.

- (e) An indemnifying party shall make payments of all amounts required to be made pursuant to the foregoing provisions of this Section 3.5 to or for the account of the indemnified party from time to time promptly upon receipt of bills or invoices relating thereto or when otherwise due or payable.
- (f) The indemnity and contribution agreements contained in this Section 3.5 shall remain in full force and effect regardless of any investigation made by or on behalf of a participating holder of Registrable Securities, its officers, directors, agents or any Person, if any, who controls such holder as aforesaid, and shall survive the Transfer of Equity Securities by such holder and the termination of this Agreement for any reason.

Section 3.6 Holdback Agreements.

Each Stockholder agrees not to sell, make any short sale of, grant any option for the purchase of, or otherwise dispose of any securities, other than those Registrable Securities included in such Registration pursuant to Section 3.1 or 3.2(a) for the seven (7) days prior to and the ninety (90) days after the effectiveness of the registration statement pursuant to which such offering shall be made (or such longer periods as may be advised by the underwriter with respect to the applicable offering but in any event not to exceed thirty (30) days prior to and ninety (90) days after the effectiveness of such registration statement). The Company agrees that it and its executive officers will be subject to the holdback period requested by the underwriters of a Required Registration, if any, pursuant to this Section 3.6 to the extent that such underwriters determine such holdback by the Company and its executive officers is reasonably necessary for the successful offering and sale of all Registrable Securities in connection with such registration.

Section 3.7 Availability of Information.

The Company shall cooperate with each Stockholder who is a holder of any Registrable Securities in supplying such information as may be reasonably necessary for such holder to complete and file any information reporting forms presently or hereafter required by the SEC as a condition to the availability of an exemption from the Securities Act for the sale of any Registrable Securities.

Section 3.8 Information Concerning Stockholders.

It shall be a condition precedent to the obligations of the Company to include the Registrable Securities of any selling Stockholder in any registration statement or prospectus, as the case may be, that such selling Stockholder shall take the actions described in this Section 3.8:

(a) each selling Stockholder that has requested inclusion of its Registrable Securities in any registration statement shall furnish to the Company in writing all information as may be necessary to make the information previously furnished to the Company by such Stockholder, in light of the circumstances under which it was made, not misleading, any other information regarding such Stockholder and the distribution of such Registrable Securities as may be required to be disclosed in the prospectus or registration statement under applicable Law or pursuant to SEC comments and any information otherwise reasonably requested from time to time by the Company to comply with applicable Law or regulations, including, without limitation, (i) the then current name and address of such Stockholder(s), (ii) the aggregate

number of Registrable Securities requested to be registered, (iii) the total number of Registrable Securities then held by such Stockholder(s), (iv) the intended means of distribution, and (v) any other information required to be disclosed with respect to such Stockholder or such Stockholder's Registrable Securities in the registration statement or related prospectus by the Securities Act;

- (b) each selling Stockholder shall promptly (i) following it becoming aware thereof, notify the Company of the occurrence of any event that makes any statement made in a registration statement or prospectus regarding such selling Stockholder untrue in any material respect or that requires the making of any changes in a registration statement or prospectus so that, in such regard, it shall not contain any untrue statement of a material fact or omit any material fact required to be stated therein or necessary to make the statements (in the case of a prospectus, in light of the circumstances under which they were made), not misleading and (ii) in connection with providing such notice, provide the Company with such information in its possession as may be required to enable the Company to prepare a supplement or post-effective amendment to any such registration statement or a supplement to such prospectus;
- (c) with respect to any registration statement for an underwritten offering, the inclusion of a Stockholder's Registrable Securities therein shall be conditioned, at the managing underwriter's request, upon the execution and delivery by such Stockholder of an underwriting agreement as may be negotiated by the Company;
- (d) any sale of any Registrable Securities by any Stockholder shall constitute a representation and warranty by such Stockholder that the prospectus delivered by such Stockholder does not as of the time of such sale contain any untrue statement of a material fact relating to the information expressly provided in writing by such Stockholder for inclusion in such prospectus and that such prospectus does not as of the time of such sale omit to state any material fact relating to the information expressly provided in writing by such Stockholder for inclusion in such prospectus necessary to make the statements in such prospectus, in light of the circumstances under which they were made, not misleading; and
- (e) no Stockholder shall use, distribute or otherwise disseminate any "free writing prospectus", as defined in Rule 405 under the Securities Act, in connection with the sale of Registrable Shares under the Shelf Registration Statement, without the prior written consent of the Company.

ARTICLE IV

BOARD OF DIRECTORS OF THE COMPANY

Section 4.1 Composition.

(a) At the Effective Time (as defined in the Merger Agreement), the Company shall expand the size of the Board so that the number of members on the Board is equal to eight (8) and shall appoint (i) one individual designated by WCAS (the "WCAS Permanent Director"), whose term ends in 2011 and (ii) another individual designated by WCAS (the "WCAS Temporary Director"), whose term ends in 2009. WCAS hereby designates Sanjay Swani as the

initial WCAS Permanent Director and Michael Donovan as the initial WCAS Temporary Director.

- (b) From and after the date hereof, the Company shall use all commercially reasonable efforts to take all necessary and desirable actions within its control (including calling special board meetings) so that:
 - (i) the number of members on the Board will not exceed eight (8) or such larger number as may be required to comply with any Laws, rules or regulations including rules and regulations concerning director independence; and
 - (ii) the WCAS Permanent Director will be elected to a seat as a director of the class whose term ends in 2011 and the WCAS Temporary Director will be elected to a seat as a director of the class whose term ends in 2009.
- (c) At the next annual meeting of the Company, and thereafter each time the applicable class of directors comes up for re-election, until the WCAS Directorship Term End Date, the Board shall recommend that the stockholders of the Company elect to the Board one (1) individual selected by WCAS to fill the seat of the WCAS Permanent Director. WCAS shall, as promptly as practicable (and in any event no later than the Selection Date, provided that the Company shall provide WCAS with written confirmation of the Selection Date 30 days prior to such date), provide the Board with written notice of the name of the individual so selected by it, together with such biographical information regarding such individual as the Nominating and Corporate Governance Committee of the Board may request. Notwithstanding the foregoing provisions of this Section 4.1(b), if the Nominating and Corporate Governance Committee of the Board shall determine in its good faith and reasonable judgment (including, without limitation, any failure of WCAS's designee to satisfy all legal and regulatory requirements as well as any and all requirements that may be set forth in the Company's Code of Business Conduct and Ethics or any other governing instruments or policies of the Company) to disqualify any designee of WCAS from service on the Board, then the Board shall not be required to recommend the election of such individual, and WCAS shall promptly designate a different individual. Nothwithstanding anything to the contrary set forth herein, if the individual selected by WCAS in accordance with this Section 4.1(c) is not able to serve on the Board or ceases to be affiliated with WCAS, then WCAS shall have right to select another individual so long as WCAS provides the Company with written notice of the name of the individual so selected by it (together with such biographical information regarding such individual as the Nominating and Corporate Governance Committee of the Board may request) at least 5 days prior to the date on which the Company distr
- (d) The WCAS Temporary Director shall resign from his or her seat as director of the Company effective as of the close of business on the earlier of the WCAS Directorship Term End Date and December 31, 2009 (it being understood that such seat may be filled by the remaining directors on the Board in accordance with Section 5.4(a) of the By-Laws of the Company). WCAS shall cause the WCAS Temporary Director to comply with this Section 4.1(c).

- (e) The WCAS Permanent Director shall resign from his or her seat as director of the Company effective immediately on the WCAS Directorship Term End Date (it being understood that such seat may be filled by the remaining directors on the Board in accordance with Section 5.4(a) of the By-Laws of the Company). WCAS shall cause the WCAS Permanent Director to comply with this Section 4.1(d).
- (f) Notwithstanding any other provision of this Section 4.1, the Company shall be entitled to (i) excuse the WCAS Permanent Director and the WCAS Temporary Director from any portion of any meeting of the Board, (A) when the Board discusses any matters directly relating to the Securities or the Transaction Documents that could, in the Board's good faith determination, present a conflict for the WCAS Permanent Director or the WCAS Temporary Director; or (B) if participation by the WCAS Permanent Director or the WCAS Temporary Director in such meeting would, in the Board's good faith determination, reasonably be expected to waive the attorney/client privilege relating to communications between the Company and its legal advisors with respect to such matters and (ii) withhold information from the WCAS Permanent Director and the WCAS Temporary Director delivered to the Board prior to any meeting of the Board if the Company believes there is a reasonable likelihood that the receipt of such information by the WCAS Permanent Director or the WCAS Temporary Director would, in the Board's good faith determination, create a conflict of interest for the WCAS Permanent Director of the WCAS Temporary Director in respect of the Securities or the Transaction Documents or, if privileged, would, in the Board's good faith determination, reasonably be expected to effectively waive the attorney/client privilege of the Company with respect thereto.

Section 4.2 Vacancy and Removal.

If, during the period commencing on the date hereof and ending on the WCAS Directorship Term End Date, a vacancy of the Board seat occupied by the WCAS Permanent Director or the WCAS Temporary Director occurs for any reason (including death, resignation or removal) the remaining directors on the Board shall fill such vacancy pursuant to Section 5.4(a) of the By-Laws of the Company with an individual nominated by WCAS who satisfies the requirements of the Nominating and Corporate Governance Committee of the Board, to hold such seat (subject to Section 4.1(d)) until the next annual meeting of the Company at which such seat is up for re-election.

Section 4.3 Board Observation Rights.

- (a) From and after January 1, 2010 until the WCAS Directorship Term End Date, WCAS shall be entitled to designate one (1) observer (the "<u>WCAS Observer</u>") to attend, as a non-voting observer, all meetings (including participation in telephonic meetings) of the Board. The Company shall reimburse the WCAS Observer for his or her reasonable out-of-pocket costs incurred in attending such meetings in person in accordance with the Company's expense reimbursement policy applicable to directors in effect from time to time.
- (b) The Company shall provide the WCAS Observer with (i) notice of all meetings of the Board and (ii) all information delivered to the Directors at the same time such information is distributed to the Board.

(c) Notwithstanding any other provision of this Section 4.3, the Company shall be entitled to (i) excuse the WCAS Observer from any portion of any meeting of the Board, (A) when the Board discusses any matters directly relating to the Securities or the Transaction Documents that could, in the Board's good faith determination, present a conflict for the WCAS Observer; or (B) if the WCAS Observer's participation in such meeting would, in the Board's good faith determination, reasonably be expected to waive the attorney/client privilege related to communications between the Company and its legal advisors and (ii) withhold information from the WCAS Observer delivered to the Board prior to any meeting of the Board if the Company believes there is a reasonable likelihood that the receipt of such information by the WCAS Observer would, in the Board's good faith determination, create a conflict of interest for the WCAS Observer in respect of the Securities or the Transaction Documents or, if privileged, would, in the Board's good faith determination, reasonably be expected to effectively waive the attorney/client privilege of the Company with respect thereto.

Section 4.4 Transfer of Preferred Stock by WCAS.

WCAS's rights under this Article IV (including, without limitation, any right WCAS may have to designate a WCAS Permanent Director, WCAS Temporary Director, or WCAS Observer pursuant to the applicable provisions of this Article IV) shall immediately terminate and expire on the date on which the Original Stockholders cease to hold, in the aggregate, at least two million (2,000,000) shares of Qualifying Securities (as adjusted to reflect stock splits, stock dividends, stock combinations, recapitalizations and like occurrences) (the "WCAS Directorship Term End Date").

ARTICLE V

MISCELLANEOUS

Section 5.1 Entire Agreement.

This Agreement, including the schedules hereto and any other documents referred to herein which form a part hereof, contains the entire understanding of the parties hereto with respect to the subject matter contained herein and therein. This Agreement supersedes all prior agreements and understandings between the parties with respect to such subject matter.

Section 5.2 Table of Contents; Captions.

The table of contents and the Article and Section captions used herein are for reference purposes only, and shall not in any way affect the meaning or interpretation of this Agreement.

Section 5.3 Counterparts.

This Agreement may be executed in two or more counterparts, all of which taken together shall constitute one instrument.

Section 5.4 Notices.

Any notice or other communication required or permitted under this Agreement shall be deemed to have been duly given (i) five (5) Business Days following deposit in the mails if sent by registered or certified mail, postage prepaid, (ii) when sent, if sent by facsimile transmission, if receipt thereof is confirmed by telephone, (iii) when delivered, if delivered personally to the intended recipient and (iv) two (2) Business Days following deposit with a nationally recognized overnight courier service, in each case addressed as follows:

If to the Company, to:

Mobile Mini, Inc. 7420 South Kyrene Road, Suite 101 Tempe, AZ 85283 Fax: (480) 894-6433

Attn: Larry Trachtenberg

and if to any of the Stockholders, to the addresses or facsimile numbers set forth opposite each of their names on Schedule I attached hereto; or such other addresses or number as shall be furnished in writing by any such party.

Section 5.5 Successors and Assigns.

This Agreement shall be binding upon and inure to the benefit of the Company, the Stockholders and their respective successors and Permitted Transferees. Any or all of the rights of a Stockholder under this Agreement may be assigned or otherwise conveyed by any Stockholder only in connection with a Transfer of Equity Securities which is in compliance with this Agreement; provided, that notwithstanding any Permitted Transfer, the rights of WCAS pursuant to Article IV of this Agreement shall not be assignable to any Person (whether or not a Permitted Transferee).

Section 5.6 Governing Law.

The interpretation and construction of this Agreement, and all matters relating hereto, shall be governed by the laws of the State of Delaware, without regard to the principles of conflicts of laws thereof.

Section 5.7 Submission to Jurisdiction.

(a) Each of the parties hereto hereby irrevocably acknowledges and consents that any legal action or proceeding brought with respect to any of the obligations arising under or relating to this Agreement may be brought in the courts of the State of New York, County of New York or in the United States District Court for the Southern District of New York and each of the parties hereto hereby irrevocably submits to and accepts with regard to any such action or proceeding, for itself and in respect of its property, generally and unconditionally, the non-exclusive jurisdiction of the aforesaid courts. Each party hereby further irrevocably waives any claim that any such courts lack jurisdiction over such party, and agrees not to plead or claim, in any legal action or proceeding with respect to this Agreement or the transactions contemplated hereby brought in any of the aforesaid courts, that any such court lacks jurisdiction over such

party. Each party irrevocably consents to the service of process in any such action or proceeding by the mailing of copies thereof by registered or certified mail, postage prepaid, to such party, at its address for notices set forth in Section 5.4, such service to become effective ten (10) days after such mailing. Each party hereby irrevocably waives any objection to such service of process and further irrevocably waives and agrees not to plead or claim in any action or proceeding commenced hereunder or under any other documents contemplated hereby that service of process was in any way invalid or ineffective. Subject to Section 5.7(b), the foregoing shall not limit the rights of any party to serve process in any other manner permitted by law. The foregoing consents to jurisdiction shall not constitute general consents to service of process in the State of New York for any purpose except as provided above and shall not be deemed to confer rights on any Person other than the respective parties to this Agreement.

- (b) Each of the parties hereto hereby waives any right it may have under the laws of any jurisdiction to commence by publication any legal action or proceeding with respect to this Agreement. To the fullest extent permitted by applicable Law, each of the parties hereto hereby irrevocably waives the objection which it may now or hereafter have to the laying of the venue of any suit, action or proceeding arising out of or relating to this Agreement in any of the courts referred to in Section 5.7(a) and hereby further irrevocably waives and agrees not to plead or claim that any such court is not a convenient forum for any such suit, action or proceeding.
- (c) The parties hereto agree that any judgment obtained by any party hereto or its successors or assigns in any action, suit or proceeding referred to above may, in the discretion of such party (or its successors or assigns), be enforced in any jurisdiction, to the extent permitted by applicable Law.
- (d) The parties hereto agree that the remedy at law for any breach of this Agreement may be inadequate and that should any dispute arise concerning any matter hereunder, this Agreement shall be enforceable in a court of equity by an injunction or a decree of specific performance. Such remedies shall, however, be cumulative and nonexclusive, and shall be in addition to any other remedies which the parties hereto may have.
- (e) The prevailing party or parties in any legal action or proceeding brought with respect to any of the obligations arising under or relating to this Agreement shall be entitled to receive from the losing party or parties all costs and expenses, including reasonable counsel fees, incurred by the prevailing party or parties.

Section 5.8 Waiver of Jury Trial.

Each of the Company and each Stockholder hereby waives, to the fullest extent permitted by applicable Law, any right it may have to a trial by jury in respect of any litigation as between the parties directly or indirectly arising out of, under or in connection with this Agreement or the transactions contemplated hereby or disputes relating hereto. Each of the Company and each Stockholder (i) certifies that no representative, agent or attorney of the Company or such Stockholder has represented, expressly or otherwise that the Company or such Stockholder, as the case may be, would not, in the event of litigation, seek to enforce the foregoing waiver and (ii) acknowledges that it, the other Stockholders and the Company have

been induced to enter into this Agreement by, among other things, the mutual waivers and certifications in this Section 5.8.

Section 5.9 Third Party Beneficiaries.

Each party hereto intends that this Agreement shall not benefit or create any right or cause of action in or on behalf of any Person other than the parties hereto.

Section 5.10 Confidentiality.

Each Stockholder hereby agrees that throughout the term of this Agreement it shall keep (and shall use all commercially reasonable efforts to cause its directors, officers, general and limited partners, employees, representatives and outside advisors and its Affiliates to keep) all non-public information received by it relating to the Company (including any such information received prior to the date hereof) confidential except information which (a) becomes known to such Stockholder from a source, other than the Company, its directors, officers, employees, representatives or outside advisors, which source, to the actual knowledge of such Stockholder, is not obligated to the Company to keep such information confidential or (b) is or becomes generally available to the public through no breach of this Agreement by such Stockholder. Each of the Company and each Stockholder agrees that (i) such non-public information may be communicated to the directors, officers, general and limited partners, employees, representatives, outside advisors and Affiliates of such Stockholder and (ii) such Stockholder will use all commercially reasonable to cause its directors, officers, general and limited partners, employees, representatives, outside advisors or Affiliates to keep such non-public information confidential. Notwithstanding the foregoing, a Stockholder may disclose non-public information if required to do so upon request for disclosure pursuant to a federal or state freedom of information statute or by a court of competent jurisdiction or by any governmental agency; provided however, that, to the extent permitted by law, prompt notice of such required disclosure be given to the Company prior to the making of such disclosure so that the Company may seek a protective order or other appropriate remedy. In the event that such protective order or other remedy is not obtained, the Stockholder required to disclose the non-public information will disclose only that portion which such party is legally required to be disclosed and

Section 5.11 Expenses.

The Company shall reimburse each of the respective members of its Board who are not employees of the Company for their reasonable travel and out-of-pocket expenses incurred in connection with their serving on the Board. Employees of the Company who incur expenses in connection with their attendance of meetings of the Board in the performance of their duties shall also be reimbursed in accordance with the Company's usual expense reimbursement policies.

Section 5.12 Amendments; Waivers.

No provision of this Agreement may be amended, modified or waived without the prior written consent of the holders of more than fifty percent (50%) of the issued and

outstanding Qualifying Securities, collectively. Notwithstanding the foregoing, the addition of parties to this Agreement in accordance with its terms shall not be deemed to be an amendment, modification or waiver requiring the consent of any Stockholder.

Section 5.13 No Strict Construction.

The parties hereto have participated jointly in the negotiation and drafting of this Agreement. In the event any ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by all parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

Section 5.14 Specific Performance.

Each of the Company and each Stockholder agrees that irreparable damages would occur to the Company or such Stockholder, as the case may be, if any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that each of the Company and each Stockholder shall be entitled to seek an injunction or injunctions to prevent actual breaches of this Agreement by the Company or the Stockholders, as the case may be, and to enforce specifically the terms and provisions hereof in the courts referenced in Section 5.7 (or, on a preliminary basis in order to preserve the status quo pending a decision of the courts referenced in Section 5.7, or in order to enforce a judgment of the courts referenced in Section 5.7, in any court of competent jurisdiction), in addition to having any other remedies to which the Company or such Stockholder is entitled at law or in equity and without the necessity of proving damages or posting a bond or other security.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first set forth above.

WCAS:

WELSH, CARSON, ANDERSON & STOWE X, L.P.

By: WCAS X Associates LLC, its General Partner

By: /s/ Sanjay Swani
Name: Sanjay Swani
Title: Managing Member

DE NICOLA HOLDINGS, L.P.

By: /s/ Anthony de Nicola

Name: Anthony de Nicola
Title: General Partner

WCAS CAPITAL PARTNERS IV, L.P.

By: WCAS CP IV Associates LLC, its General Partner

By: /s/ Sanjay Swani
Name: Sanjay Swani
Title: Managing Member

WCAS MANAGEMENT CORPORATION

By: /s/ Sanjay Swani

Name: Sanjay Swani
Title: Vice President

CALIFORNIA STATE TEACHERS' RETIREMENT SYSTEM

By:

/s/ Christopher J. Ailman Christopher J. Ailman Chief Investment Officer Name: Title:

FOXKIRK, LLC

By: NML Securities Holdings, LLC, its Sole Member

By: The Northwestern Mutual Life Insurance Company, its Sole Member

By: /s/ Richard A. Strait

Name: Richard A. Strait

Title: Its Authorized Representative

/s/ Ronald F. Valenta

RONALD F. VALENTA

GRANDVIEW HOLDINGS, L.P.

By: /s/ James S. Robertson
Name: James S. Robertson
Title: General Partner

/s/ Douglas A. Waugaman

DOUGLAS A. WAUGAMAN

/s/ Gilbert Gomez

GILBERT GOMEZ

/s/ James Martell

JAMES MARTELL

LEHMAN BROTHERS CO-INVESTMENT PARTNERS L.P.

By: Lehman Brothers Co-Investment Associates L.P., its general partner

By: Lehman Brothers Co-Investment Associates L.L.C., its general partner

By: /s/ Ashvin Rao

Name: Ashvin Rao

Title: Senior Vice President

LEHMAN BROTHERS CO-INVESTMENT CAPITAL PARTNERS L.P.

By: LB I Group Inc., its general partner

By: /s/ Ashvin Rao

Name: Ashvin Rao

Title: Senior Vice President

LEHMAN BROTHERS CO-INVESTMENT GROUP L.P.

By: LB I Group Inc., its general partner

By: /s/ Ashvin Rao

Name: Ashvin Rao

Title: Senior Vice President

THE WILSON TRUST DATED AUGUST 24, 2000

/s/ Christopher A. Wilson Christopher A. Wilson Trustee By:

Name:

Title:

/s/ William Armstead

WILLIAM ARMSTEAD

/s/ Jeffrey Kluckman

JEFFREY KLUCKMAN

/s/ Jody Miller

JODY MILLER

/s/ Allan Villegas

ALLAN VILLEGAS

MOBILE MINI, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg
Title: Executive Vice President

ABL CREDIT AGREEMENT

among

MOBILE MINI, INC.,

THE OTHER BORROWERS AND GUARANTORS PARTY HERETO,

THE VARIOUS LENDERS PARTY HERETO

and

DEUTSCHE BANK AG NEW YORK BRANCH, as ADMINISTRATIVE AGENT

Dated as of June 27, 2008

DEUTSCHE BANK SECURITIES INC. and BANC OF AMERICA SECURITIES LLC, as JOINT LEAD ARRANGERS

DEUTSCHE BANK SECURITIES INC., BANC OF AMERICA SECURITIES LLC and J.P. MORGAN SECURITIES INC.,

as JOINT BOOKRUNNERS

BANK OF AMERICA, N.A., as SYNDICATION AGENT,

And

JPMORGAN CHASE BANK, N.A., ING CAPITAL LLC and THE BANK OF NOVA SCOTIA, as CO-DOCUMENTATION AGENTS

Execution Copy

TABLE OF CONTENTS

| SECTION 1. | Definitions and Accounting Terms | Page 1 |
|------------|--|--------|
| 1.01 | Defined Terms | 1 |
| SECTION 2. | Amount and Terms of Credit | 49 |
| 2.01 | The Commitments | 49 |
| 2.02 | Minimum Amount of Each Borrowing | 54 |
| 2.03 | Notice of Borrowing | 54 |
| 2.04 | Disbursement of Funds | 55 |
| 2.05 | Notes | 57 |
| 2.06 | Conversions | 58 |
| 2.07 | Pro Rata Borrowings | 58 |
| 2.08 | Interest | 58 |
| 2.09 | Interest Periods | 59 |
| 2.10 | Increased Costs, Illegality, etc. | 60 |
| 2.11 | Compensation | 62 |
| 2.12 | Change of Lending Office | 62 |
| 2.13 | Replacement of Lenders | 62 |
| 2.14 | US Company as Agent for US Borrowers and UK Company as Agent for UK Borrower | 63 |
| 2.15 | Incremental Commitments | 65 |
| 2.16 | UK Revolving Loans; Intra-Lender Issues | 66 |
| 2.17 | Equivalent Amount | 72 |
| SECTION 3. | Letters of Credit | 72 |
| 3.01 | US Letters of Credit | 72 |
| 3.02 | UK Letters of Credit | 73 |
| 3.03 | Maximum US Letter of Credit Outstandings; Final Maturities | 74 |
| 3.04 | Maximum UK Letter of Credit Outstandings; Final Maturities | 74 |
| 3.05 | US Letter of Credit Requests; Minimum Stated Amount | 74 |
| 3.06 | UK Letter of Credit Requests; Minimum Stated Amount | 75 |
| 3.07 | US Letter of Credit Participations | 76 |
| 3.08 | UK Letter of Credit Participations | 77 |
| 3.09 | Agreement to Repay US Letter of Credit Drawings | 79 |
| 3.10 | Agreement to Repay UK Letter of Credit Drawings | 80 |
| 3.11 | Increased Costs — US Letters of Credit | 81 |
| 3.12 | Increased Costs — UK Letters of Credit | 81 |
| SECTION 4. | Commitment Commission; Fees; Reductions of Commitment | 82 |
| 4.01 | Fees | 82 |
| | i | |

| 4.02 4.03 | Voluntary Termination of Unutilized Commitments Mandatory Reduction of Commitments | <u>Page</u> 83 83 |
|--------------|---|-------------------------|
| SECTION 5. | Prepayments; Payments; Taxes | 84 |
| | | |
| 5.01 | Voluntary Prepayments | 84 |
| 5.02 | Mandatory Repayments; Cash Collateralization | 85 |
| 5.03 | Method and Place of Payment | 88 |
| 5.04 | Net Payments — US Borrowers | 90 |
| 5.05 | Tax Gross Up and Indemnities — UK Subsidiaries | 92 |
| SECTION 6. | Conditions Precedent to Credit Events on the Initial Borrowing Date | 97 |
| 6.01 | Effective Date; Notes | 97 |
| 6.02 | Officer's Certificate | 98 |
| 6.03 | Opinions of Counsel | 98 |
| 6.04 | Company Documents; Proceedings; etc. | 99 |
| 6.05 | Consummation of the Transactions; etc. | 99 |
| 6.06 | No Acquisition Agreement Material Adverse Effect | 100 |
| 6.07 | No Material Adverse Effect on US Company | 100 |
| 6.08 | US Pledge Agreement | 100 |
| 6.09 | UK Share Charge | 100 |
| 6.10 | US Security Agreement | 100 |
| 6.11 | UK Debenture | 101 |
| 6.12 | Financial Statements; Pro Forma Balance Sheet; Projections; etc. | 101 |
| 6.13 | Solvency Certificate; Insurance Certificates | 101 |
| 6.14 | Fees, etc. | 102 |
| 6.15 | Initial Borrowing Base Certificate; etc | 102 |
| 6.16 | No Defaults under Senior Note Indentures | 102 |
| 6.17 | Patriot Act | 102 |
| 6.18 | Notice of Borrowing | 102 |
| SECTION 7. | Conditions Precedent to All Credit Events | 102 |
| 7.01 | No Default; Representations and Warranties | 103 |
| 7.02 | Notice of Borrowing; Letter of Credit Request | 103 |
| SECTION 8. | Representations, Warranties and Agreements | 104 |
| 8.01 | Organization and Qualification | 104 |
| 8.02 | Power and Authority; No Violation | 104 |
| 8.03 | Legally Enforceable Agreement | 104 |
| 8.04 | Capital Structure | 104 |
| 8.05 | Names | 105 |
| 8.06 | Business Locations; Agent for Process | 105 |
| 8.07 | Title to Properties; Priority of Liens | 105 |
| 8.08 | Accounts | 105 |
| 8.09 | Equipment | 106 |
| | | |

[ABL Credit Agreement]

| | | Page |
|------------|--|------|
| 8.10 | Financial Statements; Financial Condition; Undisclosed Liabilities; Projections | 106 |
| 8.11 | Full Disclosure | 108 |
| 8.12 | Surety Obligations | 108 |
| 8.13 | Tax Returns and Payments | 108 |
| 8.14 | Dutch Credit Parties and Luxembourg Subsidiary | 108 |
| 8.15 | Intellectual Property, etc | 108 |
| 8.16 | Government Consents | 109 |
| 8.17 | Compliance with Laws | 109 |
| 8.18 | Restrictions | 109 |
| 8.19 | Litigation | 109 |
| 8.20 | No Defaults | 109 |
| 8.21 | Leases | 109 |
| 8.22 | [Intentionally Omitted] | 110 |
| 8.23 | Use of Proceeds; Margin Regulations | 110 |
| 8.24 | Compliance with ERISA | 110 |
| 8.25 | Trade Relations | 111 |
| 8.26 | Security Documents | 111 |
| 8.27 | Investment Company Act | 112 |
| 8.28 | Representations and Warranties in Other Documents | 112 |
| 8.29 | Environmental Matters | 112 |
| 8.30 | Employment and Labor Relations | 113 |
| 8.31 | Indebtedness | 113 |
| 8.32 | Insurance | 113 |
| 8.33 | Employee Benefit Plans; Non-Compete Agreements; Collective Bargaining Agreements; Existing Indebtedness Agreements | 113 |
| 8.34 | Anti-Terrorism Laws | 114 |
| 8.35 | UK Financial Assistance | 114 |
| 8.36 | UK Pensions | 114 |
| SECTION 9. | Affirmative Covenants | 114 |
| 9.01 | Information Covenants | 114 |
| 9.02 | Books, Records and Inspections; Field Examinations; Appraisals; Records and Reports of Inventory, Machinery and | |
| | Equipment | 117 |
| 9.03 | Maintenance of Property; Insurance | 118 |
| 9.04 | Administration of Equipment; Maintenance of Equipment | 119 |
| 9.05 | Existence; Franchises | 120 |
| 9.06 | Compliance with Statutes, etc. | 120 |
| 9.07 | Compliance with Environmental Laws | 120 |
| 9.08 | ERISA | 121 |
| 9.09 | End of Fiscal Years; Fiscal Quarters | 122 |
| 9.10 | Performance of Obligations | 122 |
| 9.11 | Payment of Taxes | 122 |
| 9.12 | Use of Proceeds | 122 |
| 9.13 | Additional Security; Further Assurances; etc. | 123 |
| 9.14 | Convertible Preferred Stock | 124 |

| 9.15 | Projections | <u>Page</u> 124 |
|-------------|--|--------------------|
| 9.16 | Landlord, Processor and Storage Agreements | 124 |
| 9.17 | Deposit and Brokerage Accounts | 124 |
| 9.18 | Credit Party Financial Statements | 125 |
| 9.19 | Qualifying Derivative Obligations | 125 |
| 9.20 | Centre of Main Interest | 125 |
| 9.21 | Administration of Accounts | 125 |
| 9.22 | Completion of Mobile Storage Acquisition on the Closing Date | 125 |
| | Completion of Mobile Storage Aequisition on the Grossing Bute | 120 |
| SECTION | | |
| 10. | Negative Covenants | 126 |
| 10.01 | Capital Expenditures | 126 |
| 10.02 | Liens | 126 |
| 10.03 | Sale of Assets | 128 |
| 10.04 | Restricted Payments | 128 |
| 10.05 | Indebtedness | 129 |
| 10.06 | Contingent Obligations | 131 |
| 10.07 | Advances, Investments and Loans | 131 |
| 10.08 | Transactions with Affiliates | 132 |
| 10.09 | [Reserved] | 132 |
| 10.10 | [Reserved] | 132 |
| 10.11 | Additional Negative Pledges | 132 |
| 10.12 | No Subsidiaries | 132 |
| 10.13 | Operating Leases; Off-Balance Sheet Financing | 133 |
| 10.14 | Permitted Acquisitions | 134 |
| 10.15 | Modifications of Certificate of Incorporation, By-Laws and Certain Other Agreements, etc | 135 |
| 10.16 | Limitation on Certain Restrictions on Subsidiaries | 136 |
| 10.17 | Limitation on Issuance of Equity Interests | 136 |
| 10.18 | Business; etc | 136 |
| 10.19 | [Reserved] | 136 |
| 10.20 | No Additional Deposit Accounts; etc | 136 |
| 10.21 | [Reserved] | 137 |
| 10.22 | Tax Consolidation | 137 |
| 10.23 | Fiscal Year End | 137 |
| 10.24 | Applicability of Financial Covenants | 137 |
| 10.25 | Fixed Charge Coverage Ratio | 137 |
| 10.26 | Debt Ratio | 138 |
| 10.27 | Minimum Utilization | 138 |
| SECTION 11. | Events of Default | 139 |
| 11.01 | Payment of Obligations | 139 |
| 11.02 | Misrepresenations | 139 |
| 11.03 | Breach of Specific Covenants | 139 |
| 11.04 | Breach of Other Covenants | 139 |
| 11.05 | Default Under Other Agreements | 139 |

| | | Page |
|-------------|--|------|
| 11.06 | Failure of Enforceability of Credit Documents; Security | 139 |
| 11.07 | [Reserved] | 140 |
| 11.08 | Insolvency and Related Proceedings | 140 |
| 11.09 | Business Disruption; Condemnation | 140 |
| 11.10 | ERISA | 141 |
| 11.11 | Guarantee | 141 |
| 11.12 | Criminal Forfeiture | 141 |
| 11.13 | Judgments | 141 |
| 11.14 | Change of Control | 142 |
| ECTION | | |
| 2. | The Administrative Agent and the Collateral Agent | 142 |
| 12.01 | Appointment | 142 |
| 12.02 | Nature of Duties | 142 |
| 12.03 | Lack of Reliance on the Administrative Agent and the Collateral Agent | 143 |
| 12.04 | Certain Rights of the Agents | 143 |
| 12.05 | Reliance | 143 |
| 12.06 | Indemnification | 144 |
| 12.07 | The Administrative Agent in its Individual Capacities | 144 |
| 12.08 | Holders | 144 |
| 12.09 | Resignation by the Administrative Agent | 144 |
| 12.10 | Collateral Matters | 145 |
| 12.11 | Delivery of Information | 146 |
| 12.12 | Withholding | 146 |
| 12.13 | Delegation of Duties | 147 |
| 12.14 | Quebec Security | 147 |
| SECTION 13. | Miscellaneous | 148 |
| 13.01 | Payment of Expenses, etc. | 148 |
| 13.02 | Right of Setoff | 149 |
| 13.03 | Notices | 149 |
| 13.04 | Benefit of Agreement; Assignments; Participations | 150 |
| 13.05 | No Waiver; Remedies Cumulative | 152 |
| 13.06 | Payments Pro Rata | 152 |
| 13.07 | Calculations; Computations | 153 |
| 13.08 | GOVERNING LAW; SUBMISSION TO JURISDICTION; VENUE; WAIVER OF JURY TRIAL | 153 |
| 13.09 | Counterparts | 154 |
| 13.10 | Effectiveness | 155 |
| 13.11 | Headings Descriptive | 155 |
| 13.12 | Amendment or Waiver; etc. | 155 |
| 13.13 | Survival | 156 |
| 13.14 | Domicile of Loans | 156 |
| 13.15 | Register | 157 |
| 13.16 | Confidentiality | 157 |
| 13 17 | Patriot Act | 158 |

| 13.18 | Release of Borrowers | <u>Page</u> 158 |
|-------------|---|-----------------|
| SECTION | | |
| 14. | Nature of Borrower Obligations | 158 |
| 14.01 | Nature of Borrower Obligations | 158 |
| 14.02 | Independent Obligation | 158 |
| 14.03 | Authorization | 158 |
| 14.04 | Reliance | 159 |
| 14.05 | Contribution; Subrogation | 159 |
| 14.06 | Waiver | 159 |
| SECTION 15. | Guarantee | 159 |
| 15.01 | The Guarantees | 159 |
| 15.02 | Obligations Unconditional | 160 |
| 15.03 | Reinstatement | 162 |
| 15.04 | Subrogation; Subordination | 163 |
| 15.05 | Remedies | 163 |
| 15.06 | Instrument for Payment of Money | 163 |
| 15.07 | Continuing Guarantee | 164 |
| 15.08 | General Limitation on Guarantee Obligations | 164 |
| 15.09 | Release of Guarantors | 165 |
| 15.10 | Right of Contribution | 165 |

SCHEDULES

| Schedule 1.01(a) | Lenders and Commitments |
|------------------|-------------------------|
| | |

Schedule 1.01(b) Locations — United States and Canada

Schedule 1.01(c)

Schedule 1.01(d)

Schedule 1.01(e)

Schedule 1.01(e)

Schedule 1.01(f)

Schedule 1.01(g)

Schedule 1.01(g)

Schedule 3.01(c)

Schedule 3.01(c)

Schedule 8.01

Locations — United Kingdom

Eligible Real Property

Consolidated EBITDA

Consolidated Net Cash Flow

Qualified Derivative Obligations

Existing US Letters of Credit

Qualifications to do Business

Schedule 8.04 Capital Structure

Schedule 8.05 Names

Schedule 8.06 Business Locations
Schedule 8.12 Surety Obligations
Schedule 8.18 Restrictions
Schedule 8.19 Litigation
Schedule 8.21 Leases
Schedule 8.24 ERISA Plans

Schedule 8.25 Business Relationships
Schedule 8.31 Continuing Indebtedness

Schedule 8.32 Insurance

Schedule 8.32 Employee Benefit Plans

Schedule 10.02 Liens
Schedule 10.05 Indebtedness
Schedule 10.09 Bank Accounts

Schedule 10.07 Deposits with Financial Institutions

Schedule 10.08 Transactions with Affiliates

Schedule 10.13 Operating Leases
Schedule 10.12 Subsidiaries
Schedule 10.20 Deposit Accounts

EXHIBITS

Exhibit A-1 Notice of Borrowing

Exhibit A-2 Notice of Conversion/Continuation
Exhibit B-1 Form of US Revolving Note
Exhibit B-2 Form of US Swingline Note
Exhibit B-3 Form of UK Revolving Note
Exhibit B-4 Form of UK Swingline Note

Exhibit C Form Assignment and Assumption Agreement Exhibit D Form of Incremental Commitment Agreement

Exhibit E Form of Joinder Agreement

Exhibit F Form of US Letter of Request

Exhibit G Form of UK Letter of Request

Exhibit H Form of Section 5.04(b)(ii) Certificate

Exhibit I-1-A Form of Squire Sanders Opinion (Mobile Mini Entities)
Exhibit I-1-B Form of Squire Sanders Opinion (Mobile Storage Entities)

Exhibit I-2 Form of White & Case Opinion
Exhibit I-3 Form of Kirkland & Ellis Opinion

Exhibit I-4-A Form of UK Opinion (Mobile Mini Entities)

Exhibit I-5 Form of Luxembourg Opinion Exhibit I-6 Form of Canadian Opinion

Exhibit I-7 Form of Special Delaware Counsel Opinion Exhibit J Form of Credit Party Secretary's Certificate

Exhibit K Form of US Pledge Agreement
Exhibit L Form of UK Share Charge
Exhibit M US Security Agreement
Exhibit N UK Debenture

Exhibit O Form of Solvency Certificate
Exhibit P Form of Compliance Certificate
Exhibit Q Form of Borrowing Base Certificate

Exhibit R UK Intercreditor Deed

ABL CREDIT AGREEMENT, dated as of June 27, 2008, among Mobile Mini, Inc., a Delaware corporation (the "<u>US Company</u>" and, together with each other entity that executes this Agreement as a US Borrower or that becomes a US Borrower pursuant to <u>Section 9.13(f)</u>, <u>Section 9.13(g)</u>, or <u>Section 10.12</u>, collectively, the "<u>US Borrowers</u>", and each, a "<u>US Borrower</u>"), Ravenstock MSG Limited, a limited liability company incorporated in England and Wales (the "<u>UK Company</u>"), Mobile Mini UK Limited, a corporation incorporated in England and Wales ("<u>Mobile Mini UK</u>" and together with UK Company and each other entity that executes this Agreement as a UK Borrower or that becomes a UK Borrower pursuant to <u>Section 10.12</u>, collectively, the "<u>UK Borrowers</u>", and each, a "<u>Borrower</u>" and, together with each US Borrower, collectively, the "<u>Borrowers</u>", and each, a "<u>Borrower</u>"), the Guarantors party hereto from time to time, the Lenders party hereto from time to time and Deutsche Bank AG New York Branch, as Administrative Agent. All capitalized terms used herein and defined in Section 1 are used herein as therein defined.

WITNESSETH:

WHEREAS, subject to and upon the terms and conditions set forth herein, the Joint Lead Arrangers have arranged, and the Lenders are willing to make available to the Borrowers, the senior secured revolving credit facility provided for herein;

NOW, THEREFORE, IT IS AGREED:

SECTION 1. Definitions and Accounting Terms.

1.01 <u>Defined Terms</u>. As used in this Agreement, the following terms shall have the following meanings (such meanings to be equally applicable to both the singular and plural forms of the terms defined):

"Account" shall mean an "account" as such term is defined in Article 9 of the UCC, and any and all supporting obligations in respect thereof.

"Account Debtor" shall mean each Person who is obligated on an Account.

"Acquisition" shall mean (i) the acquisition by US Company or any of its Subsidiaries of all of the issued and outstanding Securities or other equity interests of a Person, (ii) the acquisition by US Company or any of its Subsidiaries of all or substantially all of the assets of a Person or a line of business of a Person or (iii) the merger or consolidation of US Company or any of its Subsidiaries with a Person other than a Person that was a Subsidiary of US Company or such Subsidiary immediately prior to such merger.

"Acquisition Documents" shall mean collectively, (i) the Merger Agreement (ii) the Joinder Agreement in the form attached to the Merger Agreement as Exhibit A thereto, (iii) the Escrow Agreement in the form attached to the Merger Agreement as Exhibit B thereto, (iv) the Stockholders Agreement, in the form attached to the Merger Agreement as Exhibit C thereto, (v) the certificate of designations for the Convertible Preferred Stock in the form attached to the Merger Agreement as Exhibit D, (vi) the Amendment to the Amended and Restated Certificate of Incorporation of US Company in the form attached to the Merger Agreement as Exhibit E thereto, (vii) the agreements relating to the Subsequent Mergers, and all other agreements and documents relating to the Mobile Storage Acquisition.

"Additional Security Documents" shall have the meaning provided in Section 9.13.

- "Adjustable Applicable Margins" shall have the meaning provided in the definition of Applicable Margin.
- "Administrative Agent" shall mean Deutsche Bank AG New York Branch, in its capacity as Administrative Agent for the Lenders hereunder and under the other Credit Documents, and shall include any successor to the Administrative Agent appointed pursuant to Section 12.09.
- "Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlling (including, but not limited to, all directors and officers of such Person), controlled by, or under direct or indirect common control with, such Person. A Person shall be deemed to control another Person if such Person possesses, directly or indirectly, the power (a) to vote 15% or more of the securities having ordinary voting power for the election of directors (or equivalent governing body) of such Person or (b) to direct or cause the direction of the management and policies of such other Person, whether through the ownership of voting securities, by contract or otherwise; provided, however, that none of the Administrative Agent, any Lender or any of their respective Affiliates shall be considered an Affiliate of US Company or any Subsidiary thereof.
 - "Agent Advance" shall have the meaning provided in Section 2.01(h).
 - "Agent Advance Period" shall have the meaning provided in Section 2.01(h).
 - "Agents" shall mean and include the Administrative Agent, the Collateral Agent, the Syndication Agent and the Co-Documentation Agents.
 - "Aggregate Borrowing Base" shall mean as of any date of determination thereof, an amount equal to the lesser of:
- (i) the sum of (A) the amount calculated under the definition of US Borrowing Base; plus (B) an amount equal to the lesser of (x) the UK Maximum Amount and (y) the amount calculated under the definition of UK Borrowing Base (excluding from such calculation, subclause (i) thereof); or
 - (ii) the amount permitted to be outstanding under this Agreement by each of the Senior Note Indentures.
 - "Aggregate Exposure" shall mean, at any time, the sum of (a) the Aggregate UK Exposure and (b) the Aggregate US Exposure.
- "Aggregate UK Exposure" shall mean, at any time, the sum of (a) the aggregate principal amount of all UK Revolving Loans then outstanding, (b) the aggregate amount of all UK Letter of Credit Outstandings at such time (exclusive of UK Letter of Credit Outstandings which are repaid with the proceeds of, and simultaneously with the incurrence of, the respective incurrence of UK Revolving Loans), and (c) except for purposes of calculating the Applicable Commitment Commission Percentage, the aggregate principal amount of all UK Swingline Loans then outstanding (exclusive of UK Swingline Loans which are repaid with the proceeds of, and simultaneously with the incurrence of, the respective incurrence of UK Revolving Loans).
- "Aggregate US Exposure" shall mean, at any time, the sum of (a) the aggregate principal amount of all US Revolving Loans then outstanding, (b) the aggregate amount of all US Letter of Credit Outstandings at such time (exclusive of US Letter of Credit Outstandings which are repaid with the proceeds of, and simultaneously with the incurrence of, the respective incurrence of US Revolving Loans), and (c) except for purposes of calculating the Applicable Commitment Commission Percentage,

the aggregate principal amount of all US Swingline Loans then outstanding (exclusive of US Swingline Loans which are repaid with the proceeds of, and simultaneously with the incurrence of, the respective incurrence of US Revolving Loans).

"Agreement" shall mean this Credit Agreement, as modified, supplemented, amended, restated (including any amendment and restatement hereof), extended and/or renewed from time to time.

"Applicable Commitment Commission Percentage" shall mean (i) for each day on which the Aggregate Exposure is less than or equal to 50% of the Total Revolving Loan Commitment, 0.375%, and (ii) for each day during which the Aggregate Exposure exceeds 50% of the Total Revolving Loan Commitment, 0.25%.

"Applicable Law" shall mean all laws, rules and regulations applicable to the Person, conduct, transaction, covenant, Credit Document or other material contract in question, including all applicable common law and equitable principles; all provisions of all applicable state, federal and foreign constitutions, statutes, rules, regulations and legally enforceable orders of governmental bodies; and legally enforceable orders, judgments and decrees of all courts and arbitrators, in each case, in any jurisdiction.

"Applicable Margin" shall mean, for the period from the Initial Borrowing Date until the first Start Date thereafter, a percentage per annum equal to, in the case of

- (a) US Revolving Loans maintained as (i) Base Rate Loans, 1.00%, and (ii) US LIBOR Loans, 2.50%;
- (b) US Swingline Loans, 1.00%;
- (c) UK Revolving Loans, 2.50%, and
- (d) UK Swingline Loans, 2.50%.

From and after the first day of the calendar month immediately following the delivery of any certificate delivered in accordance with the first sentence of the following paragraph indicating an entitlement to a different margin for any Type of Loan than that described in the immediately preceding sentence (each, a "Start Date") to and including the applicable End Date described below, the Applicable Margins for such Type of Loan (hereinafter, the "Adjustable Applicable Margins") shall be those set forth below opposite the Debt Ratio indicated to have been achieved in any certificate delivered in accordance with the following sentence:

| Level | Debt Ratio | Revolving Loan LIBOR/EURIBOR and UK Swingline Loan Margin | US Revolving Loan and US Swingline Loan Base Rate Margin |
|-------|--|---|--|
| I | Greater than 4.50 to 1.00 | 2.75% | 1.25% |
| II | Greater than 3.50 to 1.00 but less than or equal to 4.50 to 1.00 | 2.50% | 1.00% |
| III | Less than or equal to 3.50 to 1.00 | 2.25% | 0.75% |
| | -3- | | |

The Debt Ratio used in a determination of Adjustable Applicable Margins shall be determined based on the delivery of a certificate of US Company (each, a "Quarterly Pricing Certificate") by an Authorized Officer of US Company to the Administrative Agent (with a copy to be sent by the Administrative Agent to each Lender), within 45 days after the last day of any fiscal quarter of US Company (or within 90 days after the last day of the fourth fiscal quarter of US Company), which certificate shall set forth the calculation of the Debt Ratio as at the last day of the test period ended immediately prior to the relevant Start Date and the Adjustable Applicable Margins which shall be thereafter applicable (until same are changed or cease to apply in accordance with the following sentences). The Adjustable Applicable Margins so determined shall apply, except as set forth in the succeeding sentence, from the relevant Start Date to the first day of the calendar month immediately following the date on which the next Quarterly Pricing Certificate is delivered to the Administrative Agent (the "End Date"); provided that if no such subsequent Quarterly Pricing Certificate is delivered on or prior to the date which is 45 days (or 90 days in the case of the fourth fiscal quarter of the US Company) following the last day of the test period in which the previous Start Date occurred, a new Start Date shall be deemed to have commenced and the Adjustable Applicable Margins shall automatically adjust to those that correspond to a Debt Ratio at Level I (such Adjustable Applicable Margins as so determined, the "Highest Adjustable Applicable Margins"), effective prospectively from such date until the next End Date. Notwithstanding anything to the contrary contained above in this definition, (i) the Applicable Margins and the Adjustable Applicable Margins shall be the Highest Adjustable Applicable Margins at all times during which there shall exist any Default or Event of Default and (ii) for the period from the Initial Borrowing Date th

Notwithstanding anything to the contrary contained above in this definition or elsewhere in this Agreement, if it is subsequently determined that the Debt Ratio set forth in any Quarterly Pricing Certificate delivered to the Administrative Agent is inaccurate for any reason and the result thereof is that the Lenders received interest or fees for any period based on an Adjustable Applicable Margin that is less than that which would have been applicable had the Debt Ratio been accurately determined, then, for all purposes of this Agreement, the "Adjustable Applicable Margin" for any day occurring within the applicable period shall retroactively be deemed to be the relevant percentage as based upon the accurately determined Debt Ratio for such period, and any shortfall in the interest or fees theretofore paid by the Borrowers for the relevant periods pursuant to Sections 2.08 and 4.01(b) as a result of the miscalculation of the Debt Ratio shall be deemed to be (and shall be) due and payable under the relevant provisions of Section 2.08 or 4.01(b), as applicable, within two Business Days of discovery of such inaccuracy (and shall remain due and payable until paid in full, together with all amounts owing under Section 2.08(d), in accordance with the terms of this Agreement).

"Appraised Fair Market Value" shall mean, with respect to any Real Property, the price at which a willing buyer, not an Affiliate of the seller, and a willing seller who does not have to sell, would agree to purchase and sell such Real Property, as determined by an Appraiser in an appraisal in form and substance reasonably satisfactory to Administrative Agent.

"Appraiser" shall mean an appraiser employed by Administrative Agent or an independent third party appraiser engaged by Administrative Agent, at US Company's expense.

"Approved Credit Support" shall mean an Account is (a) supported by an irrevocable letter of credit satisfactory to the Administrative Agent, in its Permitted Discretion (as to form, substance, and issuer or domestic confirming bank), that has been delivered to the Administrative Agent and is

directly drawable by the Administrative Agent, or (b) is covered by credit insurance in form, substance, and amount, and by an insurer, satisfactory to the Administrative Agent, in its Permitted Discretion.

- "Arrangers" shall mean collectively, Deutsche Bank Securities Inc. and Banc of America Securities LLC.
- "Asset Sale" shall have the meaning set forth in Section 10.03.
- "Assignment and Assumption Agreement" shall mean an Assignment and Assumption Agreement substantially in the form of Exhibit C.
- "Authorized Officer" shall mean, with respect to (a) delivering Notices of Borrowing, Notices of Conversion/Continuation and similar notices, any person or persons that has or have been authorized by the board of directors (or equivalent governing body) of the respective Borrower to deliver such notices pursuant to this Agreement and for which an appropriate incumbency certificate in form and substance satisfactory to the Administrative Agent has been delivered to the Administrative Agent, the Swingline Lender or the respective Issuing Lender, (b) delivering financial information and officer's certificates pursuant to this Agreement, the chief financial officer or the chief accounting officer of the US Company or the respective Borrower, and (c) any other matter in connection with this Agreement or any other Credit Document, any officer (or a person or persons so designated by any two officers) of the US Company or the respective Borrower.

"Bank" shall mean Deutsche Bank AG New York Branch.

"Banking Product Obligations" of the Credit Parties means any and all obligations of the Credit Parties, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired (including all renewals, extensions and modifications thereof and substitutions therefor) in connection with Banking Products of which the Administrative Agent has received written notice.

"Banking Products" means each and any of the following bank services provided to any Credit Party (or the LKE Qualified Intermediary with respect to a LKE Joint Account) by any Lender or any of its Affiliates: (a) commercial credit cards, (b) stored value cards and (c) treasury management services (including, without limitation, controlled disbursement, automated clearinghouse transactions, E-payables or comparable services, return items, overdrafts and interstate depository network services).

"Bank Product Reserve" shall mean a reserve established by the Administrative Agent from time to time in respect of the Borrowing Base Parties' liabilities or potential liabilities as part of their cash management system (including, without limitation, liabilities related to Banking Products) such as, but not limited to, reserves for returned items, customary charges for maintaining Deposit Accounts and similar items.

"Bankruptcy Code" shall mean Title 11 of the United States Code entitled "Bankruptcy" as now or hereafter in effect, or any successor thereto.

"Base Rate" shall mean, at any time, the higher of (a) the Prime Lending Rate at such time and (b) 1/2 of 1% in excess of the overnight Federal Funds Rate at such time.

"Base Rate Loan" shall mean (a) each US Swingline Loan and (b) each US Revolving Loan designated or deemed designated as such by the relevant Borrower at the time of the incurrence thereof or conversion thereto determined by reference to the Base Rate.

- "Board" shall mean the Board of Governors of the Federal Reserve System of the United States.
- "Borrower" and "Borrowers" shall have the meaning provided in the first paragraph of this Agreement.
- "Borrowing" shall mean a UK Borrowing or a US Borrowing.
- "Borrowing Base Certificate" shall have the meaning provided in Section 9.01(j).
- "Borrowing Base Party" shall mean each US Borrowing Base Party and each UK Borrowing Base Party.
- "Business Day" shall mean (a) for all purposes other than as covered by clause (b) below, any day except Saturday, Sunday and any day which shall be in New York, New York, and/or London, England, a legal holiday or a day on which banking institutions are authorized or required by law or other government action to close and (b) with respect to all notices and determinations in connection with, and payments of principal and interest on, (i) LIBOR Loans, any day which is a Business Day described in clause (a) above and which is also a day for trading by and between banks in US Dollar and Pounds Sterling deposits in the interbank eurodollar market and (ii) EURIBOR Loans, any TARGET Day.
 - "Cabin Fleet Inventory" shall mean timber accommodation units which are included in UK Borrowers' lease fleet.
 - "Canadian Dollars" and the sign "CA\$" shall each mean freely transferable lawful money of Canada.
- "Canadian Priority Payables" shall mean, at any time, with respect to any Borrowing Base Party which has employees in Canada or otherwise carries on business in Canada or which leases, sells or otherwise owns goods in Canada or has Accounts with Account Debtors located in Canada:
 - (i) the amount past due and owing by such Borrowing Base Party, or the accrued amount for which such Borrowing Base Party has an obligation to remit to a Governmental Authority in Canada or in any province, municipality or other political subdivision thereof ("Canadian Governmental Authority") or other Person pursuant to any applicable law, rule or regulation, in respect of (a) pension fund obligations; (b) unemployment insurance; (c) goods and services taxes, sales taxes, employee income taxes and other taxes payable or to be remitted or withheld; (c) workers' compensation; (e) vacation pay; and (f) other like charges and demands; in each case, in respect of which any Canadian Governmental Authority or other Person may claim a security interest, lien, trust or other claim ranking or capable of ranking in priority to or pari passu with one or more of the Liens granted in the Security Documents; and
 - (ii) the aggregate amount of any other liabilities of such Borrowing Base Party (a) in respect of which a trust has been or may be imposed on any Collateral to provide for payment or (b) which are secured by a security interest, pledge, lien, charge, right or claim on any Collateral; in each case, pursuant to any applicable law, rule or regulation and which trust, security interest, pledge, lien, charge, right or claim ranks or is capable of ranking in priority to or pari passu with one or more of the Liens granted in the Security Documents.
- "Capital Expenditures" shall mean expenditures made or liabilities incurred for the acquisition of any fixed assets (including but not limited to containers) or improvements, replacements,

substitutions or additions thereto which have a useful life of more than one year, including the total principal portion of Capitalized Lease Obligations and that portion of Investments allocable to property, plant or equipment. Capital Expenditures shall exclude (i) new and used manufactured or remanufactured portable container Inventory held for sale, (ii) proceeds of a Casualty Loss applied to the repair or replacement of the property affected by the Casualty Loss and (iii) Inventory or Equipment acquired in a Permitted Acquisition.

"<u>Capitalized Lease Obligation</u>" shall mean any Indebtedness represented by obligations under a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP.

"Cash Equivalents" shall mean, as to any Person, (a) securities issued or directly and fully guaranteed or insured by the United States or the United Kingdom or any agency or instrumentality thereof (provided that the full faith and credit of such country is pledged in support thereof or it otherwise has a equivalent credit rating) having maturities of not more than one year from the date of acquisition, (b) marketable direct obligations issued by any state of the United States or the United Kingdom or any political subdivision of such country or any public instrumentality thereof maturing within one year from the date of acquisition thereof and, at the time of acquisition, having one of the two highest ratings obtainable from either S&P or Moody's, (c) Pounds Sterling or Dollar-denominated time deposits, certificates of deposit and bankers acceptances of any Lender or any commercial bank having, or which is the principal banking subsidiary of a bank holding company having, a long-term unsecured debt rating of at least "A" or the equivalent thereof from S&P or "A2" or the equivalent thereof from Moody's with maturities of not more than one year from the date of acquisition by such Person, (d) repurchase obligations with a term of not more than seven days for underlying securities of the types described in clause (a) above entered into with any bank meeting the qualifications specified in clause (c) above, (e) commercial paper denominated in Pounds Sterling or Dollars and not convertible or exchangeable into any other securities issued by any Person incorporated in the United States or the United Kingdom rated at least A-1 or the equivalent thereof by S&P or at least P-1 or the equivalent thereof by Moody's and in each case maturing not more than one year after the date of acquisition by such Person, (f) Pounds Sterling bills of exchange eligible for rediscount at the Bank of England and accepted by any bank meeting the qualifications specified in clause (c) above, (g) investments in money market funds substantially all of whose assets are comprised of securities of the types described in clauses (a) through (f) above and which can be turned into cash on not more than 30 days' notice, or (h) any other debt security approved by the Required Lenders, and in each case described in clauses (a) through (h) above, which is not subject to any Lien (other than any Lien arising under the Security Documents).

"Cash Management Control Agreement" shall mean a "control agreement" in form and substance reasonably acceptable to the Administrative Agent and containing terms regarding the treatment of all cash and other amounts on deposit in the respective Collection Account, Disbursement Account, Designated Petty Cash Account, Designated Payroll Account or LKE Joint Account governed by such Cash Management Control Agreement consistent with the requirements of Section 5.03 hereto and Section 3.9 of the US Security Agreement (including, without limitation, any such agreement with respect to any LKE Joint Account pursuant to which the applicable Credit Party and the LKE Qualified Intermediary instruct the appropriate financial institution(s) to transfer funds from the LKE Joint Accounts to the Administrative Agent in accordance with Section 5.03 hereto).

"<u>Casualty Loss</u>" shall mean (i) the loss, damage, or destruction of any asset owned or used by US Company or any of its Subsidiaries, (ii) the condemnation, confiscation, or other taking, in whole or in part, of any such asset, or (iii) the diminishment of such asset so as to render use for its intended purpose impracticable or unreasonable.

"CERCLA" shall mean the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as the same has been amended and may hereafter be amended from time to time, 42 USC. § 9601 et seq.

"Certificate of Title" shall mean a certificate of title, certificate of ownership or other registration certificate issued or required to be issued for any asset under the certificate of title or similar laws of any jurisdiction.

"Change in Law" shall have the meaning provided in Section 11.10.

"Change of Control" shall mean (i) any "person" (as such term is used in Subsections 13(d) and 14(d) of the Securities and Exchange Act of 1934, as amended) on or after the Initial Borrowing Date is or becomes a "beneficial owner" (as defined in Rule 13d-3 under such Act), directly or indirectly, of Securities of US Company representing 25% or more of the combined voting power of US Company's then-outstanding Securities; or (ii) the existing directors for any reason cease to constitute 75% of US Company's Board of Directors or (iii) any Borrower or Guarantor ceases to be a wholly-owned Subsidiary of US Company, except as expressly permitted by the Credit Documents; or (iv) a "Change of Control" (as defined in any of the Senior Note Indentures) occurs. For purposes of this definition, "existing directors" means (x) individuals constituting US Company's Board of Directors on the Initial Borrowing Date, and (y) any subsequent director whose election by the Board of Directors or nomination for election by US Company's shareholders was approved by a vote of at least 75% of the directors then in office which directors either were directors on the Initial Borrowing Date or whose election or nomination for election was previously so approved.

"Chief Executive Office" shall mean, with respect to any Person, the location from which such Person manages the main part of its business operations or other affairs.

"Claims" shall have the meaning provided in the definition of Environmental Claims.

"Code" shall mean the Internal Revenue Code of 1986, as amended from time to time, and the regulations promulgated and rulings issued thereunder. Section references to the Code are to the Code, as in effect at the date of this Agreement and any subsequent provisions of the Code, amendatory thereof, supplemental thereto or substituted therefor.

"Co-Documentation Agent" shall mean JPMorgan Chase Bank, N.A., ING Capital LLC and The Bank of Nova Scotia, each in its capacity as documentation agent.

"Collateral" shall mean all property (whether real or personal) with respect to which any Liens have been granted (or purported to be granted) pursuant to any Security Document, including, without limitation, all Pledge Agreement Collateral, all Security Agreement Collateral, all Mortgaged Properties and all cash and Cash Equivalents delivered as collateral pursuant to Section 5.02 or Section 11.

"Collateral Agent" shall mean the Administrative Agent, in its capacity as collateral agent for the Lenders hereunder and under the other Credit Documents, and shall include any successor to the Administrative Agent appointed pursuant to Section 12.09.

"Collection Account" shall mean each account established at a Collection Bank subject to a Cash Management Control Agreement into which funds shall be transferred as provided in Section 5.03(b).

"Collection Bank" shall have the meaning provided in Section 5.03(b).

"Collective Bargaining Agreements" shall mean all collective bargaining agreements applying or relating to any employee of US Company or any of its Subsidiaries.

"Commitment Commission" shall have the meaning provided in Section 4.01(a).

"Compliance Period" shall mean any period (x) commencing on the first Business Day on which the Total Borrowing Availability is less than the Minimum Availability Amount and (y) ending on the first Business Day thereafter on which the Total Borrowing Availability has been equal to or greater than the Minimum Availability Amount for 30 consecutive days.

"Computation Date" shall mean the date on which the Equivalent Amount of any currency is determined.

"Concentration Account" shall have the meaning provided in Section 5.03(c).

"Consolidated" shall mean the consolidation in accordance with GAAP of the accounts or other items as to which such term applies.

"Consolidated EBITDA" shall mean for a period, the Consolidated Net Income of US Company and its Subsidiaries (excluding (a) extraordinary gains and (b) non-cash extraordinary losses) and without duplication (i) plus all Interest Expense, income tax expense, depreciation and amortization (including amortization of any goodwill or other intangibles) for the period, (ii) less gains or plus losses attributable to any fixed asset sales (excluding sales of containers held for lease) in the period, (iii) plus or minus any other non-cash charges which have been subtracted or added in calculating Consolidated Net Income, (iv) plus fees and expenses directly incurred in connection with the Transactions (including, without limitation, professional, legal and other advisory fess, and other costs incurred in connection with the initial funding under this Agreement), provided that such fees, costs and expenses are in an aggregate amount not to exceed \$30,000,000 (or the Equivalent Amount thereof) and are incurred on or prior to December 31, 2008, in each case to the extent deducted (and not added back) in such period in computing such Consolidated Net Income, (v) plus the amount of any one time merger related costs, including, without limitation, costs related to the closure and/or consolidation of facilities, duplicative advertising costs, integration costs, severance, relocation costs and curtailments or modifications to pension and post-retirement employee benefit plans, in each case to the extent deducted (and not added back) in such period in computing such Consolidated Net Income, provided that such fees, costs and expenses are in an aggregate amount not to exceed \$25,000,000 (or the Equivalent Amount thereof) and are incurred on or prior to June 30, 2009, and (vi) with respect to any period ending on or prior to the last date of the fourth full fiscal quarter to be completed following the Closing Date, the amount of post-Mobile Storage Acquisition net cost savings projected by the US Company in good faith to be realized as a result of specified actions commenced by the US Company and its Subsidiaries (calculated on a pro forma basis as though such cost savings had been realized on the first day of such period), net of the amount of actual benefits realized during such period from such actions, provided that (A) such cost savings are reasonably identifiable and factually supportable, (B) such actions are taken on or prior to the date that is twelve months following the Closing Date, (C) no cost savings shall be added pursuant to this clause (vi) to the extent duplicative of any expenses or charges relating to such cost savings that are included in clause (v) above with respect to such period and (D) the aggregate amount of cost savings added pursuant to this clause (vi) shall not exceed an amount equal to \$25,000,000 (it being understood and agreed that, for purposes solely of Section 6.12(b) hereof, the amount of such cost savings shall be deemed to be \$25,000,000). For all purposes other than calculating Consolidated Net Cash Flow, Consolidated EBITDA for any such period shall be calculated by giving pro forma effect to any Permitted Acquisition

and any Asset Sale specifically permitted pursuant to clauses (iv) or (v) of Section 10.03 during such period, as if such Acquisition or Asset Sale, as the case may be, had been consummated on the first day of such period, as long as US Company shall have delivered to Administrative Agent audited financial statements for such period for the Person or assets acquired or if consented to by Administrative Agent, other reasonably acceptable financial statements or other supporting documentation. Notwithstanding anything to the contrary contained herein and subject to adjustment as provided in the immediately preceding sentence with respect to Acquisitions and Asset Sales occurring following the Closing Date, Consolidated EBITDA for the fiscal quarters ended March 31, 2008, December 31, 2007 and September 30, 2007 shall be as set forth on Schedule 1.01(e).

"Consolidated Net Cash Flow" shall mean for a period, Consolidated EBITDA less the sum of (i) Unfinanced Capital Expenditures during such period plus (ii) income taxes paid in cash during such period plus (iii) Restricted Payments paid in cash during such period (other than Restricted Payments paid by a Subsidiary of US Company to a Credit Party). Notwithstanding anything to the contrary contained herein, Consolidated Net Cash Flow for the fiscal quarters ended March 31, 2008, December 31, 2007, and September 30, 2007 shall be as set forth on Schedule 1.01(f).

"Consolidated Net Income" shall mean, for any period, the net income (or loss) of US Company and its Subsidiaries determined on a consolidated basis for such period (taken as a single accounting period) in accordance with GAAP (after any deduction for minority interests); provided that the following items shall be excluded in computing Consolidated Net Income (without duplication): (a) the net income of any other Person which is not a Subsidiary of US Company or is accounted for by the equity method of accounting except to the extent of the payment of cash dividends or cash distributions by such other Person to US Company or a Subsidiary thereof during such period, and (b) the net income of any Subsidiary of US Company to the extent that the declaration or payment of cash dividends or similar cash distributions by such Subsidiary of such net income is not at the time permitted by the operation of the terms of its charter or any agreement, instrument, judgment, decree, order, statute, rule or governmental regulation applicable to such Subsidiary.

"Container Fleet Inventory." shall mean new and used manufactured or remanufactured portable and ISO containers and portable mobile offices held by US Company or another Credit Party for intended lease or rental by US Company or another Credit Party to third parties.

"Contingent Obligation" shall mean, as to any Person, any obligation of such Person as a result of such Person being a general partner of any other Person, unless the underlying obligation is expressly made non-recourse as to such general partner, and any obligation of such Person guaranteeing or intended to guarantee any Indebtedness, leases, dividends or other obligations ("primary obligations") of any other Person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of such Person, whether or not contingent, (a) to purchase any such primary obligation or any property constituting direct or indirect security therefor, (b) to advance or supply funds (i) for the purchase or payment of any such primary obligation or (ii) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (c) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation or (d) otherwise to assure or hold harmless the holder of such primary obligation against loss in respect thereof; provided, however, that the term Contingent Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business or guarantees by a parent entity of real property leases entered into by a UK Subsidiary in the ordinary course of business. The amount of any Contingent Obligation shall be deemed to be an amount equal to the stated or determinable amount of the primary obligation in respect of which such Contingent Obligation is made

or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof (assuming such Person is required to perform thereunder) as determined by such Person in good faith.

"Convertible Preferred Stock" shall mean US Company's Series A Convertible Redeemable Participating Preferred Stock to be issued in connection with the Mobile Storage Acquisition.

"Convertible Preferred Stock Documents" shall mean (i) the Stockholders Agreement, in the form attached to the Merger Agreement as Exhibit C thereto, and (ii) the certificate of designations for the Convertible Preferred Stock in the form attached to the Merger Agreement as Exhibit D, and all other agreements and documents relating to the issuance of the Convertible Preferred Stock.

"Credit Account" shall have the meaning provided in Section 5.03(g).

"Credit Documents" shall mean this Agreement, the US Pledge Agreement, the US Security Agreement, the UK Security Agreements, the Fee Letter and, after the execution and delivery thereof pursuant to the terms of this Agreement, each Note, each Incremental Commitment Agreement, each Joinder Agreement, each Mortgage and each other Security Document.

"Credit Event" shall mean the making of any Loan or the issuance of any Letter of Credit under this Agreement.

"Credit Party" shall mean each of the US Credit Parties and each of the UK Credit Parties.

"DBNY" shall mean Deutsche Bank AG New York Branch, and its permitted successors and assigns.

"Debt Ratio" shall mean as of any date of determination, the ratio of (i) Funded Debt as of such date to (ii) Consolidated EBITDA in each case for the four fiscal quarters ending on such date.

"Default" shall mean any event, act or condition which with notice or lapse of time, or both, would constitute an Event of Default.

"Defaulting Lender" shall mean any Lender with respect to which a Lender Default is in effect.

"Deposit Account" shall mean a demand, time, savings, passbook or like account with a bank, savings and loan association, credit union or like organization.

"<u>Derivative Obligations</u>" shall mean every obligation of a Person under any forward contract, futures contract, swap, option or other financing agreement or arrangement (including, without limitation, caps, floors, collars and similar agreements), the value of which is dependent upon interest rates, currency or exchange rates or valuations.

"<u>Designated Payroll Accounts</u>" shall mean those accounts set forth on Part E of <u>Schedule 10.20</u>, as well as any new payroll account established pursuant to <u>Section 10.20</u>, in each case which are maintained near a branch for any Credit Party and in which cash is transferred in accordance with the terms of this Agreement to pay ordinary course payroll expenses of such Credit Party.

"<u>Designated Petty Cash Accounts</u>" shall mean those accounts set forth on Part D of <u>Schedule 10.20</u>, as well as any new petty cash account established pursuant to <u>Section 10.20</u>, in each case

which are maintained near a branch for any Credit Party and in which cash is transferred in accordance with the terms of this Agreement to pay certain ordinary course cash expenses of such Credit Party.

- "<u>Disbursement Account</u>" shall mean each checking and/or disbursement account set forth on Part C of <u>Schedule 10.20</u>, maintained by each Borrower and each Guarantor for their respective general corporate purposes, including for the purpose of paying their trade payables and other operating expenses.
 - "Documents" shall mean, collectively, (a) the Credit Documents and (b) the Acquisition Documents.
 - "Dollars" and the sign "\$" shall each mean freely transferable lawful money of the United States.
- "<u>Domestic Subsidiary</u>" of any Person shall mean any Subsidiary of such Person incorporated or organized in the United States or any State thereof or the District or Columbia.
 - "Dutch Credit Parties" shall mean Mobile Mini Holding B.V. and Mobile Mini B.V.
 - "Effective Date" shall have the meaning provided in Section 13.10.
- "Eligible Account" shall mean those Accounts created by a Borrowing Base Party in the ordinary course of its business that arise out of its sale, lease or rental of goods or rendition of services, that comply in all material respects with each of the representations and warranties respecting Eligible Accounts made in the Credit Documents, and that are not excluded as ineligible by virtue of one or more of the excluding criteria set forth below. The Administrative Agent shall have the right to establish or modify Reserves against Eligible Accounts from time to time in its Permitted Discretion on three Business Days' prior written notice to US Company in respect of the US Borrowing Base or the UK Company in respect of the UK Borrowing Base, in each case after consultation with US Company or UK Company, as applicable. In determining the amount to be included, Eligible Accounts shall be the face amount of such Eligible Accounts and shall be calculated net of unearned revenue, charge-backs, customer deposits, unapplied cash, bonding subrogation rights to the extent not cash collateralized, any and all returns, rebates, discounts (which may, at the Administrative Agent's option, be calculated on shortest terms), service charges, customer deposits, credits, allowances or excise taxes of any nature at any time issued, owing, claimed by Account Debtors, granted, outstanding or payable in connection with such Accounts at such time. Without limiting the generality of the foregoing, unless otherwise approved in writing by Administrative Agent, no Account shall be an Eligible Account if:
 - (i) it arises out of a sale made or services rendered by a Borrowing Base Party to US Company, a Subsidiary of US Company or an Affiliate of US Company or to a Person controlled by an Affiliate of US Company; or
 - (ii) it is an Account that has payment terms longer than 60 days from the date of invoice; <u>provided</u>, <u>however</u>, that \$500,000 or the Equivalent Amount thereof may be considered Eligible Accounts with payment terms longer than 60 days but no longer than 90 days from the date of the invoice;
 - (iii) it remains unpaid more than 90 days after the original invoice date; or

- (iv) it is owed by an Account Debtor and the total unpaid Accounts of such Account Debtor exceed 10% of the net amount of all Eligible Accounts, but only to the extent of such excess; or
- (v) any covenant, representation or warranty contained in the Agreement or any Security Document with respect to such Account has been breached; or
- (vi) (1) the Account Debtor is also a creditor or supplier of the applicable Borrowing Base Party or any other Subsidiary of US Company, or the Account Debtor has disputed liability with respect to such Account, or the Account Debtor has made any claim with respect to any other Account due from such Account Debtor to the applicable Borrowing Base Party or any other Subsidiary of US Company, or the Account otherwise is or may become subject to right of setoff by the Account Debtor, <u>provided</u>, that any such Account shall be eligible to the extent such amount thereof exceeds such contract, dispute, claim, setoff or similar right; or (2) it is a portion (and only such portion) of any Account that constitutes late fees or finance charges or (3) it is less than ninety (90) days past the original invoice date and related to invoices that have been partially paid, if the US Company or the UK Company, as applicable, has reason to believe that such Account will not be fully paid; or
- (vii) the Account Debtor has commenced a voluntary case under the federal bankruptcy laws, as now constituted or hereafter amended, (or otherwise voluntarily submitted to insolvency, bankruptcy, arrangement, liquidation or equivalent proceedings in any jurisdiction) or made an assignment for the benefit of creditors, or a decree or order for relief has been entered by a court having jurisdiction in the premises in respect of the Account Debtor in an involuntary case under the federal or other similar bankruptcy, receivership, reorganization, arrangement, liquidation or insolvency laws, as now constituted or hereafter amended, or any other petition, procedure or other application for relief under the federal or other similar bankruptcy, receivership, reorganization or insolvency laws of any jurisdiction, as now constituted or hereafter amended, has been filed against the Account Debtor, declared a moratorium on any indebtedness or is the subject of a voluntary scheme of arrangement or if the Account Debtor has failed, suspended business or payments, ceased to be Solvent, or consented to or suffered a receiver, receiver manager, trustee, administrator, liquidator, compulsory manager, monitor or custodian to be appointed for it or for all or a significant portion of its assets or affairs; or
- (viii) (1) with respect to Account Debtors of any US Borrowing Base Party, it arises from a sale made or services rendered to an Account Debtor outside the United States, unless the sale is either (a) to an Account Debtor located in Ontario or any other province of Canada in which the Personal Property Security Act has been adopted in substantially the same form as currently in effect in Ontario or (b) is subject to Approved Credit Support or (2) with respect to Account Debtors of any UK Borrowing Base Party it arises from a sale made or services rendered to an Account Debtor outside the United Kingdom, unless the sale is subject to Approved Credit Support, or (3) with respect to Account Debtors of any US Borrowing Base Party, the Account Debtor is the government of any country or sovereign state other than the United States, or of any state, municipality, or other political subdivision thereof, or (4) with respect to Account Debtors of any UK Borrowing Base Party, the Account Debtor is the government of any country or sovereign state, or of any state, province, municipality, or other political subdivision thereof, or of any department, agency, public corporation, or other instrumentality thereof, unless the sale is subject to Approved Credit Support, or (4) with respect to Approved Credit Support, or of any department, agency, public corporation, or other instrumentality thereof, unless the sale is subject to Approved Credit Support,

- (ix) (1) it arises from a sale to the Account Debtor on a bill-and-hold or, consignment basis or in any other transaction wherein goods are sold pursuant to a guaranteed sale, a sale or return, a sale on approval, or any other terms by reason of which the payment by the Account Debtor may be conditional (other than, for the avoidance of doubt, a rental or lease basis) or (2) it is subject to a reserve established by US Company or any of its Subsidiaries for potential returns or refunds, to the extent of such reserve; or
- (x) the Account Debtor is the United States of America, any State or any political subdivision or department, agency or instrumentality thereof, unless the applicable Borrowing Base Party, assigns its right to payment of such Account to the Administrative Agent, in a manner satisfactory to the Administrative Agent, in its Permitted Discretion, so as to comply with the Assignment of Claims Act of 1940 (31 USC. §203 et seq., as amended) or complies with any similar applicable state or local law as the Administrative Agent may require; or
- (xi) it is not at all times subject to Collateral Agent's duly perfected, First Priority security interest and to no other Lien that is not a Permitted Lien; or
- (xii) the goods giving rise to such Account have not been delivered to and accepted by the Account Debtor or the services giving rise to such Account have not been performed by the applicable Borrowing Base Party and accepted by the Account Debtor or the Account otherwise does not represent a final sale; or
 - (xiii) the Account is evidenced by an instrument of any kind, or has been reduced to judgment; or
- (xiv) the applicable Borrowing Base Party has made any agreement with the Account Debtor for any deduction therefrom, except for discounts or allowances which are made in the ordinary course of business for prompt payment and which discounts or allowances are reflected in the calculation of the face value of each invoice related to such Account; or
 - (xv) more than 50% of the Accounts owing from the Account Debtor are not Eligible Accounts hereunder; or
 - (xvi) the Account is subject to any progress payment or other similar advance made by or for the benefit of the applicable Account Debtor; or
- (xvii) the Account evidences a lease or sale to an Account Debtor that is an individual to the extent that the aggregate of such Accounts exceeds \$750,000; or
- (xviii) the Account represents amounts which have not yet been billed to the applicable Account Debtor and the amount of such Account together with the amount of all other Accounts which represent amounts which have not yet been billed to the applicable Account Debtors to the extent such amount exceeds \$2,000,000 or the Equivalent Amount thereof, or
- (xix) except as provided in clause (x) above, with respect to which either the perfection, enforceability, or validity of the Collateral Agent's Liens in such Account, or the Collateral Agent's right or ability to obtain direct payment to the Collateral Agent of the proceeds of such Account, is governed by any national, federal, state, provincial or local statutory requirements other than those of the UCC or, in the case of the UK Borrowing Base Parties, the UK Companies Act of 1985 (as amended), or

(xx) in the case of Accounts of any US Borrowing Base Party, is not payable in Dollars or Canadian Dollars or, in the case of Accounts of any UK Borrowing Base Party, is not payable in Pounds Sterling or Euros; or

- (xxi) with respect to the UK Borrowing Base Parties, Accounts regulated by the UK Consumer Credit Act of 1974 (as amended); or
- (xxii) with respect to the UK Borrowing Base Parties, the Accounts are governed by laws other than that of England and Wales; or (xxiii) it is a LKE Account.

"<u>Eligible Cabin Fleet Inventory</u>" shall mean shall mean Eligible Goods Inventory of a UK Borrowing Base Party consisting of Cabin Fleet Inventory, valued at the lower of such Borrowing Base Party's cost or Orderly Liquidation Value.

"<u>Eligible Container Fleet Inventory</u>" shall mean Eligible Goods Inventory of a Borrowing Base Party consisting of Container Fleet Inventory, valued at the lower of such Borrowing Base Party's cost or Orderly Liquidation Value, except for custom containers that are pre-sold and ISO containers that are pre-sold, which will be valued at the lower of Borrowing Base Party's cost or sales invoice price.

"<u>Eligible Container Inventory Held For Sale</u>" shall mean Eligible Goods Inventory of a Borrowing Base Party consisting of (a) new and used manufactured or remanufactured portable and ISO containers and portable mobile offices held by a Borrowing Base Party for intended sale to third parties, containers temporarily out of service and otherwise unrefurbished ISO units and (b) up to \$10,000,000 (or the Equivalent Amount thereof) of containers used by Borrowing Base Parties in the conduct of their business (and not held for sale or lease), each of which containers in clauses (a) and (b) shall be valued at the net book value thereof.

"Eligible Goods Inventory." shall mean all of the Inventory owned by one of the Borrowing Base Parties that comply in all material respects with each of the representations and warranties respecting such Inventory made in the Credit Documents, and that are not excluded as ineligible by one or more of the excluding criteria set forth below, and which the Administrative Agent shall have, in its Permitted Discretion, deemed to be Eligible Goods Inventory. The Administrative Agent shall have the right to establish or modify Reserves against Eligible Goods Inventory from time to time in its Permitted Discretion on three Business Days' prior written notice to US Company after consultation with US Company. The amount of Eligible Goods Inventory shall be determined on a first-in, first-out basis; and Inventory "cost" shall be determined in a manner consistent with the respective Borrowing Base Parties' current and historical accounting practices unless otherwise specifically provided in this Agreement. Unless otherwise approved in writing by Administrative Agent, no Inventory shall be deemed Eligible Goods Inventory if:

- (i) it is not owned solely by a Borrowing Base Party or a Borrowing Base Party does not have good, valid and marketable title thereto; or
- (ii) it is not (A) with respect to a US Borrowing Base Party (x) located at one of the locations in the United States or Canada, in either case, set forth on Schedule 1.01(b), or from and after the date hereof, another location maintained by US Company or any of its Subsidiaries in the United States or Canada (it being understood that, with respect to Inventory with a cost in excess of \$1,000,000 located in any Province of Canada other than the Provinces of British Columbia or Ontario, such Inventory must be subject to a valid and perfected First Priority Lien

in favor of the Collateral Agent, enforceable on substantially the same basis as the Liens in favor of the Collateral Agent pursuant to the Security Documents over Inventory located in the United States (as reasonably determined by the Administrative Agent) and supported by such Additional Security Documents and opinions of counsel as the Administrative Agent may reasonably request, and with all actions required pursuant to Section 9.13(a) with respect to such Additional Security Documents having been completed), or (y) on lease with a customer in the ordinary course of business and located in the United States or Canada (it being understood that, with respect to Inventory located in Canada, such Inventory must be subject to a valid and perfected First Priority Lien in favor of the Collateral Agent, enforceable on substantially the same basis as the Liens in favor of the Collateral Agent pursuant to the Security Documents over Inventory located in the United States (as reasonably determined by the Administrative Agent) and supported by such Additional Security Documents and opinions of counsel as the Administrative Agent may reasonably request, and with all actions required pursuant to Section 9.13(a) with respect to such Additional Security Documents having been completed), or (B) with respect to a UK Borrowing Base Party (x) located at one of the locations in the United Kingdom set forth on Schedule 1.01(c), or from and after the date hereof, another location maintained by a UK Borrowing Base Party in the United Kingdom, or (y) on lease with a customer in the ordinary course of business and located in the United Kingdom;

- (iii) it is not subject to a valid and perfected First Priority Lien in favor of Administrative Agent except, with respect to Inventory stored at sites described in clause (ii) above, for Liens for unpaid rent or normal and customary warehousing charges; provided that this clause (iii) will not apply to either (A) Inventory represented by a Certificate of Title (such Inventory being subject to clause (xiii) below) or (B) Inventory with a cost not in excess of \$1,000,000 located in any Province of Canada other than the Provinces of British Columbia or Ontario, which such Inventory described in this clause (B) need not be subject to a First Priority Lien; or
 - (iv) it consists of goods returned or rejected by Borrowing Base Party's or Affiliate's customers or goods in transit to third parties; or
- (v) it is not first-quality finished goods or work in process, is obsolete, or does not otherwise conform to the representations and warranties contained in the Credit Documents; or
- (vi) it is subject to a lease which should be classified as a capital lease under GAAP or contains a purchase option for an amount less than the amount equal to the net book value; or
 - (vii) [omitted]; or
 - (viii) it can not be located at the time of Borrowing Base Party's physical inventory; or
- (ix) it is subject to any third party retention of title or romalpa provision, unless a Reserve has been established by the Administrative Agent in respect of the UK Priority Claims related thereto in an amount equal to the value of such claims; or
 - (x) it is Eligible Raw Materials Inventory or Eligible Machinery and Equipment; or
 - (xi) it is damaged or defective and is not repairable; or
 - (xii) [omitted]; or

- (xiii) it is Inventory represented (or required to be represented) by a Certificate of Title unless the Credit Parties shall have complied with Article 6 of the US Security Agreement; or
- (xiv) it has not been appraised by the Appraiser with an appraisal in form and substance satisfactory to Administrative Agent and reasonably satisfactory to the Required Lenders and it is not of an identical kind or type of Inventory that has been appraised.
 - "Eligible Inventory" shall mean Eligible Goods Inventory and Eligible Raw Materials Inventory.
- "Eligible Machinery and Equipment" shall mean all of the Equipment owned by a Borrowing Base Party in the ordinary course of business and which complies in all material respects with the representations and warranties respecting Eligible Machinery and Equipment made in the Credit Documents and that are not excluded as ineligible by virtue of one or more of the exclusionary criteria set forth below. The Administrative Agent shall have the right to establish or modify Reserves against Eligible Machinery and Equipment from time to time in its Permitted Discretion on three Business Days' prior written notice to US Company after consultation with US Company. Eligible Machinery and Equipment shall be valued at its Net Orderly Liquidation Value. Without limiting the generality of the foregoing, unless otherwise approved in writing by Administrative Agent, no Equipment shall be deemed Eligible Machinery and Equipment if:
 - (i) it is not owned solely by a Borrowing Base Party or a Borrowing Base Party does not have good, valid and marketable title thereto; or
 - (ii) it is not (A) with respect to a US Borrowing Base Party, located at one of the locations in the United States or Canada set forth on Schedule 1.01(b), or from and after the date hereof, another location maintained by US Company or one of its Subsidiaries in the United States or Canada (it being understood that, with respect to Equipment located in any Province of Canada other than the Provinces of British Columbia or Ontario, such Equipment must be subject to a valid and perfected First Priority Lien in favor of the Collateral Agent, enforceable on substantially the same basis as the Liens in favor of the Collateral Agent pursuant to the Security Documents over Equipment located in the United States (as reasonably determined by the Administrative Agent) and supported by such Additional Security Documents and opinions of counsel as the Administrative Agent may reasonably request, and with all actions required pursuant to Section 9.13(a) with respect to such Additional Security Documents having been completed), or (B) with respect to a UK Borrowing Base Party, located at one of the locations in the United Kingdom set forth on Schedule 1.01(c), or from and after the date hereof, another location maintained by a UK Borrowing Base Party in the United Kingdom; or
 - (iii) it is not subject to a valid and perfected First Priority Lien in favor of Administrative Agent except, with respect to Equipment stored at sites described in clause (c) above, for Liens for unpaid rent or normal and customary warehousing charges; <u>provided</u> that this clause (iii) will not apply to Equipment represented by a Certificate of Title (such Equipment being subject to clause (v) below); or
 - (iv) it has not been appraised by the Appraiser with an appraisal in form and substance satisfactory to Administrative Agent and reasonably satisfactory to the Required Lenders and it is not of an identical kind or type of Equipment that has been appraised; or

- (v) it is Equipment represented (or required to be represented) by a Certificate of Title or constitutes Serial Numbered Equipment unless the Credit Parties shall have complied with Article 6 of the US Security Agreement; or
 - (vi) it is Eligible Goods Inventory.
- "Eligible Raw Materials Inventory." shall mean Inventory of a Borrowing Base Party, purchased from third parties consisting of steel, lumber, plywood, paint, drywall, plumbing materials and fixtures, electrical components, insulation materials, HVAC materials, doors and windows, and fasteners, valued at the applicable Borrowing Base Party's cost (except, in the case of steel, lumber, plywood, paint, where, for purposes of fiscal year end calculations, the value will be the lower of the applicable Borrowing Base Party's cost or market), which Administrative Agent, in its Permitted Discretion, deems to be Eligible Raw Materials Inventory. The amount of Raw Materials Inventory shall be determined on a first-in, first-out basis. Without limiting the generality of the foregoing, unless otherwise approved in writing by Administrative Agent, no Inventory shall be deemed Eligible Raw Materials Inventory if:
 - (i) it is not owned solely by a Borrowing Base Party or a Borrowing Base Party does not have good, valid and marketable title thereto; or
 - (ii) it is not (A) with respect to a US Borrowing Base Party, located at one of the locations in the United States or Canada set forth on Schedule 1.01(b), or from and after the date hereof, another location maintained by US Company or one of its Subsidiaries in the United States or Canada (it being understood that, with respect to Inventory located in any Province of Canada other than the Provinces of British Columbia or Ontario, such Inventory must be subject to a valid and perfected First Priority Lien in favor of the Collateral Agent, enforceable on substantially the same basis as the Liens in favor of the Collateral Agent pursuant to the Security Documents over Inventory located in the United States (as reasonably determined by the Administrative Agent) and supported by such Additional Security Documents and opinions of counsel as the Administrative Agent may reasonably request, and with all actions required pursuant to Section 9.13(a) with respect to such Additional Security Documents having been completed), or (B) with respect to a UK Borrowing Base Party, it is not located at one of the locations in the United Kingdom set forth on Schedule 1.01(c), or from and after the date hereof, another location maintained by a UK Borrowing Base Party in the United Kingdom; or
 - (iii) it is not subject to a valid and perfected First Priority Lien in favor of Administrative Agent except, with respect to Inventory stored at sites described in clause (ii) above, for Liens for unpaid rent or normal and customary warehousing charges; or
 - (iv) it is not first-quality raw materials, is obsolete or slow moving, or does not otherwise conform to the representations and warranties contained in the Credit Documents; or
 - (v) it is Eligible Goods Inventory or Eligible Machinery and Equipment; or
 - (vi) it is subject to any third party retention of title or romalpa provision, unless a Reserve has been established by the Administrative Agent in respect of the UK Priority Claims related thereto in an amount equal to the value of such claims.

"<u>Eligible Real Property</u>" shall mean all of the Real Property owned by a US Borrowing Base Party in the ordinary course of business and which complies in all material respects with the representations and warranties respecting Real Property made in the Credit Documents and that are not excluded as ineligible by virtue of one or more of the exclusionary criteria set forth below. The

Administrative Agent shall have the right to establish or modify Reserves against Eligible Real Property from time to time in its Permitted Discretion on three Business Days' prior written notice to US Company after consultation with US Company. Eligible Real Property shall be valued at its Appraised Fair Market Value. Without limiting the generality of the foregoing, unless otherwise approved in writing by Administrative Agent, no Real Property shall be deemed Eligible Real Property if:

- (i) it is not owned solely by a US Borrowing Base Party or a US Borrowing Base Party does not have good record and valid and marketable title in fee simple thereto; or
 - (ii) it is not located in the United States; or
- (iii) it is not subject to a valid and perfected First Priority Lien, subject only to any Permitted Encumbrances, pursuant to a Mortgage in form and substance satisfactory to the Administrative Agent, in favor of Administrative Agent on behalf of itself and the Lenders; or
- (iv) it has not been appraised by the Appraiser with an appraisal in form and substance satisfactory to Administrative Agent and reasonably satisfactory to the Required Lenders; or
 - (vi) it is not designated as "Eligible Real Property" on Schedule 1.01(d), as amended from time to time with the consent of Administrative Agent; or
- (vii) it is not covered by a Mortgage Policy with respect to the Lien of the Administrative Agent and casualty and property insurance, in each case reasonably acceptable to the Agent; or
 - (viii) it is not the subject of an environmental report reasonably acceptable to the Agent;
- (ix) an opinion of counsel for the Credit Party which is the owner of the Real Property has not been delivered to the Administrative Agent, in a form, scope and substance reasonably satisfactory to the Administrative Agent and its counsel, if reasonably requested by the Administrative Agent; or
- (x) a customary certificate in a form reasonable acceptable to the Administrative Agent has not been obtained indicating that the property is not in a flood zone, or if the property is in a flood zone, appropriate insurance reasonable acceptable to the Administrative Agent has not been obtained.
- "<u>Eligible Trailer Fleet Inventory</u>" shall mean Eligible Goods Inventory consisting of Trailer Fleet Inventory, valued at the lower of cost or Orderly Liquidation Value, excluding any Inventory that is not manufactured in accordance with and does not meet all standards imposed by all requirements of law or by any governmental authority having regulatory authority over such goods or their manufacture, use, sale, or lease.
- "Eligible Transferee" shall mean and include a commercial bank, an insurance company, a finance company, a financial institution, any fund that invests in loans or any other "accredited investor" (as defined in Regulation D of the Securities Act), but in any event excluding US Company and its Subsidiaries and Affiliates.

- "Eligible UK Account" shall mean an Eligible Account of UK Company or another UK Borrowing Base Party.
- "Eligible UK Container Fleet Inventory," shall mean Eligible Container Fleet Inventory of UK Company or another UK Borrowing Base Party.
- "Eligible UK Goods Inventory" shall mean Eligible Goods Inventory of UK Company or another UK Borrowing Base Party.
- "Eligible UK Inventory" shall mean Eligible UK Goods Inventory and Eligible UK Raw Materials Inventory.
- "Eligible UK Machinery and Equipment" shall mean Eligible Machinery and Equipment of UK Company or any other UK Borrowing Base Party.
- "Eligible UK Raw Materials Inventory," shall mean Eligible Raw Materials Inventory of UK Company or another UK Borrowing Base Party.
- "<u>Eligible UK Work-In-Process Container Inventory</u>" shall mean Eligible Work-In-Process Container Inventory of UK Company or another UK Borrowing Base Party.
 - "Eligible US Account" shall mean an Eligible Account of US Company or another US Borrowing Base Party.
 - "Eligible US Container Fleet Inventory" shall mean Eligible Container Fleet Inventory of US Company or another US Borrowing Base Party.
- "Eligible US Container Inventory Held For Sale" shall mean Eligible Container Inventory Held For Sale of US Company or another US Borrowing Base Party.
 - "Eligible US Goods Inventory" shall mean Eligible Goods Inventory of US Company or another US Borrowing Base Party.
 - "Eligible US Inventory" shall mean Eligible US Goods Inventory and Eligible US Raw Materials Inventory.
 - "Eligible US Machinery and Equipment" shall mean Eligible Machinery and Equipment of US Company or another US Borrowing Base Party.
 - "Eligible US Raw Materials Inventory" shall mean Eligible Raw Materials Inventory of US Company or another US Borrowing Base Party.
- "<u>Eligible US Work-In-Process Container Inventory</u>" shall mean Eligible Work-In-Process Container Inventory of US Company or another US Borrowing Base Party.
- "<u>Eligible Work-In-Process Container Inventory</u>" shall mean Eligible Goods Inventory, valued at net book value, consisting of: (a) new and used manufactured or remanufactured portable containers, which is in the work-in-process phase of manufacturing; (b) shaped steel component parts; or (c) sub-assemblies.
- "Employee Benefit Plans" shall mean all Plans (and for each Plan that is required to file an annual report on Internal Revenue Service Form 5500-series, the most recent such report (including, to

the extent required, the related financial and actuarial statements and opinions and other supporting statements, certifications, schedules and information), and for each Plan that is a "single-employer plan" as defined in Section 4001(a)(15) of ERISA, the most recently prepared actuarial valuation therefor) and any other "employee benefit plans" as defined in Section 3(3) of ERISA, and any other material agreements, plans or arrangements, with or for the benefit of current or former employees of US Company or any of its Subsidiaries or any ERISA Affiliate (<u>provided</u> that the foregoing shall apply in the case of any multiemployer plan, as defined in 4001(a)(3) of ERISA, only to the extent that any document described herein is in the possession of US Company or any Subsidiary of US Company or any ERISA Affiliate or is reasonably available thereto from the sponsor or trustee of any such plan).

"End Date" shall have the meaning provided in the definition of Applicable Margin.

"Environmental Claims" shall mean any and all administrative, regulatory or judicial actions, suits, demands, demand letters, directives, claims, liens, notices of noncompliance or violation, investigations or proceedings relating in any way to any Environmental Law or any permit issued, or any approval given, under any such Environmental Law (hereafter, "Claims"), including, without limitation, (a) any and all Claims by Governmental Authorities for enforcement, cleanup, removal, response, remedial or other actions or damages pursuant to any applicable Environmental Law, and (b) any and all Claims by any third party seeking damages, contribution, indemnification, cost recovery, compensation or injunctive relief in connection with alleged injury or threat of injury to health, safety or the environment due to the presence of Hazardous Materials.

"Environmental Law" shall mean any Federal, state, foreign or local statute, law, rule, regulation, ordinance, code, guideline, policy and rule of common law now or hereafter in effect and in each case as amended, and any judicial or administrative interpretation thereof, including any judicial or administrative order, consent decree or judgment, relating to the environment, employee health and safety or Hazardous Materials, including, without limitation, CERCLA; the Resource Conservation and Recovery Act, 42 USC § 6901 et seq.; the Federal Water Pollution Control Act, 33 USC. § 1251 et seq.; the Toxic Substances Control Act, 15 USC. § 2601 et seq.; the Clean Air Act, 42 USC. § 7401 et seq.; the Safe Drinking Water Act, 42 USC. § 3803 et seq.; the Oil Pollution Act of 1990, 33 USC. § 2701 et seq.; the Emergency Planning and the Community Right-to-Know Act of 1986, 42 USC. § 11001 et seq.; the Hazardous Material Transportation Act, 49 USC. § 1801 et seq.; the Occupational Safety and Health Act, 29 USC. § 651 et seq.; and any state and local or foreign counterparts or equivalents in any jurisdiction, in each case as amended from time to time.

"Equipment" shall mean all machinery, apparatus, equipment, motor vehicles and other similar assets (other than Inventory) used in the operations of the US Company or any of its Subsidiaries or owned by the US Company or any of its Subsidiaries or in which the US Company or any of its Subsidiaries has an interest, whether now owned or hereafter acquired by a Borrower or any of its Subsidiaries or Affiliates and wherever located, and all parts, accessories and special tools and all increases and accessions thereto and substitutions and replacements therefore.

"<u>Equity Interests</u>" of any Person shall mean any and all shares, interests, rights to purchase, warrants, options, participation or other equivalents of or interest in (however designated) equity of such Person, including any common stock, preferred stock, any limited or general partnership interest and any limited liability company membership interest.

"Equivalent Amount" shall mean (i) the equivalent amount in US Dollars of any amount expressed in Pounds Sterling or Euros, as the case may be, as determined by Administrative Agent on the date of determination on the basis of the Spot Rate for the purchase of US Dollars with Pounds Sterling or Euros, as the case may be, on the relevant Computation Date provided for hereunder; or (ii) the equivalent

amount in Pounds Sterling or Euros, as the case may be, of any amount expressed in US Dollars as determined by Administrative Agent on the date of determination on the basis of the Spot Rate for the purchase of Pounds Sterling or Euros, as the case may be, with Dollars on the relevant Computation Date provided for hereunder.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, and the regulations promulgated and rulings issued thereunder. Section references to ERISA are to ERISA, as in effect at the date of this Agreement and any subsequent provisions of ERISA, amendatory thereof, supplemental thereto or substituted therefor.

"ERISA Affiliate" shall mean each person (as defined in Section 3(9) of ERISA) which together with any Credit Party would be deemed to be a "single employer" (a) within the meaning of Section 414(b) or (c) of the Code, and for the purpose of Section 302 of ERISA and/or Section 412, 4971, 4977 and/or each "applicable section" under Section 414(t)(2) of the Code, within the meaning of Section 414(b), (c), (m) or (o) of the Code or (b) as a result of a Credit Party being or having been a general partner of such person.

"Euro" and "€" means the single currency of the Participating Member States.

"EURIBOR" shall mean, for each Interest Period:

- (a) the Euro Inter-Bank Offered Rate displayed on the relevant Reuters Screen LIBOR01 Page (or any other page or screen that would be substituted therefor) at 11:00 a.m. (Brussels time) under the supervision of the European Banking Federation three Business Days before the first day of such Interest Period, with respect to deposits in Euros on the European interbank market amongst leading banks for a period equivalent to such Interest Period; and
- (b) if no rate is available as described above with respect to an Interest Period applicable to UK Revolving Loans, the annual rate determined by the Administrative Agent as being equal to the arithmetic mean (rounded up if necessary to the nearest four decimal places) of the rates supplied to the Administrative Agent by four (4) reference banks selected by the Administrative Agent in the European interbank market, at approximately 3:00 p.m. (Brussels time) three Business Days before the first day of such Interest Period as quoted by the respective reference banks to leading banks on the European interbank market for deposits in Euros, for a period equivalent to such Interest Period and starting on the first day of such Interest Period and for an amount comparable to the amount of such EURIBOR Loan.

If the EURIBOR is replaced by a similar or equivalent interest rate, or in the event of a change affecting the entity which publishes the EURIBOR or the terms of its publication, the corresponding modified or replacement interest rate shall automatically apply in accordance with the foregoing paragraphs and any reference herein to "EURIBOR" shall be deemed to be a reference to such rate.

"EURIBOR Loan" shall mean each UK Revolving Loan denominated in Euros bearing interest at a rate determined by reference to EURIBOR.

"Euro Denominated Revolving Loan" shall have the meaning set forth in Section 2.01(b) hereof.

"Euro Funding Capacity" shall mean at any date of determination, for any Lender, the ability of such Lender or any Affiliate or branch office thereof to fund Revolving Loans denominated in

Euros, as set forth in the records of Administrative Agent upon notification from such Lender from time to time.

- "Euro Participation" shall have the meaning set forth in Section 2.16(b).
- "Euro Participation Fee" shall have the meaning set forth in Section 2.16(j).
- "Euro Participation Settlement" shall have the meaning set forth in Section 2.16(d)(i).
- "Euro Participation Settlement Amount" shall have the meaning set forth in Section 2.16(d)(ii).
- "Euro Participation Settlement Date" shall have the meaning set forth in Section 2.16(d)(i).
- "Euro Participation Settlement Period" shall have the meaning set forth in defined in Section 2.16(d)(i).
- "Event of Default" shall have the meaning provided in Section 11.
- "Existing Indebtedness" shall have the meaning provided in Section 8.31.
- "Existing Indebtedness Agreements" shall mean all agreements evidencing or relating to Indebtedness of US Company or any of its Subsidiaries which is to remain outstanding after giving effect to the Transactions.
 - "Existing US Letter of Credit" shall have the meaning provided in Section 3.01(c).
- "Expenses" shall mean all present and future reasonable and invoiced expenses incurred by or on behalf of the Administrative Agent, the Collateral Agent or any Issuing Lender in connection with this Agreement, any other Credit Document or otherwise in its capacity as the Administrative Agent under this Agreement or the Collateral Agent under any Security Document or as an Issuing Lender under this Agreement, whether incurred heretofore or hereafter, which expenses shall include, without limitation, the cost of record searches, the reasonable fees and expenses of attorneys and paralegals, all reasonable and invoiced costs and expenses incurred by the Administrative Agent (and the Collateral Agent) in opening bank accounts, depositing checks, electronically or otherwise receiving and transferring funds, and any other charges imposed on the Administrative Agent (and the Collateral Agent) due to insufficient funds of deposited checks and the standard fee of the Administrative Agent (and the Collateral Agent) relating thereto, collateral examination fees and expenses, reasonable fees and expenses of accountants, appraisers or other consultants, experts or advisors employed or retained by the Administrative Agent and the Collateral Agent, fees and taxes related to the filing of financing statements, costs of preparing and recording any other Credit Documents, all expenses, costs and fees set forth in this Agreement and the other Credit Documents, all other fees and expenses required to be paid pursuant to any other letter agreement and all fees and expenses incurred in connection with releasing Collateral and the amendment or termination of any of the Credit Documents.

"Facing Fee" shall have the meaning provided in Section 4.01(c).

"<u>Fair Market Value</u>" shall mean, with respect to any asset (including any Equity Interests of any Person), the price at which a willing buyer, not an Affiliate of the seller, and a willing seller who does not have to sell, would agree to purchase and sell such asset, as determined in good faith by the

board of directors or other governing body or, pursuant to a specific delegation of authority by such board of directors or other governing body, a designated senior executive officer, of US Company, or the Subsidiary of US Company selling such asset.

"Federal Funds Rate" shall mean, for any period, a fluctuating interest rate equal for each day during such period to the weighted average of the rates on overnight Federal Funds transactions with members of the Federal Reserve System arranged by Federal Funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by the Administrative Agent from three Federal Funds brokers of recognized standing selected by the Administrative Agent.

"Fee Letter" shall mean that Fee Letter, dated as of February 22, 2008, by and among US Company, Deutsche Bank AG New York Branch, Deutsche Bank Securities Inc., Bank of America, N.A., Banc of America Securities LLC, JPMorgan Chase Bank, N.A., and J.P. Morgan Securities Inc.

"Fees" shall mean all amounts payable pursuant to or referred to in Section 4.01.

"FEMA" shall mean the Federal Emergency Management Agency.

"Final Maturity Date" shall mean June 27, 2013.

"<u>First Priority</u>." shall mean, with respect to any Lien purported to be created on any Collateral pursuant to any Security Document, that such Lien is prior in right to any other Lien thereon, other than any Permitted Liens applicable to such Collateral which as a matter of law have priority over the respective Liens on such Collateral created pursuant to the relevant Security Document.

"Fixed Charge Coverage Ratio" shall mean as of any date of determination, the ratio of (i) Consolidated Net Cash Flow for the four fiscal quarters ending on such date to (ii) Fixed Charges.

"Fixed Charges" shall mean as of any date of determination, the sum of Interest Expense for the four fiscal quarters ending on such date <u>plus</u> the principal payments with respect to Funded Debt (other than payments of Revolving Loans) made during the four fiscal quarters ending on such date. The foregoing notwithstanding, for the three quarter periods ending prior to June 30, 2009, Fixed Charges shall be calculated as follows: (x) for the four quarter period ending September 30, 2008, by multiplying Fixed Charges for the quarter ended September 30, 2008 by 4; (y) for the four quarter period ending December 31, 2008, by multiplying the sum of (1) Fixed Charges for the quarter ended December 31, 2008, by 2; and (z) for the four quarter period ending March 31, 2009, by multiplying the sum of (1) Fixed Charges for the quarter ended September 30, 2008, (2) Fixed Charges for the quarter ended December 31, 2008 and (3) Fixed Charges for the quarter ended March 31, 2009, by 4/3.

"Foreign Account" shall mean each account of any Credit Party established at a bank, other than in the United States, Canada or the United Kingdom set forth on Part F of Schedule 10.20.

"<u>Foreign Pension Plan</u>" shall mean any plan, fund (including, without limitation, any superannuation fund) or other similar program established or maintained outside the United States by US Company or any one or more of its Subsidiaries primarily for the benefit of employees of US Company or such Subsidiaries residing outside the United States, which plan, fund or other similar program provides, or results in, retirement income, a deferral of income in contemplation of retirement or payments to be made upon termination of employment, and which plan is not subject to ERISA or the Code.

"Foreign Subsidiary" shall mean any Subsidiary of US Company that is not a Domestic Subsidiary.

"Funded Debt" shall mean, without duplication, (i) Indebtedness arising from the lending of money by any Person to US Company or any of its Subsidiaries; (ii) Indebtedness, whether or not in any such case arising from the lending by any Person of money to US Company or any of its Subsidiaries, (1) which is represented by notes payable or drafts accepted that evidence extensions of credit, (2) which constitutes obligations evidenced by bonds, debentures, notes or similar instruments, or (3) upon which interest charges are customarily paid (other than accounts payable) or that was issued or assumed as full or partial payment for Property; (iii) Indebtedness that constitutes a Capitalized Lease Obligation; (iv) reimbursement obligations with respect to letters of credit or guaranties of letters of credit; and (v) Indebtedness of US Company or any of its Subsidiaries under any guaranty of obligations that would constitute Funded Debt under clauses (i) through (iv) hereof, if owed directly by US Company or any of its Subsidiaries. Funded Debt shall not include trade payables or accrued expenses or Indebtedness (other than Indebtedness under the Agreement) of up to the amount permitted pursuant to Section 10.05(g) incurred to finance insurance premiums.

"Funded Euro Participation" shall mean with respect to any Participating Euro Lender relating to Euro Denominated Revolving Loans funded by DBNY, (i) the aggregate amount paid by such Participating Euro Lender to DBNY pursuant to Section 2.16(b) of the Agreement in respect of such Participating Euro Lender's participation in the principal amount of Euro Denominated Revolving Loans funded by DBNY minus (ii) the aggregate amount paid to such Participating Euro Lender by DBNY pursuant to Section 2.16(b) of the Agreement in respect of its participation in the principal amount of Euro Denominated Revolving Loans funded by DBNY, excluding in each case any payments made in respect of interest accrued on the Euro Denominated Revolving Loans funded by DBNY. DBNY's Funded Euro Participation in any Euro Denominated Revolving Loans funded by DBNY shall be equal to the outstanding principal amount of such Euro Denominated Revolving Loans minus the total Funded Euro Participation of all other Lenders therein.

"Funded Pounds Sterling Participation" shall mean with respect to any Participating Pounds Lender relating to Pounds Sterling Revolving Loans funded by DBNY, (i) the aggregate amount paid by such Participating Pounds Lender to DBNY pursuant to Section 2.16(a) of the Agreement in respect of such Participating Pounds Lender's participation in the principal amount of Pounds Sterling Denominated Revolving Loans funded by DBNY minus (ii) the aggregate amount paid to such Participating Pounds Lender by DBNY pursuant to Section 2.16(a) of the Agreement in respect of its participation in the principal amount of Pounds Sterling Denominated Revolving Loans funded by DBNY, excluding in each case any payments made in respect of interest accrued on the Pounds Sterling Denominated Revolving Loans funded by DBNY. DBNY's Funded Pounds Sterling Participation in any Pounds Sterling Denominated Revolving Loans funded by DBNY shall be equal to the outstanding principal amount of such Pounds Sterling Denominated Revolving Loans minus the total Funded Pounds Sterling Participation of all other Lenders therein.

"GAAP" shall mean generally accepted accounting principles in the United States as in effect from time to time.

"General Intangibles" shall mean "general intangibles" as such term is defined in Article 9 of the UCC.

"Governmental Authority" shall mean the government of the United States, the United Kingdom, any other nation or any political subdivision thereof, whether state, provincial or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising

executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government, in each case, in any jurisdiction.

"Guarantees" shall meant the guarantees issued pursuant to Section 15.

"Guarantor" shall mean each US Guarantor and each UK Guarantor.

"<u>Hazardous Materials</u>" shall mean (a) any petroleum or petroleum products, radioactive materials, asbestos in any form that is or could become friable, urea formaldehyde foam insulation, dielectric fluid containing levels of polychlorinated biphenyls, and radon gas; (b) any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous waste," "hazardous materials," "extremely hazardous substances," "restricted hazardous waste," "toxic substances," "toxic pollutants," or "pollutants," or words of similar import, under any applicable Environmental Law; and (c) any other chemical, material or substance, the exposure to, or Release of which is prohibited, limited or regulated by any Governmental Authority.

"Highest Adjustable Applicable Margins" shall have the meaning provided in the definition of Applicable Margin.

"Incremental Commitment" shall mean, for any Lender, any Revolving Loan Commitment provided by such Lender after the Effective Date in an Incremental Commitment Agreement delivered pursuant to Section 2.15; it being understood, however, that on each date upon which an Incremental Commitment of any Lender becomes effective, such Incremental Commitment of such Lender shall be added to (and thereafter become a part of) the Revolving Loan Commitment of such Lender for all purposes of this Agreement as contemplated by Section 2.15.

"<u>Incremental Commitment Agreement</u>" shall mean each Incremental Commitment Agreement in substantially the form of Exhibit D (appropriately completed, and with such modifications as may be satisfactory to the Administrative Agent) executed and delivered in accordance with <u>Section 2.15</u>.

"<u>Incremental Commitment Date</u>" shall mean each date upon which an Incremental Commitment under an Incremental Commitment Agreement becomes effective as provided in <u>Section 2.15(b)</u>.

"Incremental Commitment Request Requirements" shall mean, with respect to any request for an Incremental Commitment made pursuant to Section 2.15, the satisfaction of each of the following conditions on the date of such request: (i) no Default or Event of Default then exists or would result therefrom and (ii) all of the representations and warranties contained herein and in the other Credit Documents are true and correct in all material respects at such time (unless stated to relate to a specific earlier date, in which case such representations and warranties shall have been true and correct in all material respects as of such earlier date).

"Incremental Commitment Requirements" shall mean, with respect to any provision of an Incremental Commitment on a given Incremental Commitment Date, the satisfaction of each of the following conditions on or prior to the effective date of the respective Incremental Commitment Agreement: (i) satisfaction of the conditions in the definition of Incremental Commitment Request Requirements (to the extent applicable) as of the effective date of such Incremental Commitment Agreement as certified by an Authorized Officer of US Company in an officer's certificate delivered to the Administrative Agent; (ii) the delivery by US Company to the Administrative Agent of an acknowledgement in form and substance reasonably satisfactory to the Administrative Agent and

executed by each other Credit Party, acknowledging that such Incremental Commitment and all Revolving Loans subsequently incurred, and Letters of Credit issued, pursuant to such Incremental Commitment shall constitute (and be included in the definitions of) Obligations and secured on a pari passu basis with the Obligations under the relevant Security Documents; (iii) the delivery by US Company to the Administrative Agent of an opinion or opinions, in form and substance reasonably satisfactory to the Administrative Agent, from counsel to the Credit Parties reasonably satisfactory to the Administrative Agent and dated such date, covering such matters incident to the transactions contemplated thereby as the Administrative Agent may reasonably request; (iv) the delivery by each Credit Party to the Administrative Agent of such other officers' certificates, board of director (or equivalent governing body) resolutions and evidence of good standing (to the extent available under Applicable Law) as the Administrative Agent shall reasonably request; and (v) the completion by each Credit Party of such other actions as the Administrative Agent may reasonably request in connection with such Incremental Loan Commitment in order to create, continue or maintain the Liens of the Collateral Agent in the Collateral and the perfection thereof (including, without limitation, any mortgage amendments, Mortgage Policies and such other documents reasonably requested by the Administrative Agent to be delivered in connection therewith).

"Incremental Lender" shall have the meaning specified in Section 2.15(b).

"Indebtedness" shall mean, as to any Person, without duplication, (a) all indebtedness of such Person for borrowed money or for the deferred purchase price of property or services, (b) the maximum amount available to be drawn or paid under all letters of credit, bankers' acceptances, bank guaranties, surety and appeal bonds and similar obligations issued for the account of such Person and all unpaid drawings and unreimbursed payments in respect of such letters of credit, bankers' acceptances, bank guaranties, surety and appeal bonds and similar obligations, (c) all indebtedness of the types described in clause (a), (b), (d), (e), (f), (g) or (h) of this definition secured by any Lien on any property owned by such Person, whether or not such indebtedness has been assumed by such Person (provided that, if the Person has not assumed or otherwise become liable in respect of such indebtedness, such indebtedness shall be deemed to be in an amount equal to the Fair Market Value of the property to which such Lien relates), (d) all Capitalized Lease Obligations of such Person, (e) all obligations of such Person to pay a specified purchase price for goods or services, whether or not delivered or accepted, i.e., take-or-pay and similar obligations, (f) all Contingent Obligations of such Person, (g) all obligations under any Interest Rate Protection Agreement, any Other Hedging Agreement or under any similar type of agreement and (h) all Off-Balance Sheet Liabilities of such Person. The Indebtedness of any Person shall include the Indebtedness of any other entity (including any partnership in which such Person is a general partner) to the extent such Person is directly liable therefor as a result of such Person's ownership interest in or other relationship with such entity, except to the extent the terms of such Indebtedness provide that such Person is not liable therefor. Notwithstanding the foregoing, Indebtedness shall not include trade payables, accrued expenses and deferred tax and other credits incurred by any Person

"Individual Exposure" of any Lender shall mean, at any time, the sum of (a) the aggregate principal amount of all Revolving Loans made by such Lender and then outstanding, (b) such Lender's RL Percentage of the aggregate principal amount of all Swingline Loans then outstanding and (c) such Lender's RL Percentage of the aggregate amount of all Letter of Credit Outstandings at such time.

"Initial Borrowing Date" shall mean the date occurring on or after the Effective Date on which the initial Borrowing of Loans occurs.

- "Interest Determination Date" shall mean, with respect to any LIBOR Loan or EURIBOR Loan, the second Business Day prior to the commencement of any Interest Period relating to such LIBOR Loan or EURIBOR Loan.
- "Interest Expense" shall mean the consolidated expense of US Company and its Subsidiaries for interest on Indebtedness, including, without limitation, amortization of original issue discount, incurrence fees (to the extent included in interest expense), the interest portion of any deferred payment obligation and the interest component of any Capitalized Lease Obligation.
 - "Interest Period" shall have the meaning provided in Section 2.09.
- "<u>Interest Rate Protection Agreement</u>" shall mean any interest rate swap agreement, interest rate cap agreement, interest collar agreement, interest rate hedging agreement or other similar agreement or arrangement.
 - "Inventory" shall mean "inventory" as such term is defined in Article 9 of the UCC.
- "Investments" shall mean all expenditures made and all liabilities incurred (including Contingent Obligations) for or in connection with the acquisition of Securities or Indebtedness of a Person, loans, advances, capital contributions or transfers of property to a Person, or acquisition of substantially all the assets of a Person. In determining the aggregate amount of Investments outstanding at any particular time, (i) a guaranty shall be valued at not less than the principal amount guaranteed and outstanding; (ii) returns of capital (but only by repurchase, redemption, retirement, repayment, liquidating dividend or liquidating distribution) shall be deducted; (iii) earnings, whether as dividends, interest or otherwise, shall not be deducted; and (iv) decreases in the market value shall not be deducted.
 - "Issuing Lender" shall mean each US Issuing Lender and each UK Issuing Lender.
 - "Joinder Agreement" shall mean a Joinder Agreement substantially in the form of Exhibit E (appropriately completed).
 - "Joint Bookrunners" shall mean collectively, Deutsche Bank Securities Inc., Banc of America Securities LLC and J.P. Morgan Securities Inc.
- "<u>Landlord Personal Property Collateral Access Agreement</u>" shall mean a Landlord Waiver and Consent Agreement in form and substance satisfactory to the Administrative Agent.
 - "L/C Supportable Obligations" shall mean each of the US L/C Supportable Obligations and the UK L/C Supportable Obligations.
- "<u>Leaseholds</u>" of any Person shall mean all the right, title and interest of such Person as lessee or licensee in, to and under leases or licenses of land, improvements and/or fixtures.
- "Lender" shall mean each financial institution listed on <u>Schedule 1.01(a)</u>, as well as any Person that becomes a "Lender" hereunder pursuant to <u>Section 2.13</u>, <u>2.15</u> or <u>13.04(b)</u>.
- "Lender Default" shall mean (a) the wrongful refusal (which has not been retracted) or the failure of a Lender to make available its portion of any Borrowing (including any Mandatory Borrowing) or to fund its portion of any unreimbursed payment under Section 3.04(c) or (b) a Lender having notified in writing US Company and/or the Administrative Agent that such Lender does not intend to comply with its obligations under Section 2.01(a) or (c), Section 2.04 or Section 3.

- "Letter of Credit" shall mean each US Letter of Credit and each UK Letter of Credit.
- "Letter of Credit Fee" shall mean each of the US Letter of Credit Fee and the UK Letter of Credit Fee.
- "Letter of Credit Outstandings" shall mean, at any time, the sum of the US Letter of Credit Outstandings and the UK Letter of Credit Outstandings.
- "LIBOR" shall mean with respect to any Interest Period, the average of interbank offered rates for deposits in US Dollars or Pounds Sterling, as applicable, having a maturity approximately equal to such Interest Period in the London eurodollar market as set forth on page 3750 (i.e., the LIBOR page), or any successor page, of the Telerate News Services, titled "British Banker Association Interest Settlement Rates" at approximately 11:00 a.m. (London time) two London Business Days prior to the first day of such Interest Period (or same day with respect to LIBOR Loans denominated in Pounds Sterling) or if such rate is not then quoted, the arithmetic average as determined by the Administrative Agent of the rates at which deposits in immediately available US Dollars or Pounds Sterling, as applicable, in an amount equal to the amount of such LIBOR Loan having a maturity approximately equal to such Interest Period are offered to four (4) reference banks to be selected by the Administrative Agent in the London interbank market, at approximately 11:00 a.m. (London time) two London Business Days prior to the first day of such Interest Period (or same day with respect to LIBOR Loans denominated in Pounds Sterling).
 - "LIBOR Loan" shall mean each US LIBOR Loan and each UK LIBOR Loan.
- "<u>Lien</u>" shall mean any mortgage, pledge, hypothecation, charge, assignment, deposit arrangement, encumbrance, lien (statutory or other), preference, priority or other security agreement of any kind or nature whatsoever (including, without limitation, any conditional sale or other title retention agreement, any financing or similar statement or notice filed under the UCC or any other similar recording or notice statute, and any lease having substantially the same effect as any of the foregoing).
- "LKE Account" shall mean any Account arising from the sale or disposal (including by auction) of Equipment or Inventory of Borrower or any of its Domestic Subsidiaries in the ordinary course of business pursuant to the LKE Master Exchange Agreement.
- "LKE Joint Account" shall mean any Deposit Account maintained jointly by US Company or any of its Domestic Subsidiaries and the LKE Qualified Intermediary as may be notified by the US Company to the Administrative Agent from time to time, in which the aggregate amount of funds deposited therein shall not at any time exceed \$25,000,000.
- "LKE Master Exchange Agreement" shall mean any master like-kind exchange agreement, between US Company or any Domestic Subsidiary thereof and the LKE Qualified Intermediary, in a form reasonably satisfactory to the Administrative Agent, together with all amendments and modifications thereof, and replacements and substitutions therefor, which have been consented to by the Administrative Agent, which consent shall not be unreasonably withheld, conditioned or delayed.
 - "LKE QI Receivables" shall mean amounts owing to the US Company or any of its Domestic Subsidiaries from the LKE Qualified Intermediary.
- "<u>LKE Proceeds</u>" shall mean the proceeds received by the LKE Qualified Intermediary from an LKE Account or from the sale of Equipment or Inventory in cash or otherwise, in each case in accordance with the LKE Master Exchange Agreement.

- "<u>LKE Qualified Intermediary</u>" shall mean DBNY or such other person as may be consented to by the Administrative Agent, such consent not to be unreasonable withheld, delayed or conditioned.
- "LKE Transaction" shall mean the sale of Equipment and Inventory and the replacement of such Equipment and Inventory with similar property in a manner that qualifies for deferred recognition of taxable gains for US federal income tax purposes and pursuant to the LKE Master Exchange Agreement.
 - "Loan" shall mean each Revolving Loan and each Swingline Loan.
- "London Business Day" shall means a day for trading by and between banks in US Dollar and Pounds Sterling deposits in the London interbank eurodollar market.
- "<u>Luxembourg Debt</u>" shall mean the intercompany Indebtedness of (i) the UK Company to the Luxembourg Subsidiary in an aggregate principal amount not to exceed £30,000,000 and (ii) the Luxembourg Subsidiary to UK-LP in an aggregate principal amount not to exceed £30,000,000.
- "<u>Luxembourg Security Agreement</u>" shall mean the receivables pledge agreement, dated as of the date hereof, and entered into by the Administrative Agent and UK-LP, as amended, restated, supplemented or otherwise modified from time to time.
 - "Luxembourg Security Documents" shall mean the Luxembourg Share Charge, the Luxembourg Security Agreement and the UK Intercreditor Deed.
- "<u>Luxembourg Share Charge</u>" shall mean the share pledge agreement, dated as of the date hereof, and entered into by the Administrative Agent, UK-LP and the Luxembourg Subsidiary, and as amended, restated, supplemented or otherwise modified from time to time.
- "<u>Luxembourg Subsidiary</u>" shall mean LIKO Luxembourg International S.a.r.l, a company organized under the laws of Luxembourg having its registered office at 74, rue de Merl, L-2146 Luxembourg, Grand Duchy of Luxembourg and registered with the Luxembourg Trade and Companies Register under number B.82639, and an indirect Wholly-Owned Subsidiary of the US Company.
- "Mandatory Redeemable Obligation" shall mean an obligation of US Company or any of its Subsidiaries (or guaranteed by any of them) which must be redeemed or repaid (a) at a fixed or determinable date, whether by operation of sinking fund or otherwise, (b) at the option of any Person other than US Company or such Subsidiary, or (c) upon the occurrence of a condition not solely within the control of US Company or such Subsidiary, such as a redemption required to be made out of future earnings.
 - "Mandatory Borrowing" shall mean each of a Mandatory US Borrowing or a Mandatory UK Borrowing.
- "Mandatory Cost" shall mean the rate of interest calculated by the Administrative Agent to compensate the Lenders for the cost (if any) of compliance with the requirements of the Bank of England, the UK Financial Services Authority and/or the European Central Bank.
 - "Mandatory UK Borrowing" shall have the meaning provided in Section 2.01(f).
 - "Mandatory US Borrowing" shall have the meaning provided in Section 2.01(e).

- "Margin Stock" shall have the meaning provided in Regulation U.
- "Material Adverse Effect" shall mean (i) a material adverse effect on the business, prospects, operations, results of operations, assets, liabilities or condition (financial or otherwise) of US Company and the other Credit Parties, taken as a whole, (ii) the impairment of the ability of US Company or any other Credit Party to perform its obligations under the Credit Documents to which it is a party or of Administrative Agent or the Lenders to enforce the Obligations or realize upon the Collateral, or (iii) a material adverse effect on the value of a material portion of the Collateral or the amount which Administrative Agent or the Lenders would receive (after giving consideration to delays in payment and costs of enforcement) in the liquidation of such Collateral.
 - "Maximum Incremental Commitment Amount" shall mean \$350,000,000.
 - "Maximum UK Letter of Credit Amount" shall have the meaning provided in Section 3.04.
 - "Maximum UK Swingline Amount" shall mean \$10,000,000 or the Equivalent Amount thereof.
 - "Maximum US Letter of Credit Amount" shall have the meaning provided in Section 3.03.
 - "Maximum US Swingline Amount" shall mean \$25,000,000.
- "Merger Agreement" shall mean the Agreement and Plan of Merger, by and among US Company, Cactus Merger Sub, Inc., Target and the target stockholder representative, dated as of February 22, 2008.
- "Mezzanine Notes" shall mean the notes issued pursuant to the Note Purchase Agreement, dated as of August 1, 2006, by and among Mobile Storage WC Holdings Corp., as issuer, WCAS Capital Partners IV, L.P. and Foxkirk, LLC, as purchasers, in the principal amount of \$110,000,000.
- "Minimum Availability Amount" shall mean, at any time, the greater of (i) \$100,000,000 and (ii) ten percent (10%) of the then Total Revolving Loan Commitment.
- "Minimum Borrowing Amount" shall mean for Revolving Loans, \$1,000,000, £1,000,000 or €1,000,000, as applicable, and increments of \$100,000, £100,000, and €100,000 in excess thereof, as applicable; there shall be no Minimum Borrowing Amount with respect to Swingline Loans.
- "Mobile Mini Indenture" shall mean the Indenture, dated May 7, 2007 by and among US Company, as issuer, any guarantors party thereto, and the Law Debenture Trust Company of New York, as trustee, relating to the 6 7/8% Senior Notes due 2015.
- "Mobile Mini Senior Notes" shall mean US Company's senior unsecured notes in the aggregate principal amount of \$150,000,000 due 2015 issued pursuant to the Mobile Mini Indenture, and on terms and conditions satisfactory to the Lenders.
- "Mobile Storage Acquisition" shall mean, collectively, (i) the merger of Mobile Mini Merger Sub, Inc, a wholly owned subsidiary of US Company, into Target, with Target as the surviving corporation, (ii) the consummation of each of the Subsequent Mergers, and (iii) the conversion of all the

issued and outstanding common stock of Target into 8,555,556 shares of Convertible Preferred Stock of US Company and \$12,500,000 in cash, subject to certain adjustments set forth in the Merger Agreement.

- "Mobile Storage Indenture" shall mean the Indenture, dated August 1, 2006 by and among Mobile Services Group, Inc. and Mobile Storage Group, Inc. as issuers, the subsidiary guarantors named therein, and Wells Fargo Bank, N.A. as trustee, relating to the 9 3/4 % Senior Notes due 2014.
- "Mobile Storage Senior Notes" shall mean Mobile Services Group, Inc.'s and Mobile Storage Group, Inc.'s senior unsecured notes in the aggregate principal amount of \$200,000,000 due 2014 issued pursuant to the Mobile Storage Indenture.
 - "Moody's" shall mean Moody's Investors Service, Inc.
- "Mortgage" shall mean a mortgage, leasehold mortgage, deed of trust, leasehold deed of trust, deed to secure debt, leasehold deed to secure debt or similar security instrument in form and substance reasonably acceptable to the Administrative Agent.
 - "Mortgage Policy" shall mean a lender's title insurance policy (Form 1992).
- "Mortgaged Property" shall mean any Real Property owned by US Company or any of its Subsidiaries which is encumbered (or required to be encumbered) by a Mortgage pursuant to the terms of this Agreement or any Security Document.
- "Motor Vehicles" shall mean Inventory or Equipment that is subject to any motor vehicle registration statute, including, without limitation, any of those statutes described in the Uniform Commercial Code Section 9-311(a)(2), as adopted in any state in which a Credit Party owns any Collateral.
- "<u>Multiemployer Plan</u>" shall mean a multiemployer plan as defined in Section 4001(a)(3) of ERISA to which contributions are, or within the immediately preceding five-year period, have been made (or have been required to have been made) by US Company or a Subsidiary of US Company or an ERISA Affiliate.
 - "NAIC" shall mean the National Association of Insurance Commissioners.
- "<u>Net Orderly Liquidation Value</u>" shall mean the Orderly Liquidation Value (net of costs and expenses reasonably estimated to be incurred in connection with such liquidation) of the Borrowing Base Parties' Eligible Machinery and Equipment.
- "Non-Compete Agreements" shall mean all non-compete agreements entered into by US Company or any of its Subsidiaries which materially restrict the activities of US Company or any of its Subsidiaries.
 - "Non-Defaulting Lender" shall mean and include each Lender, other than a Defaulting Lender.
 - "Non-U.S. Lender" shall have the meaning provided in Section 5.04(b).
- "Non-Wholly-Owned Subsidiary" shall mean, as to any Person, each Subsidiary of such Person which is not a Wholly-Owned Subsidiary of such Person.

- "Note" shall mean each US Revolving Note, US Swingline Note, UK Revolving Note and UK Swingline Note.
- "Notice of Borrowing" shall have the meaning provided in Section 2.03(a).
- "Notice of Conversion/Continuation" shall have the meaning provided in Section 2.06.
- "Notice Office" shall mean (i) for credit notices, the office of the Administrative Agent located at 60 Wall Street, New York, New York 10005, Attention: Marguerite Sutton, Telephone No.: (212) 250-6150, and Telecopier No.: (212) 797-4655, and (ii) for operational notices, the office of the Administrative Agent located at 100 Plaza One, Jersey City, New Jersey 07311, Attention: Jonathan Cohen, Telephone No.: (201) 593-2326, and Telecopier No.: (201) 593-2314, or (in either case) such other office or person as the Administrative Agent may hereafter designate in writing as such to the other parties hereto.
 - "Obligations" shall mean, collectively, the UK Obligations and the US Obligations.
- "Off-Balance Sheet Liabilities" of any Person shall mean (a) any repurchase obligation or liability of such Person with respect to accounts or notes receivable sold by such Person, (b) any liability of such Person under any sale and leaseback transactions that does not create a liability on the balance sheet of such Person, (c) any obligation under a Synthetic Lease or (d) any obligation arising with respect to any other transaction which is the functional equivalent of or takes the place of borrowing but which does not constitute a liability on the balance sheet of such Person.
- "Orderly Liquidation Value" shall mean the orderly liquidation value of the Borrowing Base Parties' Eligible Goods Inventory or Eligible Machinery and Equipment that is estimated to be recoverable in an orderly liquidation of such Eligible Goods Inventory or Eligible Machinery and Equipment, as applicable, with such value determined from time to time by reference to the most recent appraisal completed by a qualified independent third-party appraisal company (approved by the Administrative Agent in its Permitted Discretion in consultation with US Company) delivered to the Administrative Agent; provided that, unless an appraisal is required pursuant to Section 10.14, such value in respect of any Eligible Goods Inventory or Eligible Machinery and Equipment acquired in accordance with Section 10.14, may be determined by reference to an identical kind or type of such Eligible Goods Inventory or Eligible Machinery and Equipment, as applicable, that has been so appraised.
- "Other Hedging Agreements" shall mean any foreign exchange contracts, currency swap agreements or other similar agreements, or arrangements designed to protect against fluctuations in currency values.
 - "Other Taxes" shall have the meaning provided in Section 13.01.
 - "Participating Euro Lender" shall have the meaning set forth in Section 2.16(b).
 - "Participating Pounds Lender" shall have the meaning set forth in Section 2.16(a).
- "<u>Participating Member State</u>" means any member state of the European Community that adopts or has adopted the Euro as its lawful currency in accordance with the legislation of the European Community relating to economic and monetary union.
 - "Patriot Act" shall have the meaning provided in Section 8.34.

"<u>Payment Conditions</u>" means with respect to any Restricted Payment (which such term, for purposes of this definition, shall include any prepayment, repurchase or redemption of any Indebtedness permitted pursuant to the Senior Note Documents), Restricted Foreign Funding or Acquisition, each of the following conditions shall be satisfied immediately after giving effect to such Restricted Payment, Restricted Foreign Funding or Acquisition:

(i) either (a) (x) the average daily Total Borrowing Availability over the 90 days prior to the making of such Restricted Payment, Restricted Foreign Funding or Acquisition is greater than \$200,000,000, (y) the Total Borrowing Availability calculated on a pro forma basis before and immediately after giving effect to such Restricted Payment, Restricted Foreign Funding or Acquisition shall be greater than \$200,000,000, and (z) before and immediately after giving effect to such Restricted Payment, Restricted Foreign Funding or Acquisition and any Indebtedness incurred in connection therewith, US Company shall be in compliance with the financial covenant set forth in Section 10.26 (Debt Ratio) hereof on a pro forma basis (whether or not Section 10.24 hereof would then require compliance with such covenant) for the most recently ended fiscal quarter for which the financial statements in Section 9.01(b) have been delivered to the Administrative Agent; or (b) (x) the average Total Borrowing Availability over the 90 days prior to the making of such Restricted Payment, Restricted Foreign Funding or Acquisition shall be greater than \$150,000,000, (y) the Total Borrowing Availability calculated on a pro forma basis before and immediately after giving effect to such Restricted Payment, Restricted Foreign Funding or Acquisition shall be greater than \$150,000,000, and (z) before and immediately after giving effect to such Restricted Payment, Restricted Foreign Funding or Acquisition and any Indebtedness incurred in connection therewith, US Company shall be in compliance with the financial covenants set forth in Section 10.25 (Fixed Charge Coverage Ratio), Section 10.26 (Debt Ratio) and Section 10.27 (Minimum Utilization) hereof, each calculated on a pro forma basis (whether or not Section 10.24 hereof would then require compliance with such covenants) for the most recently ended fiscal quarter for which the financial statements in Section 9.01(b) have been delivered to the Administrative Agent; and

(ii) not later than three Business Days prior to the making of such Restricted Payment, Restricted Foreign Funding or Acquisition, Administrative Agent shall receive (a) a certificate of US Company, with supporting detail acceptable to Administrative Agent certifying that on the date on which such Restricted Payment, Restricted Foreign Funding or Acquisition is made, US Company has satisfied the conditions set forth in clause (i) above and (b) financial projections demonstrating that during the six month period following the making of such Restricted Payment, Restricted Foreign Funding or Acquisition, Total Borrowing Availability at all times shall not be less than the amount required to satisfy the condition set forth in clause (i) above.

"Payment Office" shall mean (i) with respect to payments in US Dollars, the office of the Administrative Agent located at 100 Plaza One, Jersey City, New Jersey 07311 or such other office as the Administrative Agent may hereafter designate in writing as such to the other parties hereto, and (ii) with respect to payments in Pounds Sterling or Euros or other currencies other than US Dollars, such account at such bank or office in London or such other place as Administrative Agent shall designate by notice to the Person required to make the relevant payment.

"PBGC" shall mean the Pension Benefit Guaranty Corporation established pursuant to Section 4002 of ERISA, or any successor thereto.

"Permitted Acquisition" shall mean an Acquisition permitted under Section 10.14 of this Agreement.

"Permitted Discretion" shall mean the commercially reasonable judgment of the Administrative Agent exercised in good faith in accordance with customary business practices for comparable asset-based lending transactions, as to any factor which such Agent reasonably determines: (a) will or reasonably could be expected to adversely affect in any material respect the value of any Eligible Accounts, Eligible Inventory, Eligible Machinery and Equipment and Eligible Real Property, the enforceability or priority of the Collateral Agent's Liens thereon or the amount which any Agent, the Lenders or any Issuing Lender would be likely to receive (after giving consideration to delays in payment and costs of enforcement) in the liquidation of such Eligible Accounts, Eligible Inventory, Eligible Machinery and Equipment or Eligible Real Property or (b) is evidence that any collateral report or financial information delivered to such Agent by any Person on behalf of US Company is incomplete, inaccurate or misleading in any material respect. In exercising such judgment, such Agent may consider, without duplication, such factors already included in or tested by the definition of Eligible Accounts, Eligible Inventory, Eligible Machinery and Equipment or Eligible Real Property, as well as any of the following: (i) changes after the Effective Date in any material respect in any concentration of risk with respect to Accounts; and (iii) any other factors arising after the Effective Date that change in any material respect the credit risk of lending to the Borrowers on the security of the Eligible Accounts, Eligible Inventory, Eligible Machinery and Equipment or Eligible Real Property.

"<u>Permitted Encumbrance</u>" shall mean, with respect to any Mortgaged Property, such exceptions to title as are set forth in the Mortgage Policy delivered with respect thereto, all of which exceptions must be acceptable to the Administrative Agent in its Permitted Discretion.

"Permitted Liens" shall have the meaning provided in Section 10.02.

"Person" shall mean any individual, partnership, joint venture, firm, corporation, association, limited liability company, trust or other enterprise or any Governmental Authority.

"Plan" shall mean any pension plan as defined in Section 3(2) of ERISA, other than a Multiemployer Plan, which is maintained or contributed to by (or to which there is an obligation to contribute of) US Company or a Subsidiary of US Company or an ERISA Affiliate, and each such plan for the five-year period immediately following the latest date on which US Company, a Subsidiary of US Company or an ERISA Affiliate maintained, contributed to or had an obligation to contribute to such plan.

"Pledge Agreements" shall mean each of the US Pledge Agreement and the UK Share Charge.

"Pounds Sterling" or "£" – lawful money of the United Kingdom.

"Pounds Sterling Denominated Revolving Loan" shall have the meaning set forth in Section 2.01(b) hereof.

"<u>Pounds Sterling Funding Capacity</u>" shall mean at any date of determination, for any Lender, the ability of such Lender or an Affiliate or branch office thereof to fund UK Revolving Loans denominated in Pounds Sterling, as set forth in the records of Administrative Agent upon notification from such Lender from time to time.

"Pounds Sterling Participation" shall have the meaning set forth in Section 2.16(a) of this Agreement.

- "Pounds Sterling Participation Fee" shall have the meaning set forth in Section 2.16(i) of this Agreement.
- "Pounds Sterling Participation Settlement" shall have the meaning set forth in Section 2.16(c)(i) of this Agreement.
- "Pounds Sterling Participation Settlement Amount" shall have the meaning set forth in Section 2.16(c)(ii) of this Agreement.
- "Pounds Sterling Participation Settlement Date" shall have the meaning set forth in Section 2.16(c)(i) of this Agreement.
- "Pounds Sterling Participation Settlement Period" shall have the meaning set forth in Section 2.16(c)(i) of this Agreement.
- "PPSA" shall mean the *Personal Property Security Act* (Ontario) and other personal property security legislation of the applicable Canadian province or provinces in respect of the Credit Parties or the Collateral (including the Civil Code of Quebec and the regulation respecting the registration of personal and movable real rights promulgated thereunder) as all such legislation now exists or may from time to time hereafter be amended, modified, recodified, supplemented or replaced, together with all rules, regulations and interpretations thereunder or related thereto.
- "Preferred Equity", as applied to the Equity Interests of any Person, means Equity Interests of such Person (other than common Equity Interests of such Person) of any class or classes (however designed) that ranks prior, as to the payment of dividends or as to the distribution of assets upon any voluntary or involuntary liquidation, dissolution or winding up of such Person, to shares of Equity Interests of any other class of such Person, and shall include any Qualified Preferred Stock of US Company.
- "Prime Lending Rate" shall mean the rate which the Administrative Agent announces from time to time as its prime lending rate, the Prime Lending Rate to change when and as such prime lending rate changes. The Prime Lending Rate is a reference rate and does not necessarily represent the lowest or best rate actually charged to any customer by the Administrative Agent, which may make commercial loans or other loans at rates of interest at, above or below the Prime Lending Rate.
- "Projections" shall mean the projections that are contained in the Confidential Information Memorandum dated May 2008 and that were prepared by or on behalf of US Company in connection with this Agreement and delivered to the Administrative Agent and the Lenders prior to the Initial Borrowing Date.
 - "Property" shall mean any interest in any kind of property, asset or undertaking, whether real, personal or mixed, or tangible or intangible.
- "Qualified Derivative Obligation" shall mean any Derivative Obligation (i) which is owing to Administrative Agent or any Affiliate of Administrative Agent or Bank; (ii) which is owing to any other Lender or any Affiliate of such a Lender and with respect to which Agent has received the notice required pursuant to Section 9.19; or (iii) which is set forth on Schedule 1.01(g).
- "Qualified Preferred Stock" shall mean any Preferred Equity of US Company, so long as the terms of any such Preferred Equity (a) do not contain any mandatory put, redemption, repayment, sinking fund or other similar provision prior to June 27, 2014, (b) do not require the cash payment of

dividends or distributions that would otherwise be prohibited by the terms of this Agreement or any other agreement or contract of US Company or any of its Subsidiaries, (c) do not contain any covenants (other than periodic reporting requirements), (d) do not grant the holders thereof any voting rights except for (i) voting rights required to be granted to such holders under Applicable Law and (ii) limited customary voting rights on fundamental matters such as mergers, consolidations, sales of all or substantially all of the assets of US Company, or liquidations involving US Company, and (e) are otherwise reasonably satisfactory to the Administrative Agent.

"Qualified Swap Termination Value" means any Swap Termination Value in respect of which (and only to the extent of the amount which) the US Company has notified the Administrative Agent on any Borrowing Base Certificate delivered in accordance with the terms hereof, and meeting the requirements of Section 9.01(j), that the Borrower intends such Swap Termination Value to constitute a Qualified Swap Termination Value for purposes of Section 5.03(d) and which amount is reasonably acceptable to the Administrative Agent.

"Quarterly Payment Date" shall mean the last Business Day of each March, June, September and December, it being understood that the first Quarterly Payment Date shall be the last Business Day of September 2008.

"Quarterly Pricing Certificate" shall have the meaning provided in the definition of Applicable Margin.

"Real Property" of any Person shall mean all the right, title and interest of such Person in and to land, improvements and fixtures, including Leaseholds.

"Refinanced Indebtedness" shall mean Indebtedness outstanding as of the Initial Borrowing Date under the (i) Second Amended and Restated Loan and Security Agreement, dated as of February 17, 2006 by and among US Company, the Administrative Agent and the lenders party thereto, as amended, (ii) Credit Agreement, dated August 1, 2006, by and among Mobile Storage Group, Inc., Mobile Services Group, Inc., Target, MSG WV Intermediary Co., The CIT Group/Business Credit, Inc., CIT Capital Securities LLC, Lehman Brothers Inc., and certain lenders party thereto, and all ancillary documents thereto, and (iii) Credit Agreement, dated August 1, 2006, by and among Mobile Storage Group, Inc., Mobile Services Group, Inc., Target, MSG WV Intermediary Co., Ravenstock MSG Limited, The CIT Group/Business Credit, Inc., CIT Capital Securities LLC, Lehman Brothers Inc., and certain lenders party thereto, and all ancillary documents thereto.

"Refinancing" shall mean the refinancing of the Refinanced Indebtedness, as described in Section 6.05(c).

"Refinancing Documents" shall mean all pay-off letters, guaranty releases, Lien releases (including, without limitation, UCC termination statements) and other documents and agreements entered into in connection with the Refinancing.

"Register" shall have the meaning provided in Section 13.15.

"Regulation D" shall mean Regulation D of the Board as from time to time in effect and any successor to all or a portion thereof establishing reserve requirements.

"Regulation T" shall mean Regulation T of the Board as from time to time in effect and any successor to all or a portion thereof.

- "Regulation U" shall mean Regulation U of the Board as from time to time in effect and any successor to all or a portion thereof.
- "Regulation X" shall mean Regulation X of the Board as from time to time in effect and any successor to all or a portion thereof.
- "Release" shall mean actively or passively disposing, discharging, injecting, spilling, pumping, leaking, leaching, dumping, emitting, escaping, emptying, pouring, seeping, migrating or the like, into or upon any land or water or air, or otherwise entering into the environment.
- "Rent Reserve" shall mean a reserve established by the Administrative Agent in respect of rent payments (not to exceed three months rent) made by a Borrowing Base Party for each location at which Eligible Inventory or Eligible Machinery and Equipment of a Borrowing Base Party is located that is not subject to a Landlord Personal Property Collateral Access Agreement (as reported to the Administrative Agent by US Company or UK Company from time to time as requested by the Administrative Agent in its Permitted Discretion.
 - "Replaced Lender" shall have the meaning provided in Section 2.13(a).
 - "Replacement Lender" shall have the meaning provided in Section 2.13(a).
- "Reportable Event" shall mean an event described in Section 4043(c) of ERISA with respect to a Plan that is subject to Title IV of ERISA other than those events as to which the 30-day notice period is waived under subsection .22, .23, .25, .27 or .28 of PBGC Regulation Section 4043.
- "Required Lenders" shall mean, at any time, Non-Defaulting Lenders the sum of whose outstanding Revolving Loan Commitments at such time (or, after the termination thereof, outstanding Revolving Loans and RL Percentages of (x) outstanding Swingline Loans at such time and (y) Letter of Credit Outstandings at such time) represents at least a majority of the sum of the Total Revolving Loan Commitment in effect at such time less the Revolving Loan Commitments of all Defaulting Lenders at such time (or, after the termination thereof, the sum of then total outstanding Revolving Loans of Non-Defaulting Lenders and the aggregate RL Percentages of all Non-Defaulting Lenders of the total outstanding Swingline Loans and Letter of Credit Outstandings at such time).

"Reserves" shall mean reserves, if any, established by the Administrative Agent from time to time hereunder in its Permitted Discretion against the Borrowing Base, including without limitation, (i) Rent Reserves, (ii) Bank Product Reserves, (iii) Unpaid Supplier Reserves, (iv) potential dilution related to Accounts, (v) damaged or defective Equipment, (vi) obsolescence of any Eligible Inventory or Eligible Machinery and Equipment, (vii) sums that the Borrowing Base Parties are or will be required to pay (such as taxes (including, without limitation, sales taxes and payroll taxes), assessments and insurance premiums) and have not yet paid, (viii) Canadian Priority Payables, (ix) amounts owing by any Borrowing Base Party to any Person to the extent secured by a Lien on, or trust over, any Collateral, (x) in the case of any UK Borrowing Base Party, reserves to reflect the prior ranking nature or dilutive effect of UK Priority Claims, (xi) reserves in respect of the aggregate amount of any and all Swap Termination Values constituting Qualified Swap Termination Values for purposes of Section 5.03(d) and (xii) such other events, conditions or contingencies as to which the Administrative Agent, in its Permitted Discretion, determines reserves should be established from time to time hereunder.

"Restricted Foreign Funding" shall have the meaning set forth in Section 10.05(e)(v).

- "Restricted Payment" shall have the meaning set forth in Section 10.04.
- "Returns" shall have the meaning provided in Section 8.13.
- "Revolving Loan" shall mean a US Revolving Loan or a UK Revolving Loan.
- "Revolving Loan Commitment" shall mean, for each Lender, the amount set forth opposite such Lender's name in Schedule 1.01(a) directly below the column entitled "Revolving Loan Commitment," as same may be (x) reduced from time to time or terminated pursuant to Sections 4.02, 4.03 and/or 11, as applicable, (y) increased from time to time pursuant to Section 2.15, or (z) adjusted from time to time as a result of assignments to or from such Lender pursuant to Section 2.13 or 13.04(b).
 - "Revolving Note" shall mean a US Revolving Note or a UK Revolving Note.
- "RL Percentage" of any Lender at any time shall mean a fraction (expressed as a percentage) the numerator of which is the Revolving Loan Commitment of such Lender at such time and the denominator of which is the Total Revolving Loan Commitment at such time, <u>provided</u> that if the RL Percentage of any Lender is to be determined after the Total Revolving Loan Commitment has been terminated, then the RL Percentages of such Lender shall be determined immediately prior (and without giving effect) to such termination.
 - "S&P" shall mean Standard & Poor's Ratings Services, a division of McGraw-Hill, Inc.
 - "SEC" shall have the meaning provided in Section 9.01(h).
 - "Sale Proceeds" shall have the meaning set forth in Section 5.02(c).
 - "Section 5.04(b)(ii) Certificate" shall have the meaning provided in Section 5.04(b)(ii).
 - "Secured Creditors" shall have the meaning assigned that term in the respective Security Documents.
- "<u>Securities</u>" shall mean all shares of stock, shares, partnership interests, membership interests, membership units or other ownership interests in any other Person and all warrants, options or other rights to acquire the same.
 - "Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.
- "Security Agreement" shall mean the US Security Agreement, the US Pledge Agreement, the UK Security Agreements and each other agreement pursuant to which a Credit Party grants a Lien to secure all or part of the Obligations.
 - "Security Agreement Collateral" shall mean all "Collateral" as defined in the Security Agreements.
- "Security Document" shall mean and include the US Security Agreement, the UK Debenture, and the US Pledge Agreement, the UK Share Charge, the Luxembourg Security Documents, each Mortgage, after the execution and delivery thereof, each Additional Security Document and any other related document, agreement or grant pursuant to which US Company or any of its Subsidiaries

grants, perfects or continues a Lien in favor of the Collateral Agent for the benefit of the Secured Creditors.

"Senior Note Documents" shall mean collectively, the Mobile Mini Indenture, the Mobile Storage Indenture, the Mobile Mini Senior Notes, the Mobile Storage Senior Notes and all other agreements, instruments, and documents delivered by US Company or any of its Subsidiaries in connection therewith.

"Senior Note Indentures" shall mean collectively, the Mobile Mini Indenture and the Mobile Storage Indenture,

"Serial Numbered Equipment" shall mean, collectively, the following types of Equipment located at any time, and from time to time, in Canada:
(i) motor vehicles within the meaning of the PPSA of Ontario; (ii) Equipment which is serial number goods within the meaning of the PPSA of Alberta; (iii) Equipment which is serial numbered goods within the meaning of the PPSA of British Columbia, Manitoba, New Brunswick, Newfoundland, Nova Scotia and Prince Edward Island and Saskatchewan; and (iv) Equipment which is road vehicles or other movable property for which a descriptive file may be opened at the Quebec provincial registry office within the meaning of the PPSA in Quebec.

"Settlement Date" shall have the meaning provided in Section 2.04(d)(i).

"Solvent" and "Solvency." shall mean, with respect to any Person on a particular date, the condition that, on such date, (a) the fair value of the property of such Person is greater than the total amount of liabilities, including contingent liabilities, of such Person, (b) the present fair salable value of the assets of such Person is not less than the amount that will be required to pay the probable liability of such Person on its debts as they become absolute and matured, (c) such Person does not intend to, and does not believe that it will, incur debts or liabilities beyond such Person's ability to pay as such debts and liabilities mature, (d) such Person is not engaged in business or a transaction, and is not about to engage in business or a transaction, for which such Person's property would constitute an unreasonably small amount of capital, and (e) in the case of any person incorporated in England and Wales only, a Person that is not "unable to pay its debts". In this context, "unable to pay its debts" means that there are no grounds on which such Person would be deemed unable to pay its debts (as defined in Section 123(1) of the Insolvency Act 1986 of England Wales (as amended by the Enterprise Act 2002 of England and Wales) on the basis that the words "proved to the satisfaction of the court" are deemed omitted from sections 123(1)(e) and 123(2) of that Act) or on which a court would be satisfied that the value of such Person's assets is less than the amount of its liabilities, taking into account its contingent and prospective liabilities (as such term would be construed for the purposes of Section 123(2) of the Insolvency Act 1986 of England and Wales (as amended by the Enterprise Act 2002 of England and Wales)). The amount of contingent liabilities (such as litigation, guaranties and pension plan liabilities) at any time shall be computed as the amount that, in light of all the facts and circumstances existing at the time, represents the amount that can be reasonably be expected to become an actual

"Specified Representations" shall mean the representations set forth in Sections 8.01, 8.02, 8.03, 8.07, 8.23, 8.34, and 8.35 of this Agreement, and Sections 2.01, 2.02 and 2.03 of the US Security Agreement, in each case, only to the extent that such representations relate to the Target and its Subsidiaries (as constituted prior to the consummation of the Mobile Storage Acquisition).

"Spot Rate" shall mean with respect to any currency, the rate quoted by Administrative Agent as the spot rate for the purchase by Administrative Agent of such currency with another currency through its foreign exchange office at approximately 11:00 a.m. (New York time) on the date of determination thereof.

"Start Date" shall have the meaning provided in the definition of Applicable Margin.

"<u>Stated Amount</u>" of each Letter of Credit shall mean, at any time, the maximum amount available to be drawn thereunder (in each case determined without regard to whether any conditions to drawing could then be met).

"Subsequent Mergers" shall mean (i) the merger of Target into US Company, with US Company as the surviving corporation, (ii) the merger of Mobile Storage WC Intermediary Co. with and into US Company, with US Company as the surviving corporation, and (iii) the merger of Mobile Storage Services Group, with and into, US Company, with US Company as the surviving corporation,

"Subsidiary" shall mean, as to any Person, (a) any corporation more than 50% of whose stock of any class or classes having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at the time stock of any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time owned by such Person and/or one or more Subsidiaries of such Person or (b) any partnership, limited liability company, association, joint venture or other entity in which such Person and/or one or more Subsidiaries of such Person has more than a 50% equity interest at the time. Unless otherwise qualified, all references to a "Subsidiary" or to "Subsidiaries" in this Agreement shall refer to a Subsidiary or Subsidiaries of US Company.

"Supermajority Lenders" shall mean those Non-Defaulting Lenders which would constitute the Required Lenders under, and as defined in, this Agreement, if the reference to "a majority" contained in the definition of Required Lenders were changed to "66%."

"Swap Termination Value" means, in respect of any one or more Qualified Derivative Obligations, after taking into account the effect of any legally enforceable netting agreement relating to such Qualified Derivative Obligations, (a) for any date on or after the date on which such Qualified Derivative Obligations have been closed out and termination value(s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in clause (a), the amount(s) determined as the mark-to-market value(s) for such Qualified Derivative Obligations, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Qualified Derivative Obligations (which may include the Administrative Agent, the Bank, a Lender or any Affiliate of any of the foregoing).

"Swingline Expiry Date" shall mean that date which is five Business Days prior to the Final Maturity Date.

"Swingline Lender" shall mean the Administrative Agent, in its capacity as Swingline Lender hereunder.

"Swingline Loan" shall mean a US Swingline Loan or a UK Swingline Loan.

"Swingline Note" shall mean a US Swingline Note or a UK Swingline Note.

"Syndication Agent" shall mean Bank of America, N.A., in its capacity as syndication agent.

"Syndication Date" shall mean that date upon which the Administrative Agent determines in its sole discretion (and notifies US Company) that the primary syndication (and resultant addition of Persons as Lenders pursuant to Section 13.04(b)) has been completed.

"Synthetic Lease" shall mean a lease transaction under which the parties intend that (a) the lease will be treated as an "operating lease" by the lessee and (b) the lessee will be entitled to various tax and other benefits ordinarily available to owners (as opposed to lessees) of like property.

"Target" shall mean Mobile Storage WC Holdings Corp., a Delaware corporation.

"TARGET" means the Trans-European Automated Real-time Gross Settlement Express Transfer payment system which utilises interlinked national real time gross settlement systems and the European Central Bank's payment mechanism and which began operations on 4 January 1999.

"TARGET2" means the Trans-European Automated Real-time Gross Settlement Express Transfer payment system which utilises a single shared platform and which was launched on 19 November 2007.

"TARGET Day" means:

- (a) until such time as TARGET is permanently closed down and ceases operations, any day on which both TARGET and TARGET2 are; and
- (b) following such time as TARGET is permanently closed down and ceases operations, any day on which TARGET2 is, open for the settlement of payments in Euros.

"Taxes" shall have the meaning provided in Section 5.04(a).

"Total Borrowing Availability" shall mean, as of any date of determination, the remainder of (i) the lesser of (x) the Total Revolving Loan Commitment at such time and (y) the Aggregate Borrowing Base at such time minus (ii) the Aggregate Exposure at such time.

"Total Revolving Loan Commitment" shall mean, at any time, the sum of the Revolving Loan Commitments of each of the Lenders at such time.

"<u>Total Unutilized Revolving Loan Commitment</u>" shall mean, at any time, an amount equal to the remainder of (a) the Total Revolving Loan Commitment in effect at such time <u>less</u> (b) the sum of (i) the aggregate principal amount of all Revolving Loans and Swingline Loans outstanding at such time <u>plus</u> (ii) the aggregate amount of all Letter of Credit Outstandings at such time.

"<u>Trailer Fleet Inventory</u>" – new and used manufactured or remanufactured Trailers held by a Borrower or a Credit Party for intended lease or rental to third parties.

"<u>Trailers</u>" shall mean over-the-road tractor trailers and trailers intended for use as storage facilities not constituting portable and ISO containers owned by US Company or any of its Subsidiaries.

"<u>Transactions</u>" shall mean, collectively, (a) the purchase of the Mezzanine Notes by US Company, (b) the consummation of the Mobile Storage Acquisition and the other transactions contemplated by the Acquisition Documents including the Subsequent Mergers, (c) the consummation of the Refinancing, (d) the execution, delivery and performance by each Credit Party of the Credit Documents to which it is a party, the incurrence of Loans on the Initial Borrowing Date and the use of proceeds thereof, and (e) the payment of all fees and expenses in connection with the foregoing.

"Type" shall mean the type of Loan determined with regard to the interest option applicable thereto, i.e., whether a Base Rate Loan, a LIBOR Loan or a EURIBOR Loan.

"UCC" shall mean the Uniform Commercial Code as from time to time in effect in the relevant jurisdiction.

"UK Borrower" and "UK Borrowers" shall have the meaning provided in the first paragraph of this Agreement.

"<u>UK Borrowing</u>" shall mean the borrowing of UK Revolving Loans from all the Lenders, or from the Swingline Lender in the case of UK Swingline Loans, on a given date having the same Interest Period.

"<u>UK Borrowing Availability</u>" shall mean, as of any date of determination, the remainder of (a) the lesser of (i) the UK Maximum Amount at such time and (ii) the UK Borrowing Base at such time minus (b) the Aggregate UK Exposure at such time.

"<u>UK Borrowing Base</u>" shall mean, as of any date of calculation, the amount calculated pursuant to the Borrowing Base Certificate most recently delivered to the Administrative Agent in accordance with <u>Section 9.01(j)</u> (but as modified as provided below in this definition), equal to, without duplication, the sum of

- (i) the amount calculated under the definition of US Borrowing Base, provided that in no event shall any asset constituting a portion of the US Borrowing Base concurrently constitute a portion of the UK Borrowing Base, plus
 - (ii) eighty-five percent (85%) of Eligible UK Accounts, plus
 - (iii) ninety percent (90%) of Eligible UK Container Fleet Inventory, plus
- (iv) the lesser of (A) (1) \$50,000,000 less (2) the amount included in such Borrowing Base Certificate with respect to clause (iii)(B) of the definition of US Borrowing Base and (B) eighty percent (80%) of Eligible Cabin Fleet Inventory, plus
- (v) the lesser of (A) (1) \$40,000,000 less (2) the amount included in such Borrowing Base Certificate with respect to clause (iv)(B) of the definition of US Borrowing Base and (B) the sum of (1) ninety percent (90%) of Eligible UK Container Inventory Held for Sale; <u>plus</u> (2) ninety percent (90%) of Eligible UK Work-in-Process Container Inventory; <u>plus</u> (3) sixty-five percent (65%) of Eligible UK Raw Materials Inventory; <u>plus</u>
- (vi) the lesser of (A) (1) \$50,000,000 less (2) the amount included in such Borrowing Base Certificate with respect to clause (v)(B) of the definition of US Borrowing Base and (B) eighty-five percent (85%) of Eligible UK Machinery and Equipment; minus
 - (vii) the amount of all Reserves then established by the Administrative Agent against the UK Borrowing Base.

The Administrative Agent shall have the right (but no obligation) to review such computations and if, in its Permitted Discretion, such computations have not been calculated in

accordance with the terms of this Agreement, the Administrative Agent shall have the right to correct any such errors in such manner it shall determine in its Permitted Discretion. The UK Borrowing Base (and the Borrowing Base Certificate then most recently delivered) may be adjusted by the Administrative Agent to reflect any new or incremental Reserves required by the Administrative Agent in its Permitted Discretion; <u>provided</u> that no change in the amount of any Reserve shall be effective until the date occurring three Business Days after written notice thereof by the Administrative Agent to the US Company and the UK Company following consultation with the US Company.

"<u>UK Borrowing Base Party</u>" shall mean each Wholly-Owned Subsidiary of US Company that is incorporated under the laws of England and Wales and that is a UK Borrower or a UK Guarantor and is incorporated in England and Wales.

"UK Credit Party" shall mean US Company and each of its Subsidiaries.

"<u>UK Debenture</u>" shall mean the debentures executed by Mobile Storage Group, Inc., the UK Company and each other UK Subsidiary with respect to the UK Obligations in favor of the Collateral Agent for the benefit of the Lenders.

"UK Drawing" shall have the meaning set forth in Section 3.10(b).

"UK Guaranteed Obligations" shall have the meaning set forth in Section 15.01(b).

"<u>UK Guarantors</u>" shall mean US Company and each Subsidiary that guarantees the UK Obligations including each Subsidiary listed under <u>Schedule 10.12</u>, and each other Subsidiary that is or becomes a party to this Agreement pursuant to <u>Section 10.12</u>. For the avoidance of doubt, US Company, each other US Borrower and each US Guarantor shall be UK Guarantors.

"<u>UK Intercreditor Deed</u>" means the intercreditor deed, dated as of the date hereof, and entered into by and among the Administrative Agent, the UK Company, the Luxembourg Subsidiary, Mobile Storage (UK), among others, as amended, restated, supplemented or otherwise modified from time to time.

"<u>UK Issuing Lender</u>" shall mean (i) each of Deutsche Bank AG, London (except as otherwise provided in <u>Section 12.09</u>) and any other Lender reasonably acceptable to the Administrative Agent and US Company or UK Company which agrees to issue UK Letters of Credit hereunder and (ii) with respect to the Existing Letters of Credit, the Lender designated as the issuer thereof on <u>Schedule 3.01(c)</u>. Any UK Issuing Lender may, in its discretion, arrange for one or more UK Letters of Credit to be issued by one or more Affiliates of such UK Issuing Lender (and such Affiliate shall be deemed to be a "UK Issuing Lender" for all purposes of the Credit Documents).

"<u>UK L/C Supportable Obligations</u>" shall mean (a) obligations of a UK Borrower with respect to workers compensation, surety bonds and other similar statutory obligations and (b) such other obligations of a UK Borrower as are reasonably acceptable to the respective UK Issuing Lender and otherwise permitted to exist pursuant to the terms of this Agreement (other than obligations in respect of (i) any Indebtedness or other obligations that are subordinated in right of payment to the UK Obligations and (ii) any Equity Interests).

"UK Letter of Credit" shall have the meaning provided in Section 3.02(a).

"UK Letter of Credit Fee" shall have the meaning provided in Section 4.01(b).

"<u>UK Letter of Credit Outstandings</u>" shall mean, at any time, the sum of (a) the Stated Amount of all outstanding UK Letters of Credit at such time and (b) the aggregate amount of all Unpaid UK Drawings in respect of all UK Letters of Credit at such time.

"UK Letter of Credit Request" shall have the meaning provided in Section 3.06(a).

"UK LIBOR Loan" shall mean each UK Revolving Loan denominated in Pounds Sterling bearing interest at a rate determined by reference to LIBOR.

"UK-LP" shall mean Mobile Storage UK Finance LP, a limited partnership formed under the laws of England and Wales.

"<u>UK Maximum Amount</u>" shall mean, as of any date of determination, the lesser of (a) Equivalent Amount of \$200,000,000 and (b) the Equivalent Amount of the Total Revolving Loan Commitment <u>minus</u> the Equivalent Amount of the Aggregate US Exposure.

"<u>UK Obligations</u>" shall mean, with respect to any UK Credit Party, all Loans, all obligations under Letters of Credit and all other advances, debts, liabilities, obligations, covenants and duties, together with all interest, fees and other charges thereon, owing, arising, due or payable from such UK Credit Party to an Agent, any Lender or any Affiliate of any Lender, or from a UK Borrower to any UK Issuing Lender, of any kind or nature, present or future, whether or not evidenced by any note, guaranty or other instrument, whether arising under the Agreement or any of the other Credit Documents or cash management services rendered in connection therewith, whether direct or indirect (including those acquired by assignment), absolute or contingent, primary or secondary, due or to become due, now existing or hereafter arising and however acquired, and any Banking Product Obligations or Qualified Derivative Obligations owing to an Agent, any Lender or any Affiliate of a Lender, but excluding the US Obligations.

"UK Participant" shall have the meaning set forth in Section 3.08(a).

"<u>UK Priority Claims</u>" means with respect to any UK Borrowing Base Party only, (a) sums which are due by way of contributions to occupational pension schemes and state scheme premiums; (b) unpaid remuneration of employees in respect of the 4-month period prior to insolvency, together with any amount owed in respect of accrued holiday; (c) an amount equal to the aggregate of (i) 50% of the first £10,000 in value of assets comprising the UK Borrowing Base of any UK Borrowing Base Party above £10,000, subject to a cap for this sub-clause (c) of £600,000 for each UK Borrowing Base Party, (e) the expenses of any administration or winding-up and (f) an amount equal to the value of suppliers' retention of title and romalpa claims with respect to Eligible Inventory.

"UK Revolving Loan" shall have the meaning set forth in Section 2.01(b).

"UK Revolving Note" shall have the meaning set forth in Section 2.05(a).

"<u>UK Revolving Loan Commitment</u>" shall mean, for each Lender, the amount set forth opposite such Lender's name in <u>Schedule 1.01(a)</u> directly below the column entitled "UK Revolving Loan Commitment," as same may be (x) reduced from time to time or terminated pursuant to <u>Sections 4.02</u>, <u>4.03</u> and/or <u>11</u>, as applicable, (y) increased from time to time pursuant to <u>Section 2.15</u>, or (z) adjusted from time to time as a result of assignments to or from such Lender pursuant to <u>Section 2.13</u> or <u>13.04(b)</u>.

- "UK Security Agreements" shall mean the UK Debenture and the UK Share Charge.
- "UK Share Charge" shall have the meaning set forth in Section 6.09.
- "UK Swingline Loans" shall have the meaning set forth in Section 2.01(c).
- "UK Swingline Note" shall have the meaning set forth in Section 2.05(a).
- "UK Subsidiary" shall mean each Subsidiary of US Company that is incorporated in England and Wales.
- "<u>Unfinanced Capital Expenditures</u>" for any period, cash expenditures made for Capital Expenditures during such period <u>less</u> the sum of (i) eighty percent (80%) of the actual cost of all additions to Container Fleet Inventory and Trailer Fleet Inventory during such period less cash received from the sale of any Container Fleet Inventory and Trailer Fleet Inventory during such period and (ii) sixty percent (60%) of the actual cost of all additions to fixed assets of US Company and its Subsidiaries during such period less cash received from the sale of any fixed assets during such period.
- "<u>Unfunded Current Liability</u>" of any Plan subject to Title IV of ERISA shall mean the amount, if any, by which the value of the accumulated plan benefits under such Plan determined on a plan termination basis in accordance with actuarial assumptions at such time consistent with those prescribed by the PBGC for purposes of Section 4044 of ERISA, exceeds the Fair Market Value of all plan assets allocable to such liabilities under Title IV of ERISA (excluding any accrued but unpaid contributions).
 - "United States" and "US" shall each mean the United States of America.
- "<u>Unpaid Supplier Reserve</u>" means, at any time, with respect to any Borrowing Base Party which carries on business in Canada or which leases or sells goods in Canada or which has Accounts with Account Debtors located in Canada or otherwise has any assets in Canada, the amount equal to the percentage applicable to Eligible Inventory in the calculation of the US Borrowing Base multiplied by the aggregate value of the Eligible Inventory which the Administrative Agent, in good faith, considers is or may be subject to a right of a supplier to repossess goods pursuant to Section 81.1 of the Bankruptcy and Insolvency Act (Canada) or any other laws of Canada or any other applicable jurisdiction granting revendication or similar rights to unpaid suppliers, in each case, where such supplier's right ranks or is capable of ranking in priority to or pari passu with one or more of the Liens granted in the Security Documents.
 - "Unpaid UK Drawing" shall have the meaning provided in Section 3.10(a).
 - "Unpaid US Drawing" shall have the meaning provided in Section 3.09(a).
- "<u>Unutilized Revolving Loan Commitment</u>" shall mean, with respect to any Lender at any time, such Lender's Revolving Loan Commitment at such time less the sum of (a) the aggregate outstanding principal amount of all Revolving Loans made by such Lender at such time and (b) such Lender's RL Percentage of the Letter of Credit Outstandings at such time.
 - "US Borrower" and "US Borrowers" shall have the meaning provided in the first paragraph of this Agreement.
- "<u>US Borrowing</u>" shall mean the borrowing of one Type of US Revolving Loan from all the Lenders, or from the Swingline Lender in the case of US Swingline Loans, on a given date (or

resulting from a conversion or conversions on such date) having in the case of US LIBOR Loans the same Interest Period; <u>provided</u> that Base Rate Loans incurred pursuant to <u>Section 2.10(b)</u> shall be considered part of the related US Borrowing of US LIBOR Loans.

"<u>US Borrowing Availability</u>" shall mean, as of any date of determination, the remainder of (i) the lesser of (x) the US Maximum Amount at such time and (y) the US Borrowing Base at such time minus (ii) the Aggregate US Exposure at such time.

"<u>US Borrowing Base</u>" shall mean, as of any date of calculation, the amount calculated pursuant to the Borrowing Base Certificate most recently delivered to the Administrative Agent in accordance with <u>Section 9.01(j)</u> (but as modified as provided below in this definition), equal to, without duplication, an amount equal to the sum of:

- (i) eighty-five percent (85%) of Eligible US Accounts, plus
- (ii) ninety percent (90%) of Eligible US Container Fleet Inventory, plus
- (iii) the lesser of (A) (1) \$50,000,000 less (2) the amount included in such Borrowing Base Certificate with respect to clause (iv)(B) of the definition of UK Borrowing Base and (B) eighty percent (80%) of Eligible Trailer Fleet Inventory, plus
- (iv) the lesser of (A) (1) \$40,000,000 less (2) the amount included in such Borrowing Base Certificate with respect to clause (v)(B) of the definition of UK Borrowing Base and (B) the sum of (1) ninety percent (90%) of Eligible US Container Inventory Held for Sale; <u>plus</u> (2) ninety percent (90%) of Eligible US Work-in-Process Container Inventory; <u>plus</u> (3) sixty-five percent (65%) of Eligible US Raw Materials Inventory; <u>plus</u>
- (v) the lesser of (A) (1) \$50,000,000 less (2) the amount included in such Borrowing Base Certificate with respect to clause (vi)(B) of the definition of UK Borrowing Base and (B) the sum of (1) eighty-five percent (85%) of Eligible US Machinery and Equipment; <u>plus</u> (2) sixty percent (60%) of Eligible Real Property; <u>minus</u>
 - (vi) the amount of all Reserves then established by the Administrative Agent against the US Borrowing Base.

The Administrative Agent shall have the right (but no obligation) to review such computations and if, in its Permitted Discretion, such computations have not been calculated in accordance with the terms of this Agreement, the Administrative Agent shall have the right to correct any such errors in such manner it shall determine in its Permitted Discretion. The US Borrowing Base (and the Borrowing Base Certificate then most recently delivered) may be adjusted by the Administrative Agent to reflect any new or incremental Reserves required by the Administrative Agent in its Permitted Discretion; provided that no change in the amount of any Reserve shall be effective until the date occurring three Business Days after written notice thereof by the Administrative Agent to the US Company after consultation with US Company.

"<u>US Borrowing Base Party</u>" shall mean US Company and each Wholly-Owned Domestic Subsidiary of US Company that is a US Borrower and/or a US Guarantor and was formed in a US jurisdiction.

"US Company" shall have the meaning provided in the first paragraph of this Agreement.

- "US Credit Party" shall mean each of the US Borrowers and each of the US Guarantors.
- "US Drawing" shall have the meaning set forth in Section 3.09(b).
- "US Guaranteed Obligations" shall have the meaning set forth in Section 15.01(a).
- "<u>US Guarantors</u>" shall mean US Company and each Domestic Subsidiary of US Company and each other Person who now or hereafter guarantees payment or performance of the whole or any part of the US Obligations, including each Domestic Subsidiary listed on <u>Schedule 10.12</u> and each other Domestic Subsidiary that is or becomes a party to this Agreement pursuant to <u>Section 10.12</u>.
- "US Issuing Lender" shall mean (i) each of Deutsche Bank AG New York Branch (except as otherwise provided in Section 12.09) and any other Lender reasonably acceptable to the Administrative Agent and US Company which agrees to issue US Letters of Credit hereunder and (ii) with respect to the Existing Letters of Credit, the Lender designated as the issuer thereof on Schedule 3.01(c). Any US Issuing Lender may, in its discretion, arrange for one or more US Letters of Credit to be issued by one or more Affiliates of such US Issuing Lender (and such Affiliate shall be deemed to be a "US Issuing Lender" for all purposes of the Credit Documents).
- "<u>US L/C Supportable Obligations</u>" shall mean (a) obligations of a US Borrower with respect to workers compensation, surety bonds and other similar statutory obligations and (b) such other obligations of a US Borrower as are reasonably acceptable to the respective US Issuing Lender and otherwise permitted to exist pursuant to the terms of this Agreement (other than obligations in respect of (i) any Indebtedness or other obligations that are subordinated in right of payment to the US Obligations and (ii) any Equity Interests).
 - "US Letter of Credit" shall have the meaning provided in Section 3.01(a).
 - "US Letter of Credit Fee" shall have the meaning provided in Section 4.01(b).
- "<u>US Letter of Credit Outstandings</u>" shall mean, at any time, the sum of (a) the Stated Amount of all outstanding US Letters of Credit at such time and (b) the aggregate amount of all Unpaid US Drawings in respect of all US Letters of Credit at such time.
 - "US Letter of Credit Request" shall have the meaning provided in Section 3.05(a).
- "<u>US LIBOR Loan</u>" shall mean each US Revolving Loan designated as such by the relevant Borrower at the time of the incurrence thereof or conversion thereto bearing interest at a rate determined by reference to LIBOR.
- "<u>US Maximum Amount</u>" shall mean, as of any date of determination, the Equivalent Amount of the Total Revolving Loan Commitment <u>minus</u> the Equivalent Amount of the Aggregate UK Exposure.
- "<u>US Obligations</u>" with respect to any US Credit Party, all Loans, all obligations arising under Letters of Credit and all other advances, debts, liabilities, obligations, covenants and duties, together with all interest, fees and other charges thereon, owing, arising, due or payable from such US Credit Party to an Agent, any Lender or any Affiliate of any Lender, or from a US Borrower to any US Issuing Lender, of any kind or nature, present or future, whether or not evidenced by any note, guaranty or other instrument, whether arising under the Agreement or any of the other Credit Documents or cash management services rendered in connection therewith, whether direct or indirect (including those

acquired by assignment), absolute or contingent, primary or secondary, due or to become due, now existing or hereafter arising and however acquired, and any Banking Product Obligations and any Qualified Derivative Obligations owing to an Agent, any Lender or any Affiliate of a Lender.

- "US Participant" shall have the meaning set forth in Section 3.07(a).
- "US Pledge Agreement" shall have the meaning set forth in Section 6.08.
- "US Pledge Agreement Collateral" shall mean all "Collateral" as defined in the US Pledge Agreement.
- "US Revolving Loans" shall have the meaning set forth in Section 2.01(a).
- "US Revolving Note" shall have the meaning set forth in Section 2.05(a).
- "<u>US Security Agreement</u>" shall mean the security agreement executed by US Company and each other US Credit Party with respect to the US Obligations in favor of the Collateral Agent for the benefit of the Lenders.
 - "US Swingline Loans" shall have the meaning set forth in Section 2.01(c).
 - "US Swingline Note" shall have the meaning set forth in Section 2.05(a).
- "<u>Voting Stock</u>" shall mean Securities of any class or classes of a corporation, limited partnership or limited liability company or any other entity the holders of which are ordinarily, in the absence of contingencies, entitled to vote with respect to the election of corporate directors (or Persons performing similar functions).
 - "Wholly-Owned Domestic Subsidiary" shall mean, as to any Person, any Domestic Subsidiary of such Person that is a Wholly-Owned Subsidiary.
- "Wholly-Owned Subsidiary" shall mean, as to any Person, (a) any corporation 100% of whose capital stock is at the time owned by such Person and/or one or more Wholly-Owned Subsidiaries of such Person and (b) any partnership, limited liability company, association, joint venture or other entity in which such Person and/or one or more Wholly-Owned Subsidiaries of such Person has a 100% equity interest at such time (other than, in the case of a Foreign Subsidiary of US Company with respect to the preceding clauses (a) and (b), directors' qualifying shares and/or other nominal amounts of shares required to be held by Persons other than US Company and its Subsidiaries under Applicable Law).

SECTION 2. Amount and Terms of Credit.

2.01 The Commitments. (a) US Revolving Loans. Subject to and upon the terms and conditions set forth herein, each Lender severally agrees to make, at any time and from time to time on or after the Initial Borrowing Date and prior to the Final Maturity Date, a revolving loan or revolving loans (each, a "US Revolving Loan" and, collectively, the "US Revolving Loans") to the US Borrowers (on a joint and several basis), which US Revolving Loans (i) shall be denominated in Dollars, (ii) shall, at the option of the respective Borrower, be incurred and maintained as, and/or converted into, Base Rate Loans or LIBOR Loans; provided that, (A) except as otherwise specifically provided in Section 2.10(c), all US Revolving Loans comprising the same Borrowing shall at all times be of the same Type and (B) unless the Administrative Agent otherwise has agreed or has determined that the Syndication Date has occurred (at which time this clause (B) shall no longer be applicable), no more

than two Borrowings of US Revolving Loans to be maintained as LIBOR Loans may be incurred prior to the 60th day after the Initial Borrowing Date (or, if later, the last day of the Interest Period applicable to the second Borrowing of LIBOR Loans referred to below), each of which Borrowings of LIBOR Loans may only have an Interest Period of one month, and the first of which Borrowings may only be made on, or within five Business Days after, the Initial Borrowing Date, and the second of which Borrowings may only be made on the last day of the Interest Period of the first such Borrowing, (iii) may be repaid and reborrowed in accordance with the provisions hereof, (iv) shall not be made (and shall not be required to be made) by any Lender in any instance where the incurrence thereof (after giving effect to the use of the proceeds thereof on the date of the incurrence thereof to repay any amounts theretofore outstanding pursuant to this Agreement) would cause the Individual Exposure of such Lender to exceed the amount of its Revolving Loan Commitment at such time and (v) shall not be made (and shall not be required to be made) by any Lender in any instance where the incurrence thereof (after giving effect to the use of the proceeds thereof on the date of the incurrence thereof to repay any amounts theretofore outstanding pursuant to this Agreement) would cause (A) the Aggregate Exposure to exceed the Total Revolving Loan Commitment, as then in effect, (B) the Aggregate US Exposure to exceed the US Maximum Amount, as then in effect, (C) the Aggregate Exposure to exceed the Aggregate Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate), or (D) the Aggregate US Exposure to exceed the US Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate).

(b) <u>UK Revolving Loans</u>. Subject to and upon the terms and conditions set forth herein, each Lender severally agrees to make, at any time and from time to time on or after the Initial Borrowing Date and prior to the Final Maturity Date, a revolving loan or revolving loans (each, a "UK Revolving Loan" and, collectively, the "UK Revolving Loans") to the UK Borrowers (on a joint and several basis), which UK Revolving Loans (i) shall, at the option of the applicable UK Borrower, be denominated in either Pounds Sterling (such Revolving Loans, "Pounds Sterling Denominated Revolving Loans") or Euros (such Revolving Loans, "Euro Denominated Revolving Loans"), (ii) shall be incurred and maintained as solely in the case of UK Revolving Loans denominated in Pounds Sterling, LIBOR Loans, and in the case of UK Revolving Loans denominated in Euros, EURIBOR Loans; provided that, (A) except as otherwise specifically provided in <u>Section 2.01(d)</u>, all UK Revolving Loans comprising the same Borrowing shall at all times be of the same Type and (B) unless the Administrative Agent otherwise has agreed or has determined that the Syndication Date has occurred (at which time this clause (B) shall no longer be applicable), no more than five Borrowings of UK Revolving Loans may be incurred prior to the 60th day after the Initial Borrowing Date (or, if later, the last day of the Interest Period applicable to the fifth Borrowing of LIBOR Loans referred to below), each of which Borrowings of UK Revolving Loans may only have an Interest Period of one month, (iii) may be repaid and reborrowed in accordance with the provisions hereof, (iv) shall not be made (and shall not be required to be made) by any Lender in any instance where the incurrence thereof (after giving effect to the use of the proceeds thereof on the date of the incurrence thereof to repay any amounts theretofore outstanding pursuant to this Agreement) would cause the Individual Exposure of such Lender to exceed the amount of its Revolving Loan Commitment at such time and (v) shall not be made (and shall not be required to be made) by any Lender in any instance where the incurrence thereof (after giving effect to the use of the proceeds thereof on the date of the incurrence thereof to repay any amounts theretofore outstanding pursuant to this Agreement) would cause (A) the Aggregate Exposure to exceed the Total Revolving Loan Commitment, as then in effect, (B) the Aggregate UK Exposure to exceed the UK Maximum Amount or the UK Revolving Loan Commitments, as then in effect, (C) the Aggregate Exposure to exceed the Aggregate Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate or (D) the Aggregate UK Exposure to exceed the UK Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate). Upon receipt by the Administrative Agent of any Notice of Borrowing pursuant to which the UK Company or any other UK Borrower proposes to borrow in Pounds Sterling or Euros, the Administrative Agent shall provide notice of the same to each Lender.

(c) US Swingline Loans. Subject to and upon the terms and conditions set forth herein, the Swingline Lender agrees to make, at any time and from time to time on or after the Initial Borrowing Date and prior to the Swingline Expiry Date, a revolving loan or revolving loans (each, a "US Swingline Loan" and, collectively, the "US Swingline Loans") to the US Borrowers (on a joint and several basis), which US Swingline Loans (i) shall be denominated in Dollars, (ii) shall be incurred and maintained as Base Rate Loans; (iii) may be repaid and reborrowed in accordance with the provisions hereof, (iv) shall not be made (and shall not be required to be made) by the Swingline Lender in any instance where the incurrence thereof (after giving effect to the use of the proceeds thereof on the date of the incurrence thereof to repay any amounts theretofore outstanding pursuant to this Agreement) would cause (A) the Aggregate Exposure to exceed the Total Revolving Loan Commitment, as then in effect, (B) the Aggregate US Exposure to exceed the US Maximum Amount, as then in effect, (C) the Aggregate Exposure to exceed the Aggregate Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate, or (D) the Aggregate US Exposure to exceed the US Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate), and (v) shall not exceed in aggregate principal amount at any time outstanding the Maximum US Swingline Amount. Notwithstanding anything to the contrary contained in this Section 2.01(c), (i) the Swingline Lender shall not be obligated to make any US Swingline Loans at a time when a Lender Default exists unless the Swingline Lender has entered into arrangements satisfactory to it and US Company to eliminate the Swingline Lender's risk with respect to the Defaulting Lender's or Defaulting Lenders' participation in such Swingline Loans, including by cash collateralizing such Defaulting Lender's or Defaulting Lenders' RL Percentage of the outstanding Swingline Loans, (ii) the Swingline Lender shall not make any US Swingline Loan after it has received written notice from any Borrower, any other Credit Party or the Required Lenders stating that a Default or an Event of Default exists and is continuing until such time as the Swingline Lender shall have received written notice (A) of rescission of all such notices from the party or parties originally delivering such notice or notices or (B) of the waiver of such Default or Event of Default by the Required Lenders, and (iii) no Swingline Lender shall be obligated to make US Swingline Loans to the extent such advance would cause such Lender's then outstanding Loans to exceed its Revolving Credit Commitment.

(d) <u>UK Swingline Loans</u>. Subject to and upon the terms and conditions set forth herein, the Swingline Lender agrees to make, at any time and from time to time on or after the Initial Borrowing Date and prior to the Swingline Expiry Date, a revolving loan or revolving loans (each, a "<u>UK Swingline Loans</u>" and, collectively, the "<u>UK Swingline Loans</u>") to the UK Borrowers (on a joint and several basis), which UK Swingline Loans (i) shall be denominated in Pounds Sterling or Euros, at the option of the applicable Borrower, (ii) shall be incurred and maintained as in the case of UK Revolving Loans denominated in Pounds Sterling, LIBOR Loans, and in the case of UK Revolving Loans denominated in Euros, EURIBOR Loans, with an Interest Period of one week; (iii) may be repaid and reborrowed in accordance with the provisions hereof, (iv) shall not be made (and shall not be required to be made) by the Swingline Lender in any instance where the incurrence thereof (after giving effect to the use of the proceeds thereof on the date of the incurrence thereof to repay any amounts theretofore outstanding pursuant to this Agreement) would cause (A) the Aggregate Exposure to exceed the Total Revolving Loan Commitment, as then in effect, (B) the Aggregate UK Exposure to exceed the UK Maximum Amount, as then in effect, (C) the Aggregate Exposure to exceed the Aggregate Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate), or (D) the Aggregate UK Exposure to exceed the UK Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate), or (D) the Aggregate UK Exposure to exceed the UK Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate), or (D) the Aggregate UK Exposure to exceed the UK Borrowing Base Certificate), or (D) the Aggregate UK Exposure to exceed the UK Borrowing Base Certificate), or (D) the Aggregate UK Exposure to exceed the UK Borrowing Base Certificate), or (D) the Aggregate UK Exposure to exceed the UK Bo

Defaulting Lender's or Defaulting Lenders' RL Percentage of the outstanding Swingline Loans, (ii) the Swingline Lender shall not make any UK Swingline Loan after it has received written notice from any Borrower, any other Credit Party or the Required Lenders stating that a Default or an Event of Default exists and is continuing until such time as the Swingline Lender shall have received written notice (A) of rescission of all such notices from the party or parties originally delivering such notice or notices or (B) of the waiver of such Default or Event of Default by the Required Lenders, and (iii) no Swingline Lender shall be obligated to make UK Swingline Loans to the extent such advance would cause such Lender's then outstanding Loans to exceed its Revolving Credit Commitment.

(e) Mandatory US Borrowings. On any Business Day, the Swingline Lender may, in its sole discretion, give notice to the Lenders that the Swingline Lender's outstanding US Swingline Loans shall be funded with one or more Borrowings of US Revolving Loans (provided that such notice shall be deemed to have been automatically given upon the occurrence of a Default or an Event of Default under Section 11.08 or upon the exercise of any of the remedies provided in the last paragraph of Section 11), in which case one or more Borrowings of US Revolving Loans constituting Base Rate Loans (each such Borrowing, a "Mandatory US Borrowing") shall be made on the immediately succeeding Business Day by all Lenders pro rata based on each such Lender's RL Percentage (determined before giving effect to any termination of the Revolving Loan Commitments pursuant to the last paragraph of Section 11) and the proceeds thereof shall be applied directly by the Swingline Lender to repay the Swingline Lender for such outstanding US Swingline Loans. Each Lender hereby irrevocably agrees to make US Revolving Loans upon one Business Day's notice pursuant to each Mandatory US Borrowing in the amount and in the manner specified in the preceding sentence and on the date specified in writing by the Swingline Lender notwithstanding (i) the amount of the Mandatory US Borrowing may not comply with the Minimum Borrowing Amount otherwise required hereunder, (ii) whether any conditions specified in Section 7 are then satisfied, (iii) whether a Default or an Event of Default then exists, (iv) the date of such Mandatory US Borrowing, (v) the amount of the US Borrowing Base or the US Maximum Amount at such time, and (vi) the amount of the Aggregate Borrowing Base or the Total Revolving Loan Commitment at such time. In the event that any Mandatory US Borrowing cannot for any reason be made on the date otherwise required above (including, without limitation, as a result of the commencement of a proceeding under the Bankruptcy Code with respect to any Borrower), then each Lender hereby agrees that it shall forthwith purchase (as of the date the Mandatory US Borrowing would otherwise have occurred, but adjusted for any payments received from any Borrower on or after such date and prior to such purchase) from the Swingline Lender such participations in the outstanding US Swingline Loans as shall be necessary to cause the Lenders to share in such US Swingline Loans ratably based upon their respective RL Percentages (determined before giving effect to any termination of the Revolving Loan Commitments pursuant to the last paragraph of Section 11), provided that (x) all interest payable on the US Swingline Loans shall be for the account of the Swingline Lender until the date as of which the respective participation is required to be purchased and, to the extent attributable to the purchased participation, shall be payable to the participant from and after such date and (y) at the time any purchase of participations pursuant to this sentence is actually made, the purchasing Lender shall be required to pay the Swingline Lender interest on the principal amount of participation purchased for each day from and including the day upon which the Mandatory US Borrowing would otherwise have occurred to but excluding the date of payment for such participation, at the overnight Federal Funds Rate for the first three days and at the interest rate otherwise applicable to US Revolving Loans maintained as Base Rate Loans hereunder for each day thereafter.

(f) <u>Mandatory UK Borrowings</u>. On any Business Day, the Swingline Lender may, in its sole discretion, give notice to the Lenders that the Swingline Lender's outstanding UK Swingline Loans shall be funded with one or more Borrowings of UK Revolving Loans (<u>provided</u> that such notice shall be deemed to have been automatically given upon the occurrence of a Default or an Event of Default under <u>Section 11.08</u> or upon the exercise of any of the remedies provided in the last paragraph of Section

- 11), in which case one or more Borrowings of UK Revolving Loans constituting LIBOR Loans or, in the case of UK Swingline Loans in Euros, EURIBOR Loans with an Interest Period of one week (each such Borrowing, a "Mandatory UK Borrowing") shall be made on the immediately succeeding Business Day by all Lenders pro rata based on each such Lender's RL Percentage (determined before giving effect to any termination of the Revolving Loan Commitments pursuant to the last paragraph of Section 11) and the proceeds thereof shall be applied directly by the Swingline Lender to repay the Swingline Lender for such outstanding UK Swingline Loans. Each Lender hereby irrevocably agrees to make UK Revolving Loans upon one Business Day's notice pursuant to each Mandatory UK Borrowing in the amount and in the manner specified in the preceding sentence and on the date specified in writing by the Swingline Lender notwithstanding (i) the amount of the Mandatory UK Borrowing may not comply with the Minimum Borrowing Amount otherwise required hereunder, (ii) whether any conditions specified in Section 7 are then satisfied, (iii) whether a Default or an Event of Default then exists, (iv) the date of such Mandatory UK Borrowing, (v) the amount of the UK Borrowing Base or the UK Maximum Amount at such time, and (vi) the amount of the Aggregate Borrowing Base or the Total Revolving Loan Commitment at such time. In the event that any Mandatory UK Borrowing cannot for any reason be made on the date otherwise required above (including, without limitation, as a result of the occurrence of any of the circumstances listed in Section 11.08 with respect to any UK Borrower), then each Lender hereby agrees that it shall forthwith purchase (as of the date the Mandatory UK Borrowing would otherwise have occurred, but adjusted for any payments received from any Borrower on or after such date and prior to such purchase) from the Swingline Lender such participations in the outstanding UK Swingline Loans as shall be necessary to cause the Lenders to share in such UK Swingline Loans ratably based upon their respective RL Percentages (determined before giving effect to any termination of the Revolving Loan Commitments pursuant to the last paragraph of Section 11), provided that (x) all interest payable on the UK Swingline Loans shall be for the account of the Swingline Lender until the date as of which the respective participation is required to be purchased and, to the extent attributable to the purchased participation, shall be payable to the participant from and after such date and (y) at the time any purchase of participations pursuant to this sentence is actually made, the purchasing Lender shall be required to pay the Swingline Lender interest on the principal amount of participation purchased for each day from and including the day upon which the Mandatory UK Borrowing would otherwise have occurred to but excluding the date of payment for such participation, at the rate for one-week LIBOR or EURIBOR (as applicable) Interest Period with respect to such UK Swingline Loan.
- (g) Notwithstanding anything to the contrary in Section 2.01(a), Section 2.01(b) or elsewhere in this Agreement, the Administrative Agent shall have the right, following reasonable consultation with US Company, to establish Reserves in such amounts, and with respect to such matters, as the Administrative Agent in its Permitted Discretion shall deem necessary or appropriate, against the US Borrowing Base and the UK Borrowing Base (which reserves shall reduce the then existing US Borrowing Base and UK Borrowing Base in an amount equal to such reserves).
- (h) (i) In the event that the Borrowers are unable to comply with the US Borrowing Base limitations set forth in Section 2.01(a), or (ii) the Borrowers are unable to comply with the conditions precedent to the making of Revolving Loans set forth in Section 7, in either case, the Lenders, subject to the immediately succeeding proviso, hereby authorize the Administrative Agent, for the account of the Lenders, to make US Revolving Loans to the US Borrowers (on a joint and several basis), solely in the event that the Administrative Agent in its Permitted Discretion deems necessary or desirable (A) to preserve or protect the Collateral, or any portion thereof, (B) to enhance the likelihood of repayment of the Obligations, or (C) to pay any other amount chargeable to the Borrowers pursuant to the terms of this Agreement or any other Credit Document, including, without limitation, Expenses and Fees, which Revolving Loans may only be made as Base Rate Loans (an "Agent Advance"), for a period commencing on the date the Administrative Agent first receives a Notice of Borrowing requesting an Agent Advance until the earliest of (x) the twentieth Business Day after such date, (y) the date the

respective Borrowers are again able to comply with the US Borrowing Base limitations and the conditions precedent to the making of US Revolving Loans, or obtains an amendment or waiver with respect thereto and (z) the date the Required Lenders instruct the Administrative Agent to cease making Agent Advances (the "Agent Advance Period"); provided that the Administrative Agent shall not make any Agent Advance to the extent that at the time of the making of such Agent Advance, the amount of such Agent Advance (I) when added to the aggregate outstanding amount of all other Agent Advances made to the US Borrowers at such time, would exceed 5.0% of the US Borrowing Base at such time (based on the Borrowing Base Certificate last delivered) (the "Agent Advance Amount") or (II) when added to the Aggregate US Exposure as then in effect (immediately prior to the incurrence of such Agent Advance), would exceed the US Maximum Amount at such time. Agent Advances may be made by the Administrative Agent in its sole discretion and the Borrowers shall have no right whatsoever to require that any Agent Advances be made. Agent Advances will be subject to periodic settlement with the Lenders pursuant to Section 2.04(d).

2.02 Minimum Amount of Each Borrowing. The aggregate principal amount of each Borrowing of Loans shall not be less than the Minimum Borrowing Amount applicable thereto. More than one Borrowing may occur on the same date, but at no time shall there be outstanding more than ten Borrowings of US LIBOR Loans (or such greater number of Borrowings of US LIBOR Loans as may be agreed to from time to time by the Administrative Agent) or more than five Borrowings of UK Revolving Loans (or such greater number of Borrowings of UK Revolving Loans as may be agreed to from time to time by the Administrative Agent).

2.03 Notice of Borrowing. (a) Whenever a Borrower desires to incur (i) LIBOR Loans or EURIBOR Loans hereunder (excluding UK Swingline Loans and UK Revolving Loans made pursuant to a Mandatory UK Borrowing), such Borrower shall give the Administrative Agent at the Notice Office at least three Business Days' prior notice of each LIBOR Loan or EURIBOR Loan to be incurred hereunder and (ii) Base Rate Loans hereunder (including Agent Advances, but excluding US Swingline Loans and US Revolving Loans made pursuant to a Mandatory US Borrowing), such Borrower shall give the Administrative Agent at the Notice Office at least one Business Day's prior notice of each Base Rate Loan to be incurred hereunder; provided that any such notice shall be deemed to have been given on a certain day only if given before 2:00 P.M. (New York City time) on such day (or, in the case of any UK Revolving Loans, 10:00 A.M. (New York City time) on such day). Each such notice (each, a "Notice of Borrowing"), except as otherwise expressly provided in Section 2.10, shall be irrevocable and shall be in writing, or by telephone promptly confirmed in writing, in the form of Exhibit A-1, appropriately completed to specify: (A) the aggregate principal amount of the Revolving Loans to be incurred pursuant to such Borrowing, (B) the date of such Borrowing (which shall be a Business Day), (C) whether the Revolving Loans made pursuant to such Borrowing constitute US Revolving Loans or UK Revolving Loans, (D) whether the Revolving Loans made pursuant to such Borrowing constitute Agent Advances (it being understood that the Administrative Agent shall be under no obligation to make such Agent Advance), (E) in the case of US Revolving Loans, whether the Revolving Loans being incurred pursuant to such Borrowing are to be initially maintained as Base Rate Loans or, to the extent permitted hereunder, LIBOR Loans and, if US LIBOR Loans, the initial Interest Period to be applicable thereto, (F) in the case of UK Revolving Loans whether EURIBOR Loans or UK LIBOR Loans, the initial Interest Period to be applicable thereto and whether denominated in Pounds Sterling or Euros, and (G) the US Borrowing Base and the UK Borrowing Base at such time (based on the Borrowing Base Certificate last delivered). Except as provided in Section 2.04(d), the Administrative Agent shall promptly give each Lender notice of such proposed Borrowing, of such Lender's proportionate share thereof and of the other matters required by the immediately preceding sentence to be specified in the Notice of Borrowing.

- (b) (i) Whenever a US Borrower desires to incur US Swingline Loans hereunder, such Borrower shall give the Swingline Lender no later than 2:00 P.M. (New York City time) on the date that a US Swingline Loan is to be incurred, written notice or telephonic notice promptly confirmed in writing of each US Swingline Loan to be incurred hereunder. Each such notice shall be irrevocable and specify in each case (A) the date of Borrowing (which shall be a Business Day), and (B) the aggregate principal amount of the US Swingline Loans to be incurred pursuant to such Borrowing.
- (ii) Whenever a UK Borrower desires to incur UK Swingline Loans hereunder, such Borrower shall give the Swingline Lender one Business Days' prior notice of each UK Swingline Loan to be incurred hereunder; <u>provided</u> that (in each case) any such notice shall be deemed to have been given on a certain day only if given before 10:00 A.M. (New York City time) on such day or such later time as shall be acceptable to the Administrative Agent (which such notice shall be written notice or telephonic notice promptly confirmed in writing). Each such notice shall be irrevocable and specify in each case (A) the date of Borrowing (which shall be a Business Day), and (B) the aggregate principal amount of the UK Swingline Loans to be incurred pursuant to such Borrowing.
- (iii) Mandatory Borrowings shall be made upon the notice specified in Sections $\underline{2.01(e)}$ or $\underline{2.01(f)}$, as applicable, with the respective Borrower irrevocably agreeing, by its incurrence of any Swingline Loan, to the making of the Mandatory Borrowings as set forth in $\underline{\text{Section 2.01(e)}}$ or $\underline{2.01(f)}$, as applicable.
- (c) Without in any way limiting the obligation of any Borrower to confirm in writing any telephonic notice of any Borrowing or prepayment of Loans, the Administrative Agent or the Swingline Lender, as the case may be, may act without liability upon the basis of telephonic notice of such Borrowing or prepayment, as the case may be, believed by the Administrative Agent or the Swingline Lender, as the case may be, in good faith to be from an Authorized Officer of such Borrower, prior to receipt of written confirmation. In each such case, such Borrower hereby waives the right to dispute the Administrative Agent's or the Swingline Lender's record of the terms of such telephonic notice of such Borrowing or prepayment of Loans, as the case may be, absent manifest error.
- 2.04 <u>Disbursement of Funds</u>. (a) <u>Disbursement of US Revolving Loans</u>. No later than 2:00 P.M. (New York City time) on the date specified in each Notice of Borrowing with respect to any US Revolving Loans (or (x) in the case of US Swingline Loans, no later than 4:00 P.M. (New York City time) on the date specified pursuant to <u>Section 2.03(c)</u> or (y) in the case of Mandatory US Borrowings, no later than 1:00 P.M. (New York City time) on the date specified in <u>Section 2.01(e)</u>), each Lender will make available its pro rata portion (determined in accordance with <u>Section 2.07</u>) of each such Borrowing requested to be made on such date. All such amounts will be made available in Dollars and in immediately available funds at the Payment Office, and the Administrative Agent will make available to the relevant US Borrower at the Payment Office the aggregate of the amounts so made available by the Lenders (or in the case of US Swingline Loans, the Swingline Lender will make available the full amount thereof).
- (b) <u>Disbursement of UK Revolving Loans</u>. No later than 2:00 P.M. (London time) on the date specified in each Notice of Borrowing with respect to any UK Revolving Loans (or (x) in the case of UK Swingline Loans, no later than 2:00 P.M. (London time) on the date specified pursuant to Section <u>2.03(d)</u> or (y) in the case of Mandatory UK Borrowings, no later than 1:00 P.M. (London time) on the date specified in Section <u>2.01(f)</u>), each Lender will make available its pro rata portion (determined in accordance with <u>Section 2.07</u>) of each such Borrowing requested to be made on such date. Subject to <u>Section 2.16</u>, all such amounts will be made available in either Pounds Sterling or Euro (as requested by the applicable UK Borrower) and in immediately available funds at the Payment Office, and the Administrative Agent will make available to the relevant UK Borrower at the Payment Office the

aggregate of the amounts so made available by the Lenders (or in the case of UK Swingline Loans, the Swingline Lender will make available the full amount thereof).

- (c) Unless the Administrative Agent shall have been notified by any Lender prior to the date of Borrowing that such Lender does not intend to make available to the Administrative Agent such Lender's portion of any Borrowing to be made on such date, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on such date of Borrowing and the Administrative Agent may (but shall not be obligated to), in reliance upon such assumption, make available to the relevant Borrower a corresponding amount. If such corresponding amount is not in fact made available to the Administrative Agent by such Lender, the Administrative Agent shall be entitled to recover such corresponding amount on demand from such Lender. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent shall promptly notify the relevant Borrower, and the relevant Borrower shall immediately pay such corresponding amount to the Administrative Agent. The Administrative Agent also shall be entitled to recover on demand from such Lender or the relevant Borrower, as the case may be, interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to the relevant Borrower until the date such corresponding amount is recovered by the Administrative Agent, at a rate per annum equal to (i) if recovered from such Lender, in the case of US Revolving Loans, the overnight Federal Funds Rate for the first three days and at the interest rate otherwise applicable to such Loans for each day thereafter and, the case of all other Loans, the interest rate applicable to such Loans and (ii) if recovered from the relevant Borrower or Borrowers, the rate of interest applicable to the respective Borrowing, as determined pursuant to Section 2.08. Nothing in this Section 2.04 shall be deemed to relieve any Lender from its obligation to make Loans hereunder or to prejudice a
- (d) Unless the Required Lenders have instructed the Administrative Agent to the contrary, the Administrative Agent on behalf of the Lenders may, but shall not be obligated to, make US Revolving Loans to the Borrower that are maintained as Base Rate Loans under Section 2.01(a) without prior notice of the proposed Borrowing to the Lenders as follows:
 - (i) The amount of each Lender's RL Percentage of US Revolving Loans shall be computed weekly (or more frequently in the Administrative Agent's sole discretion) and shall be adjusted upward or downward on the basis of the amount of outstanding US Revolving Loans as of 5:00 P.M. (New York City time) on the last Business Day of each week, or such other period specified by the Administrative Agent (each such date, a "Settlement Date"). The Lenders shall transfer to the Administrative Agent, or the Administrative Agent shall transfer to the Lenders, such amounts as are necessary so that (after giving effect to all such transfers) the amount of US Revolving Loans made by each Lender shall be equal to such Lender's RL Percentage of the aggregate amount of US Revolving Loans outstanding as of such Settlement Date. If a notice from the Administrative Agent of any such necessary transfer is received by a Lender on or prior to 12:00 Noon (New York City time) on any Business Day, then such Lender shall make transfers described above in immediately available funds no later than 3:00 P.M. (New York City time) on the day such notice was received; and if such notice is received by a Lender after 12:00 Noon (New York City time) on any Business Day, such Lender shall make such transfers no later than 1:00 P.M. (New York City time) on the next succeeding Business Day. The obligation of each of the Lenders to transfer such funds shall be irrevocable and unconditional and without recourse to, or without representation or warranty by, the Administrative Agent. Each of the Administrative Agent and each Lender agrees and the Lenders agree to mark their respective books and records on each Settlement Date to show at all times the dollar amount of their respective RL Percentage of the outstanding US Revolving Loans on such date.

- (ii) To the extent that the settlement described in preceding clause (i) shall not yet have occurred with respect to any particular Settlement Date, upon any repayment of US Revolving Loans by any Borrower prior to such settlement, the Administrative Agent may apply such amounts repaid directly to the amounts that would otherwise be made available by the Administrative Agent pursuant to this Section 2.04(d).
- (iii) Because the Administrative Agent on behalf of the Lenders may be advancing and/or may be repaid US Revolving Loans prior to the time when the Lenders will actually advance and/or be repaid US Revolving Loans, interest with respect to Revolving Loans shall be allocated by the Administrative Agent to each Lender and the Administrative Agent in accordance with the amount of US Revolving Loans actually advanced by and repaid to each Lender and the Administrative Agent and shall accrue from and including the date such US Revolving Loans are so advanced to but excluding the date such US Revolving Loans are either repaid by the Borrower in accordance with the terms of this Agreement or actually settled by the Administrative Agent or the applicable Lender as described in this Section 2.04(d).
- 2.05 Notes. (a) Each Borrower's obligation to pay the principal of, and interest on, the Loans made by each Lender shall be evidenced in the Register maintained by the Administrative Agent pursuant to Section 13.15 and shall, if requested by such Lender, also be evidenced (i) in the case of US Revolving Loans, by a promissory note duly executed and delivered by each US Borrower substantially in the form of Exhibit B-1, with blanks appropriately completed in conformity herewith (each, a "US Revolving Note" and, collectively, the "US Revolving Notes"), (ii) in the case of US Swingline Loans, by a promissory note duly executed and delivered by each US Borrower substantially in the form of Exhibit B-2, with blanks appropriately completed in conformity herewith (each, a "US Swingline Note") and, collectively, the "US Swingline Notes"), (iii) in the case of UK Revolving Loans, by a promissory note duly executed and delivered by each UK Borrower substantially in the form of Exhibit B-3, with blanks appropriately completed in conformity herewith (each, a "UK Revolving Note"), and (iv) in the case of UK Swingline Loans, by a promissory note duly executed and delivered by each UK Borrower substantially in the form of Exhibit B-4, with blanks appropriately completed in conformity herewith (each, a "UK Swingline Note").
- (b) Each Lender will note on its internal records the amount of each Loan made by it and each payment in respect thereof and prior to any transfer of any of its Notes will endorse on the reverse side thereof the outstanding principal amount of Loans evidenced thereby. Failure to make any such notation or any error in such notation shall not affect any Borrower's obligations in respect of such Loans.
- (c) Notwithstanding anything to the contrary contained above in this Section 2.05 or elsewhere in this Agreement, Notes shall only be delivered to Lenders which at any time specifically request the delivery of such Notes. No failure of any Lender to request, obtain, maintain or produce a Note evidencing its Loans to any Borrower shall affect, or in any manner impair, the obligations of any Borrower to pay the Loans (and all related Obligations) incurred by such Borrower which would otherwise be evidenced thereby in accordance with the requirements of this Agreement, and shall not in any way affect the security or guaranties therefor provided pursuant to any Credit Document. Any Lender which does not have a Note evidencing its outstanding Loans shall in no event be required to make the notations otherwise described in preceding clause (b). At any time when any Lender requests the delivery of a Note to evidence any of its Loans, the respective Borrower shall promptly execute and deliver to the respective Lender the requested Note in the appropriate amount or amounts to evidence such Loans.

2.06 Conversions. Each US Borrower shall have the option to convert, on any Business Day, all or a portion equal to at least the Minimum Borrowing Amount of the outstanding principal amount of US Revolving Loans made pursuant to one or more Borrowings of one or more Types of US Revolving Loans into a Borrowing of another Type of US Revolving Loan; provided that, (a) except as otherwise provided in Section 2.10(b), US LIBOR Loans may be converted into Base Rate Loans only on the last day of an Interest Period applicable to the US Revolving Loans being converted and no such partial conversion of LIBOR Loans shall reduce the outstanding principal amount of such US LIBOR Loans made pursuant to a single Borrowing to less than the Minimum Borrowing Amount applicable thereto, (b) unless the Required Lenders otherwise agree, Base Rate Loans may only be converted into US LIBOR Loans if no Default or Event of Default is in existence on the date of the conversion, (c) unless the Administrative Agent has otherwise agreed or has determined that the Syndication Date has occurred (at which time this clause (c) shall no longer be applicable), prior to the 60th day following the Effective Date, conversions of Base Rate Loans into US LIBOR Loans may only be made if any such conversion is effective on the first day of the first or second Interest Period referred to in clause (B) of Section 2.01(a)(ii) and so long as such conversion does not result in a greater number of Borrowings of LIBOR Loans prior to the 60th day after the Initial Borrowing Date than are permitted under Section 2.01(a)(ii), (d) no conversion pursuant to this Section 2.06 shall result in a greater number of Borrowings of US LIBOR Loans than are permitted under Section 2.02 and (e) for the avoidance of doubt, in no event shall any US Revolving Loan be converted into any UK Revolving Loan and in no event shall any UK Revolving Loan be converted into any US Revolving Loan. Each such conversion shall be effected by the respective US Borrower by giving the Administrative Agent at the Notice Office prior to 11:00 A.M. (New York City time) at least (i) in the case of conversions of Base Rate Loans into US LIBOR Loans, three Business Days' prior notice and (ii) in the case of conversions of US LIBOR Loans into Base Rate Loans, one Business Day's prior notice (each, a "Notice of Conversion/Continuation"), in each case in the form of Exhibit A-2, appropriately completed to specify the Revolving Loans to be so converted, the Borrowing or Borrowings pursuant to which such Revolving Loans were incurred and, if to be converted into US LIBOR Loans, the Interest Period to be initially applicable thereto. The Administrative Agent shall give each Lender prompt notice of any such proposed conversion affecting any of its Revolving Loans.

2.07 <u>Pro Rata Borrowings</u>. All Borrowings of Revolving Loans under this Agreement shall be incurred from the Lenders pro rata on the basis of their Revolving Loan Commitments, <u>provided</u> that all Mandatory Borrowings shall be incurred from the Lenders <u>pro rata</u> on the basis of their RL Percentages. It is understood that no Lender shall be responsible for any default by any other Lender of its obligation to make Loans hereunder and that each Lender shall be obligated to make the Loans provided to be made by it hereunder, regardless of the failure of any other Lender to make its Loans hereunder.

2.08 <u>Interest</u>. (a) Each US Borrower jointly and severally agrees to pay interest in respect of the unpaid principal amount of each Base Rate Loan from the date of Borrowing thereof until the earlier of (i) the maturity thereof (whether by acceleration or otherwise) and (ii) the conversion of such Base Rate Loan to a LIBOR Loan pursuant to <u>Section 2.06</u> or <u>2.09</u>, as applicable, at a rate per annum which shall be equal to the sum of the relevant Applicable Margin plus the Base Rate, each as in effect from time to time.

(b) Each US Borrower jointly and severally agrees to pay interest in respect of the unpaid principal amount of each US LIBOR Loan from the date of Borrowing thereof until the earlier of (i) the maturity thereof (whether by acceleration or otherwise) and (ii) the conversion of such LIBOR Loan to a Base Rate Loan pursuant to Section 2.06, 2.09 or 2.10, as applicable, at a rate per annum which shall, during each Interest Period applicable thereto, be equal to the sum of the relevant Applicable Margin as in effect from time to time during such Interest Period plus LIBOR for such Interest Period.

- (c) Each UK Borrower jointly and severally agrees to pay interest in respect of the unpaid principal amount of each UK Revolving Loan from the date of Borrowing thereof until the maturity thereof (whether by acceleration or otherwise), at a rate per annum which shall, during each Interest Period applicable thereto, be equal to the sum of the relevant Applicable Margin as in effect from time to time during such Interest Period plus LIBOR and/or EURIBOR as applicable for such Interest Period, plus the Mandatory Cost.
- (d) (i) Overdue principal and, to the extent permitted by law, overdue interest in respect of each Loan shall, in each case, bear interest at a rate per annum equal to the greater of (x) the rate which is 2% in excess of the rate then borne by such Loans and (y) in the case of any US Revolving Loans, US Swingline Loans and Agent Advances, the rate which is 2% in excess of the rate otherwise applicable to Base Rate Loans from time to time and (ii) all other overdue amounts payable hereunder and under any other Credit Document shall bear interest at a rate per annum equal to the rate which is 2% in excess of the rate applicable to Base Rate Loans from time to time. Interest that accrues under this Section 2.08(c) shall be payable on demand.
- (e) Accrued (and theretofore unpaid) interest shall be payable (i) in respect of each Base Rate Loan, (A) quarterly in arrears on each Quarterly Payment Date, (B) on the date of any repayment or prepayment in full of all outstanding Base Rate Loans, and (C) at maturity (whether by acceleration or otherwise) and, after such maturity, on demand, and (ii) in respect of each LIBOR Loan and EURIBOR Loan, (A) on the last day of each Interest Period applicable thereto and, in the case of an Interest Period in excess of three months, on each date occurring at three month intervals after the first day of such Interest Period, and (B) on the date of any repayment or prepayment (on the amount repaid or prepaid), at maturity (whether by acceleration or otherwise) and, after such maturity, on demand.
- (f) Upon each Interest Determination Date, the Administrative Agent shall determine LIBOR and EURIBOR for each Interest Period applicable to the respective LIBOR Loans and EURIBOR Loans and shall promptly notify US Company and the Lenders thereof. Each such determination shall, absent manifest error, be final and conclusive and binding on all parties hereto.
- 2.09 Interest Periods. At the time any Borrower gives any Notice of Borrowing or Notice of Conversion/Continuation in respect of the making of, or, in the case of any US Revolving Loan, conversion into, any LIBOR Loan or EURIBOR Loan (in the case of the initial Interest Period applicable thereto) or prior to 11:00 A.M. (New York City time) on the third Business Day prior to the expiration of an Interest Period applicable to such LIBOR Loan or EURIBOR Loan (in the case of any subsequent Interest Period), such Borrower shall have the right to elect the interest period (each, an "Interest Period") applicable to such LIBOR Loan or EURIBOR Loan, which Interest Period shall, at the option of the Borrower (but otherwise subject to the provisions of clause (B) of the proviso in Section 2.01(a)(ii)), be a one (or less than one month if permitted by the Administrative Agent), two, three or six or, to the extent approved by each Lender, nine or twelve month period; provided that (in each case):
 - (a) all LIBOR Loans or EURIBOR Loans comprising a Borrowing shall at all times have the same Interest Period;
 - (b) the initial Interest Period for any LIBOR Loan or EURIBOR Loan shall commence on the date of Borrowing of such LIBOR Loan or EURIBOR Loan (including, in the case of any LIBOR Loan, the date of any conversion thereto from a Base Rate Loan) and each Interest Period occurring thereafter in respect of such LIBOR Loan or EURIBOR shall commence on the day on which the next preceding Interest Period applicable thereto expires;

- (c) if any Interest Period for a LIBOR Loan begins on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period, such Interest Period shall end on the last Business Day of such calendar month;
- (d) if any Interest Period for a LIBOR Loan or EURIBOR Loan would otherwise expire on a day which is not a Business Day, such Interest Period shall expire on the next succeeding Business Day; <u>provided</u>, <u>however</u>, that if any Interest Period for a LIBOR Loan or EURIBOR Loan would otherwise expire on a day which is not a Business Day but is a day of the month after which no further Business Day occurs in such month, such Interest Period shall expire on the next preceding Business Day;
- (e) unless the Required Lenders otherwise agree, no Interest Period may be selected at any time when a Default or an Event of Default is then in existence; and
 - (f) no Interest Period in respect of any Borrowing shall be selected which extends beyond the Final Maturity Date.
- If by 11:00 A.M. (New York City time) on the third Business Day prior to the expiration of any Interest Period applicable to a Borrowing of LIBOR Loans or EURIBOR Loans, any Borrower has failed to elect, or is not permitted to elect, a new Interest Period to be applicable to such LIBOR Loans or EURIBOR Loans as provided above, such Borrower, (i) in the case of any US Revolving Loan, shall be deemed to have elected to convert such LIBOR Loans into Base Rate Loans effective as of the expiration date of such current Interest Period and (ii) in the case of any UK Revolving Loan, shall be deemed to have elected to continue such LIBOR Loans or EURIBOR Loans or EURIBOR Loans with an Interest Period of one-month, effective as of the expiration date of such current Interest Period.
- 2.10 <u>Increased Costs, Illegality, etc.</u> (a) In the event that any Lender shall have determined (which determination shall, absent manifest error, be final and conclusive and binding upon all parties hereto but, with respect to clause (i) below, may be made only by the Administrative Agent):
 - (i) on any Interest Determination Date that, by reason of any changes arising after the date of this Agreement affecting the interbank eurodollar market, adequate and fair means do not exist for ascertaining the applicable interest rate as to Loans constituting LIBOR Loans or EURIBOR Loans on the basis provided for in the definition of LIBOR or EURIBOR, as applicable; or
 - (ii) at any time, that such Lender shall incur increased costs or reductions in the amounts received or receivable hereunder or a reduction in the rate of return or on a Lender's (or its Affiliate's) overall capital, in each case, with respect to any LIBOR Loan or EURIBOR Loan because of (A) any change since the Effective Date in any Applicable Law or governmental rule, regulation, order, guideline or request (whether or not having the force of law) or in the interpretation or administration thereof and including the introduction of any new law or governmental rule, regulation, order, guideline or request, such as, but not limited to: (1) a change in the basis or rate of taxation of payment to any Lender of the principal of or interest on the Loans or the Notes or any other amounts payable hereunder (except for (x) taxes with respect to which additional amounts are paid pursuant to Section 5.04 or Section 5.05 or would be payable but for the failure to provide the forms provided in Section 5.04(b) or (y) changes in the rate of tax on, or determined by reference to, the net income or net profits of such Lender pursuant to the laws of the jurisdiction in which it is organized or in which its principal office or applicable lending office is located or any subdivision thereof or therein) or (2) a change in

official reserve requirements, but, in all events, excluding reserves required under Regulation D to the extent included in the computation of LIBOR or EURIBOR and/or (B) other circumstances arising since the Effective Date affecting such Lender, the interbank eurodollar market or the position of such Lender in such market; or

(iii) at any time, that the making or continuance of any LIBOR Loan or EURIBOR Loan has been made (A) unlawful by any law or governmental rule, regulation or order, (B) impossible by compliance by any Lender in good faith with any governmental request (whether or not having force of law) or (C) impracticable as a result of a contingency occurring after the Effective Date which materially and adversely affects the interbank eurodollar market;

then, and in any such event, such Lender (or the Administrative Agent, in the case of clause (i) above) shall promptly give notice (by telephone promptly confirmed in writing) to US Company and, except in the case of clause (i) above, to the Administrative Agent of such determination (which notice the Administrative Agent shall promptly transmit to each of the other Lenders). Thereafter (x) in the case of clause (i) above, US Revolving Loans constituting US LIBOR Loans shall no longer be available until such time as the Administrative Agent notifies US Company and the Lenders that the circumstances giving rise to such notice by the Administrative Agent no longer exist, and any Notice of Borrowing or Notice of Conversion/Continuation given by any Borrower with respect to US LIBOR Loans which have not yet been incurred (including by way of conversion) shall be deemed rescinded by such Borrower, (y) in the case of clause (ii) above, the US Borrowers and the UK Borrowers, jointly and severally, agree to pay to such Lender, as applicable, upon such Lender's written request therefor, such additional amounts (in the form of an increased rate of, or a different method of calculating, interest or otherwise as such Lender in its sole discretion shall determine) as shall be required to compensate such Lender for such increased costs or reductions in amounts received or receivable hereunder (a written notice as to the additional amounts owed to such Lender, showing in reasonable detail the basis for the calculation thereof, submitted to US Company by such Lender shall, absent manifest error, be final and conclusive and binding on all the parties hereto) and (z) in the case of clause (iii) above, the respective Borrower or Borrowers shall take one of the actions specified in Section 2.10(b) as promptly as possible and, in any event, within the time period required by law.

(b) (i) In the case of US Revolving Loans, at any time that any US Revolving Loan constituting a US LIBOR Loan is affected by the circumstances described in Section 2.10(a)(ii), the affected Borrower may, and in the case of a US LIBOR Loan affected by the circumstances described in Section 2.10(a) (iii), the affected Borrower shall, either (1) if the affected US LIBOR Loan is then being made initially or pursuant to a conversion, cancel such Borrowing by giving the Administrative Agent telephonic notice (confirmed in writing) on the same date that such Borrower was notified by the affected Lender or the Administrative Agent pursuant to Section 2.10(a)(ii) or (2) if the affected US LIBOR Loan is then outstanding, upon at least three Business Days' written notice to the Administrative Agent, require the affected Lender to convert such US Revolving Loan constituting a US LIBOR Loan into a Base Rate Loan; provided that, if more than one Lender is affected at any time, then all affected Lenders must be treated the same pursuant to this Section 2.10(b); and (ii) in the case of UK Revolving Loans, at any time that any of the UK Revolving Loans is affected by the circumstances described in Section 2.10(a)(iii) then the relevant Borrower shall repay the applicable Lender's participation in that UK Revolving Loan on the last day of the Interest Period for such UK Revolving Loan or, if earlier, the date specified by the Lender in the notice delivered to the Administrative Agent (being no earlier than the last day of any applicable grace period permitted by law). Further, at any time that any of the UK Revolving Loans is affected by the circumstances described in Section 2.10(a)(i) then the applicable LIBOR or EURIBOR, as applicable, shall be calculated in respect of each Lender as the percentage rate of interest per annum reflecting such Lender's cost of funding its participation in the UK Revolving Loans from whatever source it may reasonably select.

- (c) If any Lender determines that after the Effective Date the introduction of or any change in any Applicable Law or governmental rule, regulation, order, guideline, directive or request (whether or not having the force of law) concerning capital adequacy, or any change in interpretation or administration thereof by the NAIC or any Governmental Authority, central bank or comparable agency, will have the effect of increasing the amount of capital required or expected to be maintained by such Lender or any corporation controlling such Lender based on the existence of such Lender's Revolving Loan Commitment hereunder or its obligations hereunder, then the US Borrowers and the UK Borrowers, jointly and severally, agree to pay to such Lender, as applicable, upon its written demand therefor, such additional amounts as shall be required to compensate such Lender or such other corporation for the increased cost to such Lender or such other corporation or the reduction in the rate of return to such Lender or such other corporation as a result of such increase of capital. In determining such additional amounts, each Lender will act reasonably and in good faith and will use averaging and attribution methods which are reasonable; provided that such Lender's determination of compensation owing under this Section 2.10(c) shall, absent manifest error, be final and conclusive and binding on all the parties hereto. Each Lender, upon determining that any additional amounts will be payable pursuant to this Section 2.10(c), will give prompt written notice thereof to US Company, which notice shall show in reasonable detail the basis for calculation of such additional amounts.
- 2.11 Compensation. Each US Borrower, jointly and severally, and each UK Borrower, jointly and severally, agrees to compensate each Lender, upon its written request (which request shall set forth in reasonable detail the basis for requesting such compensation), for all losses, expenses and liabilities (including, without limitation, any loss, expense or liability incurred by reason of the liquidation or reemployment of deposits or other funds required by such Lender to fund its LIBOR Loans or EURIBOR Loans but excluding loss of anticipated profits) which such Lender may sustain: (a) if for any reason (other than a default by such Lender or the Administrative Agent) a Borrowing of, or conversion from or into, LIBOR Loans or EURIBOR Loans does not occur on a date specified therefor in a Notice of Borrowing or Notice of Conversion/Continuation (whether or not withdrawn by the respective Borrower or Borrowers or deemed withdrawn pursuant to Section 2.10(a)); (b) if any prepayment or repayment (including any prepayment or repayment made pursuant to Section 5.01, Section 5.02 or as a result of an acceleration of the Loans pursuant to Section 11) or conversion of any of its LIBOR Loans or EURIBOR Loans is not made on any date specified in a notice of prepayment given by any Borrower; or (d) as a consequence of (i) any other default by any Borrower to repay LIBOR Loans or EURIBOR Loans when required by the terms of this Agreement or any Note held by such Lender or (ii) any election made pursuant to Section 2.10(b).
- 2.12 <u>Change of Lending Office</u>. Each Lender agrees that on the occurrence of any event giving rise to the operation of <u>Section 2.10(a)(ii)</u> or (<u>iii)</u>, <u>Section 2.10(c)</u>, <u>Section 3.06</u>, <u>Section 5.04</u> or <u>Section 5.05</u> with respect to such Lender, it will, if requested by US Company or UK Company, use reasonable efforts (subject to overall policy considerations of such Lender) to designate another lending office for any Loans or Letters of Credit affected by such event; <u>provided</u> that such designation is made on such terms that such Lender and its lending office suffer no economic, legal or regulatory disadvantage, with the object of avoiding the consequence of the event giving rise to the operation of such Section. Nothing in this <u>Section 2.12</u> shall affect or postpone any of the obligations of any Borrower or the right of any Lender provided in <u>Sections 2.10</u>, <u>3.06</u>, <u>5.04</u> or <u>5.05</u>.
- 2.13 <u>Replacement of Lenders</u>. (a) If any Lender becomes a Defaulting Lender, (b) upon the occurrence of any event giving rise to the operation of Section <u>2.10(a)(ii)</u> or (<u>iii)</u>, <u>Section 2.10(c)</u>, <u>Section 3.06</u>, <u>Section 5.04</u> or <u>Section 5.05</u> with respect to any Lender which results in such Lender charging to any Borrower increased costs in excess of those being generally charged by the

other Lenders or (c) in the case of a refusal by a Lender to consent to a proposed change, waiver, discharge or termination with respect to this Agreement which has been approved by the Required Lenders as (and to the extent) provided in <u>Section 13.12(b)</u>, the US Company or the UK Company, as applicable, shall have the right, in accordance with <u>Section 13.04(b)</u>, if no Default or Event of Default then exists or would exist after giving effect to such replacement, to replace such Lender (the "<u>Replaced Lender</u>") with one or more other Eligible Transferees, none of whom shall constitute a Defaulting Lender at the time of such replacement (collectively, the "<u>Replacement Lender</u>") and each of which shall be reasonably acceptable to the Administrative Agent; <u>provided</u> that:

- (i) at the time of any replacement pursuant to this Section 2.13, the Replacement Lender shall enter into one or more Assignment and Assumption Agreements pursuant to Section 13.04(b) (and with all fees payable pursuant to said Section 13.04(b) to be paid by the Borrowers) pursuant to which the Replacement Lender shall acquire all of the Revolving Loan Commitment and outstanding Revolving Loans of, and all participations in Letters of Credit by, the Replaced Lender and, in connection therewith, shall pay to (i) the Replaced Lender in respect thereof an amount equal to the sum of (A) an amount equal to the principal of, and all accrued interest on, all outstanding Revolving Loans of the respective Replaced Lender, (B) an amount equal to all Unpaid US Drawings and Unpaid UK Drawings of such Replaced Lender, together with all then unpaid interest with respect thereto at such time and (C) an amount equal to all accrued, but theretofore unpaid, Fees owing to the Replaced Lender pursuant to Section 4.01, (ii) each US Issuing Lender and UK Issuing Lender an amount equal to such Replaced Lender's RL Percentage of any Unpaid US Drawing or Unpaid UK Drawing, as applicable) to the extent such amount was not theretofore funded by such Replaced Lender and (iii) the Swingline Lender an amount equal to such Replaced Lender's RL Percentage of any Mandatory Borrowing to the extent such amount was not theretofore funded by such Replaced Lender; and
- (ii) all obligations of the Borrowers then owing to the Replaced Lender (other than those specifically described in clause (i) above in respect of which the assignment purchase price has been, or is concurrently being, paid, but including all amounts, if any, owing under <u>Section 2.11</u>) shall be paid in full to such Replaced Lender concurrently with such replacement.

Upon receipt by the Replaced Lender of all amounts required to be paid to it pursuant to this Section 2.13, the Administrative Agent shall be entitled (but not obligated) and authorized to execute an Assignment and Assumption Agreement on behalf of such Replaced Lender, and any such Assignment and Assumption Agreement so executed by the Administrative Agent and the Replacement Lender shall be effective for purposes of this Section 2.13 and Section 13.04. Upon the execution of the respective Assignment and Assumption Agreement, the payment of amounts referred to in clauses (i) and (ii) above, recordation of the assignment on the Register by the Administrative Agent pursuant to Section 13.15 and, if so requested by the Replacement Lender, delivery to the Replacement Lender of the appropriate Note or Notes executed by the relevant Borrowers, (x) the Replacement Lender shall become a Lender hereunder and the Replaced Lender shall cease to constitute a Lender hereunder, except with respect to indemnification provisions under this Agreement (including, without limitation, Sections 2.10, 2.11, 3.06, 5.04, 5.05, 12.06, 13.01 and 13.06), which shall survive as to such Replaced Lender and (y) the RL Percentages of the Lenders shall be automatically adjusted at such time to give effect to such replacement.

2.14 <u>US Company as Agent for US Borrowers and UK Company as Agent for UK Borrower</u>. (a) Each US Borrower hereby irrevocably appoints the US Company as its agent and attorney-in-fact for all purposes under this Agreement and each other Credit Document, which appointment shall remain in full force and effect unless and until the Administrative

Agent shall have received prior written notice signed by the respective appointing Borrower that such appointment has been revoked. Each US Borrower hereby irrevocably appoints and authorizes the US Company (i) to provide the Administrative Agent with all notices with respect to Loans and Letters of Credit obtained for the benefit of any US Borrower and all other notices and instructions under this Agreement or any other Credit Document and (ii) to take such action as the US Company deems appropriate on its behalf to obtain Loans and Letters of Credit and to exercise such other powers as are reasonably incidental thereto to carry out the purposes of this Agreement and the other Credit Documents. It is understood that the handling of the Credit Account and the Collateral of the US Borrowers in a combined fashion, as more fully set forth herein, is done solely as an accommodation to the US Borrowers in order to utilize the collective borrowing powers of the US Borrowers in the most efficient and economical manner and at their request, and that the Lenders shall not incur liability to any US Borrower as a result hereof. Each US Borrower expects to derive benefit, directly or indirectly, from the handling of the Credit Account and the Collateral in a combined fashion since the successful operation of each US Borrower is dependent on the continued successful performance of the consolidated group. To induce the Lenders to do so, and in consideration thereof, each US Borrower hereby jointly and severally agrees to indemnify each Lender and hold each Lender harmless against any and all liability, expense, loss or claim of damage or injury, made against any Lender by any US Borrower or by any third party whosoever, arising from or incurred by reason of (a) the handling of the Credit Account and Collateral of the US Borrowers as herein provided, (b) the Lenders' relying on any instructions of US Company, or (c) any other action taken by the Lenders hereunder or under the other Credit Documents, except that the US Borrowers will have no liability to any Lender, Administrative Agent or the Collateral Agent with respect to any liability that has been determined by a court of competent jurisdiction in a final and non-appealable decision to have resulted solely from the gross negligence or willful misconduct of such Lender, the Administrative Agent or the Collateral Agent, as the case may be.

(b) Each UK Borrower hereby irrevocably appoints the UK Company as its agent and attorney-in-fact for all purposes under this Agreement and each other Credit Document, which appointment shall remain in full force and effect unless and until the Administrative Agent shall have received prior written notice signed by the respective appointing Borrower that such appointment has been revoked. Each UK Borrower hereby irrevocably appoints and authorizes the UK Company (i) to provide the Administrative Agent with all notices with respect to Loans and Letters of Credit obtained for the benefit of any UK Borrower and all other notices and instructions under this Agreement or any other Credit Document and (ii) to take such action as the UK Company deems appropriate on its behalf to obtain Loans and Letters of Credit and to exercise such other powers as are reasonably incidental thereto to carry out the purposes of this Agreement and the other Credit Documents. It is understood that the handling of the Credit Account and the Collateral of the UK Borrowers in a combined fashion, as more fully set forth herein, is done solely as an accommodation to the UK Borrowers in order to utilize the collective borrowing powers of the UK Borrowers in the most efficient and economical manner and at their request, and that the Lenders shall not incur liability to any UK Borrower as a result hereof. Each UK Borrower expects to derive benefit, directly or indirectly, from the handling of the Credit Account and the Collateral in a combined fashion since the successful operation of each UK Borrower is dependent on the continued successful performance of the consolidated group. To induce the Lenders to do so, and in consideration thereof, each UK Borrower hereby jointly and severally agrees to indemnify each Lender and hold each Lender harmless against any and all liability, expense, loss or claim of damage or injury, made against any Lender by any UK Borrower or by any third party whosoever, arising from or incurred by reason of (a) the handling of the Credit Account and Collateral of the UK Borrowers as herein provided, (b) the Lenders' relying on any instructions of UK Company, or (c) any other action taken by the Lenders hereunder or under the other Credit Documents, except that the UK Borrowers will have no liability to any Lender, Administrative Agent or the Collateral Agent with respect to any liability that has been determined by a court of competent jurisdiction in a final and non-appealable decision to have

resulted solely from the gross negligence or willful misconduct of such Lender, the Administrative Agent or the Collateral Agent, as the case may be.

- 2.15 Incremental Commitments. (a) So long as the Incremental Commitment Request Requirements are satisfied at the time of the delivery of the request referred to below, the Borrowers shall have the right, in coordination with the Administrative Agent as to all of the matters set forth below in this Section 2.15, but without requiring the consent of any of the Lenders, to request at any time and from time to time after the earlier to occur of (x) the date that is 60 days after the Initial Borrowing Date and (y) the Syndication Date, and prior to the Final Maturity Date, that one or more Lenders (and/or one or more other Persons which are Eligible Transferees and which will become Lenders as provided below) provide Incremental Commitments and, subject to the applicable terms and conditions contained in this Agreement, make Revolving Loans pursuant thereto, it being understood and agreed, however, that (i) no Lender shall be obligated to provide an Incremental Commitment as a result of any such request by the Borrowers, and until such time, if any, as such Lender has agreed in its sole discretion to provide an Incremental Commitment and executed and delivered to the Administrative Agent an Incremental Commitment Agreement in respect thereof as provided in clause (b) of this Section 2.15, such Lender shall not be obligated to fund any Revolving Loans in excess of its Revolving Loan Commitment as in effect prior to giving effect to such Incremental Commitment provided pursuant to this Section 2.15, (ii) any Lender (including any Eligible Transferee who will become a Lender) may so provide an Incremental Commitment without the consent of any other Lender, (iii) each Eligible Transferee who will become a Lender shall be required to be reasonably satisfactory to the Administrative Agent, (iv) each provision of Incremental Commitments on a given date pursuant to this Section 2.15 shall be in a minimum aggregate amount (for all Lenders (including any Eligible Transferee who will become a Lender)) of at least \$5,000,000 and in integral multiples of \$1,000,000 in excess thereof, (v) the aggregate amount of all Incremental Commitments provided pursuant to this Section 2.15, shall not exceed the Maximum Incremental Commitment Amount and (vi) all Revolving Loans (and all interest, fees and other amounts payable thereon), made pursuant to an Incremental Commitment shall be entitled to the benefits of the guarantees and security provided under the Credit Documents to the other Obligations on a pari passu basis.
- (b) At the time of the provision of Incremental Commitments pursuant to this Section 2.15, US Company, each US Borrower, UK Company, each UK Borrower, the Administrative Agent and each such Lender or other Eligible Transferee which agrees to provide an Incremental Commitment (each, an "Incremental Lender") shall execute and deliver to the Administrative Agent an Incremental Commitment Agreement, with the effectiveness of such Incremental Lender's Incremental Commitment to occur on the date set forth in such Incremental Commitment Agreement, which date in any event shall be no earlier than the date on which (w) all fees required to be paid in connection therewith at the time of such effectiveness shall have been paid (including, without limitation, any agreed upon up-front or arrangement fees owing to the Administrative Agent (or any affiliate thereof)), (x) all Incremental Commitment Requirements are satisfied, (y) all other conditions set forth in this Section 2.15 shall have been satisfied, and (z) all other conditions precedent that may be set forth in such Incremental Commitment Agreement shall have been satisfied. The Administrative Agent shall promptly notify each Lender as to the effectiveness of each Incremental Commitment Agreement, and at such time, (i) the Total Revolving Loan Commitment under, and for all purposes of, this Agreement shall be increased by the aggregate amount of such Incremental Commitments, (ii) Schedule 1.01(a) shall be deemed modified to reflect the revised Revolving Loan Commitments of the affected Lenders and (iii) to the extent requested by any Incremental Lender, Revolving Notes will be issued, at the expense of the Borrowers, to such Incremental Lender in conformity with the requirements of Section 2.05.
- (c) At the time of any provision of Incremental Commitments pursuant to this <u>Section 2.15</u>, the Borrowers shall, in coordination with the Administrative Agent, repay outstanding

Revolving Loans of certain of the Lenders, and incur additional Revolving Loans from certain other Lenders (including the Incremental Lenders) (even though as a result thereof such new Loans (to the extent required to be maintained as LIBOR Loans or EURIBOR Loans) may have a shorter Interest Period than the then outstanding Borrowings of the respective such Revolving Loans), in each case to the extent necessary so that all of the Lenders participate in each outstanding Borrowing of Revolving Loans <u>pro rata</u> on the basis of their respective Revolving Loan Commitments (after giving effect to any increase in the Total Revolving Loan Commitment pursuant to this <u>Section 2.15</u>) and with the Borrowers being jointly and severally obligated to pay to the respective Lenders any costs of the type referred to in <u>Section 2.11</u> and such amounts, as reasonably determined by the respective Lenders, to compensate them for funding the various Revolving Loans during an existing Interest Period (rather than at the beginning of the respective Interest Period, based upon rates then applicable thereto) in connection with any such repayment and/or incurrence. All determinations by any Lender pursuant to the preceding sentence shall, absent manifest error, be final and conclusive and binding on all parties hereto.

(d) The terms and provisions of the Revolving Loans made pursuant to the Incremental Commitments shall be identical to the Revolving Loans; provided that the yield applicable to the Revolving Loans made pursuant to the Incremental Commitments (after giving effect to all upfront or similar fees or original issue discount payable with respect to such Revolving Loans) shall not be greater than the applicable yield payable pursuant to the terms of this Agreement as amended through the date of such calculation with respect to Revolving Loans (including any upfront fees or original issue discount payable to the initial Lenders hereunder) unless the Applicable Margin with respect to the Revolving Loans is increased so as to cause the then applicable yield under this Agreement on the Revolving Loans to equal the yield then applicable to the Revolving Loans made pursuant to the Incremental Commitment (after giving effect to all upfront or similar fees or original issue discount payable with respect to the Revolving Loans) made pursuant to the Incremental Commitment.

2.16 UK Revolving Loans; Intra-Lender Issues. (a) Pounds Sterling Participations. Notwithstanding anything to the contrary contained herein, all Pounds Sterling Denominated Revolving Loans shall be made solely by the Lenders with Pounds Sterling Funding Capacity (it being understood and agreed that each Lender at its option may make any Pounds Sterling Denominated Revolving Loan under this Agreement by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that any exercise of such option shall not affect the obligation of the Borrowers to repay any such Loan in accordance with the terms of this Agreement). Each Lender that does not have Pounds Sterling Funding Capacity (a "Participating Pounds Lender") shall irrevocably and unconditionally purchase and acquire and shall be deemed to irrevocably and unconditionally purchase and acquire from DBNY, and DBNY shall sell and be deemed to sell to each such Participating Pounds Lender, without recourse or any representation or warranty whatsoever, an undivided interest and participation (a "Pounds Sterling Participation") in each Pounds Sterling Denominated Revolving Loan funded by DBNY in an amount equal to such Participating Pounds Lender's RL Percentage of the borrowing that includes such Pounds Sterling Denominated Revolving Loan. Such purchase and sale of a Pounds Sterling Participation shall be deemed to occur automatically upon the making of a Pounds Sterling Denominated Revolving Loan by DBNY, without any further notice to any Participating Pounds Lender. The purchase price payable by each Participating Pounds Lender to DBNY for each Pounds Sterling Participation purchased by it from DBNY shall be equal to 100% of the principal amount of such Pounds Sterling Participation (i.e., the product of (i) the amount of the borrowing that includes the relevant Pounds Sterling Denominated Revolving Loan and (ii) such Participating Pounds Lender's RL Percentage), and such purchase price shall be payable by each Participating Pounds Lender to DBNY in accordance with the settlement procedure set forth in Section 2.16(c) below. DBNY and Agent shall record on their books the amount of the Pounds Sterling Denominated Revolving Loans made by DBNY and each Participating Pounds Lender's Pounds Sterling Participation and Funded Pounds Sterling Participation therein, all payments in respect thereof and

interest accrued thereon and all payments made by and to each Participating Pounds Lender pursuant to this Section 2.16.

- (b) Euro Participations. Notwithstanding anything to the contrary contained herein, all Euro Denominated Revolving Loans shall be made solely by the Lenders with Euro Funding Capacity (it being understood and agreed that each Lender at its option may make any Euro Denominated Revolving Loan under this Agreement by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that any exercise of such option shall not affect the obligation of the Borrowers to repay any such Loan in accordance with the terms of this Agreement). Each Lender that does not have Euro Funding Capacity (a "Participating Euro Lender") shall irrevocably and unconditionally purchase and acquire and shall be deemed to irrevocably and unconditionally purchase and acquire from DBNY, and DBNY shall sell and be deemed to sell to each such Participating Euro Lender, without recourse or any representation or warranty whatsoever, an undivided interest and participation (a "Euro Participation") in each Euro Denominated Revolving Loan funded by DBNY in an amount equal to such Participating Euro Lender's RL Percentage of the borrowing that includes such Euro Denominated Revolving Loan. Such purchase and sale of a Euro Participation shall be deemed to occur automatically upon the making of a Euro Denominated Revolving Loan by DBNY, without any further notice to any Participating Euro Lender. The purchase price payable by each Participating Euro Lender to DBNY for each Euro Participation purchased by it from DBNY shall be equal to 100% of the principal amount of such Euro Participation (i.e., the product of (i) the amount of the borrowing that includes the relevant Euro Denominated Revolving Loan and (ii) such Participating Euro Lender's RL Percentage), and such purchase price shall be payable by each Participating Euro Lender to DBNY in accordance with the settlement procedure set forth in Section 2.16(d) below. DBNY and Agent shall record on their books the amount of the Euro Denominated Revolving Loans made by DBNY and each Participating Euro Lender's Euro Participation and Funded Euro Participation therein, all payments in respect thereof and interest accrued thereon and all payments made by and to each Participating Euro Lender pursuant to this Section 2.16.
- (c) <u>Settlement Procedures for Pounds Sterling Denominated Loan Participations</u>. Each Participating Pounds Lender's Pounds Sterling Participation in the Pounds Sterling Denominated Loans (other than Agent Advances) shall be in an amount equal to its RL Percentage of all such Pounds Sterling Denominated Loans. However, in order to facilitate the administration of the Pounds Sterling Denominated Loans made by DBNY and the Pounds Sterling Participations, settlement among DBNY and the Participating Pounds Lenders with regard to the Participating Pounds Lenders' Pounds Sterling Participations shall take place in accordance with the following:
 - (i) DBNY and the Participating Pounds Lenders shall settle (a "Pounds Sterling Participation Settlement") by payments in respect of the Pounds Sterling Participations as follows: So long as any Pounds Sterling Denominated Loans are outstanding, Pounds Sterling Participation Settlements shall be effected through Administrative Agent on such Business Days as Administrative Agent shall specify by a notice by telecopy, telephone or similar form of notice to each Participating Pounds Lender requesting such Pounds Sterling Participation Settlement (each such date on which a Pounds Sterling Participation Settlement Date on the delivered no later than 2:00 p.m. (New York time) at least one Business Day prior to the requested Pounds Sterling Participation Settlement Date; provided, that Administrative Agent shall have the option but not the obligation to specify a Pounds Sterling Participation Settlement Date and, in any event, shall not specify a Pounds Sterling Participation Settlement Date prior to the occurrence of an Event of Default; provided, further, that if (x) such Event of Default is waived in writing in accordance with the terms hereof, (y) no Obligations have yet been declared due and payable under Section 11 and (z) Administrative Agent has actual knowledge of such cure or waiver, all prior to

Administrative Agent's giving notice to the Participating Pounds Lenders of the first Pounds Sterling Participation Settlement Date under this Agreement, then Administrative Agent shall not give notice to the Participating Pounds Lenders of a Pounds Sterling Participation Settlement Date based upon such cured or waived Event of Default. If on any Pounds Sterling Participation Settlement Date the total principal amount of the Pounds Sterling Denominated Loans made or deemed made by DBNY during the period ending on (but excluding) such Pounds Sterling Settlement Date and commencing on (and including) the immediately preceding Pounds Sterling Participation Settlement Date (or the Closing Date in the case of the period ending on the first Pounds Sterling Participation Settlement Date) (each such period herein called a "Pounds Sterling Participation Settlement Period") is greater than the principal amount of Pounds Sterling Denominated Loans repaid during such Pounds Sterling Participation Settlement Period to DBNY, each Participating Pounds Lender shall pay to DBNY (through Administrative Agent), no later than 11:00 a.m. (New York time) on such Pounds Sterling Participation Settlement Date, an amount equal to such Participating Pounds Lender's ratable share of the amount of such excess. If in any Pounds Sterling Participation Settlement Period the outstanding principal amount of the Pounds Sterling Denominated Loans repaid to DBNY shall pay to each Participating Pounds Lender (through Administrative Agent) on such Pounds Sterling Participation Settlement Date an amount equal to such Participating Pounds Lender's ratable share of such excess. Pounds Sterling Participation Settlements in respect of Pounds Sterling Denominated Loans shall be made in Pounds Sterling (or the Equivalent Amount thereof) on the Pounds Sterling Participation Settlement Date for such Pounds Sterling Denominated Loans.

- (ii) If any Participating Pounds Lender fails to pay to DBNY on any Pounds Sterling Participation Settlement Date the full amount required to be paid by such Participating Pounds Lender to DBNY on such Pounds Sterling Participation Settlement Date in respect of such Participating Pounds Lender's Pounds Sterling Participation (such Participating Pounds Lender's "Pounds Sterling Participation Settlement Amount") with DBNY, DBNY shall be entitled to recover such unpaid amount from such Participating Pounds Lender, together with interest thereon (in the same respective currency or currencies as the relevant Pounds Sterling Denominated Loans) at LIBOR for UK LIBOR Loans with an Interest Period of one-week plus 2% with respect to Loans denominated in Pounds Sterling. Without limiting DBNY's rights to recover from any Participating Pounds Lender any unpaid Pounds Sterling Participation Settlement Amount payable by such Participating Pounds Lender to DBNY, Administrative Agent shall also be entitled to withhold from amounts otherwise payable to such Participating Pounds Lender an amount equal to such Participating Pounds Sterling Participation Settlement Amount owing to DBNY and apply such withheld amount to the payment of any unpaid Pounds Sterling Participation Settlement Amount owing by such Participating Pounds Lender to DBNY.
- (iii) Following the first Pounds Sterling Participation Settlement Date, Administrative Agent shall effect a Pounds Sterling Participation Settlement on each subsequent Pounds Sterling Revolving Loan Settlement Date or within 1 Business Day thereafter.
- (d) <u>Settlement Procedures for Euro Denominated Revolving Loan Participations</u>. Each Participating Pounds Lender's Euro Participation in the Euro Denominated Revolving Loans (other than Agent Advances) shall be in an amount equal to its RL Percentage of all such Euro Denominated Revolving Loans. However, in order to facilitate the administration of the Euro Denominated Revolving Loans made by DBNY and the Euro Participations, settlement among DBNY and the Participating

Pounds Lenders with regard to the Participating Pounds Lenders' Euro Participations shall take place in accordance with the following:

(i) DBNY and the Participating Pounds Lenders shall settle (a "Euro Participation Settlement") by payments in respect of the Euro Participations as follows: So long as any Euro Denominated Revolving Loans are outstanding, Euro Participation Settlements shall be effected through Agent on such Business Days as Administrative Agent shall specify by a notice by telecopy, telephone or similar form of notice to each Participating Pounds Lender requesting such Euro Participation Settlement (each such date on which a Euro Participation Settlement occurs herein called a "Euro Participation Settlement Date"), such notice to be delivered no later than 2:00 p.m. (New York time) at least one Business Day prior to the requested Euro Participation Settlement Date; provided, that Administrative Agent shall have the option but not the obligation to specify a Euro Participation Settlement Date and, in any event, shall not specify a Euro Participation Settlement Date prior to the occurrence of an Event of Default; provided, further, that if (x) such Event of Default is waived in writing in accordance with the terms hereof, (y) no Obligations have yet been declared due and payable under Section 11 and (z) Administrative Agent has actual knowledge of such cure or waiver, all prior to Administrative Agent's giving notice to the Participating Pounds Lenders of the first Euro Participation Settlement Date under this Agreement, then Administrative Agent shall not give notice to the Participating Pounds Lenders of a Euro Participation Settlement Date based upon such cured or waived Event of Default. If on any Euro Participation Settlement Date the total principal amount of the Euro Denominated Revolving Loans made or deemed made by DBNY during the period ending on (but excluding) such Euro Settlement Date and commencing on (and including) the immediately preceding Euro Participation Settlement Date (or the Closing Date in the case of the period ending on the first Euro Participation Settlement Date) (each such period herein called a "Euro Participation Settlement Period") is greater than the principal amount of Euro Denominated Revolving Loans repaid during such Euro Participation Settlement Period to DBNY, each Participating Pounds Lender shall pay to DBNY (through Agent), no later than 11:00 a.m. (New York time) on such Euro Participation Settlement Date, an amount equal to such Participating Pounds Lender's ratable share of the amount of such excess. If in any Euro Participation Settlement Period the outstanding principal amount of the Euro Denominated Revolving Loans repaid to DBNY in such period exceeds the total principal amount of the Euro Denominated Revolving Loans made or deemed made by DBNY during such period, DBNY shall pay to each Participating Pounds Lender (through Agent) on such Euro Participation Settlement Date an amount equal to such Participating Pounds Lender's ratable share of such excess. Euro Participation Settlements in respect of Euro Denominated Revolving Loans shall be made in Euro (or the Equivalent Amount thereof) on the Euro Participation Settlement Date for such Euro Denominated Revolving Loans.

(ii) If any Participating Pounds Lender fails to pay to DBNY on any Euro Participation Settlement Date the full amount required to be paid by such Participating Pounds Lender to DBNY on such Euro Participation Settlement Date in respect of such Participating Pounds Lender's Euro Participation (such Participating Pounds Lender's "Euro Participation Settlement Amount") with DBNY, DBNY shall be entitled to recover such unpaid amount from such Participating Pounds Lender, together with interest thereon (in the same respective currency or currencies as the relevant Euro Denominated Revolving Loans) at EURIBOR for EURIBOR Loans plus 2% with respect to Loans denominated in Euros. Without limiting DBNY's rights to recover from any Participating Pounds Lender any unpaid Euro Participation Settlement Amount payable by such Participating Pounds Lender to DBNY, Administrative Agent shall also be entitled to withhold from amounts otherwise payable to such Participating Pounds Lender an amount equal to such Participating Pounds Lender's unpaid Euro Participation Settlement

Amount owing to DBNY and apply such withheld amount to the payment of any unpaid Euro Participation Settlement Amount owing by such Participating Pounds Lender to DBNY.

- (iii) Following the first Euro Participation Settlement Date, Administrative Agent shall effect a Euro Participation Settlement on each subsequent Euro Revolving Loan Settlement Date or within 1 Business Day thereafter.
- (e) <u>Obligations Irrevocable</u>. The obligations of each Participating Pounds Lender or each Participating Euro Lender, as applicable, to purchase from DBNY a participation in each Pounds Sterling Denominated Revolving Loan or Euro Denominated Revolving Loan made by DBNY and to make payments to DBNY with respect to such participation, in each case as provided herein, shall be irrevocable and not subject to any qualification or exception whatsoever, including any of the following circumstances:
 - (i) any lack of validity or enforceability of this Agreement or any of the other Credit Documents or of any Revolving Loans, against UK Company, any UK Borrower or any Guarantor;
 - (ii) the existence of any claim, setoff, defense or other right which UK Company, any UK Borrower or any UK Guarantor may have at any time in respect of any Revolving Loans;
 - (iii) any application or misapplication of any proceeds of any Revolving Loans;
 - (iv) the surrender or impairment of any security for any Revolving Loans;
 - (v) the occurrence of any Default or Event of Default;
 - (vi) the commencement or pendency of any events specified in Section 11.08 hereof, in respect of a Credit Party or any other Person; or
 - (vii) the failure to satisfy the applicable conditions precedent set forth in <u>Sections 6</u> or <u>7</u> hereof.
- (f) <u>Recovery or Avoidance of Payments</u>. In the event any payment by or on behalf of UK Company, any UK Borrower or any other Credit Party received by Administrative Agent with respect to any Pounds Sterling Denominated Revolving Loan or Euro Denominated Revolving Loan made by DBNY is thereafter set aside, avoided or recovered from Administrative Agent in connection with any Insolvency Proceeding or due to any mistake of law or fact, each Participating Pounds Lender and each Participating Euro Lender shall, upon written demand by Administrative Agent, pay to DBNY (through Administrative Agent) such Participating Pounds Lender's RL Percentage or such Participating Euro Lender's RL Percentage, as the case may be, of such amount set aside, avoided or recovered, together with interest at the rate and in the currency required to be paid by DBNY or Administrative Agent upon the amount required to be repaid by it.
- (g) <u>Indemnification by Participating Pounds Lenders</u>. Each Participating Pounds Lender agrees to indemnify DBNY (to the extent not reimbursed by UK Company or UK Borrower and without limiting the obligations of UK Company and UK Borrower hereunder or under any other Credit Document) ratably for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including attorneys' fees) or disbursements of any kind and nature whatsoever that may be imposed on, incurred by or asserted against DBNY in any way relating to or arising out of any Pounds Sterling Denominated Revolving Loans or any participations by DBNY in any UK Letters of

Credit denominated in Pounds Sterling or related L/C Supportable Obligations or any action taken or omitted by DBNY in connection therewith; provided that no Participating Pounds Lender shall be liable for any of the foregoing to the extent it arises from the gross negligence or willful misconduct of DBNY. Without limiting the foregoing, each Participating Pounds Lender agrees to reimburse DBNY promptly upon demand for such Participating Pounds Lender's ratable share of any costs or expenses payable by the Borrowers to DBNY in respect of the Pounds Sterling Denominated Revolving Loans to the extent that DBNY is not promptly reimbursed for such costs and expenses by the Borrowers. The agreement contained in this Section 2.16(g) shall survive payment in full of all UK Revolving Loans.

- (h) <u>Indemnification by Participating Euro Lenders</u>. Each Participating Euro Lender agrees to indemnify DBNY (to the extent not reimbursed by UK Company or UK Borrower and without limiting the obligations of UK Company and UK Borrower hereunder or under any other Credit Document) ratably for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including attorneys' fees) or disbursements of any kind and nature whatsoever that may be imposed on, incurred by or asserted against DBNY in any way relating to or arising out of any Euro Denominated Revolving Loans or any participations by DBNY in any UK Letters of Credit denominated in Euros or related L/C Supportable Obligations or any action taken or omitted by DBNY in connection therewith; provided that no Participating Euro Lender shall be liable for any of the foregoing to the extent it arises from the gross negligence or willful misconduct of DBNY. Without limiting the foregoing, each Participating Euro Lender agrees to reimburse DBNY promptly upon demand for such Participating Euro Lender's ratable share of any costs or expenses payable by the Borrowers to DBNY in respect of the Euro Denominated Revolving Loans to the extent that DBNY is not promptly reimbursed for such costs and expenses by the Borrowers. The agreement contained in this <u>Section 2.16(h)</u> shall survive payment in full of all UK Revolving Loans.
- (i) <u>Pounds Sterling Denominated Revolving Loan Participation Fee</u>. In consideration for each Participating Pounds Lender's participation in the Pounds Sterling Denominated Revolving Loans made by DBNY, DBNY agrees to pay to Administrative Agent for the account of each Participating Pounds Lender, as and when DBNY receives payment of interest on its Pounds Sterling Denominated Revolving Loans, a fee (the "<u>Pounds Sterling Participation Fee</u>") at a rate per annum equal to the Applicable Margin on such UK Revolving Loans minus 0.25% on the unfunded Pounds Sterling Participation of such Participating Pounds Lender in such Pounds Sterling Denominated Revolving Loans of DBNY. The Pounds Sterling Participation Fee in respect of any unfunded Pounds Sterling Participation in a Pounds Sterling Denominated Revolving Loan shall be payable to Administrative Agent in Pounds Sterling When interest on such Pounds Sterling Denominated Revolving Loan is received by DBNY. If DBNY does not receive payment in full of such interest, the Pounds Sterling Participation Fee in respect of the unfunded Pounds Sterling Participation in such Pounds Sterling Denominated Revolving Loans shall be reduced proportionately. Any amounts payable under this <u>Section 2.16(i)</u> by Administrative Agent to the Participating Pounds Lenders shall be paid in Pounds Sterling (or the US Dollar equivalent thereof as determined by the Administrative Agent in its sole discretion).
- (j) <u>Euro Denominated Revolving Loan Participation Fee</u>. In consideration for each Participating Pounds Lender's participation in the Euro Denominated Revolving Loans made by DBNY, DBNY agrees to pay to Administrative Agent for the account of each Participating Pounds Lender, as and when DBNY receives payment of interest on its Euro Denominated Revolving Loans, a fee (the "<u>Euro Participation Fee</u>") at a rate per annum equal to the Applicable Margin on such UK Revolving Loans minus 0.25% on the unfunded Euro Participation of such Participating Pounds Lender in such Euro Denominated Revolving Loans of DBNY. The Euro Participation Fee in respect of any unfunded Euro Participation in a Euro Denominated Revolving Loan shall be payable to Administrative Agent in Euros when interest on such Euro Denominated Revolving Loan is received by DBNY. If DBNY does not receive payment in full of such interest, the Euro Participation Fee in respect of the unfunded Euro

Participation in such Euro Denominated Revolving Loans shall be reduced proportionately. Any amounts payable under this <u>Section 2.16(j)</u> by Administrative Agent to the Participating Pounds Lenders shall be paid in Euro (or the US Dollar equivalent thereof as determined by the Administrative Agent in its sole discretion).

2.17 Equivalent Amount. For purposes of this Agreement, the Equivalent Amount of each Loan not denominated in Dollars shall be calculated on the date when any such Loan is made, such Letter of Credit is issued, and at such other times (no less frequently than weekly) as designated by the Administrative Agent. Such Equivalent Amount shall remain in effect until the same is recalculated by the Administrative Agent as provided above and notice of such recalculation is delivered to the US Company, it being understood that until such notice of such recalculation is delivered, the Dollar Equivalent shall be that Dollar Equivalent as last reported to US Company by the Administrative Agent. The Administrative Agent shall promptly notify US Company and the Lenders of each such determination of the Equivalent Amount of each such Loan.

SECTION 3. Letters of Credit.

- 3.01 <u>US Letters of Credit</u>. (a) Subject to and upon the terms and conditions set forth herein, any US Borrower may request that a US Issuing Lender issue, at any time and from time to time on and after the Initial Borrowing Date and prior to the 30th day prior to the Final Maturity Date, for the joint and several account of the US Borrowers and for the benefit of (x) any holder (or any trustee, agent or other similar representative for any such holders) of US L/C Supportable Obligations, an irrevocable standby letter of credit, in a form customarily used by such US Issuing Lender or in such other form as is reasonably acceptable to such US Issuing Lender, and (y) sellers of goods to the US Company or any of its Subsidiaries, an irrevocable trade letter of credit, in a form customarily used by such US Issuing Lender or in such other form as has been approved by such US Issuing Lender (each such letter of credit, a "<u>US Letter of Credit</u>") (although without limiting the joint and several nature of the US Borrowers' obligations in respect of the US Letters of Credit, any particular US Letter of Credit may name only one or more US Borrowers as the account party therein). All US Letters of Credit shall be issued on a sight basis only.
- (b) Subject to and upon the terms and conditions set forth herein, each US Issuing Lender agrees that it will, at any time and from time to time on and after the Initial Borrowing Date and prior to the 30th day prior to the Final Maturity Date, following its receipt of the respective US Letter of Credit Request, issue for the joint and several account of the US Borrowers, one or more US Letters of Credit as are permitted to remain outstanding hereunder without giving rise to a Default or an Event of Default; provided that no US Issuing Lender shall be under any obligation to issue any US Letter of Credit of the types described above if at the time of such issuance:
 - (i) any order, judgment or decree of any Governmental Authority or arbitrator shall purport by its terms to enjoin or restrain such US Issuing Lender from issuing such US Letter of Credit or any requirement of law applicable to such US Issuing Lender or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such US Issuing Lender shall prohibit, or request that such US Issuing Lender refrain from, the issuance of letters of credit generally or such US Letter of Credit in particular or shall impose upon such Issuing Lender with respect to such US Letter of Credit any restriction or reserve or capital requirement (for which such US Issuing Lender is not otherwise compensated hereunder) not in effect with respect to such US Issuing Lender on the date hereof, or any unreimbursed loss, cost or expense which was not applicable or in effect with respect to such US Issuing Lender as of the date hereof and which such US Issuing Lender reasonably and in good faith deems material to it; or

- (ii) such US Issuing Lender shall have received from such US Borrower, any other Credit Party or the Required Lenders prior to the issuance of such US Letter of Credit, notice of the type described in the second sentence of Section 3.05(b).
- (c) <u>Schedule 3.01(c)</u> contains a description of letters of credit that were issued by a Lender for the account of US Company prior to the Initial Borrowing Date and which remain outstanding on the Initial Borrowing Date (and setting forth, with respect to each such letter of credit, (i) the name of the issuing lender, (ii) the letter of credit number, (iii) the name of the account party, (iv) the stated amount (which shall be in Dollars), (v) the name of the beneficiary, (vi) the expiry date and (vii) whether such letter of credit constitutes a standby letter of credit or a trade letter of credit). The US Company hereby acknowledges and agrees that each such letter of credit, including any extension or renewal thereof in accordance with the terms thereof and hereof, an "Existing US Letter of Credit") shall constitute a "US Letter of Credit" for all purposes of this Agreement and, notwithstanding anything to the contrary stated in any such Existing US Letter of Credit (including, without limitation, the account party named therein), shall be deemed issued on the Initial Borrowing Date for the account of the US Company.
- 3.02 <u>UK Letters of Credit</u>. (a) Subject to and upon the terms and conditions set forth herein, any UK Borrower may request that a UK Issuing Lender issue, at any time and from time to time on and after the Initial Borrowing Date and prior to the 30th day prior to the Final Maturity Date, for the joint and several account of the UK Borrowers and for the benefit of (x) any holder (or any trustee, agent or other similar representative for any such holders) of UK L/C Supportable Obligations, an irrevocable standby letter of credit, in a form customarily used by such UK Issuing Lender or in such other form as is reasonably acceptable to such UK Issuing Lender, and (y) sellers of goods to the UK Company or any of its Subsidiaries, an irrevocable trade letter of credit, in a form customarily used by such UK Issuing Lender or in such other form as has been approved by such UK Issuing Lender (each such letter of credit, a "UK Letter of Credit") (although without limiting the joint and several nature of the UK Borrowers' obligations in respect of the UK Letters of Credit, any particular UK Letter of Credit may name only one or more UK Borrowers as the account party therein). All UK Letters of Credit shall be issued on a sight basis only.
- (b) Subject to and upon the terms and conditions set forth herein, each UK Issuing Lender agrees that it will, at any time and from time to time on and after the Initial Borrowing Date and prior to the 30th day prior to the Final Maturity Date, following its receipt of the respective UK Letter of Credit Request, issue for the joint and several account of the UK Borrowers, one or more UK Letters of Credit as are permitted to remain outstanding hereunder without giving rise to a Default or an Event of Default; <u>provided</u> that no UK Issuing Lender shall be under any obligation to issue any UK Letter of Credit of the types described above if at the time of such issuance:
 - (i) any order, judgment or decree of any Governmental Authority or arbitrator shall purport by its terms to enjoin or restrain such UK Issuing Lender from issuing such UK Letter of Credit or any requirement of law applicable to such UK Issuing Lender or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such UK Issuing Lender shall prohibit, or request that such UK Issuing Lender refrain from, the issuance of letters of credit generally or such UK Letter of Credit in particular or shall impose upon such UK Issuing Lender with respect to such UK Letter of Credit any restriction or reserve or capital requirement (for which such UK Issuing Lender is not otherwise compensated hereunder) not in effect with respect to such UK Issuing Lender on the date hereof, or any unreimbursed loss, cost or expense which was not applicable or in effect with respect to such UK Issuing Lender as of the date hereof and which such UK Issuing Lender reasonably and in good faith deems material to it; or

(ii) such UK Issuing Lender shall have received from such UK Borrower, any other Credit Party or the Required Lenders prior to the issuance of such UK Letter of Credit, notice of the type described in the second sentence of Section 3.06(b).

3.03 Maximum US Letter of Credit Outstandings; Final Maturities. Notwithstanding anything to the contrary contained in this Agreement, (a) no US Letter of Credit shall be issued (or required to be issued) if the Stated Amount of such US Letter of Credit, when added to the US Letter of Credit Outstandings (exclusive of Unpaid US Drawings which are repaid on the date of, and prior to the issuance of, the respective US Letter of Credit) at such time would exceed \$50,000,000 (the "Maximum US Letter of Credit Amount"), (b) no US Letter of Credit shall be issued (or required to be issued) at any time when the Aggregate US Exposure exceeds (or would after giving effect to such issuance exceed) either (i) the US Maximum Amount at such time or (ii) the US Borrowing Base at such time (based on the Borrowing Base Certificate last delivered), or when the Aggregate Exposure exceeds (or would after giving effect to such issuance exceed) either (x) the Total Revolving Loan Commitment or (y) the Aggregate Borrowing Base at such time (based on the Borrowing Base Certificate last delivered), (c) each US Letter of Credit shall be denominated in Dollars, (d) each standby US Letter of Credit shall by its terms terminate on or before the earlier of (i) the date which occurs 12 months after the date of the issuance thereof (although any such standby US Letter of Credit shall be extendible for successive periods of up to 12 months, but, in each case, not beyond the 5th Business Day prior to the Final Maturity Date) and (ii) 5 Business Days prior to the Final Maturity Date and (e) each trade US Letter of Credit shall by its terms terminate on or before the earlier of (i) the date which occurs 180 days after the date of issuance thereof and (ii) 5 Business Days prior to the Final Maturity Date.

3.04 Maximum UK Letter of Credit Outstandings; Final Maturities. Notwithstanding anything to the contrary contained in this Agreement, (a) no UK Letter of Credit shall be issued (or required to be issued) if the Stated Amount of such UK Letter of Credit, when added to the UK Letter of Credit Outstandings (exclusive of Unpaid UK Drawings which are repaid on the date of, and prior to the issuance of, the respective UK Letter of Credit) at such time would exceed \$20,000,000 or the Equivalent Amount thereof (the "Maximum UK Letter of Credit Amount"), (b) no UK Letter of Credit shall be issued (or required to be issued) at any time when the Aggregate UK Exposure exceeds (or would after giving effect to such issuance exceed) either (i) the UK Maximum Amount at such time or (ii) the UK Borrowing Base at such time (based on the Borrowing Base Certificate last delivered), (c) each UK Letter of Credit shall be denominated in Pounds Sterling or Euros, (d) each standby UK Letter of Credit shall by its terms terminate on or before the earlier of (i) the date which occurs 12 months after the date of the issuance thereof (although any such standby UK Letter of Credit shall be extendible for successive periods of up to 12 months, but, in each case, not beyond the 5th Business Day prior to the Final Maturity Date) and (ii) 5 Business Days prior to the Final Maturity Date.

3.05 <u>US Letter of Credit Requests; Minimum Stated Amount</u>. (a) Whenever any US Borrower desires that a US Letter of Credit be issued for its account, such US Borrower shall give the Administrative Agent and the respective US Issuing Lender at least five Business Days' (or such shorter period as is acceptable to such US Issuing Lender) written notice thereof (including by way of facsimile). Each notice shall be in the form of Exhibit F, appropriately completed (each, a "<u>US Letter of Credit Request</u>").

(b) The making of each US Letter of Credit Request shall be deemed to be a representation and warranty by the respective US Borrower to the Lenders that such US Letter of Credit may be issued in accordance with, and will not violate the requirements of, Section 3.03. Unless the respective US Issuing Lender has received notice from any US Borrower, any other US Credit Party or

the Required Lenders before it issues a US Letter of Credit that one or more of the conditions specified in Section 6 or 7 are not then satisfied, or that the issuance of such US Letter of Credit would violate Section 3.03, then such US Issuing Lender shall, subject to the terms and conditions of this Agreement, issue the requested US Letter of Credit for the account of the respective US Borrower in accordance with such US Issuing Lender's usual and customary practices. Upon the issuance of or modification or amendment to any standby US Letter of Credit, each US Issuing Lender shall promptly notify the respective US Borrower and the Administrative Agent, in writing of such issuance, modification or amendment and such notice shall be accompanied by a copy of such US Letter of Credit or the respective modification or amendment thereto, as the case may be. Promptly after receipt of such notice the Administrative Agent shall notify the US Participants, in writing, of such issuance, modification or amendment. On the first Business Day of each week, each US Issuing Lender shall furnish the Administrative Agent with a written (including via facsimile) report of the daily aggregate outstandings of US Letters of Credit issued by such US Issuing Lender for the immediately preceding week. Notwithstanding anything to the contrary contained in this Agreement, in the event that a Lender Default exists with respect to a Lender, no US Issuing Lender shall be required to issue any US Letter of Credit unless such US Issuing Lender has entered into arrangements satisfactory to it and the US Company to eliminate such US Issuing Lender's risk with respect to the participation in US Letters of Credit by the Defaulting Lender or Lenders, including by cash collateralizing such Defaulting Lender's or Lenders' RL Percentage of the US Letter of Credit Outstandings.

(c) The initial Stated Amount of each US Letter of Credit shall not be less than \$25,000 or such lesser amount as is acceptable to the respective US Issuing Lender.

3.06 <u>UK Letter of Credit Requests; Minimum Stated Amount</u>. (a) Whenever any UK Borrower desires that a UK Letter of Credit be issued for its account, such UK Borrower shall give the Administrative Agent and the respective UK Issuing Lender at least five Business Days' (or such shorter period as is acceptable to such UK Issuing Lender) written notice thereof (including by way of facsimile). Each notice shall be in the form of Exhibit G, appropriately completed (each, a "<u>UK Letter of Credit Request</u>").

(b) The making of each UK Letter of Credit Request shall be deemed to be a representation and warranty by the respective UK Borrower to the Lenders that such UK Letter of Credit may be issued in accordance with, and will not violate the requirements of, Section 3.04. Unless the respective UK Issuing Lender has received notice from any UK Borrower, any other UK Credit Party or the Required Lenders before it issues a UK Letter of Credit that one or more of the conditions specified in Section 6 or 7 are not then satisfied, or that the issuance of such UK Letter of Credit would violate Section 3.04, then such UK Issuing Lender shall, subject to the terms and conditions of this Agreement, issue the requested UK Letter of Credit for the account of the respective UK Borrower in accordance with such UK Issuing Lender's usual and customary practices. Upon the issuance of or modification or amendment to any standby UK Letter of Credit, each UK Issuing Lender shall promptly notify the respective UK Borrower and the Administrative Agent, in writing of such issuance, modification or amendment and such notice shall be accompanied by a copy of such UK Letter of Credit or the respective modification or amendment thereto, as the case may be. Promptly after receipt of such notice the Administrative Agent shall notify the UK Participants, in writing, of such issuance, modification or amendment. On the first Business Day of each week, each UK Issuing Lender shall furnish the Administrative Agent with a written (including via facsimile) report of the daily aggregate outstandings of UK Letters of Credit issued by such UK Issuing Lender for the immediately preceding week.

Notwithstanding anything to the contrary contained in this Agreement, in the event that a Lender Default exists with respect to a Lender, no UK Issuing Lender shall be required to issue any UK Letter of Credit unless such UK Issuing Lender has entered into arrangements satisfactory to it and the UK Company to eliminate such UK Issuing Lender's risk with respect to

Defaulting Lender or Lenders, including by cash collateralizing such Defaulting Lender's or Lenders' RL Percentage of the UK Letter of Credit Outstandings.

- (c) The initial Stated Amount of each UK Letter of Credit shall not be less than \$25,000, £25,000 or \$25,000 or such lesser amount as is acceptable to the respective UK Issuing Lender.
- 3.07 <u>US Letter of Credit Participations.</u> (a) Immediately upon the issuance by an US Issuing Lender of any US Letter of Credit, such US Issuing Lender shall be deemed to have sold and transferred to each Lender, and each such Lender (in its capacity under this <u>Section 3.07</u>, a "<u>US Participant</u>") shall be deemed irrevocably and unconditionally to have purchased and received from such US Issuing Lender, without recourse or warranty, an undivided interest and participation, to the extent of such US Participant's RL Percentage, in such US Letter of Credit, each drawing or payment made thereunder and the obligations of the US Borrowers under this Agreement with respect thereto, and any security therefor or guaranty pertaining thereto. Upon any change in the Revolving Loan Commitments or RL Percentages of the Lenders pursuant to <u>Section 2.13</u>, <u>2.15</u> or <u>13.04(b)</u>, it is hereby agreed that, with respect to all outstanding US Letters of Credit and Unpaid US Drawings relating thereto, there shall be an automatic adjustment to the participations pursuant to this <u>Section 3.07</u> to reflect the new RL Percentages of the assignor and assignee Lender, as the case may be.
- (b) In determining whether to pay under any US Letter of Credit, no US Issuing Lender shall have any obligation relative to the other Lenders other than to confirm that any documents required to be delivered under such US Letter of Credit appear to have been delivered and that they appear to substantially comply on their face with the requirements of such US Letter of Credit. Any action taken or omitted to be taken by a US Issuing Lender under or in connection with any US Letter of Credit issued by it shall not create for such US Issuing Lender any resulting liability to any US Borrower, any other US Credit Party, any Lender or any other Person unless such action is taken or omitted to be taken with gross negligence or willful misconduct on the part of such US Issuing Lender (as determined by a court of competent jurisdiction in a final and non-appealable decision).
- (c) In the event that a US Issuing Lender makes any payment under any US Letter of Credit issued by it and the US Borrowers shall not have reimbursed such amount in full to such US Issuing Lender pursuant to Section 3.09(a), such US Issuing Lender shall promptly notify the Administrative Agent, which shall promptly notify each US Participant of such failure, and each US Participant shall promptly and unconditionally pay to such US Issuing Lender the amount of such US Participant's RL Percentage of such unreimbursed payment in Dollars in immediately available funds. If the Administrative Agent so notifies, prior to 12:00 Noon (New York City time) on any Business Day, any US Participant required to fund a payment under a US Letter of Credit, such US Participant shall make available to the respective US Issuing Lender in Dollars such US Participant's Percentage of the amount of such payment on such Business Day in immediately available funds. If and to the extent such US Participant shall not have so made its RL Percentage of the amount of such payment available to the respective US Issuing Lender, such US Participant agrees to pay to such US Issuing Lender, forthwith on demand such amount, together with interest thereon, for each day from such date until the date such amount is paid to such US Issuing Lender at the overnight Federal Funds Rate for the first three days and at the interest rate applicable to Loans that are maintained as Base Rate Loans for each day thereafter. The failure of any US Participant to make available to a US Issuing Lender its Percentage of any payment under any US Letter of Credit issued by such US Issuing Lender shall not relieve any other US Participant of its obligation hereunder to make available to such US Issuing Lender its RL Percentage of any payment under any US Letter of Credit on the date required, as specified above, but no US Participant shall be responsible for the failure of any other US Participant to make available to such US Issuing Lender such other US Participant's R

- (d) Whenever a US Issuing Lender receives a payment of a reimbursement obligation as to which it has received any payments from the US Participants pursuant to clause (c) above, such US Issuing Lender shall pay to each such US Participant which has paid its RL Percentage thereof, in Dollars and in same day funds, an amount equal to such US Participant's share (based upon the proportionate aggregate amount originally funded by such US Participant to the aggregate amount funded by all US Participants) of the principal amount of such reimbursement obligation and interest thereon accruing after the purchase of the respective participations.
- (e) Upon the request of any US Participant, each US Issuing Lender shall furnish to such US Participant copies of any standby US Letter of Credit issued by it and such other documentation as may reasonably be requested by such US Participant.
- (f) The obligations of the US Participants to make payments to each US Issuing Lender with respect to US Letters of Credit shall be irrevocable and not subject to any qualification or exception whatsoever and shall be made in accordance with the terms and conditions of this Agreement under all circumstances, including, without limitation, any of the following circumstances:
 - (i) any lack of validity or enforceability of this Agreement or any of the other Credit Documents;
 - (ii) the existence of any claim, setoff, defense or other right which US Company or any of its Subsidiaries may have at any time against a beneficiary named in a US Letter of Credit, any transferee of any US Letter of Credit (or any Person for whom any such transferee may be acting), the Administrative Agent, any US Participant, or any other Person, whether in connection with this Agreement, any US Letter of Credit, the transactions contemplated herein or any unrelated transactions (including any underlying transaction between US Company or any Subsidiary of US Company and the beneficiary named in any such US Letter of Credit);
 - (iii) any draft, certificate or any other document presented under any US Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect;
 - (iv) the surrender or impairment of any security for the performance or observance of any of the terms of any of the Credit Documents; or
 - (v) the occurrence of any Default or Event of Default.
- 3.08 <u>UK Letter of Credit Participations</u>. (a) Immediately upon the issuance by an UK Issuing Lender of any UK Letter of Credit, such UK Issuing Lender shall be deemed to have sold and transferred to each Lender, and each such Lender (in its capacity under this <u>Section 3.08</u>, a "<u>UK Participant</u>") shall be deemed irrevocably and unconditionally to have purchased and received from such UK Issuing Lender, without recourse or warranty, an undivided interest and participation, to the extent of such UK Participant's RL Percentage, in such UK Letter of Credit, each drawing or payment made thereunder and the obligations of the UK Borrowers under this Agreement with respect thereto, and any security therefor or guaranty pertaining thereto. Upon any change in the Revolving Loan Commitments or RL Percentages of the Lenders pursuant to <u>Section 2.13</u>, <u>2.15</u> or <u>13.04(b)</u>, it is hereby agreed that, with respect to all outstanding UK Letters of Credit and Unpaid UK Drawings relating thereto, there shall be an automatic adjustment to the participations pursuant to this <u>Section 3.08</u> to reflect the new RL Percentages of the assignor and assignee Lender, as the case may be.

- (b) In determining whether to pay under any UK Letter of Credit, no UK Issuing Lender shall have any obligation relative to the other Lenders other than to confirm that any documents required to be delivered under such UK Letter of Credit appear to have been delivered and that they appear to substantially comply on their face with the requirements of such UK Letter of Credit. Any action taken or omitted to be taken by a UK Issuing Lender under or in connection with any UK Letter of Credit issued by it shall not create for such UK Issuing Lender any resulting liability to any UK Borrower, any other UK Credit Party, any Lender or any other Person unless such action is taken or omitted to be taken with gross negligence or willful misconduct on the part of such UK Issuing Lender (as determined by a court of competent jurisdiction in a final and non-appealable decision).
- (c) In the event that a UK Issuing Lender makes any payment under any UK Letter of Credit issued by it and the UK Borrowers shall not have reimbursed such amount in full to such UK Issuing Lender pursuant to Section 3.10(a), such UK Issuing Lender shall promptly notify the Administrative Agent, which shall promptly notify each UK Participant of such failure, and each UK Participant shall promptly and unconditionally pay to such UK Issuing Lender the amount of such UK Participant's RL Percentage of such unreimbursed payment in Pounds Sterling or Euros, as applicable, in immediately available funds. If the Administrative Agent so notifies, prior to 12:00 Noon (London time) on any Business Day, any UK Participant required to fund a payment under a UK Letter of Credit, such UK Participant shall make available to the respective UK Issuing Lender in Pounds Sterling or Euros, as applicable, such UK Participant's Percentage of the amount of such payment on such Business Day in immediately available funds. If and to the extent such UK Participant shall not have so made its RL Percentage of the amount of such payment available to the respective UK Issuing Lender, such UK Participant agrees to pay to such UK Issuing Lender, forthwith on demand such amount, together with interest thereon, for each day from such date until the date such amount is paid to such UK Issuing Lender interest rate applicable to Loans that are maintained as UK LIBOR Loans or EURIBOR Loans, as applicable with an Interest Period of one-week for each day thereafter. The failure of any UK Participant to make available to a UK Issuing Lender its Percentage of any payment under any UK Letter of Credit issued by such UK Issuing Lender shall not relieve any other UK Participant of its obligation hereunder to make available to such UK Issuing Lender its RL Percentage of any payment under any UK Letter of Credit on the date required, as specified above, but no UK Participant shall be responsible for the failure of any other UK Participant to make available t
- (d) Whenever a UK Issuing Lender receives a payment of a reimbursement obligation as to which it has received any payments from the UK Participants pursuant to clause (c) above, such UK Issuing Lender shall pay to each such UK Participant which has paid its RL Percentage thereof, in Pounds Sterling or Euros, as applicable and in same day funds, an amount equal to such UK Participant's share (based upon the proportionate aggregate amount originally funded by such UK Participant to the aggregate amount funded by all UK Participants) of the principal amount of such reimbursement obligation and interest thereon accruing after the purchase of the respective participations.
- (e) Upon the request of any UK Participant, each UK Issuing Lender shall furnish to such UK Participant copies of any standby UK Letter of Credit issued by it and such other documentation as may reasonably be requested by such UK Participant.
- (f) The obligations of the UK Participants to make payments to each UK Issuing Lender with respect to UK Letters of Credit shall be irrevocable and not subject to any qualification or exception whatsoever and shall be made in accordance with the terms and conditions of this Agreement under all circumstances, including, without limitation, any of the following circumstances:

- (i) any lack of validity or enforceability of this Agreement or any of the other Credit Documents;
- (ii) the existence of any claim, setoff, defense or other right which US Company or any of its Subsidiaries may have at any time against a beneficiary named in a UK Letter of Credit, any transferee of any UK Letter of Credit (or any Person for whom any such transferee may be acting), the Administrative Agent, any UK Participant, or any other Person, whether in connection with this Agreement, any UK Letter of Credit, the transactions contemplated herein or any unrelated transactions (including any underlying transaction between US Company or any Subsidiary of US Company and the beneficiary named in any such UK Letter of Credit);
- (iii) any draft, certificate or any other document presented under any UK Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect;
 - (iv) the surrender or impairment of any security for the performance or observance of any of the terms of any of the Credit Documents; or
 - (v) the occurrence of any Default or Event of Default.
- 3.09 Agreement to Repay US Letter of Credit Drawings. (a) Each US Borrower hereby jointly and severally agrees to reimburse each US Issuing Lender, by making payment to the Administrative Agent in Dollars in immediately available funds at the Payment Office, for any payment or disbursement made by such US Issuing Lender under any US Letter of Credit issued by it (each such amount, so paid until reimbursed by the respective US Borrower, an "Unpaid US Drawing"), not later than one Business Day following receipt by the respective US Borrower of notice of such payment or disbursement (provided that no such notice shall be required to be given if a Default or an Event of Default under Section 11.08 shall have occurred and be continuing, in which case the Unpaid US Drawing shall be due and payable immediately without presentment, demand, protest or notice of any kind (all of which are hereby waived by the US Borrowers)), with interest on the amount so paid or disbursed by such US Issuing Lender, to the extent not reimbursed prior to 12:00 Noon (New York City time) on the date of such payment or disbursement (or to the extent such Unpaid US Drawing is repaid with a US Borrowing of US Revolving Loans constituting Base Rate Loans pursuant to (and to the extent permitted by) clause (ii) of the proviso below), from and including the date paid or disbursed to but excluding the date such US Issuing Lender was reimbursed by the respective US Borrower therefor at a rate per annum equal to the Base Rate as in effect from time to time plus the Applicable Margin as in effect from time to time for Loans that are maintained as Base Rate Loans; provided, however, to the extent such amounts are not reimbursed prior to 12:00 Noon (New York City time) on the third Business Day following the receipt by the respective US Borrower of notice of such payment or disbursement or following the occurrence of a Default or an Event of Default under Section 11.08, interest shall thereafter accrue on the amounts so paid or disbursed by such US Issuing Lender (and until reimbursed by the US Borrowers) at a rate per annum equal to the Base Rate as in effect from time to time plus the Applicable Margin for Loans that are maintained as Base Rate Loans as in effect from time to time plus 2%, with such interest to be payable on demand. Each US Issuing Lender shall give the respective US Borrower prompt written notice of each US Drawing under any US Letter of Credit issued by it; provided that the failure to give any such notice shall in no way affect, impair or diminish the US Borrowers' obligations hereunder.
- (b) The joint and several obligations of the US Borrowers under this <u>Section 3.09</u> to reimburse each US Issuing Lender with respect to drafts, demands and other presentations for payment under US Letters of Credit issued by it (each, a "<u>US Drawing</u>") (including, in each case, interest thereon)

shall be absolute and unconditional under any and all circumstances and irrespective of any setoff, counterclaim or defense to payment which US Company, any US Borrower or any Subsidiary of any US Borrower may have or have had against any Lender (including in its capacity as a US Issuing Lender or as a US Participant), including, without limitation, any defense based upon the failure of any drawing under a US Letter of Credit to conform to the terms of the US Letter of Credit or any nonapplication or misapplication by the beneficiary of the proceeds of such US Drawing; provided, however, that no US Borrower shall be obligated to reimburse any US Issuing Lender for any wrongful payment made by such US Issuing Lender under a US Letter of Credit issued by it as a result of acts or omissions constituting willful misconduct or gross negligence on the part of such US Issuing Lender (as determined by a court of competent jurisdiction in a final and non-appealable decision).

3.10 Agreement to Repay UK Letter of Credit Drawings. (a) Each UK Borrower hereby jointly and severally agrees to reimburse each UK Issuing Lender, by making payment to the Administrative Agent in Pounds Sterling or Euros, as applicable in immediately available funds at the Payment Office, for any payment or disbursement made by such UK Issuing Lender under any UK Letter of Credit issued by it (each such amount, so paid until reimbursed by the respective UK Borrower, an "Unpaid UK Drawing"), not later than one Business Day following receipt by the respective UK Borrower of notice of such payment or disbursement (provided that no such notice shall be required to be given if a Default or an Event of Default under Section 11.08 shall have occurred and be continuing, in which case the Unpaid UK Drawing shall be due and payable immediately without presentment, demand, protest or notice of any kind (all of which are hereby waived by the UK Borrowers)), with interest on the amount so paid or disbursed by such UK Issuing Lender, to the extent not reimbursed prior to 12:00 Noon (London time) on the date of such payment or disbursement (or to the extent such Unpaid UK Drawing is repaid with a UK Borrowing of UK Revolving Loans pursuant to (and to the extent permitted by) clause (ii) of the proviso below), from and including the date paid or disbursed to but excluding the date such UK Issuing Lender was reimbursed by the respective UK Borrower therefor at a rate per annum equal to LIBOR or EURIBOR, as applicable, for Loans with a one-week Interest Period as in effect from time to time plus the Applicable Margin as in effect from time to time for UK Revolving Loans; provided, however, to the extent such amounts are not reimbursed prior to 12:00 Noon (London time) on the third Business Day following the receipt by the respective UK Borrower of notice of such payment or disbursement or following the occurrence of a Default or an Event of Default under Section 11.08, interest shall thereafter accrue on the amounts so paid or disbursed by such UK Issuing Lender (and until reimbursed by the UK Borrowers) at a rate per annum equal to LIBOR or EURIBOR, as applicable, for Loans with a one-week Interest Period as in effect from time to time plus the Applicable Margin for Loans that are maintained as UK Revolving Loans as in effect from time to time plus 2%, with such interest to be payable on demand. Each UK Issuing Lender shall give the respective UK Borrower prompt written notice of each UK Drawing under any UK Letter of Credit issued by it; provided that the failure to give any such notice shall in no way affect, impair or diminish the UK Borrowers' obligations hereunder.

(b) The joint and several obligations of the UK Borrowers under this Section 3.10 to reimburse each UK Issuing Lender with respect to drafts, demands and other presentations for payment under UK Letters of Credit issued by it (each, a "UK Drawing") (including, in each case, interest thereon) shall be absolute and unconditional under any and all circumstances and irrespective of any setoff, counterclaim or defense to payment which US Company, any UK Borrower or any Subsidiary of US Company may have or have had against any Lender (including in its capacity as a UK Issuing Lender or as a UK Participant), including, without limitation, any defense based upon the failure of any drawing under a UK Letter of Credit to conform to the terms of the UK Letter of Credit or any nonapplication or misapplication by the beneficiary of the proceeds of such UK Drawing; provided, however, that no UK Borrower shall be obligated to reimburse any UK Issuing Lender for any wrongful payment made by such UK Issuing Lender under a UK Letter of Credit issued by it as a result of acts or omissions constituting

willful misconduct or gross negligence on the part of such UK Issuing Lender (as determined by a court of competent jurisdiction in a final and non-appealable decision).

3.11 Increased Costs – US Letters of Credit. If at any time after the Effective Date, the introduction of or any change in any Applicable Law, rule, regulation, order, guideline or request or in the interpretation or administration thereof by the NAIC or any Governmental Authority charged with the interpretation or administration thereof, or compliance by any US Issuing Lender or any US Participant with any request or directive by the NAIC or by any such Governmental Authority (whether or not having the force of law), shall either (a) impose, modify or make applicable any reserve, deposit, capital adequacy or similar requirement against letters of credit issued by any US Issuing Lender or participated in by any US Participant, or (b) impose on any US Issuing Lender or any US Participant any other conditions relating, directly or indirectly, to this Agreement or any US Letter of Credit; and the result of any of the foregoing is to increase the cost to any US Issuing Lender or any US Participant of issuing, maintaining or participating in any US Letter of Credit, or reduce the amount of any sum received or receivable by any US Issuing Lender or any US Participant hereunder or reduce the rate of return on its capital with respect to US Letters of Credit (except for changes in the rate of tax on, or determined by reference to, the net income or net profits of such US Issuing Lender or such US Participant pursuant to the laws of the jurisdiction in which it is organized or in which its principal office or applicable lending office is located or any subdivision thereof or therein), then, upon the delivery of the certificate referred to below to the US Company by any US Issuing Lender or any US Participant (a copy of which certificate shall be sent by such US Issuing Lender or such US Participant to the Administrative Agent), the US Borrowers jointly and severally agree to pay to such US Issuing Lender or such US Participant such additional amount or amounts as will compensate such US Issuing Lender or such US Participant for such increased cost or reduction in the amount receivable or reduction on the rate of return on its capital. Any US Issuing Lender or any US Participant, upon determining that any additional amounts will be payable to it pursuant to this Section 3.11, will give prompt written notice thereof to the US Company, which notice shall include a certificate submitted to the US Company by such US Issuing Lender or such US Participant (a copy of which certificate shall be sent by such US Issuing Lender or such US Participant to the Administrative Agent), setting forth in reasonable detail the basis for the calculation of such additional amount or amounts necessary to compensate such US Issuing Lender or such US Participant. The certificate required to be delivered pursuant to this Section 3.11 shall, absent manifest error, be final and conclusive and binding on the US Borrowers.

3.12 Increased Costs — UK Letters of Credit. If at any time after the Effective Date, the introduction of or any change in any Applicable Law, rule, regulation, order, guideline or request or in the interpretation or administration thereof by the NAIC or any Governmental Authority charged with the interpretation or administration thereof, or compliance by any UK Issuing Lender or any UK Participant with any request or directive by the NAIC or by any such Governmental Authority (whether or not having the force of law), shall either (a) impose, modify or make applicable any reserve, deposit, capital adequacy or similar requirement against letters of credit issued by any UK Issuing Lender or participated in by any UK Participant, or (b) impose on any UK Issuing Lender or any UK Participant or any UK Letter of Credit; and the result of any of the foregoing is to increase the cost to any UK Issuing Lender or any UK Participant of issuing, maintaining or participating in any UK Letter of Credit, or reduce the amount of any sum received or receivable by any UK Issuing Lender or any UK Participant hereunder or reduce the rate of return on its capital with respect to UK Letters of Credit (except for changes in the rate of tax on, or determined by reference to, the net income or net profits of such UK Issuing Lender or such UK Participant pursuant to the laws of the jurisdiction in which it is organized or in which its principal office or applicable lending office is located or any subdivision thereof or therein), then, upon the delivery of the certificate referred to below to the US Company or UK Company by any UK Issuing Lender or any UK Participant (a copy of which certificate shall be sent by such UK Issuing Lender or such UK Participant to

the Administrative Agent), the UK Borrowers jointly and severally agree to pay to such UK Issuing Lender or such UK Participant such additional amount or amounts as will compensate such UK Issuing Lender or such UK Participant for such increased cost or reduction in the amount receivable or reduction on the rate of return on its capital. Any UK Issuing Lender or any UK Participant, upon determining that any additional amounts will be payable to it pursuant to this Section 3.12, will give prompt written notice thereof to the US Company, which notice shall include a certificate submitted to the US Company by such UK Issuing Lender or such UK Participant (a copy of which certificate shall be sent by such UK Issuing Lender or such UK Participant to the Administrative Agent), setting forth in reasonable detail the basis for the calculation of such additional amount or amounts necessary to compensate such UK Issuing Lender or such UK Participant. The certificate required to be delivered pursuant to this Section 3.12 shall, absent manifest error, be final and conclusive and binding on the UK Borrowers.

SECTION 4. Commitment Commission; Fees; Reductions of Commitment.

- 4.01 Fees. (a) The US Borrowers jointly and severally agree to pay to the Administrative Agent for distribution to each Lender a commitment commission (the "Commitment Commission") for the period from and including the Effective Date to and including the Final Maturity Date (or such earlier date on which the Total Revolving Loan Commitment has been terminated) computed at a rate per annum equal to the Applicable Commitment Commission Percentage of the Unutilized Revolving Loan Commitment of such Lender as in effect from time to time. Accrued Commitment Commission shall be due and payable quarterly in arrears on each Quarterly Payment Date and on the date upon which the Total Revolving Loan Commitment is terminated.
- (b) The US Borrowers jointly and severally agree to pay to the Administrative Agent for distribution to each Lender (based on each such Lender's respective RL Percentage) a fee in respect of each US Letter of Credit (the "US Letter of Credit Fee") for the period from and including the date of issuance of such US Letter of Credit to and including the date of termination or expiration of such US Letter of Credit, computed at a rate per annum equal to the Applicable Margin as in effect from time to time during such period with respect to US Revolving Loans that are maintained as LIBOR Loans on the daily Stated Amount of each such US Letter of Credit. The Borrowers jointly and severally agree to pay to the Administrative Agent for distribution to each Lender (based on each such Lender's respective RL Percentage) a fee in respect of each UK Letter of Credit (the "UK Letter of Credit Fee") for the period from and including the date of issuance of such UK Letter of Credit to and including the date of termination or expiration of such UK Letter of Credit, computed at a rate per annum equal to the Applicable Margin as in effect from time to time during such period with respect to LIBOR Loans or EURIBOR Loans on the daily Stated Amount of each such UK Letter of Credit. The UK Letter of Credit Fee shall be payable in Pounds Sterling or Euros, as applicable, or at the request of the Administrative Agent, Dollars, with the applicable exchange rate calculated in a manner acceptable to the Administrative Agent. Accrued Letter of Credit Fees shall be due and payable quarterly in arrears on the first day of each fiscal quarter hereafter and on the first day on or after the termination of the Total Revolving Loan Commitment upon which no Letters of Credit remain outstanding.
- (c) The US Borrowers jointly and severally agree to pay to each US Issuing Lender, for its own account, a facing fee in respect of each US Letter of Credit issued by such Issuing Lender and the Borrowers jointly and severally agree to pay to each UK Issuing Lender, for its own account, a facing fee in respect of each UK Letter of Credit issued by such Issuing Lender (in each case, the "Facing Fee") for the period from and including the date of issuance of such Letter of Credit to and including the date of termination or expiration of such Letter of Credit, equal to the greater of (i) \$500.00 or the Equivalent Amount thereof and (ii) 0.125% per annum on the daily Stated Amount of such Letter of Credit. Accrued Facing Fees shall be due and payable monthly in arrears on the first day of each month hereafter and upon

the first day on or after the termination of the Total Revolving Loan Commitment, upon which no Letters of Credit remain outstanding.

- (d) The US Borrowers jointly and severally agree to pay to each US Issuing Lender, for its own account, upon each payment under, issuance of, or amendment to, any US Letter of Credit issued by it, such amount as shall at the time of such event be the administrative charge and the reasonable expenses which such US Issuing Lender is generally imposing in connection with such occurrence with respect to letters of credit. The Borrowers jointly and severally agree to pay to each UK Issuing Lender, for its own account, upon each payment under, issuance of, or amendment to, any UK Letter of Credit issued by it, such amount as shall at the time of such event be the administrative charge and the reasonable expenses which such UK Issuing Lender is generally imposing in connection with such occurrence with respect to letters of credit.
- (e) The US Borrowers jointly and severally agree to pay to the Administrative Agent such fees as may have been, or are hereafter, agreed to in writing from time to time by US Company or any of its Domestic Subsidiaries and the Administrative Agent. The Borrowers jointly and severally agree to pay to the Administrative Agent such fees as may have been, or are hereafter, agreed to in writing from time to time by UK Company or any of the Foreign Subsidiaries and the Administrative Agent.
- 4.02 <u>Voluntary Termination of Unutilized Commitments</u>. (a) Upon at least three Business Days' prior written notice to the Administrative Agent at the Notice Office (which notice the Administrative Agent shall promptly transmit to each of the Lenders), the US Company shall have the right, at any time or from time to time, without premium or penalty to terminate the Total Unutilized Revolving Loan Commitment in whole, or reduce it in part, pursuant to this <u>Section 4.02(a)</u>, in an integral multiple of \$1,000,000 in the case of partial reductions to the Total Unutilized Revolving Loan Commitment; <u>provided</u> that each such reduction shall apply proportionately to permanently reduce the Revolving Loan Commitment of each Lender.
- (b) In the event of certain refusals by a Lender to consent to certain proposed changes, waivers, discharges or terminations with respect to this Agreement which have been approved by the Required Lenders as (and to the extent) provided in Section 13.12(b), the US Company shall have the right, subject to obtaining the consents required by Section 13.12(b), upon five Business Days' prior written notice to the Administrative Agent at the Notice Office (which notice the Administrative Agent shall promptly transmit to each of the Lenders), to terminate the entire Revolving Loan Commitment of such Lender, so long as all Loans, together with accrued and unpaid interest, Fees and all other amounts, owing to such Lender (including all amounts, if any, owing pursuant to Section 2.11) are repaid concurrently with the effectiveness of such termination (at which time Schedule 1.01(a) shall be deemed modified to reflect such changed amounts) and such Lender's RL Percentage of all outstanding Letters of Credit is cash collateralized in a manner satisfactory to the Administrative Agent and the respective Issuing Lenders, and at such time such Lender shall no longer constitute a "Lender" for purposes of this Agreement, except with respect to indemnifications under this Agreement (including, without limitation, Sections 2.10, 2.11, 3.06, 5.04, 5.05, 12.06, 13.01 and 13.06), which shall survive as to such repaid Lender.

4.03 <u>Mandatory Reduction of Commitments</u>. (a) The Total Revolving Loan Commitment (and the Revolving Loan Commitment of each Lender) shall terminate in its entirety on June 27, 2008, unless the Initial Borrowing Date has occurred on or prior to such date.

(b) In addition to any other mandatory commitment reductions pursuant to this <u>Section 4.03</u>, the Total Revolving Loan Commitment (and the Revolving Loan Commitment of each Lender) shall terminate in its entirety on the Final Maturity Date.

SECTION 5. Prepayments; Payments; Taxes.

5.01 Voluntary Prepayments. (a) Each Borrower shall have the right to prepay the Loans, without premium or penalty, in whole or in part at any time and from time to time on the following terms and conditions: (i) such Borrower shall give the Administrative Agent prior to 11:00 A.M. (New York City time) at the Notice Office (A) at least one Business Day's prior written notice (or telephonic notice promptly confirmed in writing) of its intent to prepay Base Rate Loans (or same day notice in the case of a prepayment of Swingline Loans) and (B) at least three Business Days' prior written notice (or telephonic notice promptly confirmed in writing) of its intent to prepay LIBOR Loans or EURIBOR Loans, which notice (in each case) shall specify whether Revolving Loans or Swingline Loans shall be prepaid, the amount of such prepayment and the Types of Loans to be prepaid and, in the case of LIBOR Loans or EURIBOR Loans, the specific Borrowing or Borrowings pursuant to which such LIBOR Loans or EURIBOR Loans were made, and which notice the Administrative Agent shall, promptly transmit to each of the Lenders; (ii) (x) each partial prepayment of Revolving Loans pursuant to this Section 5.01(a) shall be in an aggregate principal amount of at least \$500,000, £500,000 or €500,000, as applicable (or such lesser amount as is acceptable to the Administrative Agent) and (y) each partial prepayment of Swingline Loans pursuant to this Section 5.01(a) shall be in an aggregate principal amount of at least \$100,000, £100,000 or €100,000, as applicable (or such lesser amount as is acceptable to the Administrative Agent in any given case); provided that if any partial prepayment of US LIBOR Loans made pursuant to any Borrowing shall reduce the outstanding principal amount of US LIBOR Loans made pursuant to such Borrowing to an amount less than the Minimum Borrowing Amount applicable thereto, then such Borrowing may not be continued as a Borrowing of US LIBOR Loans (and same shall automatically be converted into a Borrowing of Base Rate Loans) and any election of an Interest Period with respect thereto given by such Borrower shall have no force or effect, and if any partial prepayment of UK Revolving Loans made pursuant to any Borrowing shall reduce the outstanding principal amount of UK Revolving Loans made pursuant to such Borrowing to an amount less than the Minimum Borrowing Amount applicable thereto, then such Borrowing shall automatically be converted into a Borrowing of UK Revolving Loans with an Interest Period of one week; (iii) each prepayment pursuant to this Section 5.01(a) in respect of any Loans made pursuant to a Borrowing shall be applied pro rata among such Loans; provided that at such Borrower's election in connection with any prepayment of Revolving Loans pursuant to this Section 5.01(a), such prepayment shall not, so long as no Default or Event of Default then exists, be applied to any Revolving Loan of a Defaulting Lender; and (iv) the respective Borrower shall pay all compensation due with respect to such prepayment specified in <u>Section 2.11</u> concurrently with such prepayment.

(b) In the event of certain refusals by a Lender to consent to certain proposed changes, waivers, discharges or terminations with respect to this Agreement which have been approved by the Required Lenders as (and to the extent) provided in Section 13.12(b), the Borrowers may, upon five Business Days' prior written notice to the Administrative Agent at the Notice Office (which notice the Administrative Agent shall promptly transmit to each of the Lenders), repay all Loans of such Lender, together with accrued and unpaid interest, Fees and all other amounts then owing to such Lender (including all amounts, if any, owing pursuant to Section 2.11) in accordance with, and subject to the requirements of Section 13.12(b), so long as (i) in the case of the repayment of Revolving Loans of any Lender pursuant to this clause (b), (A) the Revolving Loan Commitment of such Lender is terminated concurrently with such repayment pursuant to Section 4.02(b) (at which time Schedule 1.01(a) shall be deemed modified to reflect the changed Revolving Loan Commitments) and (B) such Lender's RL Percentage of all outstanding Letters of Credit is cash collateralized in a manner satisfactory to the

Administrative Agent and the respective Issuing Lenders and (ii) the consents, if any, required by <u>Section 13.12(b)</u> in connection with the repayment pursuant to this clause (b) shall have been obtained.

- 5.02 Mandatory Repayments; Cash Collateralization. (a) (i) On any day on which the Aggregate US Exposure exceeds (A) the US Borrowing Base at such time (based on the Borrowing Base Certificate last delivered) (other than during an Agent Advance Period) and/or (B) the US Maximum Amount at such time, then in each case, the US Borrowers jointly and severally shall repay on such day the principal of US Swingline Loans and, after all US Swingline Loans have been repaid in full or if no US Swingline Loans are outstanding, US Revolving Loans in an amount equal to such excess. If, after giving effect to the repayment of all outstanding US Swingline Loans and US Revolving Loans, the aggregate amount of the US Letter of Credit Outstandings exceeds (A) the US Borrowing Base at such time (based on the Borrowing Base Certificate last delivered) and/or (B) the US Maximum Amount at the time, then in each case, the US Borrowers jointly and severally shall pay to the Administrative Agent at the Payment Office on such day an amount of cash and/or Cash Equivalents equal to the amount of such excess, such cash and/or Cash Equivalents to be held as security for all US Obligations of the US Borrowers to each applicable US Issuing Lender and the Lenders hereunder in a cash collateral account to be established by, and under the sole dominion and control of, the Administrative Agent. Any such prepayment under this Section 5.02(a)(i) shall be applied to reduce the outstanding principal balance of the US Revolving Loans, but shall not permanently reduce the Revolving Loan Commitments;
- (ii) On any day on which the aggregate amount of the US Letter of Credit Outstandings exceeds the Maximum US Letter of Credit Amount, the US Borrowers jointly and severally shall pay to the Administrative Agent at the Payment Office on such day an amount of cash and/or Cash Equivalents equal to the amount of such excess, such cash and/or Cash Equivalents to be held as security for all US Obligations of the US Borrowers to each applicable US Issuing Lender and the Lenders hereunder in a cash collateral account to be established by, and under the sole dominion and control of, the Administrative Agent;
- (iii) On any day (including on any date on which Equivalent Amounts are determined pursuant to Section 2.17) on which the Aggregate UK Exposure exceeds (A) the UK Borrowing Base at such time (based on the Borrowing Base Certificate last delivered) (other than during an Agent Advance Period) and/or (B) the UK Maximum Amount at such time, then in each case, the Borrowers jointly and severally shall repay on such day the principal of UK Swingline Loans and, after all UK Swingline Loans have been repaid in full or if no UK Swingline Loans are outstanding, UK Revolving Loans in an amount equal to such excess. If, after giving effect to the repayment of all outstanding UK Swingline Loans and UK Revolving Loans, the aggregate amount of the UK Letter of Credit Outstandings exceeds (A) the UK Borrowing Base at such time (based on the Borrowing Base Certificate last delivered) and/or (B) the UK Maximum Amount at the time, then in each case, the Borrowers jointly and severally shall pay to the Administrative Agent at the Payment Office on such day an amount of cash and/or Cash Equivalents equal to the amount of such excess, such cash and/or Cash Equivalents to be held as security for all UK Obligations of the UK Borrowers to each applicable UK Issuing Lender and the Lenders hereunder in a cash collateral account to be established by, and under the sole dominion and control of, the Administrative Agent. Any such prepayment under this Section 5.02(a)(iii) shall be applied to reduce the outstanding principal balance of the UK Revolving Loans, but shall not permanently reduce the Revolving Loan Commitments;
- (iv) On any day (including on any date on which Equivalent Amounts are determined pursuant to Section 2.17) on which the aggregate amount of the UK Letter of Credit Outstandings exceeds the Maximum UK Letter of Credit Amount, the UK Borrowers jointly and severally shall pay to the Administrative Agent at the Payment Office on such day an amount of cash and/or Cash Equivalents equal to the amount of such excess, such cash and/or Cash Equivalents to be held as security for all UK

Obligations of the UK Borrowers to each applicable UK Issuing Lender and the Lenders hereunder in a cash collateral account to be established by, and under the sole dominion and control of, the Administrative Agent;

- (b) (i) If US Company or any Domestic Subsidiary issues any additional Indebtedness (other than intercompany Indebtedness) or obtains proceeds from any capital contributions or the issuance of any Equity Interests, US Company shall pay to Administrative Agent for the ratable benefit of Lenders, when and as received by any US Borrower or Domestic Subsidiary thereof and as a mandatory prepayment of the US Obligations, or if none are outstanding, the UK Obligations, a sum equal to 100% of the net cash proceeds to US Company or such Domestic Subsidiary of the issuance of such Indebtedness or Equity Interest or capital contribution. Any such prepayment shall be applied to reduce the outstanding principal balance of the US Revolving Loans, or if none are outstanding, the UK Revolving Loans, but shall not permanently reduce the Revolving Loan Commitments or the UK Revolving Loan Commitments; and (ii) if UK Company or any other UK Subsidiary issues any additional Indebtedness (other than intercompany Indebtedness) or obtains proceeds from any capital contributions or the issuance any Equity Interests in a manner permitted under this Agreement, UK Company shall pay to Administrative Agent for the ratable benefit of Lenders, when and as received by any UK Borrower or other UK Subsidiary thereof and as a mandatory prepayment of the UK Obligations, a sum equal to 100% of the net cash proceeds to UK Company or such other UK Subsidiary of the issuance of such Indebtedness or Equity Interest or capital contribution. Any such prepayment shall be applied to reduce the outstanding principal balance of the UK Revolving Loans, but shall not permanently reduce the UK Revolving Loan Commitments. If the proceeds of the issuance of Securities are to be used to redeem or repurchase Senior Notes in accordance with Section 10.04 hereof, such proceeds shall be paid to Administrative Agent for application to the Revolving Loans but, subject to the terms of this Agreement, may be reborrowed for such redemption or repurchase.
- (c) (i) Except as provided in Section 10.03(i), (iv) or (ix) if US Company or any of its Domestic Subsidiaries sells or otherwise disposes of any of its Equipment or Real Property or other Collateral or assets, or if a Casualty Loss occurs with respect to any of the Collateral, US Company shall, unless otherwise agreed by the Required Lenders, pay to Administrative Agent for the ratable benefit of Lenders as and when received by US Company or such Domestic Subsidiary and as a mandatory prepayment of the Loans, as herein provided, a sum equal to the proceeds (including insurance payments and condemnation awards but net of costs and taxes incurred in connection with such sale or event) ("Sale Proceeds") received by US Company or such Domestic Subsidiary from such sale or Casualty Loss. The applicable prepayment shall be applied to reduce the outstanding principal balance of the US Revolving Loans, or if none are outstanding, the UK Revolving Loans, but, except as provided below, shall not permanently reduce the Revolving Loan Commitments or the UK Revolving Loan Commitments; provided that any sale or Casualty Loss of Inventory, Equipment or Eligible Real Property by a US Credit Party shall reduce the US Borrowing Base to the extent of such applicable Property's contribution to the US Borrowing Base. Such reduction shall be effective on the date of consummation of the sale or receipt of proceeds of a Casualty Loss. If US Borrower and its Domestic Subsidiaries do not reinvest the proceeds of any sales or other dispositions of assets within 364 days after receipt of such proceeds in assets used in their business and would be required to make a "Net Proceeds Offer" (as defined in any Senior Note Indenture), then the Revolving Loan Commitments shall be automatically permanently reduced by an amount equal to the uninvested portion of such proceeds on the 364th day after receipt of such proceeds.

(ii) If US Company or any of its Domestic Subsidiaries sells or otherwise disposes of any of its Equipment or Real Property or other Collateral or assets pursuant to Section 10.03 (ix) or US Company or any of its Domestic Subsidiaries or the LKE Qualified Intermediary otherwise receives any LKE Proceeds, US Company shall, unless otherwise agreed by the Required Lenders, pay to

Administrative Agent for the ratable benefit of Lenders as and when received by US Company or such Domestic Subsidiary or the LKE Qualified Intermediary and as a mandatory prepayment of the Loans, as herein provided, the Sale Proceeds received by US Company or such Domestic Subsidiary or the LKE Qualified Intermediary from such sale; provided that solely in the event that the LKE Qualified Intermediary shall have received and retained (pending any permitted reinvestment described below) in the LKE Joint Account any LKE Proceeds (in an aggregate amount not to exceed at any time \$25,000,000) in accordance with any LKE Master Exchange Agreement, no such prepayment shall be required (A) for the period commencing on the date of the applicable sale and ending on the earliest of (I) 180 days thereafter, (II) an Event of Default and (III) the commencement of a Compliance Period (such period, the "LKE Period") and (B) with respect to all or a portion of such LKE Proceeds if and only to the extent that such LKE Proceeds are used within the applicable LKE Period to acquire assets in accordance with a LKE Transaction; provided further that it is expressly understood and agreed that such prepayment shall be required with respect to any such LKE Proceeds that cause (or, if so deposited in the LKE Joint Account, would cause) the aggregate amount of funds deposited in the LKE Joint Account to exceed at any time \$25,000,000. The applicable prepayment shall be applied to reduce the outstanding principal balance of the US Revolving Loans, or if none are outstanding, the UK Revolving Loans, but, except as provided below, shall not permanently reduce the Revolving Loan Commitments or the UK Revolving Loan Commitments; provided that any sale of Inventory, Equipment or Eligible Real Property by a US Credit Party shall reduce the US Borrowing Base to the extent of such applicable Property's contribution to the US Borrowing Base. Such reduction shall be effective on the date of consummation of the sale or receipt of proceeds of a Casualty Loss. If US Borrower and its Domestic Subsidiaries do not reinvest the proceeds of any sales or other dispositions of such assets within 364 days after receipt of such proceeds in assets used in their business and would be required to make a "Net Proceeds Offer" (as defined in any Senior Note Indenture), then the Revolving Loan Commitments shall be automatically permanently reduced by an amount equal to the uninvested portion of such proceeds on the 364th day after receipt of such proceeds.

(iii) Except as provided in Section 10.03(i) or (iv), if UK Company or any other UK Subsidiary sells or otherwise disposes of any of its Equipment or Real Property or other Collateral or assets, or if a Casualty Loss occurs with respect to any of the Collateral, UK Company shall, unless otherwise agreed by the Required Lenders, pay to Administrative Agent for the ratable benefit of Lenders as and when received by UK Company or such other UK Subsidiary and as a mandatory prepayment of the UK Revolving Loans, as herein provided, a sum equal to the Sale Proceeds received by UK Company or such UK Subsidiary from such sale or Casualty Loss. The applicable prepayment shall be applied to reduce the outstanding principal balance of the UK Revolving Loans, but, except as provided below, shall not permanently reduce the Revolving Loan Commitments or the UK Revolving Loan Commitments; provided that any sale or Casualty Loss of Inventory or Equipment by a UK Borrowing Base Party shall reduce the UK Borrowing Base to the extent of such applicable Property's contribution to the UK Borrowing Base. Such reduction shall be effective on the date of consummation of the sale or receipt of proceeds of a Casualty Loss. If UK Borrower and/or the other UK Subsidiaries do not reinvest the proceeds of any sales or other dispositions of assets within 364 days after receipt of such proceeds in assets used in their business and would be required to make a "Net Proceeds Offer" (as defined in any Senior Note Indenture), then the UK Revolving Loan Commitments shall be automatically permanently reduced by an amount equal to the uninvested portion of such proceeds on the 364th day after receipt of such proceeds.

(d) With respect to each repayment of Loans required by this <u>Section 5.02</u>, the Borrowers may designate the Types of Loans which are to be repaid and, in the case of LIBOR Loans or EURIBOR Loans, the specific Borrowing or Borrowings pursuant to which such LIBOR Loans or EURIBOR Loans were made; <u>provided</u> that: (i) repayments of LIBOR Loans or EURIBOR Loans pursuant to this <u>Section 5.02</u> made on a day other than the last day of an Interest Period applicable thereto

shall be subject to Section 2.11; (ii) if any repayment of LIBOR Loans or EURIBOR Loans made pursuant to a single US Borrowing shall reduce the outstanding LIBOR Loans or EURIBOR Loans made pursuant to such US Borrowing to an amount less than the Minimum Borrowing Amount applicable thereto, such US Borrowing shall be automatically converted into a Borrowing of Base Rate Loans; and (iii) each repayment of any Loans made pursuant to a Borrowing shall be applied pro rata among the Lenders holding such Loans. In the absence of a designation by a Borrower as described in the preceding sentence, the Administrative Agent shall, subject to the above, make such designation in its sole discretion.

- (e) In addition to any other mandatory repayments pursuant to this <u>Section 5.02</u>, (i) all then outstanding Swingline Loans shall be repaid in full on the Swingline Expiry Date and (ii) all then outstanding Revolving Loans shall be repaid in full on the Final Maturity Date.
- (f) In addition to any other mandatory repayments pursuant to this <u>Section 5.02</u>, each Swingline Loan will be repaid (for the avoidance of doubt, such repayment may be made with proceeds from Revolving Loans) no later than the seventh day following the incurrence thereof; <u>provided</u> that, if the seventh day is not a Business Day, such repayment shall be made on the next succeeding Business Day.
- 5.03 Method and Place of Payment. (a) Except as otherwise specifically provided herein, all payments under this Agreement and under any Note shall be made to the Administrative Agent for the account of the Lender or Lenders entitled thereto not later than 12:00 Noon (New York City time) on the date when due and shall be made, with respect to US Obligations, in US Dollars, and with respect to UK Obligations, in Pounds Sterling or Euros, as applicable, in immediately available funds at the Payment Office. Whenever any payment to be made hereunder or under any Note shall be stated to be due on a day which is not a Business Day, the due date thereof shall be extended to the next succeeding Business Day and, with respect to payments of principal, interest shall be payable at the applicable rate during such extension.
- (b) Each US Credit Party shall, along with the Collateral Agent and certain financial institutions selected by US Company and approved by the Administrative Agent (the "Collection Banks"), enter into on or prior to the Initial Borrowing Date (as such date may be extended from time to time by the Administrative Agent in its sole discretion) and thereafter maintain separate Cash Management Control Agreements. Each US Credit Party shall instruct all Account Debtors of such US Credit Party, as the case may be, to remit all payments to the applicable "P.O. Boxes", "Lockbox Addresses" or a sub-account of the Collection Account of the applicable Collection Bank with respect to all Accounts of such Account Debtor, which remittances shall be collected by the applicable Collection Bank and deposited in the applicable Collection Account. All amounts received by any US Credit Party and any Collection Bank in respect of any Account, in addition to all other cash received from any other source, shall upon receipt be deposited into a Collection Account or directly into the Concentration Account.
- (c) All amounts held in all of the Collection Accounts, Disbursement Accounts, Designated Petty Cash Accounts, Designated Payroll Accounts and LKE Joint Accounts with respect to each US Credit Party shall be wired by the close of business on each Business Day into one or more concentration accounts with the Collateral Agent and/or one or more other institutions reasonably acceptable to the Administrative Agent (each, a "Concentration Account") unless such amounts are otherwise required or permitted to be applied pursuant to Section 5.02; provided, however, (x) so long as no Default and no Event of Default then exists, a balance of up to \$4,000,000 in the aggregate may be maintained in all Disbursement Accounts, Designated Petty Cash Accounts and Designated Payroll Accounts and (y) so long as no Default and no Event of Default and no Compliance Period then exists,

the LKE Qualified Intermediary may retain during any applicable LKE Period in any LKE Joint Account any LKE Proceeds not required by Section 5.02(c) (ii) to be used to prepay the US Revolving Loans. Except as, and to the extent, provided in the proviso to the immediately preceding sentence, all of the Collection Accounts, Disbursement Accounts, Designated Petty Cash Accounts and Designated Payroll Accounts shall be "zero" balance accounts. So long as no Event of Default or Compliance Period then exists, the US Credit Parties shall be permitted to transfer cash from the Concentration Accounts to the Disbursement Accounts, the Designated Petty Cash Accounts and the Designated Payroll Accounts to be used for working capital and general corporate purposes, all subject to the requirements of this Section 5.03(c) and pursuant to procedures and arrangements to be determined by the Administrative Agent. So long as no Default or Event of Default or Compliance Period then exists, the US Credit Parties and the LKE Qualified Intermediary shall be permitted to transfer cash from the LKE Joint Accounts to be used to acquire assets in accordance with a LKE Transaction, subject to the requirements of this Section 5.03(c) and pursuant to procedures and arrangements to be determined by the Administrative Agent. If an Event of Default or Compliance Period exists, all collected amounts held in the Concentration Accounts shall be applied as provided in Section 5.03(d).

- (d) During the continuance of a Compliance Period or an Event of Default, all collected amounts held in the Concentration Accounts shall be distributed and applied on a daily basis in the following order (in each case, to the extent the Administrative Agent has actual knowledge of the amounts owing or outstanding as described below and after giving effect to the application of any such amounts otherwise required to be applied pursuant to Section 5.02(b), (c) or (d) constituting proceeds from any Collateral otherwise required to be applied pursuant to the terms of the respective Security Document):
 - (i) <u>first</u>, to the payment (on a ratable basis) of any outstanding Expenses actually due and payable to the Administrative Agent and/or the Collateral Agent under any of the Credit Documents and to repay or prepay outstanding Loans advanced by the Administrative Agent on behalf of the Lenders pursuant to Sections 2.01(h) and 2.04(c);
 - (ii) <u>second</u>, to the extent all amounts referred to in preceding clause (i) have been paid in full, to pay (on a ratable basis) all outstanding Expenses actually due and payable to each Issuing Lender under any of the Credit Documents and to repay all outstanding Unpaid US Drawings and Unpaid UK Drawings and all interest thereon;
 - (iii) <u>third</u>, to the extent all amounts referred to in preceding clauses (i) and (ii) have been paid in full, to pay (on a ratable basis) all accrued and unpaid interest actually due and payable to the Administrative Agent, the Issuing Lenders and the Lenders under any of the Credit Documents;
 - (iv) <u>fourth</u>, to the extent all amounts referred to in preceding clauses (i) through (iii), inclusive, have been paid in full, to repay the outstanding principal of Revolving Loans (whether or not then due and payable) and the Qualified Swap Termination Value under Qualified Derivative Obligations, ratably among the Lenders and the counterparties holding any such Qualified Derivative Obligations, in proportion to the respective amounts described in this clause (iv) held by them; and
 - (v) <u>fifth</u>, to the extent all amounts referred to in preceding clauses (i) through (iv), inclusive, have been paid in full, to pay (on a ratable basis) all other outstanding Obligations then due and payable to the Administrative Agent, the Collateral Agent, the Lenders and the other holders thereof, under any of the Credit Documents or otherwise (including, without limitation, the Swap Termination Value of any Qualified Derivative Obligations (to the extent not paid

pursuant to clause (iv) above) and any Banking Product Obligations).

- (e) Each UK Subsidiary shall have entered into a cash management system in form and substance satisfactory to the Administrative Agent, including as required under clauses 4.1 and 10 of the UK Debenture.
- (f) Without limiting the provisions set forth in Section 13.15, the Administrative Agent shall maintain accounts on its books in the name of each Borrower (collectively, the "Credit Account") in which each Borrower will be charged with all loans and advances made by the Lenders to the respective Borrower for the respective Borrower's account, including the Loans, the Letter of Credit Outstandings, and the Fees, Expenses and any other Obligations relating thereto. Each Borrower will be credited, in accordance with this Section 5.03, with all amounts received by the Lenders from such Borrower or from others for its account, including, as set forth above, all amounts received by the Administrative Agent and applied to the Obligations. In no event shall prior recourse to any Accounts or other Collateral be a prerequisite to the Administrative Agent's right to demand payment of any Obligation upon its maturity. Further, the Administrative Agent shall have no obligation whatsoever to perform in any respect any of the Borrowers' or Guarantors' contracts or obligations relating to the Accounts.

5.04 Net Payments — US Borrowers. (a) All payments made by or on behalf of the US Borrowers hereunder and under any Note will be made without setoff, counterclaim or other defense. All such payments will be made free and clear of, and without deduction or withholding for, any present or future taxes, levies, imposts, duties, fees, assessments or other charges of whatever nature now or hereafter imposed by any jurisdiction or by any political subdivision or taxing authority thereof or therein with respect to such payments (but excluding, except as provided in the second succeeding sentence, (i) any tax imposed on or measured by the net income or profits (including branch profits taxes) and franchise or similar taxes imposed in lieu of income taxes of a Lender that holds US Obligations pursuant to the laws of the jurisdiction in which it is organized or the jurisdiction in which the principal office or applicable lending office of such Lender is located or any subdivision thereof or therein, and (ii) in the case of a Non-U.S. Lender that holds US Obligations (other than an assignee pursuant to a request by a Borrower under Section 2.13), any withholding tax that is imposed on amounts payable to such Non-U.S. Lender at the time such Non-U.S. Lender becomes a party hereto or is attributable to such Non-U.S. Lender's failure (other than as a result of a Change in Law) to comply with Section 5.04(b)) and all interest, penalties or similar liabilities with respect to such non-excluded taxes, levies, imposts, duties, fees, assessments or other charges or withholding of a similar nature (including any penalty or interest payable in connection with any failure to pay or any delay in paying any of the same)(all such non-excluded taxes, levies, imposts, duties, fees, assessments or other charges or withholding of a similar nature (including any penalty or interest payable in connection with any failure to pay or any delay in paying any of the same) being referred to collectively as "Taxes"). If any Taxes are so levied or imposed, the US Borrowers jointly and severally agree to pay the full amount of such Taxes, and such additional amounts as may be necessary so that every payment of all amounts due under this Agreement or under any Note, after withholding or deduction for or on account of any Taxes, will not be less than the amount provided for herein or in such Note. If any amounts are payable in respect of Taxes pursuant to the preceding sentence, the US Borrowers jointly and severally agree to reimburse each Lender that holds US Obligations, upon the written request of such Lender, for taxes imposed on or measured by the net income or profits (including branch profits taxes) and franchise or similar taxes imposed in lieu of income taxes of such Lender pursuant to the laws of the jurisdiction in which such Lender is organized or in which the principal office or applicable lending office of such Lender is located or under the laws of any political subdivision or taxing authority of any such jurisdiction in which such Lender is organized or in which the principal office or applicable lending office of such Lender is located and for any withholding of taxes as such Lender shall determine are payable by, or withheld from, such

Lender, in respect of such amounts so paid to or on behalf of such Lender pursuant to the preceding sentence and in respect of any amounts paid to or on behalf of such Lender pursuant to this sentence. The US Borrowers will furnish to the Administrative Agent within 45 days after the date the payment of any Taxes is due pursuant to Applicable Law certified copies of tax receipts evidencing such payment by such US Borrowers. The US Borrowers jointly and severally agree to indemnify and hold harmless each Lender that holds US Obligations, and reimburse such Lender upon its written request, for the amount of any Taxes so levied or imposed and paid by such Lender.

(b) Each Lender that is not a United States person (as such term is defined in Section 7701(a)(30) of the Code) for US Federal income tax purposes (each, a "Non-U.S. Lender"), which holds US Obligations, agrees to deliver to US Company and the Administrative Agent on or prior to the Effective Date or, in the case of a Non-U.S. Lender, which holds US Obligations, and that is an assignee or transferee of an interest under this Agreement pursuant to Section 2.13 or 13.04(b) (unless the respective Lender was already a Lender hereunder immediately prior to such assignment or transfer), on the date of such assignment or transfer to such Lender in each case, only if such Lender is legally entitled to do so, (i) two accurate and complete original signed copies of Internal Revenue Service Form W-8ECI or Form W-8BEN (with respect to an exemption under an income tax treaty) (or successor forms) certifying to such Lender's entitlement as of such date to an exemption from United States withholding tax with respect to payments to be made under this Agreement and under any Note, (ii) if such Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code and cannot deliver either Internal Revenue Service Form W-8ECI or Form W-8BEN (or any successor forms) pursuant to clause (i) above, (x) a certificate substantially in the form of Exhibit H (any such certificate, a "Section 5.04(b)(ii) Certificate") and (y) two accurate and complete original signed copies of Internal Revenue Service Form W-8BEN (with respect to the portfolio interest exemption) (or successor form) certifying to such Lender's entitlement as of such date to an exemption from United States withholding tax with respect to payments of interest to be made under this Agreement and under any Note, or (iii) Internal Revenue Service Form W-8IMY (together with any applicable forms listed in (i) and (ii) of this Section. In addition, each Lender that holds US Obligations agrees that from time to time after the Effective Date, when a lapse in time or change in circumstances renders the previous certification obsolete or inaccurate in any material respect, such Lender will deliver (only if such Non-U.S. Lender is legally entitled to do so) to US Company and the Administrative Agent two new accurate and complete original signed copies of Internal Revenue Service Form W-8ECI, Form W-8BEN (with respect to the benefits of any income tax treaty), or Form W-8BEN (with respect to the portfolio interest exemption) and a Section 5.04(b)(ii) Certificate, as the case may be, and such other forms (including Internal Revenue Service Form W-8IMY (together with any applicable forms listed in (i) and (ii) of this Section)) as may be required in order to confirm or establish the entitlement of such Lender to a continued exemption from or reduction in United States withholding tax with respect to payments under this Agreement and any Note, or such Lender shall immediately notify US Company and the Administrative Agent of its inability to deliver any such Form or Certificate, in which case such Lender shall not be required to deliver any such Form or Certificate pursuant to this Section 5.04(b). Notwithstanding anything to the contrary contained in Section 5.04(a), but subject to Section 13.04(b) and the immediately succeeding sentence, each US Borrower shall be entitled, to the extent it is required to do so by law, to deduct or withhold income or similar taxes imposed by the United States (or any political subdivision or taxing authority thereof or therein) from interest, Fees or other amounts payable hereunder for the account of (i) any Non-U.S. Lender that holds US Obligations to the extent that such Lender has not provided to US Company US Internal Revenue Service Forms that establish a complete exemption from such deduction or withholding and (ii) any Lender that is not a Non-U.S. Lender and that holds US Obligations to the extent that such Lender has not provided to US Company Internal Revenue Services Form W-9 (or successor forms), or other documentation reasonably satisfactory to US Company, certifying to such Lender's exemption from back-up withholding, to the extent required pursuant to Section 5.04(c). Notwithstanding anything to the contrary contained in the preceding sentence or elsewhere in this Section 5.04 and except as set forth in Section 13.04(b), the US

Borrowers jointly and severally agree to pay any additional amounts and to indemnify each Lender that holds US Obligations in the manner set forth in Section 5.04(a) (without regard to the identity of the jurisdiction requiring the deduction or withholding) in respect of any amounts deducted or withheld by it as described in the immediately preceding sentence as a result of any changes that are effective after the Effective Date in any Applicable Law, treaty, governmental rule, regulation, guideline or order, or in the interpretation thereof, relating to the deducting or withholding of income or similar taxes.

(c) Each Lender that is a United States person (as such term is defined in Section 7701(a)(30) of the Code) for US Federal income tax purposes, which holds US Obligations, agrees to deliver to US Company and the Administrative Agent, to the extent requested by either US Company or the Administrative Agent, two accurate and complete original signed copies of Internal Revenue Services Form W-9 (or successor forms), or other documentation reasonably satisfactory to US Company, certifying to such Lender's exemption from back-up withholding.

5.05 Tax Gross Up and Indemnities — UK Subsidiaries.

(a) Definitions.

In this Section:

"ITA" means the United Kingdom Income Tax Act 2007.

"Party" means a party to this Agreement.

"<u>Protected Party</u>" means a Lender which is or will be subject to any liability, or required to make any payment, for or on account of Tax in relation to a sum received or receivable (or any sum deemed for the purposes of Tax to be received or receivable) under a Credit Document.

"Qualifying Lender" means:

(i) a Lender (other than a Lender within sub-paragraph (ii) below) which is beneficially entitled to interest payable to that Lender in respect of an advance under a Credit Document and is:

(A) a Lender:

- (1) which is a bank (as defined for the purpose of section 879 of the ITA) making an advance under a Credit Document; or
- (2) in respect of an advance made under a Credit Document by a person that was a bank (as defined for the purpose of section 879 of the ITA) at the time that that advance was made,

and which is within the charge to United Kingdom corporation tax as respects any payments of interest made in respect of that advance; or

- (B) a Lender which is:
 - (1) a company resident in the United Kingdom for United Kingdom tax purposes;
 - (2) a partnership each member of which is:

- (a) a company so resident in the United Kingdom; or
- (b) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account in computing its chargeable profits (for the purposes of section 11(2) of the Taxes Act) the whole of any share of interest payable in respect of that advance that falls to it by reason of sections 114 and 115 of the Taxes Act:
- (3) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account interest payable in respect of that advance in computing the chargeable profits (for the purposes of section 11(2) of the Taxes Act) of that company; or
- (C) a Treaty Lender; or
- (ii) a building society (as defined for the purpose of section 880 of the ITA) making an advance under a Credit Document).
- "Taxes Act" means the United Kingdom Income and Corporation Taxes Act 1988.
- "<u>Tax Confirmation</u>" means a confirmation by a Lender that the person beneficially entitled to interest payable to that Lender in respect of an advance under a Credit Document is either:
 - (i) a company resident in the United Kingdom for United Kingdom tax purposes;
 - (ii) a partnership each member of which is:
 - (A) a company so resident in the United Kingdom; or
 - (B) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account in computing its chargeable profits (for the purposes of section 11(2) of the Taxes Act) the whole of any share of interest payable in respect of that advance that falls to it by reason of sections 114 and 115 of the Taxes Act; or
 - (iii) a company not so resident in the United Kingdom which carries on a trade in the United Kingdom through a permanent establishment and which brings into account interest payable in respect of that advance in computing the chargeable profits (for the purposes of section 11(2) of the Taxes Act) of that company.

"Tax Credit" means a credit against, relief or remission for, or repayment of any Tax.

"Tax Deduction" means a deduction or withholding for or on account of Tax from a payment under a Credit Document.

"<u>Tax Payment</u>" means either the increase in a payment made by a UK Subsidiary to a Lender under Section 5.05(b) (*Tax gross-up*) or a payment under Section 5.05(c) (*Tax indemnity*).

"Treaty Lender" means a Lender which:

- (i) is treated as a resident of a Treaty State for the purposes of the Treaty; and
- (ii) does not carry on a business in the United Kingdom through a permanent establishment with which that Lender's participation in the Loan is effectively connected.

"<u>Treaty State</u>" means a jurisdiction having a double taxation agreement (a "**Treaty**") with the United Kingdom which makes provision for full exemption from tax imposed by the United Kingdom on interest.

"UK Non-Bank Lender" means:

- (i) where a Lender becomes a Party on the day on which this Agreement is entered into, a Lender listed on Schedule 1.01(a); and
- (ii) where a Lender becomes a Party after the day on which this Agreement is entered into, a Lender which gives a Tax Confirmation in the Assignment and Assumption Agreement which it executes on becoming a Party.

"VAT" means value added tax as provided for in the United Kingdom Value Added Tax Act 1994 and any other tax of a similar nature.

- (b) Tax Gross-Up.
- (i) Each UK Subsidiary shall make all payments to be made by it without any Tax Deduction, unless a Tax Deduction is required by law.
- (ii) The UK Company shall promptly upon becoming aware that a UK Subsidiary must make a Tax Deduction (or that there is any change in the rate or the basis of a Tax Deduction) notify the Administrative Agent accordingly. Similarly, a Lender shall notify the Administrative Agent on becoming so aware in respect of a payment payable to that Lender. If the Administrative Agent receives such notification from a Lender it shall notify the UK Company and that UK Subsidiary.
- (iii) If a Tax Deduction is required by law to be made by a UK Subsidiary, the amount of the payment due from that UK Subsidiary shall be increased to an amount which (after making any Tax Deduction) leaves an amount equal to the payment which would have been due if no Tax Deduction had been required.
- (iv) A UK Subsidiary is not required to make an increased payment to a Lender under paragraph (b)(iii) above for a Tax Deduction in respect of tax imposed by the United Kingdom from a payment of interest on a Loan, if on the date on which the payment falls due:
 - (A) the payment could have been made to the relevant Lender without a Tax Deduction if it was a Qualifying Lender, but on that date that Lender is not or has ceased to be a Qualifying Lender other than as a result of any change after the date it became a Lender under this Agreement in (or in

- the interpretation, administration, or application of) any law or Treaty, or any published practice or concession of any relevant taxing authority; or
- (B) (1) the relevant Lender is a Qualifying Lender solely under sub-paragraph (i)(B) of the definition of Qualifying Lender;
 - (2) an officer of H.M. Revenue & Customs has given (and not revoked) a direction (a "<u>Direction</u>") under section 931 of the ITA (as that provision has effect on the date on which the relevant Lender became a Party) which relates to that payment and that Lender has received from that UK Subsidiary or the UK Company a certified copy of that Direction; and
 - (3) the payment could have been made to the Lender without any Tax Deduction in the absence of that Direction; or
- (C) the relevant Lender is a Qualifying Lender solely under sub-paragraph (i)(B) of the definition of Qualifying Lender and it has not, other than by reason of any change after the date of this Agreement in (or in the interpretation, administration, or application of) any law, or any published practice or concession of any relevant taxing authority, given a Tax Confirmation to the UK Company; or
- (D) the relevant Lender is a Treaty Lender and the UK Subsidiary making the payment is able to demonstrate that the payment could have been made to the Lender without the Tax Deduction had that Lender complied with its obligations under paragraph (g) below.
- (v) If a UK Subsidiary is required to make a Tax Deduction, that UK Subsidiary shall make that Tax Deduction and any payment required in connection with that Tax Deduction within the time allowed and in the minimum amount required by law.
- (vi) Within thirty days of making either a Tax Deduction or any payment required in connection with that Tax Deduction, the UK Subsidiary making that Tax Deduction shall deliver to the Administrative Agent for the Lender entitled to the payment evidence reasonably satisfactory to that Lender that the Tax Deduction has been made or (as applicable) any appropriate payment paid to the relevant taxing authority.
- (vii) A Treaty Lender and each UK Subsidiary which makes a payment to which that Treaty Lender is entitled shall co-operate in completing any procedural formalities necessary for that UK Subsidiary to obtain authorisation to make that payment without a Tax Deduction.
- (viii) A UK Non-Bank Lender which becomes a Party on the day on which this Agreement is entered into gives a Tax Confirmation to the UK Company by entering into this Agreement.
- (ix) A UK Non-Bank Lender shall promptly notify the UK Company and the Administrative Agent if there is any change in the position from that set out in the Tax Confirmation.
 - (c) Tax Indemnity.

- (i) The UK Company shall (within three Business Days of demand by the Administrative Agent) pay to a Protected Party an amount equal to the loss, liability or cost which that Protected Party determines will be or has been (directly or indirectly) suffered for or on account of Tax by that Protected Party in respect of a Credit Document
 - (ii) Paragraph (a) above shall not apply:
 - (A) with respect to any Tax assessed on a Lender:
 - (1) under the law of the jurisdiction in which that Lender is incorporated or, if different, the jurisdiction (or jurisdictions) in which that Lender is treated as resident for tax purposes; or
 - (2) under the law of the jurisdiction in which that Lender's lending office is located in respect of amounts received or receivable in that jurisdiction,

if that Tax is imposed on or calculated by reference to the net income received or receivable (but not any sum deemed to be received or receivable) by that Lender; or

- (B) to the extent a loss, liability or cost:
 - (1) (A) is compensated for by an increased payment under Section 5.05(b) (*Tax gross-up*); or
 - (2) would have been compensated for by an increased payment under Section 5.05(b) (*Tax gross-up*) but was not so compensated solely because one of the exclusions in paragraph (d) of Section 5.05(b) (*Tax gross-up*) applied.
- (iii) A Protected Party making, or intending to make a claim under paragraph (a) above shall promptly notify the Administrative Agent of the event which will give, or has given, rise to the claim, following which the Administrative Agent shall notify the UK Company.
 - (iv) A Protected Party shall, on receiving a payment from a UK Subsidiary under this Section 5.05(c), notify the Administrative Agent.
 - (d) <u>Tax Credit</u>.

If a UK Subsidiary makes a Tax Payment and the relevant Lender determines that:

- (a) a Tax Credit is attributable either to an increased payment of which that Tax Payment forms part, or to that Tax Payment; and
- (b) that Lender has obtained, utilised and retained that Tax Credit,

the Lender shall pay an amount to the UK Subsidiary which that Lender determines will leave it (after that payment) in the same after-Tax position as it would have been in had the Tax Payment not been required to be made by the UK Subsidiary.

(e) Stamp Taxes.

The UK Company shall pay and, within three Business Days of demand, indemnify each Lender against any cost, loss or liability that Lender incurs in relation to all stamp duty, registration and other similar Taxes payable in respect of any Credit Document.

(f) Value Added Tax.

- (i) All amounts set out, or expressed to be payable under a Credit Document by any Party to a Lender which (in whole or in part) constitute the consideration for VAT purposes shall be deemed to be exclusive of any VAT which is chargeable on such supply, and accordingly, subject to paragraph (c) below, if VAT is chargeable on any supply made by any Lender to any Party under a Credit Document, that Party shall pay to the Lender (in addition to and at the same time as paying the consideration) an amount equal to the amount of the VAT (and such Lender shall promptly provide an appropriate VAT invoice to such Party).
- (ii) If VAT is chargeable on any supply made by any Lender (the "<u>Supplier</u>") to any other Lender (the "<u>Recipient</u>") under a Credit Document, and any Party (the "<u>Relevant Party</u>") is required by the terms of any Credit Document to pay an amount equal to the consideration for such supply to the Supplier (rather than being required to reimburse the Recipient in respect of that consideration), such Party shall also pay to the Supplier (in addition to and at the same time as paying such amount) an amount equal to the amount of such VAT. The Recipient will promptly pay to the Relevant Party an amount equal to any credit or repayment from the relevant tax authority which it reasonably determines relates to the VAT chargeable on that supply.
- (iii) Where a Credit Document requires any Party to reimburse a Lender for any costs or expenses, that Party shall also at the same time pay and indemnify the Lender against all VAT incurred by the Lender in respect of the costs or expenses to the extent that the Lender reasonably determines that neither it nor any other member of any group of which it is a member for VAT purposes is entitled to credit or repayment from the relevant tax authority in respect of the VAT.

SECTION 6. Conditions Precedent to Credit Events on the Initial Borrowing Date. The obligation of each Lender to make Loans, and the obligation of each Issuing Lender to issue Letters of Credit, on the Initial Borrowing Date, are subject at the time of the making of such Loans or the issuance of such Letters of Credit to the satisfaction of the following conditions (provided that the Loans made on the Initial Borrowing Date in order to (I) fund the purchase from the holders thereof all of the outstanding Mezzanine Notes and (II) repay in full, together with all fees and other amounts owing thereon, all Refinanced Indebtedness outstanding under the Indebtedness described in clause (i) of the definition of Refinanced Indebtedness (in each case, which such Loans (the "Pre-Merger Loans") shall be funded prior to the remaining Loans to be made on the Initial Borrowing Date) shall (x) not be subject to the conditions set forth in Sections 6.03(a)(ii), 6.03(d)(ii), 6.03(e), 6.05(b), 6.05(c), 6.05(d) (other than, in the case of Section 6.05(d), with respect to Indebtedness described in clause (i) of the definition of Refinanced Indebtedness) and (y) not be subject to the conditions set forth in Sections 6.04, 6.08, 6.09, 6.10 and 6.11 in so far as they pertain to Target and Target's Subsidiaries):

6.01 Effective Date; Notes. On or prior to the Initial Borrowing Date, (a) the Effective Date shall have occurred as provided in Section 13.10 and (b) there shall have been delivered to the Administrative Agent for the account of each of the Lenders that has requested same the appropriate Revolving Notes executed by the Borrowers and if requested by the Swingline Lender, the appropriate Swingline Notes executed by the Borrowers, in each case, in the amount, maturity and as otherwise provided herein.

6.02 <u>Officer's Certificate</u>. On the Initial Borrowing Date, the Administrative Agent shall have received a certificate, dated the Initial Borrowing Date and signed on behalf of US Company by the chairman of the board, the chief executive officer, the president or any vice president of US Company, certifying on behalf of US Company that all of the conditions have been satisfied on such date.

- 6.03 Opinions of Counsel. (a) On the Initial Borrowing Date, the Administrative Agent shall have received from Squire Sanders & Dempsey LLP, counsel to the Credit Parties, opinions addressed to the Administrative Agent, the Collateral Agent and each of the Lenders and dated the Initial Borrowing Date covering the matters (i) set forth in Exhibit I-1-A and such other matters incident to the transactions contemplated herein as the Administrative Agent may reasonably request and (ii) set forth in Exhibit I-1-B and such other matters incident to the transactions contemplated herein as the Administrative Agent may reasonably request.
- (b) On the Initial Borrowing Date, the Administrative Agent shall have received from White & Case LLP, counsel to the Credit Parties, an opinion addressed to the Administrative Agent, the Collateral Agent and each of the Lenders and dated the Initial Borrowing Date covering the matters set forth in Exhibit I-2 and such other matters incident to the transactions contemplated herein as the Administrative Agent may reasonably request.
- (c) On the Initial Borrowing Date, the Administrative Agent shall have received from Kirkland & Ellis LLP, counsel to the Target shareholders, an opinion addressed to the Administrative Agent, the Collateral Agent and each of the Lenders and dated the Initial Borrowing Date covering the matters set forth in Exhibit I-3 and such other matters incident to the transactions contemplated herein as the Administrative Agent may reasonably request.
- (d) On the Initial Borrowing Date, the Administrative Agent shall have received from Squire, Sanders & Dempsey, UK counsel to the Credit Parties, an opinion addressed to the Administrative Agent, the Collateral Agent and each of the Lenders and dated the Initial Borrowing Date covering the matters (i) set forth in Exhibit I-4-A and such other matters incident to the transactions contemplated herein as the Administrative Agent may reasonably request...
- (e) On the Initial Borrowing Date, the Administrative Agent shall have received from Linklaters LLP, Luxembourg counsel to the Credit Parties, an opinion addressed to the Administrative Agent, the Collateral Agent and each of the Lenders and dated the Initial Borrowing Date covering the matters set forth in Exhibit I-5 and such other matters incident to the transactions contemplated herein as the Administrative Agent may reasonably request.
- (f) On the Initial Borrowing Date, the Administrative Agent shall have received from Lang Michener LLP, Ontario and British Columbia counsel to the Credit Parties, an opinion addressed to the Administrative Agent, the Collateral Agent and each of the Lenders and dated the Initial Borrowing Date covering the matters set forth in Exhibit I-6 and such other matters incident to the transactions contemplated herein as the Administrative Agent may reasonably request.
- (g) On the Initial Borrowing Date, the Administrative Agent shall have received from Morris James LLP, special Delaware counsel, an opinion addressed to the Administrative Agent, the Collateral Agent, each of the Lenders and the other parties named therein, and dated the Initial Borrowing Date covering the matters set forth in Exhibit I-7 and such other matters incident to the transactions contemplated herein as the Administrative Agent may reasonably request.

- 6.04 <u>Company Documents; Proceedings; etc.</u> (a) On the Initial Borrowing Date, the Administrative Agent shall have received a certificate from each Credit Party, dated the Initial Borrowing Date, signed by the chairman of the board, the chief executive officer, the president, a director, or any vice president of such Credit Party, and attested to by the secretary or any assistant secretary of such Credit Party, in the form of Exhibit J with appropriate insertions, together with copies of the certificate and memorandum and articles of incorporation or association and by-laws (or other equivalent organizational documents), as applicable, of such Credit Party and the resolutions of such Credit Party referred to in such certificate, and each of the foregoing shall be in form and substance reasonably acceptable to the Administrative Agent.
- (b) On the Initial Borrowing Date, all limited liability company and legal proceedings and all instruments and agreements in connection with the transactions contemplated by this Agreement and the other Documents shall be reasonably satisfactory in form and substance to the Administrative Agent, and the Administrative Agent shall have received all information and copies of all documents and papers, including records of limited liability company proceedings, governmental approvals, good standing certificates (if applicable) and bring-down telegrams or facsimiles, if any, which the Administrative Agent reasonably may have requested in connection therewith, such documents and papers where appropriate to be certified by proper limited liability company or Governmental Authorities.
- 6.05 <u>Consummation of the Transactions; etc.</u> (a) On the Initial Borrowing Date and prior to the Mobile Storage Acquisition, US Company shall have purchased from the holders thereof all of the outstanding Mezzanine Notes.
- (b) On the Initial Borrowing Date, the Mobile Storage Acquisition (other than the portion thereof constituting the Subsequent Mergers) shall have been consummated in accordance with the terms and conditions of the Acquisition Documents and all Applicable Laws, and no provision of any Acquisition Document shall have been altered, amended or otherwise changed or supplemented or any condition therein waived if such alteration, amendment, change, supplement or waiver would be materially adverse to the interest of the Lenders, in any such case without the prior written consent of the Joint Bookrunners. On or prior to the Initial Borrowing Date, (i) the Administrative Agent shall have received true and correct copies of all Acquisition Documents, in each case certified as such by an Authorized Officer of US Company, and (ii) all such Acquisition Documents shall be in full force and effect.
- (c) On the Initial Borrowing Date, the Convertible Preferred Stock shall have been issued in accordance with the Convertible Preferred Stock Documents and all Applicable Laws and no provision of any Convertible Preferred Stock Document shall have been altered, amended or otherwise changed or supplemented or any condition therein waived if such alteration, amendment, change, supplement or waiver would be materially adverse to the interest of the Lenders, in any such case without the prior written consent of the Joint Bookrunners. On or prior to the Initial Borrowing Date, (i) the Administrative Agent shall have received true and correct copies of all Convertible Preferred Stock Documents, in each case certified as such by an Authorized Officer of US Company, and all such Convertible Preferred Stock Documents shall be in full force and effect.
- (d) On or prior to the Initial Borrowing Date, all Refinanced Indebtedness shall have been repaid in full, together with all fees and other amounts owing thereon, all commitments thereunder shall have been terminated and all letters of credit issued pursuant thereto shall have been terminated or cash collateralized in a manner reasonably acceptable to the Administrative Agent. On the Initial Borrowing Date, all security interests in respect of, and Liens securing, the Refinanced Indebtedness created pursuant to the security documentation relating thereto shall have been terminated and released, and the Administrative Agent shall have received all such releases as may have been requested by the

Administrative Agent, which releases shall be in form and substance reasonably satisfactory to the Administrative Agent. Without limiting the foregoing, there shall have been delivered to the Administrative Agent (i) proper termination statements (Form UCC-3 or the appropriate equivalent in each relevant jurisdiction) for filing under the UCC or equivalent statute or regulation of each relevant jurisdiction where a financing statement or application for registration (Form UCC-1 or the appropriate equivalent in each relevant jurisdiction) was filed with respect to US Company, Target or any of their respective Subsidiaries in connection with the Liens created with respect to the Refinanced Indebtedness, (ii) terminations or reassignments of any security interest in, or Lien on, any patents, trademarks, copyrights, or similar interests of US Company, Target or any of their respective Subsidiaries on which filings have been made and (iii) terminations of all mortgages, leasehold mortgages, hypothecs and deeds of trust created with respect to property of US Company or any of their respective Subsidiaries, in each case, to secure the obligations under the Refinanced Indebtedness, all of which shall be in form and substance reasonably satisfactory to the Administrative Agent. After giving effect to the Transactions, US Company and its Subsidiaries shall have no outstanding Indebtedness, Indebtedness convertible into Equity Interests, or other preferred Equity Interests except as set forth on Schedule 8.31.

6.06 No Acquisition Agreement Material Adverse Effect. There shall not have occurred any event, development or circumstance since December 31, 2007 that has caused or could reasonably be expected to cause a "Material Adverse Effect" as defined in the Merger Agreement (as in effect on February 22, 2008), with respect to the Target and its subsidiaries, taken as a whole.

6.07 No Material Adverse Effect on US Company. There shall not have occurred any event, development or circumstance since December 31, 2007 that has caused or could reasonably be expected to cause a material adverse effect on the business, operations, results of operations, assets, liabilities or financial condition of US Company and its Subsidiaries (excluding Target and its Subsidiaries as constituted immediately prior the Mobile Storage Acquisition), taken as a whole.

6.08 <u>US Pledge Agreement</u>. On the Initial Borrowing Date, each US Credit Party shall have duly authorized, executed and delivered the US Pledge Agreement in the form of <u>Exhibit K</u> (as amended, restated, modified and/or supplemented from time to time, the "<u>US Pledge Agreement</u>") and shall have delivered to the Collateral Agent, as pledgee thereunder, all of the US Pledge Agreement Collateral, if any, referred to therein and then owned by such US Credit Party, (a) endorsed in blank in the case of promissory notes constituting US Pledge Agreement Collateral and (b) together with executed and undated endorsements for transfer in the case of Equity Interests constituting certificated US Pledge Agreement Collateral, along with evidence that all other actions necessary or, in the reasonable opinion of the Collateral Agent, desirable, to perfect and protect the Liens purported to be created by the US Pledge Agreement have been taken and the US Pledge Agreement shall be in full force and effect.

6.09 <u>UK Share Charge</u>. On the Initial Borrowing Date, each US Subsidiary directly owning Securities in a UK Subsidiary shall have duly authorized, executed and delivered the UK Share Charge in the form of Exhibit L (as amended, restated, modified and/or supplemented from time to time, the "<u>UK Share Charge</u>") and shall have delivered to the Collateral Agent, as pledgee thereunder, all share certificates and blank stock transfer forms required to be delivered thereunder, along with evidence that all other actions necessary or, in the reasonable opinion of the Collateral Agent, desirable, to perfect and protect the Liens purported to be created by the UK Share Charge have been taken and the UK Share Charge shall be in full force and effect.

6.10 <u>US Security Agreement</u>. On the Initial Borrowing Date, each US Credit Party shall have duly authorized, executed and delivered the US Security Agreement in the form of Exhibit M, together with:

- (a) proper financing statements (Form UCC-1 or the equivalent) fully executed for filing under the UCC or other appropriate filing offices of each jurisdiction as may be necessary or, in the reasonable opinion of the Collateral Agent, desirable, to perfect the Liens purported to be created by the US Security Agreement;
- (b) certified copies of requests for information or copies (Form UCC-11), or equivalent reports as of a recent date, listing all effective financing statements that name US Company, Target or any of their respective Subsidiaries as debtor and that are filed in the jurisdictions referred to in clause (a) above and in such other jurisdictions in which Collateral is located on the Initial Borrowing Date, together with copies of such other financing statements that name US Company, Target or any of their respective Subsidiaries as debtor (none of which shall cover any of the Collateral except (i) to the extent evidencing Permitted Liens or (ii) those in respect of which the Collateral Agent shall have received termination statements (Form UCC-3) or such other termination statements as shall be required by local law fully executed for filing); provided that failure to obtain the results of such Lien searches shall not limit the availability of the Loans on the Initial Borrowing Date, so long as any assets that may be subject to such Lien searches are not included in the calculation of the US Borrowing Base; and
- (c) evidence of the completion of all other recordings and filings of, or with respect to, the US Security Agreement as may be necessary or, in the reasonable opinion of the Collateral Agent, desirable, to perfect and protect the Liens intended to be created by the US Security Agreement, including control agreements, access agreements and similar third party agreements as the Administrative Agent shall reasonably request, and the US Security Agreement shall be in full force and effect.
- 6.11 <u>UK Debenture</u>. On the Initial Borrowing Date, Mobile Storage Group, Inc., and each UK Subsidiary shall have duly authorized, executed and delivered the UK Debenture in the form of Exhibit N, together with all share certificates, blank stock transfer forms and notices of security to third parties required to be delivered thereunder along with that all other actions necessary or, in the reasonable opinion of the Collateral Agent, desirable, to perfect and protect the Liens intended to be created by the UK Debenture have been taken and the UK Debenture shall be in full force and effect.
- 6.12 <u>Financial Statements</u>; <u>Pro Forma Balance Sheet</u>; <u>Projections</u>; <u>etc.</u> (a) On or prior to the Initial Borrowing Date, the Arrangers shall have received (i) audited financial statements of each of US Company and its Subsidiaries and Target and its Subsidiaries for the fiscal year ended December 31, 2007, (ii) unaudited quarterly financial statements of US Company and its Subsidiaries and Target and its Subsidiaries for the most recent fiscal quarter ended at least 45 days prior to the Initial Borrowing Date, and (iii) unaudited monthly financial statements of US Company and its Subsidiaries and Target and its Subsidiaries for the most recent month ended at least 30 days prior to the Initial Borrowing Date, unless such month end shall also be a quarter end, in which case only the quarterly financial statements described in (ii) above need be delivered).
- (b) On the Initial Borrowing Date, the Administrative Agent shall have received a pro forma balance sheet and related statement of income of US Company and Target as of and for the twelve-month period ending the most recent quarter-end occurring at least 45 days prior to the Initial Funding Date in form reasonably satisfactory to the Arrangers which demonstrates a Debt Ratio of no more than 4.35:1.00.
 - 6.13 Solvency Certificate; Insurance Certificates. On the Initial Borrowing Date, the Administrative Agent shall have received:

- (a) a solvency certificate from the chief financial officer of US Company in the form of Exhibit O; and
- (b) certificates of insurance complying with the requirements of Section 9.03 for the business and properties of US Company and its Subsidiaries, in form and substance reasonably satisfactory to the Administrative Agent and naming the Collateral Agent as an additional insured and/or as loss payee, as applicable, and stating that such insurance shall not be canceled or materially revised without at least 30 days' prior written notice by the insurer to the Collateral Agent.
- 6.14 Fees, etc. On the Initial Borrowing Date, the Borrowers shall have paid to the Administrative Agent (and its relevant affiliates), the Collateral Agent and each Lender all costs, fees and expenses (including, without limitation, reasonable legal fees and expenses) and other compensation contemplated hereby and in the Fee Letter payable to the Administrative Agent (and/or its relevant affiliates), the Collateral Agent or such Lender to the extent then invoiced.
- 6.15 <u>Initial Borrowing Base Certificate</u>; <u>etc</u>. (a) On the Initial Borrowing Date, the Administrative Agent shall have received the initial Borrowing Base Certificate meeting the requirements of <u>Section 9.01(j)</u>, which shall be calculated after giving effect to the Transactions (and the Credit Events hereunder); and
- (b) On the Initial Borrowing Date, after giving effect to the Transactions (and the Credit Events hereunder), the Total Borrowing Availability shall equal or exceed \$200,000,000 as shown in the Borrowing Base Certificate delivered pursuant to (a) above.
- 6.16 No Defaults under Senior Note Indentures. There shall not exist (on a pro forma basis before and immediately after giving effect to the Transactions) (i) any default or event of default under the Mobile Storage Indenture or (ii) any event of default under the Mobile Mini Indenture, in each case relating to non-payment, bankruptcy, material judgments or negative covenants.
- 6.17 <u>Patriot Act</u>. On or prior to the Initial Borrowing Date, the Lenders shall have received from the Credit Parties, to the extent requested, all documentation and other information required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act.
- 6.18 Notice of Borrowing. Concurrent with the making of the Pre-Merger Loans, the Administrative Agent shall have received an irrevocable Notice of Borrowing for each other Loan to be made on the Initial Borrowing Date necessary to consummate the Mobile Storage Acquisition on the Closing Date and meeting the requirements of Section 2.03(a) provided that it is understood and agreed that the making of such additional loans will be subsequent to the making of the Pre-Merger Loans.

In determining the satisfaction of the conditions specified in this Section 6, to the extent any item is required to be satisfactory to any Lender, such item shall be deemed satisfactory to each Lender which has not notified the Administrative Agent in writing prior to the occurrence of the Initial Borrowing Date that the respective item or matter does not meet its satisfaction.

SECTION 7. <u>Conditions Precedent to All Credit Events</u>. The obligation of each Lender to make Loans (including Loans made on the Initial Borrowing Date), and the obligation of each Issuing Lender to issue Letters of Credit (including Letters of Credit issued on the Initial Borrowing

Date), are subject, at the time of each such Credit Event (except as hereinafter indicated), to the satisfaction of the following conditions:

7.01 No Default; Representations and Warranties. At the time of each such Credit Event and also after giving effect thereto (a) except with respect to the Borrowing on the Initial Borrowing Date, there shall exist no Default or Event of Default, (b) except as set forth in the immediately succeeding paragraph below, all representations and warranties contained herein and in the other Credit Documents shall be true and correct in all material respects with the same effect as though such representations and warranties had been made on the date of such Credit Event (it being understood and agreed that any representation or warranty which by its terms is made as of a specified date shall be required to be true and correct in all material respects only as of such specified date), other than, with respect to the Borrowing on the Initial Borrowing Date only, the representations set forth in Section 8.10(e), Section 8.19 (but only to the extent that such action, suit, proceeding or investigation relates to the Mobile Storage Acquisition), and Section 8.20, (c) the Aggregate Exposure shall not exceed the Total Revolving Loan Commitment, as then in effect, (d) the Aggregate US Exposure shall not exceed the US Maximum Amount, as then in effect, (e) the Aggregate Exposure shall not exceed the Aggregate Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate) (except in the case of Agent Advances), (g) the Aggregate US Exposure shall not exceed the US Borrowing Base at such time (based on the most recently delivered Borrowing Base Certificate) (except in the case of Agent Advances) and (h) the Aggregate UK Exposure shall not exceed the UK Borrowing Base at such time (based on the most recently delivered Borrowing Base Occurrently delivered Borrowing Base Certificate).

Notwithstanding the foregoing, the only representations concerning the Target and its Subsidiaries (as constituted prior to the Mobile Storage Acquisition) in the Credit Documents which shall be a condition to the making of Loans on the Initial Borrowing Date shall be (i) such of the representations concerning the Target and is Subsidiaries set forth in the Merger Agreement, but only to the extent that US Company has the right to terminate its obligations under the Merger Agreement as a result of the failure of such representations to be true and correct (determined without regard to whether any notice is required to be delivered) and (ii) the Specified Representations.

7.02 Notice of Borrowing; Letter of Credit Request. (a) Prior to the making of each Loan (other than a Swingline Loan or a Revolving Loan made pursuant to a Mandatory Borrowing), the Administrative Agent shall have received a Notice of Borrowing meeting the requirements of Section 2.03(a). Prior to the making of each Swingline Loan, the Swingline Lender shall have received the notice referred to in Section 2.03(b)(i) or (ii) as applicable.

(b) Prior to the issuance of each Letter of Credit, the Administrative Agent and the respective Issuing Lender shall have received a US Letter of Credit Request meeting the requirements of <u>Section 3.05</u> or a UK Letter of Credit Request meeting the requirements of <u>Section 3.06</u>.

The acceptance of the benefits of each Credit Event shall constitute a representation and warranty by each of US Company and the Borrowers to the Administrative Agent and each of the Lenders that all the conditions specified in Section 6 (with respect to Credit Events on the Initial Borrowing Date) and in this Section 7 (with respect to Credit Events on or after the Initial Borrowing Date) and applicable to such Credit Event are satisfied as of that time. All of the Notes, certificates, legal opinions and other documents and papers referred to in Section 6 and in this Section 7, unless otherwise specified, shall be delivered to the Administrative Agent at the Notice Office for the account of each of the Lenders and, except for the Notes, in sufficient counterparts or copies for each of the Lenders requesting same and shall be in form and substance reasonably satisfactory to the Administrative Agent and the Required Lenders.

SECTION 8. Representations, Warranties and Agreements. In order to induce the Agent and the Lenders to enter into this Agreement and to make the Loans, and issue (or participate in) the Letters of Credit as provided herein, each of US Company and each other Credit Party makes the following representations, warranties and agreements, in each case after giving effect to the Transactions, all of which shall survive the execution and delivery of this Agreement and the Notes and the making of the Loans and the issuance of the Letters of Credit, with the occurrence of each Credit Event on or after the Initial Borrowing Date being deemed to constitute a representation and warranty that the matters specified in this Section 8 are true and correct in all material respects on and as of the Initial Borrowing Date and on the date of each such other Credit Event (it being understood and agreed that any representation or warranty which by its terms is made as of a specified date shall be required to be true and correct in all material respects only as of such specified date).

8.01 <u>Organization and Qualification</u>. US Company and each other Credit Party is a corporation, limited partnership or limited liability company duly organized, validly existing and in good standing (or the foreign equivalent of "good standing", if any, in the relevant foreign jurisdiction) under the laws of the jurisdiction of its incorporation or organization. US Company and each other Credit Party is duly qualified and is authorized to do business and is in good standing (or the foreign equivalent of "good standing", if any, in the relevant foreign jurisdiction) as a limited liability company, limited partnership or corporation, as applicable, in each state or jurisdiction listed on <u>Schedule 8.01</u> hereto and in all other states and jurisdictions in which the failure of such Credit Party to be so qualified would reasonably be expected to have a Material Adverse Effect.

8.02 Power and Authority; No Violation. US Company and each other Credit Party (a) is duly authorized, has the capacity and is empowered to enter into, execute, deliver and perform this Agreement and each of the other Credit Documents to which it is a party and (b) has the power and authority to own its property and assets and to transact the business in which it is engaged and presently proposes to engage. The execution, delivery and performance of this Agreement and each of the other Credit Documents have been duly authorized by all necessary corporate or other relevant action and do not and will not (i) require any consent or approval of the shareholders of US Company or any of the shareholders, partners or members, as the case may be, of any other Credit Party other than such consents and approvals which have obtained prior to the Initial Borrowing Date; (ii) contravene US Company's or any other Credit Party's charter, articles or certificate of incorporation, partnership agreement, certificate of formation, by-laws, limited liability company agreement, operating agreement or other organizational documents (as the case may be); (iii) violate, or cause US Company or any other Credit Party to be in default under, any provision of any law, rule, regulation, order, writ, judgment, injunction, decree, determination or award in effect having applicability to US Company or any other Credit Party; (iv) result in a breach of or constitute a default under any indenture or loan or credit agreement or any other agreement, lease or instrument to which US Company or any other Credit Party is a party or by which it or its properties may be bound or affected; or (v) result in, or require, the creation or imposition of any Lien (other than Permitted Liens) upon or with respect to any of the properties now owned or hereafter acquired by US Company or any other Credit Party.

8.03 <u>Legally Enforceable Agreement</u>. This Agreement is, and each of the other Credit Documents when delivered under this Agreement will be, a legal, valid and binding obligation of each of US Company and each other Credit Party, in each case to the extent it is a party thereto, enforceable against it in accordance with its respective terms, except as limited by applicable bankruptcy or insolvency laws, and by general principles of equity.

8.04 <u>Capital Structure</u>. <u>Schedule 8.04</u> hereto states, as of the date hereof, (i) the correct name of each of the Subsidiaries of US Company, its jurisdiction of incorporation or organization and the percentage of its Voting Stock owned by US Company or a Subsidiary of US

Company, (ii) the name of each of US Company's and each other Credit Party's corporate or joint venture relationships and the nature of the relationship, (iii) the number and nature of all outstanding Securities of US Company and the number, nature and holder of Securities of each other Credit Party and (iv) the number of issued and treasury Securities of US Company. US Company and each other Credit Party has good title to all of the Securities it purports to own of each of such Subsidiaries, free and clear in each case of any Lien other than Permitted Liens. All such Securities have been duly issued and are fully paid and non-assessable. As of the date hereof, there are no outstanding options to purchase, or any rights or warrants to subscribe for, or any commitments or agreements to issue or sell any Securities or obligations convertible into, or any powers of attorney relating to any Securities of any of US Company's direct or indirect Subsidiaries. Except as set forth on Schedule 8.04, as of the date hereof, there are no outstanding agreements or instruments binding upon any of Company's or any other Credit Party's partners, members or shareholders, as the case may be, relating to the ownership of its Securities. On the Initial Borrowing Date, 100% of the Equity Interests of each Credit Party (other than US Company) are owned directly or indirectly by US Company.

8.05 Names. Neither US Company nor any other Credit Party has been known as or has used any legal, fictitious or trade names except those listed on Schedule 8.05 hereto as such Schedule may be amended in connection with a Permitted Acquisition. Except pursuant to the Transactions or in connection with an Acquisition permitted hereunder consummated after the date hereof, neither US Company nor any other Credit Party has been the surviving entity of a merger or consolidation or has acquired all or substantially all of the assets of any Person. US Company's and each other Credit Party's respective states of incorporation or organization, type of organization and organizational identification number are set forth on Schedule 8.04 or Schedule 8.05, as such Schedule may be amended in connection with a Permitted Acquisition. The respective exact legal names of US Company and each other Credit Party are set forth on Schedule 8.04, as such Schedule may be amended in connection with a Permitted Acquisition.

8.06 <u>Business Locations</u>; <u>Agent for Process</u>. Each of US Company's and each other Credit Party's chief executive office and other places of business are as listed on <u>Schedule 8.06</u> hereto, as updated from time to time by US Company. During the preceding one-year period, neither US Company nor any other Credit Party has had an office or place of business other than as listed on <u>Schedule 8.06</u>. All tangible Collateral is and will at all times be kept by US Company and each other Credit Party in accordance with the applicable Security Agreement. Except as shown on <u>Schedule 8.06</u>, as of the date hereof, no Inventory is stored with a bailee, distributor, warehouseman or similar party, nor is any Inventory consigned to any Person.

8.07 <u>Title to Properties</u>; <u>Priority of Liens</u>. Except with respect to the fact that the LKE Joint Accounts are jointly owned by US Company or one of its Domestic Subsidiaries and the LKE Qualified Intermediary pursuant to the terms of the LKE Master Exchange, US Company and each other Credit Party has good record, indefeasible and marketable title to and fee simple ownership of real property owned by it, or valid leasehold interests in, all of its leased real property, and good title to all of the Collateral and all of its other Property, in each case, free and clear of all Liens except Permitted Liens. US Company and each other Credit Party has paid or discharged all lawful claims which, if unpaid, might become a Lien against any of US Company's or such Credit Party's Properties that is not a Permitted Lien. The Liens granted to the Collateral Agent under the Security Documents are First Priority Liens, subject only to Permitted Liens.

8.08 <u>Accounts</u>. Agent may rely, in determining which Accounts are Eligible Accounts, on all statements and representations made by US Company with respect to any Account or Accounts of US Company or any other Credit Party. With respect to each of such Accounts, whether or not such Account is an Eligible Account, unless otherwise disclosed to Agent in writing:

- (a) It is genuine and in all respects what it purports to be, and it is not evidenced by a judgment;
- (b) It arises out of a completed, bona fide sale and delivery of goods or rendition of services by US Company or the applicable Credit Party, in the ordinary course of its business and in accordance with the terms and conditions of all purchase orders, contracts or other documents relating thereto and forming a part of the contract between US Company or the applicable Credit Party and the Account Debtor and the Account Debtor is not an Affiliate of US Company or any other Credit Party;
 - (c) It is for a liquidated amount maturing as stated in the invoice covering such sale or rendition of services;
- (d) There are no facts, events or occurrences which in any way impair the validity or enforceability of any Accounts or tend to reduce the amount payable thereunder from the face amount of the invoice and statements delivered or made available to Agent with respect thereto;
- (e) To US Company's knowledge, the Account Debtor thereunder (1) had the capacity to contract at the time any contract or other document giving rise to the Account was executed and (2) such Account Debtor is Solvent; and
- To US Company's knowledge, there are no proceedings or actions which are threatened or pending against the Account Debtor thereunder which might result in any material adverse change in such Account Debtor's financial condition or the collectibility of such Account (other than non-material disputes involving de minimis amounts arising in the ordinary course of business).
- 8.09 <u>Equipment</u>. The Equipment of US Borrower and each other Credit Party is in good operating condition and repair, ordinary wear and tear excepted.
- 8.10 Financial Statements; Financial Condition; Undisclosed Liabilities; Projections. (a) (i) The audited Consolidated balance sheet of US Company as at the last day of its fiscal year ended December 31, 2007 and the related Consolidated statements of income, retained earnings, stockholders' equity and cash flows of US Company for its fiscal year ended on such date, copies of which have been furnished to the Lenders prior to the Initial Borrowing Date, present fairly in all material respects the consolidated financial position of US Company at the date of said financial statements and the Consolidated results of its operations for the respective period covered thereby. The unaudited Consolidated balance sheet of US Company as at the last day of its fiscal quarter ended March 31, 2008 and the related Consolidated statements of income, retained earnings, stockholders' equity and cash flows of US Company for the three-month period ended on such date, copies of which in each case have been furnished to the Lenders prior to the Initial Borrowing Date, present fairly in all material respects the Consolidated financial condition of US Company at the date of said financial statements and the Consolidated results of its operations for the period covered thereby. All of the foregoing historical financial statements have been prepared in accordance with GAAP consistently applied except to the extent provided in the notes to said financial statements and subject, in the case of the unaudited financial statements, to normal year-end audit adjustments (all of which are of a recurring nature and none of which, individually or in the aggregate, would be material) and the absence of footnotes.
- (ii) The audited Consolidated balance sheet of Target as at the last day of its fiscal year ended December 31, 2007 and the related Consolidated statements of income, retained earnings, stockholders' equity and cash flows of Target for its fiscal year ended on such date, copies of which have been furnished to the Lenders prior to the Initial Borrowing Date, present fairly in all material respects the

Consolidated financial position of Target at the date of said financial statements and the Consolidated results of its operations for the respective period covered thereby. The unaudited Consolidated balance sheet of Target as at the last day of its fiscal quarter ended March 31, 2008 and the related consolidated statements of income, retained earnings, stockholders' equity and cash flows of Target for the three-month period ended on such date, copies of which in each case have been furnished to the Lenders prior to the Initial Borrowing Date, present fairly in all material respects the Consolidated financial condition of Target at the date of said financial statements and the Consolidated results of its operations for the period covered thereby. All of the foregoing historical financial statements have been prepared in accordance with GAAP consistently applied except to the extent provided in the notes to said financial statements and subject, in the case of the unaudited financial statements, to normal year-end audit adjustments (all of which are of a recurring nature and none of which, individually or in the aggregate, would be material) and the absence of footnotes.

- (iii) The <u>pro_forma</u> Consolidated financial statements of US Company and its Subsidiaries as of March 31, 2008 (after giving effect to the Transactions), copies of which have been furnished to the Lenders prior to the Initial Borrowing Date, present a good faith estimate of both the <u>pro_forma</u> Consolidated financial condition of US Company and its Subsidiaries as of such dates and the <u>pro_forma</u> Consolidated results of operations of Company and its Subsidiaries for the periods covered thereby.
- (b) On and as of the Initial Borrowing Date, and after giving effect to the Transactions and to all Indebtedness (including the Loans) being incurred or assumed and Liens created by the Credit Parties in connection therewith, US Company and each of its Subsidiaries is or are Solvent.
- (c) Except as disclosed in the financial statements delivered pursuant to Section <u>8.10(a)</u>, and except purchase accounting reserves required pursuant to GAAP and except for the Indebtedness incurred under this Agreement, there were as of the Initial Borrowing Date no liabilities or obligations with respect to US Company or any of its Subsidiaries of any nature whatsoever (whether absolute, accrued, contingent or otherwise and whether or not due) known to US Company or any of its Subsidiaries (which, either individually or in the aggregate, could reasonably be expected to be material to US Company and its Subsidiaries taken as a whole. As of the Initial Borrowing Date, US Company does not know of any basis for the assertion against it or any of its Subsidiaries of any liability or obligation of any nature whatsoever that is not fully disclosed in the financial statements delivered pursuant to Section 8.10(a) or referred to in the immediately preceding sentence which, either individually or in the aggregate, could reasonably be expected to be material to US Company and its Subsidiaries taken as a whole.
- (d) The Projections delivered to the Administrative Agent and the Lenders prior to the Initial Borrowing Date have been prepared in good faith and are based on reasonable assumptions, and there are no statements or conclusions in the Projections which are based upon or include information known to US Company or its Subsidiaries to be misleading in any material respect or which fail to take into account material information known to US Company or its Subsidiaries regarding the matters reported therein. On the Initial Borrowing Date, US Company and its Subsidiaries believe that the Projections are reasonable and attainable, it being recognized by the Lenders, however, that projections as to future events are not to be viewed as facts and that the actual results during the period or periods covered by the Projections may differ from the projected results and such differences may be material.
- (e) After giving effect to the Transactions, since December 31, 2007, nothing has occurred that has had, or could reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect with respect to the US Company and its Subsidiaries, taken as a whole.

- (f) As of the date hereof, the fiscal year of US Company and each of its Subsidiaries ends on December 31 of each year.
- 8.11 <u>Full Disclosure</u>. The financial statements referred to in <u>Section 8.10</u> hereof do not, nor does this Agreement or any other written statement of US Company to Administrative Agent or any Lender, contain any untrue statement of a material fact or omit a material fact necessary to make the statements contained therein or herein not misleading. There is no fact which US Company has failed to disclose to Administrative Agent or any Lender in writing which would reasonably be expected to have a Material Adverse Effect.
- 8.12 <u>Surety Obligations</u>. Except as set forth on <u>Schedule 8.12</u>, as of the date hereof, neither US Company nor any other Credit Party is obligated as surety or indemnitor under any surety or similar bond or other contract issued or entered into to assure payment, performance or completion of performance of any undertaking or obligation of any Person.
- 8.13 Tax Returns and Payments. Each of US Company and each of its Subsidiaries has timely filed or caused to be timely filed with the appropriate Governmental Authority all returns, statements, forms and reports for taxes (the "Returns") required to be filed by, or with respect to the income, properties or operations of, US Company and/or any of its Subsidiaries. The Returns accurately reflect all liability for taxes of US Company and its Subsidiaries, as applicable, for the periods covered thereby. Each of US Company and each of its Subsidiaries has paid all taxes and assessments payable by it which have become due, other than those that are being contested in good faith and adequately disclosed and for which adequate reserves have been established in accordance with GAAP. There is no action, suit, proceeding, investigation, audit or claim now pending or, to the knowledge of US Company or any other Credit Party, threatened by any Governmental Authority regarding any taxes relating to US Company or any of its Subsidiaries. Neither US Company nor any of its Subsidiaries has incurred, nor will any of them incur, any tax liability in connection with the Transactions or any other transactions contemplated hereby (it being understood that the representation contained in this sentence does not cover any future tax liabilities of US Company or any of its Subsidiaries arising as a result of the operation of their businesses in the ordinary course of business).
- 8.14 <u>Dutch Credit Parties and Luxembourg Subsidiary</u>. (a) Each of the Dutch Credit Parties conducts (and shall conduct) no operations and has (and shall have) no assets and no liabilities, in each case, individually or in the aggregate, with a fair market value in excess of the Equivalent Amount of €3,000,000, other than in connection with its Obligations hereunder and other than, with respect to Mobile Mini Holding B.V. only, Equity Interests of Mobile Mini B.V.
- (b) The Luxembourg Subsidiary conducts no operations and has no liabilities or assets other than in connection with the Luxembourg Debt (and shall not conduct any operations or have liabilities or assets) other than in connection with the Luxembourg Debt and in connection with its Obligations hereunder.
- 8.15 Intellectual Property, etc. Each of US Company and each of its Subsidiaries owns or has the right to use all the patents, trademarks, permits, domain names, service marks, trade names, copyrights, licenses, franchises, inventions, trade secrets, proprietary information and know-how of any type, whether or not written (including, but not limited to, rights in computer programs and databases) and formulas, or rights with respect to the foregoing, and has obtained assignments of all leases, licenses and other rights of whatever nature, necessary for the present conduct of its business, without any known conflict with the rights of others which, or the failure to own or have which, as the case may be, could reasonably be expected, either individually or in the aggregate, to have a Material Adverse Effect.

- 8.16 Government Consents. US Company and each of its Subsidiaries has, and is in good standing (or the foreign equivalent of "good standing", if any, in the relevant foreign jurisdiction) with respect to, all governmental consents, approvals, licenses, authorizations, permits, certificates, inspections and franchises necessary to continue to conduct its business as heretofore or proposed to be conducted by it and to own or lease and operate its Properties as now owned or leased by it.
- 8.17 <u>Compliance with Laws</u>. US Company and each of its Subsidiaries has duly complied in all material respects with, and its Properties, business operations and leaseholds are in compliance in all material respects with, the provisions of all federal, state, local, foreign and other laws, rules and regulations applicable to US Company or such Subsidiary, as applicable, its Properties or the conduct of its business, and there have been no citations, notices or orders of noncompliance issued to US Company or any of its Subsidiaries under any such law, rule or regulation. US Company and each of its Subsidiaries has established and maintains an adequate monitoring system to insure that it remains in compliance in all material respects with all federal, state, local, foreign and other rules, laws and regulations applicable to it. No Inventory has been produced by US Company or any of its Subsidiaries in violation of the Fair Labor Standards Act (29 USC. §201 et seq.), as amended.
- 8.18 <u>Restrictions</u>. Neither US Company nor any other Credit Party is a party or subject to any contract or agreement which restricts its right or ability to incur Indebtedness, other than as set forth on <u>Schedule 8.18</u> hereto, none of which prohibit the execution of or compliance with this Agreement or the other Credit Documents by US Company or any other Credit Party, as applicable. Except as permitted in this Agreement, none of the Collateral is subject to contractual obligations that may restrict or inhibit Agent's rights or abilities to sell or dispose of the Collateral or any part thereof after the occurrence and during the continuance of an Event of Default.
- 8.19 <u>Litigation</u>. Except as set forth on <u>Schedule 8.19</u> hereto, there are no actions, suits, proceedings or investigations pending, or to the knowledge of US Company or any other Credit Party, threatened, against or involving US Company or any of its Subsidiaries, or the business, operations, Properties, prospects, profits or condition of US Company or any of its Subsidiaries which, singly or in the aggregate, would reasonably be expected to have a Material Adverse Effect. Neither US Company nor any of its Subsidiaries is in default with respect to any order, writ, injunction, judgment, decree or rule of any court, governmental authority or arbitration board or tribunal, which, singly or in the aggregate, would reasonably be expected to have a Material Adverse Effect.
- 8.20 No Defaults. No event has occurred and no condition exists which would, upon or after the execution and delivery of this Agreement or each Credit Party's performance hereunder, constitute a Default or an Event of Default. Neither US Company nor any other Credit Party is in default in (and no event has occurred and no condition exists which constitutes, or which the passage of time or the giving of notice or both would constitute, a default in) the payment of any Indebtedness to any Person in excess of the lesser of \$5,000,000 or the Equivalent Amount thereof or that amount which would have a Material Adverse Effect. All Obligations are permitted under the Senior Note Indentures and the other Senior Note Documents.
- 8.21 <u>Leases</u>. <u>Schedule 8.21</u> hereto is a complete listing of all capitalized and operating personal property leases of US Company and the other Credit Parties and all real property leases of US Company and the other Credit Parties. US Company and each other Credit Party is in full compliance with all of the terms of each of its respective capitalized and operating leases, except where the failure to so comply would not reasonably be expected to have a Material Adverse Effect.

8.22 [Intentionally Omitted].

8.23 <u>Use of Proceeds; Margin Regulations</u>. (a) All proceeds of the Loans will be used for the working capital, Capital Expenditures, Permitted Acquisitions and general corporate purposes of US Company and its Subsidiaries, provided that the proceeds of the Loans shall not be used for purposes which would constitute unlawful financial assistance for the purposes of Sections 151 to 158 of the United Kingdom Companies Act of 1985 (as amended or otherwise re-enacted from time to time).

(b) No part of any Credit Event (or the proceeds thereof) will be used to purchase or carry any Margin Stock or to extend credit for the purpose of purchasing or carrying any Margin Stock. Neither the making of any Loan nor the use of the proceeds thereof nor the occurrence of any other Credit Event will violate or be inconsistent with the provisions of Regulation T, U or X.

8.24 Compliance with ERISA. (a) Schedule 8.24 sets forth a listing of each Plan as of the Initial Borrowing Date. Except as could not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect: (i) each Plan (and each related trust, insurance contract or fund) is in compliance with its terms and with all Applicable Laws, including without limitation ERISA and the Code; (ii) each Plan (and each related trust, if any) which is intended to be qualified under Section 401(a) of the Code has received an effective determination letter from the Internal Revenue Service to the effect that it meets the requirements of Sections 401(a) and 501(a) of the Code or is comprised of a master or prototype plan that is the subject of an effective favorable opinion letter from the Internal Revenue Service; (iii) no Reportable Event has occurred; (iv) no Multiemployer Plan is insolvent (as defined under ERISA), is in reorganization (as defined under ERISA) or has been terminated (within the meaning of Title IV of ERISA); (v) no Plan has an Unfunded Current Liability; (vi) no Plan which is subject to Section 412 of the Code or Section 302 of ERISA has an accumulated funding deficiency, within the meaning of such sections of the Code or ERISA, or has applied for or received a waiver of an accumulated funding deficiency or an extension of any amortization period, within the meaning of Section 412 of the Code or Section 303 or 304 of ERISA; (vii) all contributions required to be made with respect to a Plan and Multiemployer Plan each have been timely made; (viii) neither US Owner nor any Subsidiary of US Owner nor any ERISA Affiliate has incurred any liability (including any indirect, contingent or secondary liability) to or on account of a Plan pursuant to Section 406, 409, 502(i), 502(i), 4062, 4063, 4064 or 4069 of ERISA or Section 401(a)(29), 4971 or 4975 of the Code, or to or on account of a Multiemployer Plan pursuant to Section 515, 4201, 4204 or 4212 of ERISA or expects to incur any such liability under any of the foregoing sections with respect to any Plan or Multiemployer Plan; (ix) no event has occurred or condition exists which presents a risk to US Company or any Subsidiary of US Company or any ERISA Affiliate of incurring a liability to or on account of a Plan or Multiemployer Plan pursuant to the foregoing provisions of ERISA and the Code; (x) no proceedings have been instituted to terminate or appoint a trustee to administer any Plan which is subject to Title IV of ERISA; (xi) no action, suit, proceeding, hearing, audit or investigation with respect to the administration, operation or the investment of assets of any Plan (other than routine claims for benefits) is pending, expected or threatened; (xii) there has been no violation of the applicable requirements of Section 404 or 405 of ERISA or the exclusive benefit rule under Section 401(a) of the Code by any fiduciary or disqualified person with respect to any Plan for which US Company or any Subsidiary of US Company may be directly or indirectly liable; (xiii) neither US Company nor any Subsidiary of US Company has filed, or is considering filing, an application under the Internal Revenue Service Employee Plans Compliance Resolution System (the "EPCRS") or the Department of Labor's Voluntary Fiduciary Correction Program (the "YFCP") with respect to any Plan; (xiv) to the knowledge of US Company and the Credit Parties, no plan administrator or a "plan official" (as defined under VFCP) of any Multiemployer Plan has filed, or is considering filing, an application under the EPCRS or the VFCP with respect to any Multiemployer Plan; (xv) using actuarial assumptions and computation methods consistent

with Part 1 of subtitle E of Title IV of ERISA, US Company and its Subsidiaries and its ERISA Affiliates would not incur any liabilities with respect to any Multiemployer Plans in the event of a complete withdrawal therefrom; (xvi) each group health plan (as defined in Section 607(1) of ERISA or Section 4980B(g)(2) of the Code) which covers or has covered employees or former employees of US Company, any Subsidiary of US Company, or any ERISA Affiliate has at all times been operated in compliance with the provisions of Part 6 of subtitle B of Title I of ERISA and Section 4980B of the Code; (xvii) each group health plan (as defined in 45 Code of Federal Regulations Section 160.103) which covers or has covered employees or former employees of US Company or any Subsidiary of US Company has at all times been operated in compliance with the provisions of the Health Insurance Portability and Accountability Act of 1996 and the regulations promulgated thereunder; (xviii) no lien imposed under the Code or ERISA on the assets of US Company or any Subsidiary of US Company or any ERISA Affiliate exists on account of any Plan or Multiemployer Plan and no event or condition has occurred or exists that could reasonably be expected to result in the imposition of any such lien; (xix) US Company and its Subsidiaries may cease contributions to or terminate any Plan maintained by any of them without incurring any liability; and (xx) US Company and its Subsidiaries do not maintain or contribute to any employee welfare benefit plan (as defined in Section 3(1) of ERISA) which provides benefits to retired employees or other former employees (other than as required by Section 601 of ERISA) or any Plan the obligations with respect to which could reasonably be expected to have a Material Adverse Effect on the ability of US Company and its Subsidiaries to perform their obligations under the Credit Documents to which they are a party.

(b) Except as could not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect: (i) each Foreign Pension Plan has been maintained in compliance with its terms and with the requirements of any and all Applicable Laws, statutes, rules, regulations and orders and has been maintained, where required, in good standing with applicable regulatory authorities; (ii) all contributions required to be made with respect to a Foreign Pension Plan have been timely made; neither US Company nor any of its Subsidiaries has incurred any obligation in connection with the termination of, or withdrawal from, any Foreign Pension Plan; and (iii) the present value of the accrued benefit liabilities (whether or not vested) under each Foreign Pension Plan, determined as of the end of US Company's most recently ended fiscal year on the basis of actuarial assumptions, each of which is reasonable, did not exceed the current value of the assets of such Foreign Pension Plan allocable to such benefit liabilities.

8.25 <u>Trade Relations</u>. Except as set forth on <u>Section 8.25</u>, there exists no actual or, to any Credit Party's knowledge, threatened termination, cancellation or limitation of, or any modification or change in, the business relationship between US Company or any other Credit Party and any customer or any group of customers whose purchases individually or in the aggregate are material to the business of US Company and the other Credit Parties (taken as a whole), or with any material supplier, except in each case, where the same would not reasonably be expected to have a Material Adverse Effect, and there exists no present condition or state of facts or circumstances which would prevent US Company or any other Credit Party from conducting such business after the consummation of the transaction contemplated by this Agreement in substantially the same manner in which it has heretofore been conducted.

8.26 Security Documents. (a) The provisions of the Security Agreements are effective to create in favor of the Collateral Agent for the benefit of the Secured Creditors a legal, valid and enforceable security interest in all right, title and interest of the Credit Parties in all of the Security Agreement Collateral, and the Collateral Agent, for the benefit of the Secured Creditors, has (or within 10 days following the Initial Borrowing Date will have) a fully perfected security interest in all right, title and interest in all of the Security Agreement Collateral described therein, subject to no other Liens other than Permitted Liens. The recordation of (i) the Grant of Security Interest in US Patents and

- (ii) the Grant of Security Interest in US Trademarks in the respective form attached to the US Security Agreement, in each case in the United States Patent and Trademark Office, together with filings on Form UCC-1 made pursuant to the Security Agreement, will create, to the extent as may be perfected by such filings and recordation, a perfected security interest in the United States trademarks and patents covered by the Security Agreement, and the recordation of the Grant of Security Interest in US Copyrights in the form attached to the US Security Agreement with the United States Copyright Office, together with filings on Form UCC-1 made pursuant to the US Security Agreement, to the extent as may be perfected by such filings and recordation, a perfected security interest in the United States copyrights covered by the US Security Agreement.
- (b) The Liens created under the US Pledge Agreement in favor of the Collateral Agent, as pledgee thereunder, for the benefit of the Secured Creditors, constitute perfected Liens in all US Pledge Agreement Collateral, subject to no Liens of any other Person. No filings or recordings are required in order to perfect (or maintain the perfection or priority of) the Liens created in the US Pledge Agreement Collateral under the US Pledge Agreement other than with respect to that portion of the US Pledge Agreement Collateral constituting General Intangibles under the UCC.
- (c) Each Mortgage creates, as security for the obligations purported to be secured thereby, a valid and enforceable perfected security interest in and mortgage lien on the respective Mortgaged Property in favor of the Collateral Agent (or such other trustee as may be required or desired under local law) for the benefit of the Secured Creditors, superior and prior to the rights of all third Persons (except that the security interest and mortgage lien created on such Mortgaged Property may be subject to the Permitted Encumbrances related thereto) and subject to no other Liens.
- 8.27 <u>Investment Company Act</u>. Neither US Company nor any of its Subsidiaries is an "investment company" or a company "controlled" by an "investment company," within the meaning of the Investment Company Act of 1940, as amended.
- 8.28 <u>Representations and Warranties in Other Documents</u>. All representations and warranties set forth in the other Credit Documents were true and correct in all material respects at the time as of which such representations and warranties were made (or deemed made) and shall be true and correct in all material respects as of the Initial Borrowing Date as if such representations or warranties were made on and as of such date (it being understood and agreed that any such representation or warranty which by its terms is made as of a specified date shall be true and correct in all material respects as of such specified date).
- 8.29 Environmental Matters. (a) Each of US Company and each of its Subsidiaries is in compliance in material respects with all applicable Environmental Laws and the requirements of any permits issued under such Environmental Laws. There are no pending or, to the knowledge of US Company or any other Credit Party, threatened Environmental Claims against US Company or any of its Subsidiaries or any Real Property owned, leased or operated by US Company or any of its Subsidiaries of any Real Property formerly owned, leased or operated by US Company or any of its Subsidiaries but no longer owned, leased or operated by US Company or any of its Subsidiaries). There are no facts, circumstances, conditions or occurrences with respect to the business or operations of US Company or any of its Subsidiaries, or any Real Property owned, leased or operated by US Company or any of its Subsidiaries (including any Real Property formerly owned, leased or operated by US Company or any of its Subsidiaries) or, to the knowledge of US Company or any other Credit Party, any property adjoining or adjacent to any such Real Property that could be reasonably expected (i) to form the basis of an Environmental Claim against US Company or any of its Subsidiaries or any Real Property owned,

leased or operated by US Company or any of its Subsidiaries or (ii) to cause any Real Property owned, leased or operated by US Company or any of its Subsidiaries to be subject to any restrictions on the ownership, lease, occupancy or transferability of such Real Property by US Company or any of its Subsidiaries under any applicable Environmental Law.

- (b) To the knowledge of US Company or any of its Subsidiaries, Hazardous Materials have not at any time been generated, used, treated or stored on, or transported to or from, or Released on or from, any Real Property owned, leased or operated by US Company or any of its Subsidiaries or, to the knowledge of US Company or any other Credit Party, any property adjoining or adjacent to any Real Property, where such generation, use, treatment, storage, transportation or Release has violated or could be reasonably expected to violate any applicable Environmental Law or give rise to an Environmental Claim.
- (c) Notwithstanding anything to the contrary in this <u>Section 8.29</u>, the representations and warranties made in this <u>Section 8.29</u> shall be untrue only if the effect of any or all conditions, violations, claims, restrictions, failures and noncompliances of the types described above could, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 8.30 Employment and Labor Relations. Neither US Company nor any of its Subsidiaries is engaged in any unfair labor practice that could, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. There is (a) no unfair labor practice complaint pending against US Company or any of its Subsidiaries or, to the knowledge of US Company or any other Credit Party, threatened against any of them, before the National Labor Relations Board, and no grievance or arbitration proceeding arising out of or under any collective bargaining agreement is so pending against US Company or any of its Subsidiaries or, to the knowledge of US Company or any other Credit Party, threatened against any of them, (b) no strike, labor dispute, slowdown or stoppage pending against US Company or any of its Subsidiaries or, to the knowledge of US Company or any other Credit Party, threatened against US Company or any of its Subsidiaries, (c) no union representation question exists with respect to the employees of US Company or any of its Subsidiaries, (d) no equal employment opportunity charges or other claims of employment discrimination are pending or, to the knowledge of US Company or any other Credit Party, threatened against US Company or any of its Subsidiaries, and (e) no wage and hour department investigation has been made of US Company or any of its Subsidiaries, except (with respect to any matter specified in clauses (a) (e) above, either individually or in the aggregate) such as could not reasonably be expected to have a Material Adverse Effect.
- 8.31 <u>Indebtedness</u>. <u>Schedule 8.31</u> sets forth a list of all Indebtedness (including Contingent Obligations) of US Company and its Subsidiaries as of the Initial Borrowing Date and which is to remain outstanding after giving effect to the Transaction (excluding the Obligations), in each case showing the aggregate principal amount thereof and the name of the respective borrower and any Credit Party or any of its Subsidiaries which directly or indirectly guarantees such debt (all such non-excluded Indebtedness, the "<u>Existing Indebtedness</u>").
- 8.32 <u>Insurance</u>. <u>Schedule 8.32</u> sets forth a listing of all insurance maintained by US Company and its Subsidiaries as of the Initial Borrowing Date, with the amounts insured (and any deductibles) set forth therein.
- 8.33 <u>Employee Benefit Plans; Non-Compete Agreements; Collective Bargaining Agreements; Existing Indebtedness Agreements</u>. <u>Schedule 8.33</u> sets forth a list of all Employee Benefit Plans, Non-Compete Agreements, Collective Bargaining Agreements and Existing Indebtedness Agreements maintained by US Company and its Subsidiaries as of the Initial Borrowing Date and which are to remain in effect after giving effect to the Transactions.

- 8.34 <u>Anti-Terrorism Laws</u>. US Company and its Subsidiaries are in compliance with the Uniting and Strengthening of America by Providing the Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001 (the "Patriot Act").
- 8.35 <u>UK Financial Assistance</u>. Neither the execution, delivery and performance of this Agreement (including without limitation the guarantee in Section 15) and each of the other Credit Documents and Senior Note Documents to which it is a party nor the incurrence of any obligations or liabilities (actual or contingent) thereunder by any of the UK Subsidiaries constitutes or will constitute unlawful financial assistance for the purposes of sections 151 to 158 (inclusive) of the United Kingdom Companies Act 1985 (as amended or otherwise re-enacted from time to time).
- 8.36 <u>UK Pensions</u>. To the knowledge of US Company or any of its Subsidiaries, no UK Subsidiary has ever participated in a UK defined benefit pension plan or been associated or connected with the employer in relation to a UK defined benefit pension plan.
- SECTION 9. <u>Affirmative Covenants</u>. Each of US Company and each other Credit Party hereby covenants and agrees that on and after the Initial Borrowing Date and until the Total Revolving Loan Commitment and all Letters of Credit have terminated and the Loans, Notes and Unpaid US Drawings and Unpaid UK Drawings (in each case together with interest thereon), Fees and all other Obligations (other than indemnities described in <u>Section 13.13</u> which are not then due and payable) incurred hereunder and thereunder, are paid in full in cash:
 - 9.01 Information Covenants. US Company will furnish to each Lender:
 - (a) Monthly Reports. As soon as available, but not later than 30 days after the end of each month hereafter, excluding the last month of each fiscal quarter of US Company's, unaudited interim financial statements (including, but not limited to, balance sheet, income statement and statement of cash flows) of US Company and its Subsidiaries as of the end of such month and of the portion of the fiscal year then elapsed, on a Consolidated basis, certified by the principal financial officer or principal accounting officer of US Company as prepared in accordance with GAAP and fairly presenting in all material respects the financial position and results of operations of US Company and its Subsidiaries for such month and period (subject only to changes from audit and year-end adjustments and except that such statements need not contain notes) and, at the request of the Administrative Agent, unaudited interim financial statements on a consolidating basis (A) with respect to US Company and its Domestic Subsidiaries, on the one hand and (B) the US Company's Foreign Subsidiaries, on the other hand, in each case in a form consistent with US Company's historical practices of preparation of consolidating financial statements.
 - (b) <u>Quarterly Financial Statements</u>. As soon as available, but not later than 45 days after the end of each fiscal quarter of US Company, excluding the last quarter of US Company's fiscal year, unaudited quarterly financial statements (including, but not limited to, balance sheet, income statement and statement of cash flows) of US Company and its Subsidiaries as of the end of such fiscal quarter, on a Consolidated basis, certified by the principal financial officer or principal accounting officer of US Company as prepared in accordance with GAAP and fairly presenting in all material respects the financial position and results of operations of US Company and its Subsidiaries for such fiscal quarter and period (subject only to changes from audit and year-end adjustments and except that such statements need not contain notes) and, at the request of the Administrative Agent, unaudited interim financial statements on a consolidating basis (A) with respect to US Company and its Domestic Subsidiaries, on the one hand and (B) the US

Company's Foreign Subsidiaries, on the other hand, in each case in a form consistent with US Company's historical practices of preparation of consolidating financial statements. In addition, at the time of the delivery of the foregoing financial statements, US Company will deliver to the Collateral Agent such information and/or documents relating to intellectual property of the Credit Parties as may be required pursuant to the terms of the Security Agreements.

(c) <u>Annual Financial Statements</u>. As soon as available, but not later than 90 days after the close of each fiscal year of US Company, unqualified (except for a qualification for a change in accounting principles with which the accountant concurs) audited financial statements (including, but not limited to, balance sheet, income statement and statement of cash flows) of US Company and its Subsidiaries as of the end of such year, on a Consolidated basis, certified by a firm of independent certified public accountants of recognized standing selected by US Company but reasonably acceptable to Administrative Agent, together with, at the request of the Administrative Agent, unaudited consolidating balance sheets, income statements and statements of cash flows (A) with respect to US Company and its Domestic Subsidiaries, on the one hand and (B) the US Company's Foreign Subsidiaries, on the other hand, and, within a reasonable time thereafter a copy of any management letter issued in connection therewith.

(d) [Intentionally Omitted].

- (e) Compliance Certificates. Concurrently with the delivery of the financial statements described in clause (c) of this Section 9.01, US Company shall forward to Agent a copy of the accountants' letter to US Company's management (if any) that is prepared in connection with such financial statements. Concurrently with the delivery of the financial statements described in paragraph (b) and (c) of this Section 9.01, or more frequently if reasonably requested by Agent, US Company shall cause to be prepared and furnished to Agent a Compliance Certificate in the form of Exhibit P hereto executed by the Chief Financial Officer or principal accounting officer of US Company. To the extent any deliverable described in Section 9.01(b) or (c) is contained in a 10-Q or 10-K which is delivered to the Agent and each Lender pursuant to Section 9.01(i), US Company's obligation to deliver such item shall be deemed satisfied.
- (f) Notice of Default, Litigation and Material Adverse Effect. (i) Promptly, and in any event within three Business Days after any officer of US Company or any of its Subsidiaries obtains knowledge thereof, notice of the occurrence of any event which constitutes a Default or an Event of Default, and (ii) promptly, and in any event within five Business Days after any officer of US Company or any of its Subsidiaries obtains knowledge thereof, notice of (A) any litigation or governmental investigation or proceeding pending against US Company or any of its Subsidiaries which, either individually or in the aggregate, has had, or could reasonably be expected to have, a Material Adverse Effect, or (B) any other event, change or circumstance that has had, or could reasonably be expected to have, a Material Adverse Effect.
- (g) <u>Notice of Material Changes</u>. Promptly notify Agent in writing of the occurrence of any event or the existence of any fact that, in either case, is known to US Company or any other Credit Party, which renders any representation or warranty in this Agreement or any of the other Credit Documents inaccurate, incomplete or misleading in any material respect as of the date made or remade. In addition, US Company agrees to provide Agent with (i) 10 Business Days' prior written notice of (1) any change in the legal name of US Company or any other Credit Party, (2) the adoption by US Company or any other Credit Party of any new fictitious name or trade name and (3) any change in the chief executive office of US Company or any other Credit Party, and (ii) prompt written notice of any change in the information disclosed in any

Schedule hereto (which notice shall be deemed given in respect of information set forth within any periodic report filed by US Company with the SEC pursuant to Section 13 or 15 of the Securities Exchange Act of 1934, as amended, upon delivery of notice to the Administrative Agent of such filing), in each case after giving effect to the materiality limits and Material Adverse Effect qualifications contained therein.

- (h) Other Reports and Filings. Promptly after the filing or delivery thereof, copies of all financial information, proxy materials, reports, and, with respect to (ii) below, notices, if any, which US Company or any of its Subsidiaries shall (i) publicly file with the SEC, or any national securities exchange, or (ii) deliver to holders (or any trustee, agent or other representative therefor) of the Convertible Preferred Stock or any of its material Indebtedness pursuant to the terms of the documentation governing the same, in each case, if the same is not available in the SEC's EDGAR database or, if so available, US Company has not delivered notice of such filing with the SEC to the Administrative Agent.
- (i) <u>Environmental Matters</u>. Promptly after any officer of US Company or any of its Subsidiaries obtains knowledge thereof, notice of one or more of the following environmental matters to the extent that such environmental matters, either individually or when aggregated with all other such environmental matters, could reasonably be expected to have a Material Adverse Effect:
 - (i) any pending or threatened Environmental Claim against US Company or any of its Subsidiaries or any Real Property owned, leased or operated by US Company or any of its Subsidiaries;
 - (ii) any condition or occurrence on or arising from any Real Property owned, leased or operated by US Company or any of its Subsidiaries that (A) results in noncompliance by US Company or any of its Subsidiaries with any applicable Environmental Law or (B) could reasonably be expected to form the basis of an Environmental Claim against US Company or any of its Subsidiaries or any such Real Property;
 - (iii) any condition or occurrence on any Real Property owned, leased or operated by US Company or any of its Subsidiaries that could reasonably be expected to cause such Real Property to be subject to any restrictions on the ownership, lease, occupancy, use or transferability by US Company or any of its Subsidiaries of such Real Property under any Environmental Law; and
 - (iv) the taking of any removal or remedial action in response to the actual or alleged presence of any Hazardous Material on any Real Property owned, leased or operated by US Company or any of its Subsidiaries as required by any Environmental Law or any governmental or other administrative agency; provided that in any event US Company shall deliver to each Lender all notices received by US Company or any of its Subsidiaries from any government or governmental agency under, or pursuant to, CERCLA which identify US Company or any of its Subsidiaries as potentially responsible parties for remediation costs or which otherwise notify US Company or any of its Subsidiaries of potential liability under CERCLA.

All such notices shall describe in reasonable detail the nature of the claim, investigation, condition, occurrence or removal or remedial action and US Company' or such Subsidiary's response thereto. In addition to the foregoing, prior to inclusion of any Eligible Real Property in the US Borrowing Base,

provide the Administrative Agent with environmental reports, in form and substance satisfactory to the Administrative Agent and Required Lenders and from a firm satisfactory to Administrative Agent, relating to the properties owned by US Company or any of its Subsidiaries.

- (j) Borrowing Base Certificate. (i) On the Initial Borrowing Date, (ii) not later than 5:00 P.M. (New York time) on or before the 15th day of each month thereafter (or at such other times as the Administrative Agent may reasonably request), (iii) at the time of the consummation of a Permitted Acquisition, (vi) on a weekly basis during a Compliance Period, and (v) at such other times as US Company may elect, a borrowing base certificate setting forth the Borrowing Base (in each case with supporting calculations in reasonably detail) substantially in the form of Exhibit Q (each, a "Borrowing Base Certificate"), which shall be prepared (A) as of May 31, 2008 in the case of the initial Borrowing Base Certificate and (B) as of the last Business Day of the preceding month in the case of each subsequent Borrowing Base Certificate (but adjusted, in the case of a Borrowing Base Certificate delivered in connection with a Permitted Acquisition, to reflect any Eligible Accounts, Eligible Inventory, Eligible Machinery and Equipment and Eligible Real Property acquired by a Borrowing Base Party pursuant to such Permitted Acquisition) (or, if any such Borrowing Base Certificate is delivered more frequently than monthly, as of the last Business Day of the week preceding such delivery). Each such Borrowing Base Certificate shall include the then applicable, marked-to-market Swap Termination Value which the Borrower intends be treated as a Qualified Swap Termination Value for purposes of Section 5.03(d), and all such supporting information as may be reasonably requested from time to time by the Administrative Agent.
- (k) <u>Notice of Compliance Period</u>. Promptly, and in any event within two Business Days after any officer of US Company or any of its Subsidiaries obtains knowledge thereof, notice of the commencement of a Compliance Period.
- (l) <u>Material Real Property</u>. Promptly upon, and in any event within ten Business Days after, US Company or any other Credit Party acquires any Real Property the Fair Market Value of which is equal to or greater than \$2,500,000 or the Equivalent Amount thereof, notice of such acquisition, together with US Company' good faith determination of the Fair Market Value thereof.
- (m) <u>Other Information</u>. From time to time, such other information or documents (financial or otherwise) with respect to US Company or any of its Subsidiaries as the Administrative Agent or any Lender (through the Administrative Agent) may reasonably request.
- 9.02 Books, Records and Inspections; Field Examinations; Appraisals; Records and Reports of Inventory, Machinery and Equipment. (a) US Company will, and will cause each of its Subsidiaries to, keep proper books of record and accounts in which full, true and correct entries in conformity with GAAP of all financial transactions in relation to its business and activities. US Company will, and will cause each of its Subsidiaries to, permit representatives of Agent, and during the continuation of any Compliance Period, Default or Event of Default, any Lender, from time to time, as often as may be reasonably requested, but only during normal business hours, (i) to visit and inspect the Properties of US Company and each of its Subsidiaries (including the Collateral), inspect, audit and make extracts from their books and records, and discuss with their officers, their employees and their independent accountants, US Company's and each of its Subsidiaries' business, assets, liabilities, financial condition, business prospects and results of operations and (ii) to verify Eligible Accounts, Eligible Machinery and Equipment, Eligible Inventory and Eligible Real Property. Neither Agent nor any Lender shall have any duty to make any such inspection and shall not incur any liability by reason of its failure to conduct or delay in conducting any such inspection. Agent, if no Default or Event of Default

then exists, shall give US Company reasonable prior notice of any such inspection or audit. Without limiting the foregoing, US Company will participate and will cause its key management personnel to participate in a meeting with Agent and Lenders at least once during each year or more frequently, as Agent may reasonably request (except that during the continuation of an Event of Default such meetings may be held more frequently as requested by Agent or Required Lenders), which meeting(s) shall be held at such times and such places as may be reasonably requested by Agent.

- (b) In addition to Section 9.02(a), (i) in the case of succeeding sub-clause (x), at least once during each fiscal year of US Company, (ii) in the case of succeeding sub-clause (y), at least once, and at the discretion of the Administrative Agent, at least twice, in each fiscal year of US Company, and (iii) in the case of either succeeding sub-clause (x) or (y) at any time that a Compliance Period is in effect or any Event of Default exists, as often as the Administrative Agent may reasonably request, US Company will, and will cause each of its Subsidiaries to, permit officers and designated representatives of the Administrative Agent or any third-party appraiser or consultant reasonably satisfactory to the Administrative Agent and the Collateral Agent to visit and inspect (at the Borrowers' joint and several expense), under guidance of officers of US Company or such Subsidiary, any of the properties of US Company or such Subsidiary and to verify Eligible Accounts, Eligible Machinery and Equipment, Eligible Inventory and/or Eligible Real Property in order to complete (x) an appraisal of the Eligible Goods Inventory, Eligible Machinery and Equipment and/or Eligible Real Property of the Borrowing Base Parties and (y) a collateral examination of the Borrowing Base Parties, and in connection therewith US Company shall provide Administrative Agent and any field examiner or appraiser reasonable access to the books and records and the Collateral and shall cooperate with such field examiner or appraiser with respect to the foregoing.
- (c) When reasonably requested by Administrative Agent, US Company shall, and cause each of its Subsidiaries to, provide the following to Administrative Agent, with a copy to any Lender which requests delivery (which, at US Company's election, may be made by email or other electronic means of communication, or by web posting) of such reports: a report of Eligible Container Fleet Inventory and Eligible Trailer Fleet Inventory by category and by item (in detail), a report of Inventory, based upon a physical count, which shall describe Inventory of the Borrowing Base Parties by category and by item (in detail) and report the then appraised value of such Inventory, and a report of Equipment which shall describe Borrower's and Guarantors' Equipment (in detail) and report the then appraised value of such Equipment.
- (d) US Company shall, and shall cause its Subsidiaries to, keep records of its Inventory and Equipment, which records shall be complete and accurate in all material respects. Borrower shall furnish to Agent and Lenders updates of Schedule 1.01(b) and Inventory and Equipment reports concurrently with the delivery of each Borrowing Base Certificate or more frequently as requested by Administrative Agent, which reports will be in such other format and detail as Administrative Agent shall request and shall include (a) a current list of all locations of Inventory and Equipment of US Company and its Subsidiaries and (b) a list of all Inventory and Equipment of US Company and its Subsidiaries which are Motor Vehicles, which list shall specify the certificate of title holder (or the equivalent), the vehicle identification number (or equivalent) and the state or province (or equivalent) in which such Inventory or Equipment is located. US Company shall conduct a physical inventory of all container Inventory on premises owned or leased by US Company or any of its Subsidiaries no less frequently than monthly and shall provide to Administrative Agent on request a report based on each such physical inventory promptly thereafter, together with such supporting information as Administrative Agent shall reasonably request.

9.03 <u>Maintenance of Property; Insurance</u>. (a) US Company will, and will cause each of its Subsidiaries to, (i) keep all property necessary to the business of US Company and

its Subsidiaries in good working order and condition, ordinary wear and tear excepted and subject to the occurrence of casualty events, (ii) maintain with financially sound and reputable insurance companies insurance on all such property and against all such risks as is consistent and in accordance with industry practice for companies similarly situated owning similar properties and engaged in similar businesses as US Company and its Subsidiaries, and (iii) furnish to the Administrative Agent, upon its request therefor, full information as to the insurance carried. Such insurance shall include physical damage insurance on all real and personal property (whether now owned or hereafter acquired) on an all risk basis and business interruption insurance. The provisions of this Section 9.03 shall be deemed supplemental to, but not duplicative of, the provisions of any Security Documents that require the maintenance of insurance. In addition to the foregoing, US Company and the Borrowers acknowledge and agree that (x) the Administrative Agent has the right, on an annual basis, to review the insurance then being maintained by US Company and its Subsidiaries and to require US Company and its Subsidiaries to increase their levels of coverage from that which then exists to the extent that the Administrative Agent has a reasonable basis to require same and (y) it will, within 30 days following such a request by the Administrative Agent, obtain such increased insurance coverage.

- (b) US Company will, and will cause each of its Subsidiaries to, at all times keep its property insured in favor of the Collateral Agent, and all policies or certificates (or certified copies thereof) with respect to such insurance (and any other insurance maintained by US Company and/or such Subsidiaries) (i) shall be endorsed to the Collateral Agent's satisfaction for the benefit of the Collateral Agent (including, without limitation, by naming the Collateral Agent as loss payee and/or additional insured), (ii) shall state that such insurance policies shall not be canceled without at least 30 days' prior written notice thereof by the respective insurer to the Collateral Agent, (iii) shall provide that the respective insurers irrevocably waive any and all rights of subrogation with respect to the Collateral Agent and the other Secured Creditors, and (iv) shall be deposited with the Collateral Agent.
- (c) If US Company or any of its Subsidiaries shall fail to maintain insurance in accordance with this <u>Section 9.03</u>, or if US Company or any of its Subsidiaries shall fail to so endorse and deposit all policies or certificates with respect thereto, the Administrative Agent shall have the right (but shall be under no obligation) to procure such insurance and the Borrowers jointly and severally agree to reimburse the Administrative Agent for all costs and expenses of procuring such insurance.
- 9.04 Administration of Equipment; Maintenance of Equipment. (a) US Company shall, and shall cause its Subsidiaries to, keep records of its Equipment which shall be complete and accurate in all material respects itemizing and describing the kind, type, quality, quantity and book value of its Equipment and all dispositions made in accordance with this Agreement, and US Company shall, and shall cause its Subsidiaries to, furnish Administrative Agent with a current schedule containing the foregoing information on at least an annual basis and more often if reasonably requested by Administrative Agent. Promptly after the reasonable request therefore by Administrative Agent, US Company shall deliver to Administrative Agent any and all evidence of ownership, if any, of any Equipment.
 - (b) US Company shall, and shall cause its Subsidiaries to, make or cause to be made all necessary replacements of and repairs to Equipment so that the operating efficiency thereof shall be maintained and preserved, reasonable wear and tear excepted, except where the failure to so maintain the same would not reasonably be expected to have a Material Adverse Effect. US Company will not, and will not allow any other Credit Party to, permit any Equipment to become affixed to any Real Property leased to US Company or any other Credit Party so that an interest arises therein under the real estate laws of the applicable jurisdiction unless the landlord of such Real Property has executed a landlord waiver or leasehold mortgage in favor of and in form reasonably acceptable to Administrative Agent, and US Company will not permit, nor will it

allow any other Credit Party to permit, any of the Equipment of US Company or any other Credit Party to become an accession to any personal Property other than Equipment that is subject to First Priority (except for Permitted Liens) Liens in favor of Administrative Agent.

9.05 Existence; Franchises. US Company will, and will cause each of its Subsidiaries to, do or cause to be done, all things necessary to preserve and keep in full force and effect its existence and its material rights, franchises, licenses, permits, copyrights, trademarks and patents; provided, however, that nothing in this Section 9.05 shall prevent (a) sales of assets and other transactions by US Company or any of its Subsidiaries in accordance with Section 10.03, (b) the withdrawal by US Company or any of its Subsidiaries of its qualification as a foreign limited liability company (or other applicable entity) in any jurisdiction if such withdrawal could not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, or (c) dissolution of the Luxembourg Subsidiary so long as, in the case of clause (c) only, all of the following conditions are met: (1) no Default or Event of Default shall exist at the time of such dissolution and after giving effect to such dissolution, (2) any Subsidiary of the UK Company that assumes any of the rights or obligations under the Luxembourg Debt in connection with the transactions that effect such dissolution shall become a UK Guarantor and shall become a party to each of the Credit Documents to which the UK Company, UK-LP or the Luxembourg Subsidiary, as applicable, was a party immediately prior to the dissolution of the Luxembourg Subsidiary, (3) all other Liens under the Security Documents in favor of the Administrative Agent immediately prior to the dissolution of the Luxembourg Subsidiary shall remain perfected, and the Borrowers shall cause any Subsidiary of the UK Company that assumes any rights or obligations under the Luxembourg Debt in connection with the transactions that effect such dissolution to execute and deliver to the Administrative Agent such documents, instruments, financing statements, and amendments to Credit Documents as the Administrative Agent may reasonably request to continue the perfection of such Liens, (4) such dissolution shall not result in any Indebtedness permitted to be incurred pursuant to, and incurred in compliance with, Section 10.04(1) hereof; and (5) such dissolution shall otherwise not create any covenants, undertakings or obligations on the part of any Credit Party any more onerous than the covenants, undertakings or obligations contained in the Luxembourg Debt.

9.06 <u>Compliance with Statutes, etc.</u> US Company will, and will cause each of its Subsidiaries to, comply with all applicable statutes, regulations and orders of, and all applicable restrictions imposed by, all Governmental Authorities in respect of the conduct of its business and the ownership of its property in any jurisdiction (including applicable statutes, regulations, orders and restrictions relating to environmental standards and controls), except such non-compliances as could not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

9.07 Compliance with Environmental Laws. (a) US Company will comply, and will cause each of its Subsidiaries to comply, with all Environmental Laws and permits applicable to, or required by, the ownership, lease or use of its Real Property now or hereafter owned, leased or operated by US Company or any of its Subsidiaries, except such noncompliances as could not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, and will promptly pay or cause to be paid all costs and expenses incurred in connection with such compliance, and will keep or cause to be kept all such Real Property free and clear of any Liens imposed pursuant to such Environmental Laws. Neither US Company nor any of its Subsidiaries will generate, use, treat, store, Release or dispose of, or permit the generation, use, treatment, storage, Release or disposal of Hazardous Materials on any Real Property now or hereafter owned, leased or operated by US Company or any of its Subsidiaries, or transport or permit the transportation of Hazardous Materials to or from any such Real Property, except for Hazardous Materials generated, used, treated, stored, Released or disposed of at any such Real Properties in compliance in all material respects with all applicable Environmental Laws and as required in connection with the normal operation, use and maintenance of the business or operations of US Company or any of its Subsidiaries.

(b) (i) After the receipt by the Administrative Agent or any Lender of any notice of the type described in Section 9.01(i), (ii) at any time that US Company or any of its Subsidiaries are not in compliance with Section 9.07(a) or (iii) in the event that the Administrative Agent or the Lenders have exercised any of the remedies pursuant to the last paragraph of Section 11, the Borrowers will provide, at the sole joint and several expense of the Borrowers and at the request of the Administrative Agent, an environmental site assessment report concerning any Real Property owned, leased or operated by US Company or any of its Subsidiaries, prepared by an environmental consulting firm reasonably approved by the Administrative Agent, indicating the presence or absence of Hazardous Materials and the potential cost of any removal or remedial action in connection with such Hazardous Materials on such Real Property. If the Borrowers fail to provide the same within 30 days after such request was made, the Administrative Agent may order the same, the cost of which shall be borne by the Borrowers on a joint and several basis, and the Borrowers shall grant and hereby grant to the Administrative Agent and the Lenders and their respective agents access to such Real Property and specifically grant the Administrative Agent and the Lenders an irrevocable non-exclusive license, subject to the rights of tenants, to undertake such an assessment at any reasonable time upon reasonable notice to US Company, all at the sole joint and several expense of the Borrowers.

9.08 ERISA. (a) As soon as possible and, in any event, within ten (10) days after US Company, any Subsidiary of US Company or any ERISA Affiliate knows or has reason to know of the occurrence of any of the following ERISA matters, US Company will deliver to each of the Lenders a certificate of any Authorized Officer of US Company setting forth the full details as to such occurrence and the action, if any, that US Company, such Subsidiary or such ERISA Affiliate is required or proposes to take, together with any notices required or proposed to be given or filed by US Company, such Subsidiary, the Plan administrator or such ERISA Affiliate to or with the PBGC or any other Governmental Authority, or a Plan or Multiemployer Plan participant and any notices received by US Company, such Subsidiary or ERISA Affiliate from the PBGC or any other Governmental Authority, or a Plan or Multiemployer Plan participant with respect thereto: that a Reportable Event has occurred (except to the extent that US Company has previously delivered to the Lenders a certificate and notices (if any) concerning such event pursuant to the next clause hereof); that a contributing sponsor (as defined in Section 4001(a)(13) of ERISA) of a Plan subject to Title IV of ERISA is subject to the advance reporting requirement of PBGC Regulation Section 4043.61 (without regard to subparagraph (b)(1) thereof), and an event described in subsection ..62, .63, .64, .65, .66, .67 or .68 of PBGC Regulation Section 4043 is reasonably expected to occur with respect to such Plan within the following 30 days; that an accumulated funding deficiency, within the meaning of Section 412 of the Code or Section 302 of ERISA, has been incurred or an application may be or has been made for a waiver or modification of the minimum funding standard (including any required installment payments) or an extension of any amortization period under Section 412 of the Code or Section 303 or 304 of ERISA with respect to a Plan or Multiemployer Plan; that any contribution required to be made with respect to a Plan, Multiemployer Plan or Foreign Pension Plan has not been timely made; that a Plan has been terminated or a Multiemployer Plan has been reorganized, partitioned or declared insolvent under Title IV of ERISA; that a Plan has an Unfunded Current Liability; that proceedings have been instituted to terminate or appoint a trustee to administer a Plan which is subject to Title IV of ERISA; that a proceeding has been instituted pursuant to Section 515 of ERISA to collect a delinquent contribution to a Multiemployer Plan; that US Company, any Subsidiary of US Company or any ERISA Affiliate will or may incur any liability (including any indirect, contingent, or secondary liability) to or on account of the termination of or withdrawal from a Plan under Section 4062, 4063, 4064 or 4069 of ERISA, or to or on account of a Multiemployer Plan under Section 4201, 4204 or 4212 of ERISA or with respect to a Plan under Section 401(a)(29), 4971, 4975 or 4980 of the Code or Section 409, 502(i) or 502(l) of ERISA or with respect to a group health plan (as defined in Section 607(1) of ERISA, Section 4980B(g)(2) of the Code or 45 Code of Federal Regulations Section 160.103) under Section 4980B of the Code and/or the Health Insurance Portability and Accountability Act of 1996; or that US Company or any Subsidiary of US Company may incur any liability pursuant to

any employee welfare benefit plan (as defined in Section 3(1) of ERISA) that provides benefits to retired employees or other former employees (other than as required by Section 601 of ERISA). US Company will deliver to each of the Lenders (i) a copy of each funding waiver request filed with the Internal Revenue Service or any other Governmental Authority with respect to any Plan and all communications received by US Company, any Subsidiary of US Company or any ERISA Affiliate from the Internal Revenue Service or any other Governmental Authority with respect to such Plan of US Company, any Subsidiary of US Company or any ERISA Affiliate, (ii) copies of any records, documents or other information that must be furnished to the PBGC with respect to any Plan pursuant to Section 4010 of ERISA and (iii) upon the request of the Administrative Agent, a complete copy of the annual report (on Internal Revenue Service Form 5500-series) of each Plan (including, to the extent required, the related financial and actuarial statements and opinions and other supporting statements, certifications, schedules and information) required to be filed with the US Department of Labor. In addition to any certificates or notices delivered to the Lenders pursuant to the first sentence hereof, copies of annual reports and any records, documents or other information required to be furnished to the PBGC, and any material notices received by US Company, any Subsidiary of US Company or any ERISA Affiliate with respect to any Plan or Foreign Pension Plan or received from any government agency or plan administrator or sponsor or trustee with respect to any Multiemployer Plan, shall be delivered to the Lenders no later than ten (10) days after the date such annual report has been filed or such records, documents and/or information has been furnished to the PBGC or such notice has been received by US Company, the Subsidiary or the ERISA Affiliate, as applicable.

- (b) US Company and each of its applicable Subsidiaries shall ensure that all Foreign Pension Plans administered by it or into which it makes payments obtains or retains (as applicable) registered status under and as required by Applicable Law and is administered in a timely manner in all respects in compliance with all Applicable Laws, except where the failure to do any of the foregoing, either individually or in the aggregate, could not be reasonably likely to result in a Material Adverse Effect.
- 9.09 End of Fiscal Years; Fiscal Quarters. US Company will cause (i) its and each of its Subsidiaries' fiscal years to end on December 31 of each calendar year and (ii) its and each of its Subsidiaries' fiscal quarters to end on March 31, June 30, September 30 and December 31 of each calendar year.
- 9.10 <u>Performance of Obligations</u>. US Company will, and will cause each of its Subsidiaries to, perform all of its obligations under the terms of each mortgage, indenture, security agreement, loan agreement or credit agreement and each other agreement, contract or instrument by which it is bound, except such non-performances as could not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 9.11 <u>Payment of Taxes</u>. US Company will pay and discharge, and will cause each of its Subsidiaries to pay and discharge, all material taxes, assessments and governmental charges or levies imposed upon it or upon its income or profits or upon any properties belonging to it, and all lawful claims which, if unpaid, might become a Lien or charge upon any properties of US Company or any of its Subsidiaries not otherwise permitted under <u>Section 10.02</u>; <u>provided</u> that neither US Company nor any of its Subsidiaries shall be required to pay any such tax, assessment, charge, levy or claim which is being contested in good faith and by proper proceedings if it has maintained adequate reserves with respect thereto in accordance with GAAP.
 - 9.12 <u>Use of Proceeds</u>. The Borrowers will use the proceeds of the Loans only as provided in <u>Section 8.23</u>.

9.13 Additional Security; Further Assurances; etc. (a) US Company will, and will cause each other Credit Party to, grant to the Collateral Agent for the benefit of the Secured Creditors Liens, hypothecs and Mortgages in such assets and owned Real Property of US Company and such other Credit Party as are not covered by the original Security Documents and as may be reasonably requested from time to time by the Administrative Agent or the Required Lenders (collectively, the "Additional Security Documents"). All such Liens, hypothecs and Mortgages shall be granted pursuant to documentation reasonably satisfactory in form and substance to the Administrative Agent and shall constitute valid and enforceable perfected Liens, hypothecations and Mortgages superior to and prior to the rights of all third Persons and enforceable against third parties and subject to no other Liens except for Permitted Liens or, in the case of Real Property, the Permitted Encumbrances related thereto (provided that Administrative Agent's Liens on the LKE Joint Accounts need not be perfected). The Additional Security Documents or instruments related thereto shall have been duly recorded or filed in such manner and in such places as are required by law to establish, perfect, preserve and protect the Liens in favor of the Collateral Agent required to be granted pursuant to the Additional Security Documents and all taxes, fees and other charges payable in connection therewith shall have been paid in full. In connection with the delivery of any Mortgage, US Company will, and will cause each other Credit Party to, to the extent reasonably requested from time to time by the Administrative Agent or the Required Lenders, deliver (i) a Mortgage Policy issued by a title insurer reasonably satisfactory to the Administrative Agent, in form and substance and in an amount reasonably satisfactory to the Administrative Agent insuring that the Mortgage is a valid and enforceable First Priority Lien on the respective property other than Permitted Encumbrances, (ii) a then current A.L.T.A. survey, certified to the Administrative Agent by a licensed surveyor sufficient to allow the issuer of the Mortgage Policy to issue such Mortgage Policy without a survey exception, (iii) an environmental site assessment prepared by a qualified firm reasonably acceptable to the Agent, in form and substance satisfactory to the Administrative Agent and (iv) a certificate in a form reasonably acceptable to the Administrative Agent indicating that the property is not in a flood zone, or if the property is in a flood zone, evidence that appropriate insurance reasonable acceptable to the Administrative Agent has been obtained. Notwithstanding the foregoing, this Section 9.13(a) shall not apply to (and US Company and the other Credit Parties shall not be required to grant a Mortgage in) any Real Property the Fair Market Value of which is less than \$2,500,000 or the Equivalent Amount thereof.

(b) US Company will, and will cause each of the other Credit Parties to, at the expense of US Company, make, execute, endorse, acknowledge, file and/or deliver to the Collateral Agent from time to time such vouchers, invoices, schedules, confirmatory assignments, conveyances, financing statements, transfer endorsements, powers of attorney, certificates, real property surveys, reports, landlord waivers, bailee agreements, control agreements and other assurances or instruments and take such further steps relating to the Collateral covered by any of the Security Documents as the Collateral Agent may reasonably require. Furthermore, US Company will, and will cause the other Credit Parties to, deliver to the Collateral Agent such opinions of counsel, title insurance and other related documents as may be reasonably requested by the Collateral Agent to assure itself that this Section 9.13 has been complied with.

(c) If the Administrative Agent or the Required Lenders reasonably determine that they are required by law or regulation to have appraisals prepared in respect of any Real Property of US Company and the other Credit Parties constituting Collateral, US Company will, at its own expense, provide to the Administrative Agent appraisals which satisfy the applicable requirements of the Real Estate Appraisal Reform Amendments of the Financial Institution Reform, Recovery and Enforcement Act of 1989, as amended, and which shall otherwise be in form and substance reasonably satisfactory to the Administrative Agent.

- (d) US Company agrees that each action required by clauses (a) through (c) of this Section 9.13 shall be completed as soon as possible, but in no event later than 60 days after such action is requested to be taken by the Administrative Agent or the Required Lenders; provided that, in no event will US Company or any of its Subsidiaries be required to take any action, other than using its commercially reasonable efforts, to obtain consents from third parties with respect to its compliance with this Section.
- (e) Notwithstanding anything contained in this <u>Section 9.13</u> or this Agreement to the contrary, in no event shall (i) the assets of any of the UK Company or any other Foreign Subsidiary be pledged to secure the US Obligations or (b) more than 65% of the voting securities of any Foreign Subsidiary held by the US Company or any Domestic Subsidiary be pledged to secure the US Obligations.
- (f) Each Subsidiary that is a Domestic Subsidiary (other than any holding company or shell company without any assets constituting a portion of the Borrowing Base and other than any other Subsidiary reasonably agreed by the Administrative Agent) shall become a party to this Agreement as a US Borrower on or prior to the date which is 21 days following the Closing Date, in each case by executing and delivering to the Administrative Agent a counterpart of a Joinder Agreement, and delivering to the Administrative Agent a counterpart to each US Revolving Note and the US Swingline Note.
- (g) Each Subsidiary that is a UK Subsidiary (other than any holding company or shell company without any assets constituting a portion of the Borrowing Base and other than any other Subsidiary reasonably agreed by the Administrative Agent) shall become a party to this Agreement as a UK Borrower on or prior to the date which is 21 days following the Closing Date, in each case by executing and delivering to the Administrative Agent a counterpart of a Joinder Agreement and delivering to the Administrative Agent a counterpart to each UK Revolving Note and the UK Swingline Note.
- 9.14 <u>Convertible Preferred Stock</u>. US Company will pay all dividends on its Convertible Preferred Stock solely through the issuance of additional shares of such Convertible Preferred Stock (but not in cash or Cash Equivalents or other Property).
- 9.15 <u>Projections</u>. US Company shall, no later than 60 days after the end of each fiscal year of US Company, deliver to Administrative Agent Projections of US Company and each of its Subsidiaries for the forthcoming three (3) fiscal years, month by month (including, but not limited to, projected balance sheets, income statements, statements of cash flows and Total Borrowing Availability, US Borrowing Availability, UK Borrowing Availability and calculations of projected covenant compliance other than compliance with <u>Section 10.27</u> (Minimum Utilization).
- 9.16 <u>Landlord, Processor and Storage Agreements</u>. US Company shall provide Administrative Agent on request with copies of all agreements between US Company or any other Credit Party and any landlord, processor, distributor, warehouseman or consignee which owns any premises at which any Collateral may, from time to time, be kept.
- 9.17 <u>Deposit and Brokerage Accounts</u>. For each deposit account or brokerage account that US Company or any other Credit Party that is a Domestic Subsidiary of US Company at any time opens or maintains (including, without limitation, any LKE Joint Account, but excluding any account for which such a Cash Management Control Agreement is expressly not required pursuant to Section 3.9 of the US Security Agreement), US Company shall, at Administrative Agent's request and option, pursuant to an Cash Management Control Agreeement in form and substance satisfactory to Administrative Agent, cause the depository bank or securities intermediary, as applicable,

to agree to comply at any time with instructions from Administrative Agent to such depository bank or securities intermediary, as applicable, directing the disposition of funds from time to time credited to such deposit or brokerage account, without further consent of US Company or such other Credit Party, during the continuance of a Compliance Period, a Default or an Event of Default.

- 9.18 <u>Credit Party Financial Statements</u>. US Company shall deliver or cause to be delivered to Administrative Agent financial statements, if any, for each other Credit Party (to the extent not consolidated or combined with the financial statements delivered to Administrative Agent under <u>Section 9.01</u>) in form and substance reasonably satisfactory to Agent at such intervals and covering such time periods as Administrative Agent may request.
- 9.19 <u>Qualifying Derivative Obligations</u>. Prior to or concurrently with US Company or any other Credit Party entering into any Derivative Obligation or any modification of such Derivative Obligation with any Lender or any Affiliate of a Lender (other than Administrative Agent or any Affiliate of Administrative Agent), US Company shall provide, or shall cause such Lender or such Affiliate to provide, written notice to Administrative Agent specifying, in a manner reasonably acceptable to Administrative Agent, the terms and conditions of such Derivative Obligations. In addition, any Lender that enters into any Derivative Obligation or any modification of a Derivative Obligation with a Credit Party may provide notice thereof to the Administrative Agent.
- 9.20 <u>Centre of Main Interest</u>. Each UK Subsidiary and the Luxembourg Subsidiary shall maintain its centre of main interest for purposes of Recital 13 of EC Regulation No. 1346/2000 on Insolvency Proceedings within respectively, the United Kingdom and Luxembourg.
- 9.21 Administration of Accounts. (a) Borrower shall, and shall cause each of its Subsidiaries to, keep accurate and complete records of its Accounts and all payments and collections thereon and shall submit to Administrative Agent on such periodic basis as Agent shall request a sales and collections report for the preceding period, in form consistent with the reports currently prepared by US Company with respect to such information. When requested by Administrative Agent, from and after the date hereof, US Company shall deliver to Administrative Agent a detailed aging of all of Accounts of US Company and its Subsidiaries, and upon Administrative Agent's request therefore, copies of proof of delivery and the original copy of all documents, including, without limitation, repayment histories and present status reports relating to the Accounts so scheduled and such other matters and information relating to the status of then existing Accounts as Administrative Agent shall reasonably request.
- (b) If an Account includes a charge for any tax payable to any governmental taxing authority, Administrative Agent is authorized, in its sole discretion, to pay the amount thereof to the proper taxing authority for the account of US Company or its Subsidiaries and to charge US Company therefore, except for taxes that (i) are being actively contested in good faith and by appropriate proceedings and with respect to which US Company or such Subsidiary maintains reasonable reserves on its books therefore and (ii) would not reasonably be expected to result in any Lien other than a Permitted Lien. In no event shall Administrative Agent or any Lender be liable for any taxes to any governmental taxing authority that may be due by US Company or any of its Subsidiaries or Affiliates.
- 9.22 <u>Completion of Mobile Storage Acquisition on the Closing Date</u>. (a) Concurrent with the making of the Pre-Merger Loans, the US Company shall deliver to the Administrative Agent an irrevocable Notice of Borrowing for each other Loan to be made on the Initial Borrowing Date necessary to consummate the Mobile Storage Acquisition on the Closing Date; (b) the US Company shall cause the Mobile Storage Acquisition, including without limitation each Subsequent

Merger, to have been completed on the Closing Date; and (c) the US Company shall cause (i) all Refinanced Indebtedness to have been repaid in full (together with all fees and other amounts owing thereon), all commitments thereunder to have been terminated and all letters of credit issued pursuant thereto to have been terminated or cash collateralized in a manner reasonably acceptable to the Administrative Agent, and (ii) all security interests in respect of, and Liens securing, the Refinanced Indebtedness created pursuant to the security documentation relating thereto to have been terminated and released in a manner reasonably satisfactory to the Administrative Agent, in each case as of the Closing Date.

SECTION 10. Negative Covenants. Each of US Company and each other Credit Party hereby covenants and agrees that on and after the Effective Date and until the Total Revolving Loan Commitment and all Letters of Credit have terminated and the Loans, Notes and Unpaid US Drawings and Unpaid UK Drawings (in each case, together with interest thereon), Fees and all other Obligations (other than any indemnities described in Section 13.13 which are not then due and payable) incurred hereunder and thereunder, are paid in full in cash:

10.01 <u>Capital Expenditures</u>. US Company and its Subsidiaries shall not make payments for Capital Expenditures (net of sales of Eligible Container Fleet Inventory) in excess of \$150,000,000 or the Equivalent Amount in any fiscal year; <u>provided</u>, that as long as no Event of Default shall have occurred and be continuing, US Company and its Subsidiaries may carry forward and add to the next year's limitation amount (but not beyond such next year) the unused portion of the limitation on Capital Expenditures for the prior year, up to a maximum of one hundred percent (100%) of the prior year's limitation amount; and <u>provided</u>, <u>further</u>, that the amount set forth in this <u>Section 10.01</u> shall be increased by an amount equal to three hundred percent (300%) of the net cash proceeds received by US Company from any sale of equity Securities of US Company less such amount of such net cash proceeds used to redeem or repurchase Senior Notes in compliance with this Agreement (the "<u>CapEx Equity Increase</u>"), and the unused portion of any CapEx Equity Increase may be carried forward to any subsequent fiscal year. US Company and its Subsidiaries shall not make any Capital Expenditures that are not directly related to the business conducted on the Initial Borrowing Date by US Company and its Subsidiaries.

10.02 <u>Liens</u>. US Company will not, and will not permit any of its Subsidiaries to, create, incur, assume or suffer to exist any Lien upon or with respect to any property or assets (real or personal, tangible or intangible) of US Company or any of its Subsidiaries, whether now owned or hereafter acquired, or sell any such property or assets subject to an understanding or agreement, contingent or otherwise, to repurchase such property or assets (including sales of accounts receivable with recourse to US Company or any of its Subsidiaries), or assign any right to receive income or permit the filing of any financing statement under the UCC or any other similar notice of Lien under any similar recording or notice statute; <u>provided</u> that the provisions of this <u>Section 10.02</u> shall not prevent the creation, incurrence, assumption or existence of the following (Liens described below are herein referred to as "<u>Permitted Liens</u>"):

- (a) Liens granted to Collateral Agent for the benefit of the Lenders under the Security Documents to secure the Obligations;
- (b) (i) Liens listed on <u>Schedule 10.02</u> and (ii) other Liens purporting to relate to less than \$500,000 individually or \$5,000,000 in the aggregate, relating to assets acquired in connection with the transactions under the Merger Agreement, if such Liens are being contested in good faith by US Company or a Subsidiary;

- (c) Liens for taxes not yet due or being contested in good faith and by appropriate proceedings to the extent permitted under this Agreement;
- (d) Purchase Money Liens and Leases;
- (e) Liens of warehousemen, mechanics, materialmen, workers, repairmen, common carriers, or landlords, liens for taxes, assessments or other governmental charges, and other similar Liens (other than Liens arising under ERISA) arising by operation of law for amounts that are not yet due and payable or which are being diligently contested in good faith by a Credit Party, and for which adequate reserves are maintained by US Company for their payment in accordance with GAAP;
- (f) Attachment or judgment Liens not to exceed an aggregate of \$2,000,000 or the Equivalent Amount thereof excluding in each case amounts (i) bonded to the reasonable satisfaction of Administrative Agent or (ii) covered by insurance to the reasonable satisfaction of Administrative Agent;
- (g) Deposits or pledges to secure obligations under worker's compensation, social security or similar laws, or under unemployment insurance, not to exceed an aggregate of \$5,000,000 or the Equivalent Amount thereof;
- (h) Deposits or pledges to secure bids, tenders, contracts (other than contracts for the payment of money), leases, statutory obligations, surety and appeal bonds and other obligations of like nature arising in the ordinary course of business not to exceed an aggregate of \$5,000,000 or the Equivalent Amount thereof;
- (i) Easements, rights of way, restrictions and other similar encumbrances incurred in the ordinary course of business which, in the aggregate, are not substantial in amount and which do not materially detract from the value of the property subject thereto or materially interfere with the ordinary conduct of the business of US Company or any of its Subsidiaries;
- (j) Liens securing Indebtedness described in <u>Section 10.05(d)</u> which has been refinanced so long as such refinanced Indebtedness is not secured by any collateral which did not secure the Indebtedness prior to such refinancing;
 - (k) Liens securing Indebtedness described in Section 10.05(j) and (k);
- (l) Extensions and renewals of any of the foregoing so long as the aggregate amount of extended or renewed Liens are not increased and are on terms and conditions no more restrictive than the terms and conditions of the Liens extended or renewed;
 - (m) Liens securing Capitalized Lease Obligations permitted pursuant to Section 10.05(l);
 - (n) Liens securing Indebtedness of a Foreign Subsidiary permitted under Section Section 10.05(e)(iii); and
 - (o) Liens securing Capitalized Lease Obligations permitted pursuant to Section 10.05(m).

10.03 Sale of Assets. US Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, sell, lease, assign, transfer or otherwise dispose of any assets (an "Asset Sale") other than (i) Inventory (including containers held for lease) in the ordinary course of business, (ii) individual items of Collateral with a book value of less than \$2,500,000 or the Equivalent Amount thereof in the aggregate during any fiscal year, (iii) obsolete or worn out property disposed of in the ordinary course of business, (iv) transfers of Inventory and Equipment from US Company to a Credit Party, or from one Credit Party to another Credit Party or to US Company, (v) sales of Trailers acquired in Permitted Acquisitions or owned by US Company or a Credit Party on the date hereof, (vi) so long as no Default or Event of Default has occurred and is continuing, to the extent permitted by the Senior Note Indentures, sales of container Inventory held for lease pursuant to sale and leaseback transactions (with the "leaseback" portion of such transaction being permitted pursuant to Section 10.05(m)), provided that the Orderly Liquidation Value of such container Inventory sold during the term of this Agreement shall not exceed \$50,000,000 or the Equivalent Amount thereof, (vii) dispositions of assets not otherwise addressed by this Section 10.03 with an aggregate Fair Market Value not in excess of \$2,500,000 or the Equivalent Amount thereof in the aggregate in any fiscal year; provided that, with respect to clauses (ii), (iii), (iv), (v), (vi) and (vii), (a) such dispositions are for fair value, (b) the aggregate consideration is paid in full in cash at the time of disposition and is either reinvested in the business of US Company or its Subsidiaries (subject to the limitations of this Agreement) or used to repay Revolving Loans in accordance with Section 5.02(c), (viii) sales of Equipment which US Company or a Credit Party will lease back under a capital lease permitted under Section 10.05(d) or an operating lease permitted under Section 10.13 and (ix) sales or other dispositions of Inventory or Equipment to the extent that (i) such property is exchanged for credit against the purchase price of similar replacement property or (ii) the proceeds of such sale or other disposition are applied to the purchase price of such replacement property (or otherwise applied in accordance with Section 5.02(c)(ii)), in each case as part of a LKE Transaction and pursuant to the terms and conditions of the LKE Master Exchange Agreement.

10.04 Restricted Payments. US Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly:

(a) declare or pay any dividend (other than dividends payable solely in common stock of US Company) on, or make any payment on account of, or set apart assets for a sinking or other analogous fund for, the purchase, redemption, defeasance, retirement or other acquisition of, any shares of any class of Securities of US Company or any warrants, options or rights to purchase any such Securities, whether now or hereafter outstanding, or make any other distribution in respect thereof, either directly or indirectly, whether in cash or property or in obligations of US Company or any of its Subsidiaries (each of the foregoing, a "Restricted Payment"); provided, (i) any Subsidiary may declare and pay dividends or distributions to US Company or any other Domestic Subsidiary of US Company which is a US Credit Party; (ii) any Foreign Subsidiary may declare and pay dividends or distributions to UK Company or any other Foreign Subsidiary, (iii) US Company may purchase on the open market or in privately-negotiated transactions Securities consisting of its common stock for an aggregate amount not to exceed \$10,000,000 if, (A) both before and after giving effect to such purchase, no Default or Event of Default exists or would result therefrom and US Company has Total Borrowing Availability of at least \$100,000,000, (B) all shares of such Securities so purchased are thereafter immediately cancelled or shall have the status of treasury stock of US Company and (C) if the Restricted Payment under this clause (iii) is to be made with the proceeds of a Loan, the request for such Loan shall be made under a separate request for borrowing and shall be accompanied by calculations in reasonable detail evidencing that US Company may make any Restricted Payment in compliance with this Section 10.04(a); and (iv) US Company may make any Restricted Payment permitted pursuant to the Senior Note Documents so long as at the time of such Restricted Payment, the Payment Conditions shall have been satisfied; or

(b) make any payment or prepayment of principal of, or any prepayment of interest on, or any redemption (including, without limitation, by making payments to a sinking or analogous fund), repurchase or defeasance of, any Indebtedness (other than the Obligations) or of any Mandatory Redeemable Obligation; provided that (i) any Subsidiary may make payments on account of Indebtedness owing to US Company or any other Credit Party, (ii) on or prior to August 1, 2010, US Company may repurchase or redeem up to 35% of the aggregate principal amount of the Senior Notes outstanding on the Initial Borrowing Date and pay accrued interest and premium thereon with the proceeds of the issuance of US Company's Securities in an "Equity Offering" under and as defined in the Senior Note Indentures if, both before and after giving effect to such repurchase or redemption, (x) no Default or Event of Default exists and (y) US Company has Total Borrowing Availability of at least \$100,000,000; (iii) US Company and its Subsidiaries may make scheduled principal and interest payments on Indebtedness permitted under Sections 10.05(a), (b), (c), (d), (g), (h), (i), (j), (k) and (l) and scheduled interest payments on the Senior Notes and (iv) US Company may prepay, repurchase or redeem any Indebtedness permitted to be prepaid, repurchased or redeemed pursuant to the Senior Note Documents so long as at the time of such prepayments, repurchases or redemptions, the Payment Conditions shall have been satisfied.

10.05 <u>Indebtedness</u>. US Company will not, and will not permit any of its Subsidiaries to, contract, create, incur, assume or suffer to exist any Indebtedness, except:

- (a) Indebtedness under the Credit Documents and Derivative Obligations under which a Lender (or its Affiliate) is the counterparty incurred in the ordinary course of business;
 - (b) Unsecured Derivative Obligations incurred in the ordinary course of business;
- (c) Indebtedness described on <u>Schedule 10.05</u>, and any refinancing of such Indebtedness, so long as the aggregate principal amount of the Indebtedness so refinanced shall not be increased and the refinancing shall be on terms and conditions no more restrictive than the terms and conditions of the Indebtedness to be refinanced;
- (d) Indebtedness, including Capitalized Lease Obligations, secured by purchase money liens on Equipment the title to or leasehold interest in which is acquired after the Initial Borrowing Date, not to exceed \$20,000,000 or the Equivalent Amount thereof in the aggregate (irrespective of when due) outstanding at any one time ("Purchase Money Liens and Leases") so long as each Purchase Money Lien or Lease shall attach or relate only to the property (and accessions thereto and proceeds thereof) to be acquired or the acquisition cost of which is financed through leasing and the principal amount of the debt incurred (including the principal component of lease payments) shall not exceed one hundred percent (100%) of the purchase price of the item or items of equipment;
- (e) (i) Indebtedness consisting of loans and advances by US Company or any Subsidiary of US Company to US Company or any Domestic Subsidiary that is a Credit Party; (ii) Indebtedness consisting of loans and advances by the UK Company or any Foreign Subsidiary to UK Company or any Foreign Subsidiary that is a Credit Party; (iii) Indebtedness consisting of loans and advances by the US Company or any Domestic Subsidiary to UK Company or any Foreign Subsidiary that is a Credit Party, in an amount, together with Investments under Section 10.07(b)(iii), not to exceed \$25,000,000 or the Equivalent Amount thereof at any time outstanding; (iv) Indebtedness of any Foreign Subsidiary (other than a Credit Party) for which none of a US Borrower, a UK Borrower or any Credit Party has provided credit support (by guarantee, granting of Liens on its assets or otherwise) in an amount not to exceed \$10,000,000 or

the Equivalent Amount thereof at any time outstanding, and (v) other Indebtedness consisting of loans and advances to Foreign Subsidiaries, which together with Investments made by any Credit Party in Foreign Subsidiaries under Section 10.07(b)(iv) (such loans and advances to Foreign Subsidiaries, together with Investments made under Section 10.07(b)(iv), "Restricted Foreign Funding"), does not exceed \$10,000,000 or the Equivalent Amount thereof in the aggregate at any time outstanding; provided that (X) all loans and advances described under (i), (ii), (iii) and (v) are evidenced by an intercompany note, which is pledged to Administrative Agent and are subordinated to the Obligations on terms and conditions reasonably acceptable to the Administrative Agent and (Y) US Company may make any Restricted Foreign Funding in excess of the limitations in the foregoing clause (v) which are otherwise permitted pursuant to the Senior Note Documents so long as at the time of such Restricted Foreign Funding, the Payment Conditions shall have been satisfied.

- (f) Indebtedness under the Senior Notes, and any refinancings thereof, in an aggregate principal balance for all such Indebtedness not to exceed \$350,000,000 at any time outstanding so long as, with respect to any refinancings thereof, the Indebtedness is on terms at least as favorable to the Lenders as the Senior Notes, in the reasonable judgment of the Administrative Agent;
- (g) Indebtedness (other than Indebtedness under this Agreement) incurred to finance insurance premiums, not to exceed \$5,000,000 or the Equivalent Amount thereof in any fiscal year;
- (h) unsecured Indebtedness which matures no earlier than six (6) months after the Final Maturity Date so long as (i) the Credit Parties shall be in compliance with the Debt Ratio then applicable to the most recently completed fiscal quarter as set forth in Section 10.26 (Debt Ratio) calculated on a pro forma basis (whether or not then tested) before and immediately after giving effect to the incurrence of such Indebtedness, (ii) before and immediately after giving effect to the incurrence of such Indebtedness and any contemporaneous use of the proceeds thereof, no Default or Event of Default has occurred and is continuing or would be created thereby, and (iii) such Indebtedness is not prohibited by the Senior Note Documents;
- (i) unsecured Indebtedness in an aggregate principal amount not to exceed \$25,000,000 or the Equivalent Amount thereof so long as (i) before and immediately after giving effect to the incurrence of such Indebtedness and any contemporaneous use of the proceeds thereof, no Default or Event of Default has occurred and is continuing or would be created thereby, and (ii) such Indebtedness is not prohibited by the Senior Note Documents.
- (j) Indebtedness which is secured by a Lien junior to the Lien granted to Administrative Agent for the benefit of the Lenders under the Security Documents, and (i) such Indebtedness matures no earlier than six (6) months after the end of the Term, (ii) the Credit Parties shall be in compliance with the Debt Ratio then applicable to the most recently completed fiscal quarter as set forth in Section 10.26 (Debt Ratio) calculated on a pro forma basis (whether or not then tested) before and immediately after giving effect to the incurrence of such Indebtedness, (iii) the creditors with respect to such Indebtedness enter into an intercreditor agreement in form and substance satisfactory to Administrative Agent, (iv) before and immediately after giving effect to the incurrence of such Indebtedness and any contemporaneous use of the proceeds thereof, no Default or Event of Default has occurred and is continuing or would be created thereby, and (v) such Indebtedness is not prohibited by the Senior Notes Documents;

- (k) Indebtedness which is secured by a first or second priority Lien on the assets of the US Company or other Credit Party, in an aggregate principal amount for all Credit Parties not to exceed \$5,000,000 or the Equivalent Amount thereof so long as (i) the creditors with respect to such Indebtedness enter into an intercreditor agreement in form and substance satisfactory to Administrative Agent, (ii) before and immediately after giving effect to the incurrence of such Indebtedness and any contemporaneous use of the proceeds thereof, no Default or Event of Default has occurred and is continuing or would be created thereby, and (iii) such Indebtedness is not prohibited by the Senior Notes Documents;
- (l) the Luxembourg Debt; provided that (A) such Indebtedness will be evidenced by a revolving credit facility agreement in the form existing as at the date hereof with claims thereunder assigned in favor of the Administrative Agent and shall be subordinated to the Obligations of the Credit Parties on terms and conditions satisfactory to the Administrative Agent and (B) the creditors with respect to such Indebtedness shall have entered into and delivered to the Administrative Agent for the benefit of the Lenders the UK Intercreditor Deed in the form attached hereto as Exhibit R; and
- (m) Capitalized Lease Obligations in respect of leasehold interest in container Inventory which is sold in a sale and leaseback transaction (with the sale portion of such transaction being permitted pursuant to Section 10.03(vi)) so long as each lease shall attach or relate only to the property subject to such sale leaseback transaction and the principal amount of the debt incurred (including the principal component of lease payments) shall not exceed one hundred percent (100%) of the purchase price of the item or items of container Inventory.
- 10.06 <u>Contingent Obligations</u>. Neither US Company nor any of its Subsidiaries shall directly or indirectly incur, assume, or suffer to exist any Contingent Obligation, excluding indemnities given in connection with the sale of Inventory or other asset dispositions permitted hereunder and Contingent Obligations for Indebtedness permitted to be incurred under <u>Section 10.05</u> hereof
- 10.07 <u>Advances, Investments and Loans</u>. US Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, make any Investment in any Person, whether in cash, securities, or other property of any kind including, without limitation, any Subsidiary or Affiliate of US Company, other than:
 - (a) Advances or loans (but not sales on open account on ordinary course of business terms) made in the ordinary course of business, including those made to finance the sale of Inventory, not to exceed \$250,000 or the Equivalent Amount thereof outstanding at any one time to any one Person and \$750,000 or the Equivalent Amount thereof in the aggregate outstanding at any one time;
 - (b) Investments (including in the form of loans or advances permitted by Section 10.4(e)): (i) by US Company or any Subsidiary of US Company in US Company or any Domestic Subsidiary that is a Credit Party, (ii) by the UK Company or any Foreign Subsidiary in UK Company or any Foreign Subsidiary that is a UK Credit Party; provided that all Investments in such Foreign Subsidiaries, together with Indebtedness of Foreign Subsidiaries permitted under Section 10.05(e)(iii), shall not exceed \$25,000,000 or the Equivalent Amount thereof at any time outstanding, and (iv) in Foreign Subsidiaries, which together with Indebtedness incurred in reliance on Section 10.05(e)(v), does not exceed \$10,000,000 or the Equivalent Amount thereof in the aggregate at any time outstanding; provided that (X) all

Investments in the form of loans and advances described under (i), (ii), (iii) and (iv) are evidenced by an intercompany note in accordance with Section 10.05(e)(v) and (Y) US Company may make any Restricted Foreign Funding in excess of the limitations set forth in the foregoing clause (iv) which are otherwise permitted pursuant to the Senior Note Documents so long as at the time of such Restricted Foreign Funding, the Payment Conditions shall have been satisfied;

- (c) Cash Equivalents;
- (d) Permitted Acquisitions;
- (e) Deposits with financial institutions, disclosed on <u>Schedule 10.07</u>, and which are insured by the Federal Deposit Insurance Corporation ("FDIC") or a similar federal insurance program; <u>provided</u>, <u>however</u>, that US Company may, in the ordinary course of its business, maintain in its disbursement accounts from time to time amounts in excess of then applicable FDIC or other program insurance limits; and
 - (f) Such other Investments as Required Lenders may approve in writing in their sole discretion.

10.08 <u>Transactions with Affiliates</u>. US Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, enter into any transaction with, including, without limitation, the purchase, sale or exchange of property or the rendering of any service to, any Subsidiary or Affiliate of US Company, except (a) the transactions in existence on the Initial Borrowing Date as described on <u>Schedule 10.08</u>, (b) transactions between or among US Company and its wholly-owned Subsidiaries which are Credit Parties and (c) transactions in the ordinary course of and pursuant to the reasonable requirements of US Company's or such Subsidiary's or Affiliate's business, as the case may be, and upon fair and reasonable terms no less favorable to US Company or such Subsidiary than could be obtained in a comparable arm's-length transaction with an unaffiliated Person.

10.09 [Reserved].

10.10 [Reserved].

10.11 <u>Additional Negative Pledges</u>. US Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, create or otherwise cause or suffer to exist or become effective, (i) any prohibition or restriction (including any agreement to provide equal and ratable security to any other Person in the event a Lien is granted to or for the benefit of Administrative Agent and the Lenders) on the creation or existence of any Lien upon the assets of US Company or its Subsidiaries or (ii) any contractual obligation which may restrict or inhibit Administrative Agent's rights or ability to sell or otherwise dispose of the Collateral or any part thereof after the occurrence of an Event of Default, other than pursuant to the Senior Note Documents as in effect on the date on which the Senior Notes were issued.

10.12 <u>No Subsidiaries</u>. US Company shall not, directly or indirectly, form or acquire any new Subsidiaries, except (a) in connection with Permitted Acquisitions in compliance with <u>Section 10.14</u> and (b) if each of the following conditions is met:

(i) each new Subsidiary is a Wholly-Owned Subsidiary of US Company created to conduct business in a specific jurisdiction;

- (ii) both before and after giving effect to the creation of such Subsidiary and the transfer of any assets from US Company to such Subsidiary, all representations and warranties of US Company and its Subsidiaries contained in any Credit Document are true and correct, on and as of such date as if made as of such date (except (x) such revisions as are necessary to reflect the formation of such new Subsidiary and (y) to the extent a representation and warranty was made only as of a specified date, such representation and warranty shall have been true and correct as of such date), no Default or Event of Default shall have occurred and be continuing, and US Company and its Subsidiaries shall be Solvent;
- (iii) US Company shall have delivered to Administrative Agent written notice at least fifteen (15) Business Days prior to consummation of any transfer of assets to, or acquisition of assets by, such new Subsidiary, describing in reasonable detail the proposed new Subsidiary and its assets;
- (iv) any such new Subsidiary shall have executed and delivered to Administrative Agent such Security Documents (or joinders thereto, in form and substance satisfactory to Agent) and other documents as are necessary (or advisable in Administrative Agent's judgment) under Applicable Law in order to grant Administrative Agent for the benefit of the Lenders a perfected First Priority security interest and Lien in the assets of, and ownership interests in, such Subsidiary and US Company or the applicable Credit Party shall execute and deliver an amendment to the applicable pledge agreement in form and substance satisfactory to Administrative Agent, together with stock certificates and promissory notes and other instruments endorsed in blank, to pledge its equity interests in such new Subsidiary and all intercompany Loans to such Subsidiary;
- (v) each such new Subsidiary that is a Domestic Subsidiary shall become a party to this Agreement, either as a US Borrower or a US Guarantor, as determined by the Administrative Agent or the Required Lenders, in each case by executing and delivering to the Administrative Agent a counterpart of a Joinder Agreement, and in the event that the Subsidiary is to become a US Borrower hereunder, by executing and delivering to the Administrative Agent a counterpart to the US Revolving Note and the US Swingline Note;
- (vi) (i) each such new Subsidiary that is a UK Subsidiary shall become a party to this Agreement, either as a UK Borrower or a UK Guarantor, as determined by the Administrative Agent or the Required Lenders, in each case by executing and delivering to the Administrative Agent a counterpart of a Joinder Agreement, and in the event that the Subsidiary is to become a UK Borrower hereunder, by executing and delivering to the Administrative Agent a counterpart to the UK Revolving Note and the UK Swingline Note, and (ii) each such new Subsidiary that is a Foreign Subsidiary other than a UK Subsidiary shall become a party to this Agreement as a UK Guarantor by executing and delivering to the Administrative Agent a counterpart of a Joinder Agreement; and
- (vii) if required by Administrative Agent, Administrative Agent shall have received opinions of counsel, in form and substance satisfactory to it, as to the due execution, delivery and enforceability of the Credit Documents executed by such new Subsidiary, together with such evidences of solvency, certificates, Certificates of Title, and other documents and instruments reasonably requested by Administrative Agent.
- 10.13 Operating Leases; Off-Balance Sheet Financing. Neither US Company nor any of its Subsidiaries shall directly or indirectly incur, create, assume or suffer to exist any liabilities for operating leases or other indebtedness or liabilities not reflected as such on their financial statements other than liabilities described on Schedule 10.13, and any refinancing of such liabilities, so long as the aggregate amount thereof so refinanced shall not be increased and the refinancing shall be on terms and conditions no more restrictive than the terms and conditions of the liabilities to be refinanced; provided, however, that US Company and its Subsidiaries may incur liabilities in connection with

operating leases of real property (including office and yard space) and office Equipment in the ordinary course of business and of other Equipment with values of up to \$10,000,000 or the Equivalent Amount thereof in any fiscal year (exclusive of Equipment acquired under operating leases executed prior to the Initial Borrowing Date and listed on Schedule 10.13) (and up to 50% of any amount not incurred in any fiscal year may be carried over to the next fiscal year).

- 10.14 <u>Permitted Acquisitions</u>. US Company shall not, and shall not permit any of its Subsidiaries to, make an Acquisition (other than the Mobile Storage Acquisition) unless each of the following conditions is satisfied:
 - (a) such Acquisition is made by US Company or another Credit Party;
- (b) such Acquisition shall be consensual and, if required under state law, shall have been approved by the board of directors or other governing body of the Person to be acquired (if there is such a governing body) and shall be permitted by the Senior Note Documents;
- (c) both before and after giving effect to such Acquisition, all representations and warranties of US Company and its Subsidiaries contained in any Credit Document are true and correct on such date as if made as of such date (except to the extent that a representation and warranty was made only as of a specified date, such representation and warranty shall have been true and correct as of such date) and no Default or Event of Default shall have occurred and be continuing, and Administrative Agent shall receive a certificate of US Company to such effect on the date on which such Acquisition is consummated;
- (d) both before and after giving effect to such Acquisition and the incurrence of Indebtedness in connection therewith, US Company and its Subsidiaries (including any Subsidiary acquired in such Acquisition) shall be Solvent;
- (e) no Default or Event of Default shall have occurred and be continuing or shall be created thereby and the Payment Conditions shall have been satisfied, each calculated on a pro forma basis (whether or not Section 10.24 hereof would then require compliance with such covenant) immediately after giving effect to such payments and any Indebtedness incurred in connection therewith, and Administrative Agent shall receive a certificate of US Company to such effect on the date on which such Acquisition is consummated;
- (f) if a Revolving Loan is to be made in connection with such Acquisition, Administrative Agent shall have received a Notice of Borrowing and, if US Company desires to include the assets to be acquired in the US Borrowing Base or the UK Borrowing Base for such Revolving Loan, a Borrowing Base Certificate:
- (g) as soon as reasonably practicable following consummation of the Acquisition, Administrative Agent shall have received such financing statements, filings, Certificates of Title and other Security Documents as required (or advisable in Administrative Agent's judgment) to create and perfect Liens on any assets to be acquired, including assets of any new Subsidiary, together with evidence (including Lien search results) satisfactory to Administrative Agent that such Liens are first and prior Liens subject only to Permitted Liens;
 - (h) all new Subsidiaries formed or acquired in such Permitted Acquisition shall be Wholly-Owned Subsidiaries, directly or indirectly, of US Company;

- (i) the business and assets to be acquired in such Acquisition shall be acquired free and clear of all Liens (other than Permitted Liens);
- (j) any new Domestic Subsidiary shall become a Borrower or Guarantor and all new Domestic Subsidiaries shall execute and deliver to Administrative Agent such Security Documents as are required to be executed by a Guarantor (or joinder agreements in form and substance satisfactory to Administrative Agent) and such other documents as are necessary (or advisable in Administrative Agent's judgment) under Applicable Law in order to grant Administrative Agent for the benefit of the Lenders a perfected First Priority security interest and Lien in the assets of, and ownership interests in, such Subsidiary (subject only to Permitted Liens); and US Company or its Subsidiary, as applicable, shall execute and deliver an amendment to the Pledge Agreement in form and substance satisfactory to Administrative Agent, together with stock certificates and promissory notes and other instruments endorsed in blank in accordance therewith;
- (k) prior to inclusion of any assets in the US Borrowing Base or the UK Borrowing Base which causes the US Borrowing Base or the UK Borrowing Base to increase by more than \$20,000,000 or the Equivalent Amount thereof, Administrative Agent shall have received appraisals, in form and substance reasonably satisfactory to Administrative Agent, of all Inventory, Equipment and Real Property to be included in the Borrowing Base and shall have completed such review of Accounts, Inventory, Equipment and Real Property as it deems necessary or desirable for inclusion in the applicable Borrowing Base:
- (l) the Person or business to be acquired is engaged in the business conducted by US Company and its Subsidiaries immediately prior to the Initial Borrowing Date or similar activities related or incidental thereto; and
- (m) in the case of any Acquisition with a purchase price of \$10,000,000 or the Equivalent Amount thereof or more, on or prior to the date of such Acquisition, Administrative Agent shall have received, in form and substance satisfactory to Administrative Agent, all acquisition documents related thereto and certificates, and other documents and instruments reasonably requested by Administrative Agent, which collectively shall confirm, to Administrative Agent's reasonable satisfaction, that the conditions set forth herein have been satisfied.
- 10.15 <u>Modifications of Certificate of Incorporation, By-Laws and Certain Other Agreements, etc</u>. US Company will not, and will not permit any of its Subsidiaries to:
- (a) amend, modify, change or waive any term or provision of any Senior Note Document, unless such amendment, modification, change or waiver could not reasonably be expected to be adverse to the interests of the Lenders;
- (b) amend, modify, change or waive any term or provision of any Acquisition Document unless such amendment, modification, change or waiver could not reasonably be expected to be adverse to the interests of the Lenders in any material respect; or
- (c) amend, modify or change its certificate or articles of incorporation (including, without limitation, by the filing or modification of any certificate or articles of designation), certificate of formation, limited liability company agreement or by-laws (or the equivalent organizational documents), as applicable, or any agreement entered into by it with respect to its capital stock or other Equity Interests (including the Convertible Preferred Stock, any shareholders' agreement or any Qualified Preferred Stock), or enter into any new agreement with respect to its capital stock or other Equity Interests, unless

such amendment, modification, change or other action contemplated by this clause (c) could not reasonably be expected to be adverse to the interests of the Lenders in any material respect and the terms of any such amendment, modification, change or other action will not violate any of the other provisions of this Agreement or any other Credit Document.

10.16 Limitation on Certain Restrictions on Subsidiaries. US Company will not, and will not permit any of its Subsidiaries to, directly or indirectly, create or otherwise cause or suffer to exist or become effective any encumbrance or restriction on the ability of any such Subsidiary to (a) pay dividends or make any other distributions on its capital stock or any other Equity Interest or participation in its profits owned by US Company or any of its Subsidiaries, or pay any Indebtedness owed to US Company or any of its Subsidiaries, (b) make loans or advances to US Company or any of its Subsidiaries or (c) transfer any of its properties or assets to US Company or any of its Subsidiaries, except for such encumbrances or restrictions existing under or by reason of (i) Applicable Law, (ii) this Agreement and the other Credit Documents, (iii) customary provisions restricting subletting or assignment of any lease governing any leasehold interest of US Company or any of its Subsidiaries, (iv) customary provisions restricting assignment of any licensing agreement (in which US Company or any of its Subsidiaries is the licensee) or other contract entered into by US Company or any of its Subsidiaries in the ordinary course of business, and (v) restrictions on the transfer of any asset pending the close of the sale of such asset.

10.17 <u>Limitation on Issuance of Equity Interests</u>. (a) US Company will not, and will not permit any of its Subsidiaries to, issue (i) any Preferred Equity other than issuance by US Company of the Convertible Preferred Stock or Qualified Preferred Stock or (ii) any redeemable common stock or other redeemable common Equity Interests other than common stock or other redeemable common Equity Interests that is or are redeemable at the sole option of US Company or such Subsidiary, as the case may be.

(b) US Company will not permit any of its Subsidiaries to issue any capital stock or other Equity Interests (including by way of sales of treasury stock) or any options or warrants to purchase, or securities convertible into, capital stock or other Equity Interests, except (i) for transfers and replacements of then outstanding shares of capital stock or other Equity Interests, (ii) for stock splits, stock dividends and other issuances which do not decrease the percentage ownership of US Company or any of its Subsidiaries in any class of the capital stock or other Equity Interests of such Subsidiary, (iii) in the case of Foreign Subsidiaries of US Company, to qualifying directors to the extent required by Applicable Law and for other nominal share issuances to Persons other than US Company and its Subsidiaries to the extent required under Applicable Law and (iv) for issuances by Subsidiaries of US Company which are newly created or acquired in accordance with the terms of this Agreement.

10.18 <u>Business; etc</u>. US Company will not, and will not permit any of its Subsidiaries to, engage directly or indirectly in any business other than the businesses engaged in by US Company and its Subsidiaries as of the Initial Borrowing Date and reasonable extensions thereof and businesses ancillary or complimentary thereto.

10.19 [Reserved].

10.20 No Additional Deposit Accounts; etc. US Company will not, and will not permit any other Credit Party to, directly or indirectly, open, maintain or otherwise have any checking, savings, deposit, securities or other accounts at any bank or other financial institution where cash or Cash Equivalents are or may be deposited or maintained with any Person, other than (a) the Concentration Accounts set forth on Part A of Schedule 10.20, (b) the Collection Accounts set forth on Part B of Schedule 10.20, (c) the Disbursement Accounts set forth on Part C of Schedule 10.20, (d) the

Designated Petty Cash Accounts set forth on Part D of Schedule 10.20, (e) the Designated Payroll Accounts set forth on Part E of Schedule 10.20 and (f) the Foreign Accounts set forth on Part F of Schedule 10.20; provided that any US Credit Party and any UK Subsidiary may open a new Concentration Account, Collection Account, Disbursement Account, Designated Petty Cash Account, Designated Payroll Account or LKE Joint Account not set forth in such Schedule 10.20, so long as prior to opening any such account (i) the Administrative Agent has consented in writing to such opening (which consent shall not be unreasonably withheld or delayed), (ii) US Company has delivered an updated Schedule 10.20 to the Administrative Agent listing such new account and (iii) in the case of any new Concentration Account, Collection Account, Disbursement Account, Designated Petty Cash Account, Designated Payroll Account or LKE Joint Account, (A) opened by a US Credit Party, the financial institution with which such account is opened, together with US Company or such other US Credit Party which has opened such account and the Collateral Agent have executed and delivered to the Administrative Agent a Cash Management Control Agreement reasonably acceptable to the Administrative Agent if and to the extent required by the US Security Agreement and (B) opened by a UK Subsidiary, such Subsidiary has complied with its obligations under clauses 4.1 and 10 of the UK Debenture.

10.21 [Reserved].

10.22 <u>Tax Consolidation</u>. US Company shall not file or consent to the filing of any consolidated income tax return with any Person other than US Company's Subsidiaries.

10.23 Fiscal Year End. US Company shall not change, or permit any Subsidiary of US Company to change, its fiscal year end.

10.24 <u>Applicability of Financial Covenants</u>. During the term of this agreement, and thereafter for so long as there are any Obligations outstanding, US Company covenants that if a Compliance Period is in effect, then as of the end of the most recent fiscal quarter for which financial statements shall have been required to be delivered pursuant to <u>Section 9.01(b)</u> and (c) and as of the end of each subsequent fiscal quarter until a Compliance Period is not in effect, US Company shall comply with all of the financial covenants set forth in <u>Sections 10.25</u>, <u>10.26</u> and <u>10.27</u> hereof. If GAAP changes from the basis used in preparing the audited financial statements delivered to Administrative Agent by US Company on or before the Initial Borrowing Date, US Company will provide Administrative Agent with certificates demonstrating compliance with such financial covenants and will include, at the election of US Company or upon the request of Administrative Agent, calculations setting forth the adjustments necessary to demonstrate how US Company is in compliance with such financial covenants based upon GAAP as in effect on the Initial Borrowing Date.

10.25 <u>Fixed Charge Coverage Ratio</u>. Subject to the provisions of <u>Section 10.24</u>, as of the end of each fiscal quarter set forth below, US Company and its Subsidiaries shall maintain a Fixed Charge Coverage Ratio of not less than the ratio set forth below opposite such date:

| Fiscal Quarter Ended on | Fixed Charge Coverage Ratio |
|----------------------------|--------------------------------|
| June 30, 2008 | 2.00:1.00 |
| September 30, 2008 | 2.00:1.00 |
| December 31, 2008 | 2.00:1.00 |

| Fiscal Quarter Ended on | Fixed Charge Coverage Ratio |
|-------------------------------|--------------------------------|
| March 31, 2009 | 2.00:1.00 |
| June 30, 2009 | 2.00:1.00 |
| September 30, 2009 | 2.25:1.00 |
| December 31, 2009 | 2.25:1.00 |
| March 31, 2010 and thereafter | 2.50:1.00 |

10.26 <u>Debt Ratio</u>. Subject to the provisions of <u>Section 10.24</u>, as of the end of each fiscal quarter, US Company and its Subsidiaries shall maintain a Debt Ratio of not more than the ratio set forth below opposite such date:

| Fiscal Quarter | |
|-------------------------------|------------|
| Ended on | Debt Ratio |
| June 30, 2008 | 5.00:1.00 |
| September 30, 2008 | 5.00:1.00 |
| December 31, 2008 | 5.00:1.00 |
| March 31, 2009 | 5.00:1.00 |
| June 30, 2009 | 5.00:1.00 |
| September 30, 2009 | 4.75:1.00 |
| December 31, 2009 | 4.75:1.00 |
| March 31, 2010 and thereafter | 4.50:1.00 |

10.27 <u>Minimum Utilization</u>. Subject to the provisions of <u>Section 10.24</u>:

- (a) US Company and the other Credit Parties shall maintain minimum utilization rates for each fiscal quarter, calculated at the end of each such quarter as the average amount during such quarter, and calculated as the number of units of Eligible Container Fleet Inventory of US Company and the other Credit Parties which is then subject to valid, current rental or lease agreements between US Company or a Credit Party and the renters or lessees thereof, divided by the aggregate number of units of Eligible Container Fleet Inventory of US Company and the other Credit Parties, of not less than seventy-six percent (76%) for any fiscal quarter; and
- (b) (i) the number of units of the Eligible Container Fleet Inventory of US Company and the other Credit Parties which is then subject to valid, current rental or lease agreements

between US Company or a Credit Party and the renters or lessees thereof, <u>divided by</u> (ii) sum of (A) the aggregate number of units of the Eligible Container Fleet Inventory of US Company and the Credit Parties, and (B) the number of units of the Eligible Container Inventory Held For Sale of US Company and the other Credit Parties, of not less than seventy-one percent (71%) in any fiscal quarter; <u>provided</u>, that for the purposes of calculation of compliance with this <u>Section 10.27</u>, the aggregate of the number of units of Eligible Container Inventory Held For Sale, as a percentage of the sum of clauses (A) and (B) above, shall not exceed five percent (5%).

SECTION 11. Events of Default. The occurrence of one or more of the following events shall constitute an "Event of Default":

- 11.01 <u>Payment of Obligations</u>. Any Credit Party shall fail to pay (i) any principal or interest on the Loans or any reimbursement obligation in respect of any Letter of Credit on the due date thereof or (ii) any other Obligation within two (2) Business Days of the date the same becomes due and payable;
- 11.02 <u>Misrepresentations</u>. Any representation, warranty or other statement made or furnished to Administrative Agent or any Lender by or on behalf of US Company, or any other Credit Party in this Agreement, any of the other Credit Documents or any instrument, certificate or financial statement furnished in compliance with or in reference thereto proves to have been false or misleading in any material respect when made, furnished or remade;
- 11.03 <u>Breach of Specific Covenants</u>. US Company shall (i) fail or neglect to perform, keep or observe any covenant contained in <u>Sections 9.03</u> (Maintenance of Property; Insurance), <u>9.02</u> (Books, Records and Inspections; Field Examinations; Appraisals), <u>9.01(j)</u> (Borrowing Base Certificates), <u>9.22</u> (Completion of Mobile Storage Acquisition) and Section 10 (Negative Covenants) hereof on the date that US Company is required to perform, keep or observe such covenant or (ii) fail or neglect to perform, keep or observe any covenant contained in <u>Section 9.01</u> (Information Covenants) other than 9.01(j) hereof within ten (10) Business Days following the date on which US Company is required to perform, keep or observe such covenant;
- 11.04 <u>Breach of Other Covenants</u>. Any Credit Party shall fail or neglect to perform, keep or observe any covenant contained in this Agreement (other than a covenant which is dealt with specifically elsewhere in <u>Section 11</u> hereof) or any other Credit Document and the breach of such other covenant is not cured to Administrative Agent's satisfaction by the earlier to occur of ten (10) Business Days after (i) the date US Company or such Subsidiary thereof or Credit Party knew or should have known of such occurrence and (ii) the date of giving of notice thereof by Administrative Agent to US Company.
- 11.05 <u>Default Under Other Agreements</u>. A default or event of default shall occur (and continue beyond any applicable grace period) under any note, agreement or instrument evidencing any other Indebtedness of US Company or any of its Subsidiaries, which default or event of default permits the acceleration of its maturity, <u>provided</u> that the aggregate principal amount of all such Indebtedness for which the default or event of default has occurred exceeds \$20,000,000 or the Equivalent Amount thereof;
- 11.06 <u>Failure of Enforceability of Credit Documents; Security.</u> Any material covenant, agreement or Obligation of US Company or any other Credit Party contained in or evidenced by any of the Credit Documents shall cease to be enforceable, or shall be determined to be unenforceable, in accordance with its terms; US Company or any other Credit Party shall deny or disaffirm any of its material Obligations under any of the Credit Documents or any Liens granted in

connection therewith; or, any Liens granted in any of the Collateral shall be determined to be void, voidable, invalid or unperfected, are subordinated or not given the priority contemplated by this Agreement (except where such circumstance arises as a result of any action or inaction by any Lender);

11.07 [Reserved].

11.08 Insolvency and Related Proceedings, US Company, any Subsidiary of US Company or any other Credit Party shall cease to be Solvent or shall suffer the appointment of a receiver, trustee, custodian or similar fiduciary, or shall make an assignment for the benefit of creditors, or any petition for an order for relief shall be filed by or against a US Credit Party under the federal bankruptcy laws and such proceeding shall continue for more than 30 days, or a Credit Party shall make any offer of settlement, extension or composition to their respective unsecured creditors generally; or, with respect to the UK Subsidiaries: (in addition to the preceding provisions of this Section 11.08 such provisions not to be deemed to otherwise limit the following) (i) such UK Subsidiary stops or suspends or threatens or announces an intention to stop or suspend payment of its debts or is for the purpose of section 123(1) of the Insolvency Act 1986 of England and Wales (on the basis that the words "proved to the satisfaction of the court" are deemed omitted from section 123(1)(e)) or any other Applicable Law deemed to be unable or shall admit in writing its inability to pay its debts as they fall due or shall become insolvent or a moratorium is declared in respect of its indebtedness; (ii) a petition is presented or meeting convened or application made for the purpose of appointing an administrator (either in or out of court) or receiver or other similar officer of, or for the making of an administration order in respect of, any UK Subsidiary and (A) (other than in the case of a petition to appoint an administrator) such petition or application is not discharged within 14 days; or (B) in the case of a petition to appoint an administrator, the Administrative Agent is not satisfied that it will be discharged before it is heard; (iii) any UK Subsidiary convenes a meeting of its creditors generally or proposes or makes any arrangement or composition with, or any assignment for the benefit of, its creditors generally; (iv) any UK Subsidiary enters into any negotiations for or in connection with the re-scheduling, restructuring or readjustment of any Indebtedness by reason of, or with a view to avoiding, financial difficulties; (v) any meeting of any UK Subsidiary is convened for the purpose of considering any resolution for (or to petition for) its winding up or any UK Subsidiary passes such a resolution; (vi) a petition is presented for the winding-up of any UK Subsidiary (other than a frivolous or vexatious petition discharged within 14 days of being presented or any other petition which is contested on bona fide grounds and discharged at least 7 days before its hearing date); or (vi) any order is made or resolution passed or other action taken for the suspension of payments, protection from creditors or bankruptcy or insolvency of any UK Subsidiary;

11.09 <u>Business Disruption</u>; <u>Condemnation</u>. There shall occur a cessation of a substantial part of the business of US Company, any Subsidiary of US Company or any Credit Party for a period which materially adversely affects the capacity of US Company and its Subsidiaries, taken as a whole, to continue its business on a profitable basis; or US Company, any Subsidiary of US Company or any Credit party shall suffer the loss or revocation of any material license or permit now held or hereafter acquired by US Company, any Subsidiary of US Company or any Credit Party which is necessary to the continued or lawful operation of its business; or US Company, any Subsidiary of US Company or any Credit Party shall be enjoined, restrained or in any way prevented by court, governmental or administrative order from conducting all or any material part of its business affairs; or any material lease or agreement pursuant to which US Company, any Subsidiary of US Company or any Credit Party leases, uses or occupies any Property shall be canceled or terminated prior to the expiration of its stated term, except any such lease or agreement the cancellation or termination of which would not reasonably be expected to have a Material Adverse Effect; or any material portion of the Collateral shall be taken through condemnation or the value of such Property shall be impaired through condemnation;

11.10 ERISA. The occurrence of any of the following: any Plan shall fail to satisfy the minimum funding standard required for any plan year or part thereof under Section 412 of the Code or Section 302 of ERISA or a waiver of such standard or extension of any amortization period is sought or granted under Section 412 of the Code or Section 303 or 304 of ERISA, a contribution failure occurs with respect to any Plan sufficient to give rise to a lien under Section 302(f) of ERISA on assets of any Credit Party or any ERISA Affiliate, a Reportable Event shall have occurred, a contributing sponsor (as defined in Section 4001(a)(13) of ERISA) of a Plan subject to Title IV of ERISA shall be subject to the advance reporting requirement of PBGC Regulation Section 4043.61 (without regard to subparagraph (b)(1) thereof) and an event described in subsection .62, .63, .64, .65, .66, .67 or .68 of PBGC Regulation Section 4043 shall be reasonably expected to occur with respect to such Plan within the following 30 days, any Plan which is subject to Title IV of ERISA shall have had or is likely to have a trustee appointed to administer such Plan, any Plan which is subject to Title IV of ERISA or any Multiemployer Plan is or shall have been or is likely to be terminated or to be the subject of termination proceedings under ERISA, any Plan shall have an Unfunded Current Liability, a contribution required to be made with respect to a Plan or a Foreign Pension Plan has not been timely made, US Company or any Subsidiary of US Company or any ERISA Affiliate has incurred or is likely to incur any liability to or on account of a Plan under Section 406, 409, 502(i), 502(i), 4062, 4063, 4064 or 4069 of ERISA or Section 401(a)(29), 4971 or 4975 of the Code or on account of a Multiemployer Plan under Section 515, 4201, 4204 or 4212 of ERISA or on account of a group health plan (as defined in Section 607(1) of ERISA, Section 4980B(g)(2) of the Code or 45 Code of Federal Regulations Section 160.103) under Section 4980B of the Code and/or the Health Insurance Portability and Accountability Act of 1996, or US Company or any Subsidiary of US Company has incurred or is likely to incur liabilities pursuant to one or more employee welfare benefit plans (as defined in Section 3(1) of ERISA) that provide benefits to retired employees or other former employees (other than as required by Section 601 of ERISA) a "default," within the meaning of Section 4219(c)(5) of ERISA, shall occur with respect to any Plan; any Applicable Law, rule or regulation is adopted, changed or interpreted, or the interpretation or administration thereof is changed, in each case after the date hereof, by any Governmental Authority (a "Change in Law"), or, as a result of a Change in Law, an event occurs following a Change in Law, with respect to or otherwise affecting any Plan; in each case, which event or events, either individually or in the aggregate, has had, or could reasonably be expected to have, a Material Adverse Effect;

- 11.11 <u>Guarantee</u>. Any Guarantee or any provision thereof shall cease to be in full force or effect as to any Guarantor (except as a result of a release of any Guarantor in accordance with the terms thereof), or any Guarantor or any Person acting for or on behalf of such Guarantor shall deny or disaffirm such Guarantor's obligations under the Guarantee to which it is a party or any Guarantor shall default in the due performance or observance of any term, covenant or agreement on its part to be performed or observed pursuant to the Guarantee to which it is a party;
- 11.12 <u>Criminal Forfeiture</u>. US Company, any Subsidiary of US Company or any Credit Party shall be criminally indicted or convicted under any law that could lead to a forfeiture of any Property of US Company, any Subsidiary of US Company or any other Credit Party;
- 11.13 <u>Judgments</u>. Any money judgments, writ of attachment or similar processes (collectively, "<u>Judgments</u>") are issued or rendered against US Company, any Subsidiary of US Company or any other Credit Party, or any of their respective Property (i) in the case of money judgments in an amount of \$10,000,000 or the Equivalent Amount thereof or more for any single judgment, attachment or process or \$20,000,000 or the Equivalent Amount thereof or more for all such judgments, attachments or processes in the aggregate, in each case in excess of any applicable insurance with respect to which the insurer has admitted liability, and (ii) in the case of non-monetary Judgments, such Judgments (in the aggregate) would reasonably be expected to have a Material Adverse Effect, in each case which Judgment is not stayed, released or discharged within 30 days; or

11.14 Change of Control. A Change of Control shall occur;

then, and in any such event, and at any time thereafter, if any Event of Default shall then be continuing, the Administrative Agent, upon the written request of the Required Lenders, shall by written notice to the US Company, take any or all of the following actions, without prejudice to the rights of the Administrative Agent, any Lender or the holder of any Note to enforce its claims against any Credit Party (provided that, if an Event of Default specified in Section 11.08 shall occur with respect to any Credit Party, the result which would occur upon the giving of written notice by the Administrative Agent as specified in clauses (a) and (b) below, shall occur automatically without the giving of any such notice): (a) declare the Total Revolving Loan Commitment terminated, whereupon the Revolving Loan Commitment of each Lender shall forthwith terminate immediately and any Commitment Commission shall forthwith become due and payable without any other notice of any kind; (b) declare the principal of and any accrued interest in respect of all Loans and the Notes and all Obligations owing hereunder and thereunder to be, whereupon the same shall become, forthwith due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby waived by each Credit Party; (c) terminate any Letter of Credit which may be terminated in accordance with its terms; (d) direct the Borrowers to pay (and the Borrowers jointly and severally agree that upon receipt of such notice, or upon the occurrence of an Event of Default specified in Section 11.08 with respect to any Borrower, they will pay) to the Collateral Agent at the Payment Office such additional amount of cash or Cash Equivalents, to be held as security by the Collateral Agent, as is equal to the aggregate Stated Amount of all Letters of Credit issued for the account of the Borrowers and then outstanding; (e) enforce, as Collateral Agent, all of the Liens and security interests created pursuant to the Security Documents; and (f) apply any c

SECTION 12. The Administrative Agent and the Collateral Agent.

12.01 <u>Appointment</u>. The Lenders hereby irrevocably designate and appoint Deutsche Bank AG New York Branch as Administrative Agent (for purposes of this <u>Section 12</u> and <u>Section 13.01</u>, the term "Administrative Agent" also shall include DBNY in its capacity as Collateral Agent pursuant to the Security Documents) to act as specified herein and in the other Credit Documents. Each Lender hereby irrevocably authorizes, and each holder of any Note by the acceptance of such Note shall be deemed irrevocably to authorize the Administrative Agent to take such action on its behalf under the provisions of this Agreement, the other Credit Documents and any other instruments and agreements referred to herein or therein and to exercise such powers and to perform such duties hereunder and thereunder as are specifically delegated to or required of the Administrative Agent by the terms hereof and thereof and such other powers as are reasonably incidental thereto. The Lenders hereby authorize the Collateral Agent (and any sub-agents appointed in accordance with Section 12.13 below), at its option and in its sole discretion, to execute powers of attorney in favor of US Company and/or its Subsidiaries with respect to the release of Liens on Collateral constituting Motor Vehicles which are sold in accordance with the terms of the Credit Documents. The Administrative Agent may perform any of its respective duties hereunder or under any other Credit Document by or through its officers, directors, agents, employees or affiliates.

12.02 Nature of Duties. (a) The Administrative Agent shall not have any duties or responsibilities except those expressly set forth in this Agreement and in the other Credit Documents. Neither the Administrative Agent nor any of its officers, directors, agents, employees or affiliates shall be liable for any action taken or omitted by it or them hereunder or under any other Credit Document or in connection herewith or therewith, unless caused by its or their gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable decision). The duties of the Administrative Agent shall be mechanical and administrative in nature; the Administrative Agent shall not have by reason of this Agreement or any other Credit Document a

fiduciary relationship in respect of any Lender or the holder of any Note; and nothing in this Agreement or in any other Credit Document, expressed or implied, is intended to or shall be so construed as to impose upon the Administrative Agent any obligations in respect of this Agreement or any other Credit Document except as expressly set forth herein or therein.

(b) Notwithstanding any other provision of this Agreement or any provision of any other Credit Document, each of the Lead Arrangers, the Joint Bookrunners, the Syndication Agent and the Co-Documentation Agents are named as such for recognition purposes only, and each in its respective capacity as such shall have no powers, duties, responsibilities or liabilities with respect to this Agreement or the other Credit Documents or the transactions contemplated hereby and thereby except as explicitly provided for herein; it being understood and agreed that each of the Lead Arrangers, the Joint Bookrunners, the Syndication Agent and Co-Documentation Agents shall be entitled to all indemnification and reimbursement rights in favor of the Administrative Agent as, and to the extent, provided for under Sections 12.06 and 13.01. Without limitation of the foregoing, each of the Lead Arrangers, the Joint Bookrunners, the Syndication Agent and Co-Documentation Agents shall not, solely by reason of this Agreement or any other Credit Documents, have any fiduciary relationship in respect of any Lender or any other Person.

12.03 Lack of Reliance on the Administrative Agent and the Collateral Agent. Independently and without reliance upon the Administrative Agent, each Lender and the holder of each Note, to the extent it deems appropriate, has made and shall continue to make (a) its own independent investigation of the financial condition and affairs of US Company and its Subsidiaries in connection with the making and the continuance of the Loans and the participation in the Letters of Credit and the taking or not taking of any action in connection herewith and (b) its own appraisal of the creditworthiness of US Company and its Subsidiaries and, except as expressly provided in this Agreement, the Administrative Agent shall not have any duty or responsibility, either initially or on a continuing basis, to provide any Lender or the holder of any Note with any credit or other information with respect thereto, whether coming into its possession before the making of the Loans or the issuance of any Letter of Credit or at any time or times thereafter. The Administrative Agent shall not be responsible to any Lender or the holder of any Note for any recitals, statements, information, representations or warranties herein or in any document, certificate or other writing delivered in connection herewith or for the execution, effectiveness, genuineness, validity, enforceability, perfection, collectibility, priority or sufficiency of this Agreement or any other Credit Document or the financial condition of US Company or any of its Subsidiaries or be required to make any inquiry concerning either the performance or observance of any of the terms, provisions or conditions of this Agreement or any other Credit Document, or the financial condition of US Company or any of its Subsidiaries or the existence or possible existence of any Default or Event of Default.

12.04 <u>Certain Rights of the Agents</u>. If the Administrative Agent shall request instructions from the Required Lenders with respect to any act or action (including failure to act) in connection with this Agreement or any other Credit Document, the Administrative Agent shall be entitled to refrain from such act or taking such action unless and until the Administrative Agent shall have received instructions from the Required Lenders; and the Administrative Agent shall not incur liability to any Lender by reason of so refraining. Without limiting the foregoing, neither any Lender nor the holder of any Note shall have any right of action whatsoever against the Administrative Agent as a result of the Administrative Agent acting or refraining from acting hereunder or under any other Credit Document in accordance with the instructions of the Required Lenders.

12.05 <u>Reliance</u>. The Administrative Agent shall be entitle d to rely, and shall be fully protected in relying, upon any note, writing, resolution, notice, statement, certificate, telex, teletype or telecopier message, cablegram, radiogram, order or other document or telephone message

signed, sent or made by any Person that the Administrative Agent believed to be the proper Person, and, with respect to all legal matters pertaining to this Agreement and any other Credit Document and its duties hereunder and thereunder, upon advice of counsel selected by the Administrative Agent.

12.06 <u>Indemnification</u>. To the extent the Administrative Agent (or any affiliate thereof) is not reimbursed and indemnified by US Company or the Borrowers, the Lenders will reimburse and indemnify the Administrative Agent (and any affiliate thereof) in proportion to their respective "percentage" as used in determining the Required Lenders (determined as if there were no Defaulting Lenders) for and against any and all liabilities, obligations, losses, damages, penalties, claims, actions, judgments, costs, expenses or disbursements of whatsoever kind or nature which may be imposed on, asserted against or incurred by the Administrative Agent (or any affiliate thereof) in performing its respective duties hereunder or under any other Credit Document or in any way relating to or arising out of this Agreement or any other Credit Document; <u>provided</u> that no Lender shall be liable for any portion of such liabilities, obligations, losses, damages, penalties, claims, actions, judgments, suits, costs, expenses or disbursements resulting from the Administrative Agent's (or such affiliates' thereof) gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable decision).

12.07 The Administrative Agent in its Individual Capacities. With respect to their obligation to make Loans, or issue or participate in Letters of Credit, under this Agreement, the Administrative Agent shall have the rights and powers specified herein for a "Lender" and may exercise the same rights and powers as though it were not performing the duties specified herein; and the term "Lender," "Required Lenders," "Supermajority Lenders," "holders of Notes" or any similar terms shall, unless the context clearly indicates otherwise, include the Administrative Agent in its respective individual capacities. The Administrative Agent and its affiliates may accept deposits from, lend money to, and generally engage in any kind of banking, investment banking, trust or other business with, or provide debt financing, equity capital or other services (including financial advisory services) to any Credit Party or any Affiliate of any Credit Party (or any Person engaged in a similar business with any Credit Party or any Affiliate thereof) as if they were not performing the duties specified herein, and may accept fees and other consideration from any Credit Party or any Affiliate of any Credit Party for services in connection with this Agreement and otherwise without having to account for the same to the Lenders.

12.08 <u>Holders</u>. The Administrative Agent may deem and treat the payee of any Note as the owner thereof for all purposes hereof unless and until a written notice of the assignment, transfer or endorsement thereof, as the case may be, shall have been filed with the Administrative Agent. Any request, authority or consent of any Person who, at the time of making such request or giving such authority or consent, is the holder of any Note shall be conclusive and binding on any subsequent holder, transferee, assignee or endorsee, as the case may be, of such Note or of any Note or Notes issued in exchange therefor.

12.09 Resignation by the Administrative Agent. (a) The Administrative Agent may resign from the performance of all its respective functions and duties hereunder and/or under the other Credit Documents at any time by giving 30 days prior written notice to the Lenders and, unless a Default or an Event of Default under Section 11.08 then exists, the US Company. Any such resignation by an Administrative Agent hereunder shall also constitute its resignation as an Issuing Lender and the Swingline Lender, in which case the resigning Administrative Agent (x) shall not be required to issue any further Letters of Credit or make any additional Swingline Loans hereunder and (y) shall maintain all of its rights as Issuing Lender or Swingline Lender, as the case may be, with respect to any Letters of Credit issued by it, or Swingline Loans made by it, prior to the date

of such resignation. Such resignation shall take effect upon the appointment of a successor Administrative Agent pursuant to clauses (b) and (c) below or as otherwise provided below.

- (b) Upon any such notice of resignation by the Administrative Agent, the Required Lenders shall appoint a successor Administrative Agent hereunder and under the other Credit Documents who shall be a commercial bank or trust company reasonably acceptable to US Company, which acceptance shall not be unreasonably withheld or delayed (<u>provided</u> that US Company's approval shall not be required if an Event of Default then exists).
- (c) If a successor Administrative Agent shall not have been so appointed within such 30 day period, the Administrative Agent, with the consent of US Company (which consent shall not be unreasonably withheld or delayed, <u>provided</u> that US Company's consent shall not be required if an Event of Default then exists), shall then appoint a successor Administrative Agent who shall serve as Administrative Agent hereunder and under the other Credit Documents until such time, if any, as the Required Lenders appoint a successor Administrative Agent as provided above.
- (d) If no successor Administrative Agent has been appointed pursuant to clause (b) or (c) above by the 40th day after the date such notice of resignation was given by the Administrative Agent, the Administrative Agent's resignation shall become effective and the Required Lenders shall thereafter perform all the duties of the Administrative Agent hereunder and/or under any other Credit Document until such time, if any, as the Required Lenders appoint a successor Administrative Agent as provided above.
- (e) Upon a resignation of the Administrative Agent pursuant to this <u>Section 12.09</u>, the Administrative Agent shall remain indemnified to the extent provided in this Agreement and the other Credit Documents and the provisions of this <u>Section 12</u> (and the analogous provisions of the other Credit Documents) shall continue in effect for the benefit of the Administrative Agent for all of its actions and inactions while serving as the Administrative Agent hereunder and under the other Credit Documents.
- 12.10 Collateral Matters. (a) Each Lender authorizes and directs the Collateral Agent to enter into the Security Documents for the benefit of the Lenders and the other Secured Creditors. Each Lender hereby agrees, and each holder of any Note by the acceptance thereof will be deemed to agree, that, except as otherwise set forth herein, any action taken by the Required Lenders in accordance with the provisions of this Agreement and the Security Documents, and the exercise by the Required Lenders of the powers set forth herein or therein, together with such other powers as are reasonably incidental thereto, shall be authorized and binding upon all of the Lenders. The Collateral Agent is hereby authorized on behalf of all of the Lenders, without the necessity of any notice to or further consent from any Lender, from time to time prior to an Event of Default, to take any action with respect to any Collateral or Security Documents which may be necessary to perfect and maintain perfected the security interest in and liens upon the Collateral granted pursuant to the Security Documents. Without prejudice to the foregoing, each of the Administrative Agent and the Lenders hereby acknowledges, agrees and accepts that the Collateral Agent holds Collateral which is the subject of the UK Security Agreements as trustee for and on behalf of the Lenders in accordance with the terms of the declaration of trust set out in the UK Security Agreements and that the terms of its appointment, and such trust, shall be as set out (or referred to) in the UK Security Agreements and this Agreement.
- (b) The Lenders hereby authorize the Collateral Agent, at its option and in its discretion, to release any Lien granted to or held by the Collateral Agent upon any Collateral (i) upon termination of the Total Revolving Loan Commitment (and all Letters of Credit) and payment and satisfaction of all of the Obligations (other than inchoate indemnification obligations) at any time arising under or in respect of this Agreement or the Credit Documents or the transactions contemplated hereby or

thereby, (ii) constituting property being sold or otherwise disposed of (to Persons other than US Company and its Subsidiaries) upon the sale or other disposition thereof in compliance with Section 10.03, (iii) if approved, authorized or ratified in writing by the Required Lenders (or all of the Lenders hereunder, to the extent required by Section 13.12(a)) or (iv) as otherwise may be expressly provided in the relevant Security Documents. Lenders hereby irrevocably authorize the Collateral Agent, at its option and in its discretion, to subordinate any Lien granted to Collateral Agent on Collateral if required by the holder of any Indebtedness (including Capitalized Lease Obligations) secured by Purchase Money Liens and Leases permitted hereunder or Liens permitted by Section 10.02(o). Upon request by the Administrative Agent at any time, the Lenders will confirm in writing the Collateral Agent's authority to release particular types or items of Collateral pursuant to this Section 12.10.

- (c) The Collateral Agent shall have no obligation whatsoever to the Lenders or to any other Person to assure that the Collateral exists or is owned by any Credit Party or is cared for, protected or insured or that the Liens granted to the Collateral Agent herein or pursuant hereto have been properly or sufficiently or lawfully created, perfected, protected or enforced or are entitled to any particular priority, or to exercise or to continue exercising at all or in any manner or under any duty of care, disclosure or fidelity any of the rights, authorities and powers granted or available to the Collateral Agent in this Section 12.10, or in any of the Security Documents, it being understood and agreed that in respect of the Collateral, or any act, omission or event related thereto, the Collateral Agent may act in any manner it may deem appropriate, in its sole discretion, given the Collateral Agent's own interest in the Collateral as one of the Lenders and that the Collateral Agent shall have no duty or liability whatsoever to the Lenders, except for its gross negligence or willful misconduct (as determined by a court of competent jurisdiction in a final and non-appealable decision).
- (d) The Administrative Agent shall hold the benefit of the Liens created by the Luxembourg Security Agreement as agent for the Secured Creditors in accordance with article 2 (4) of the Luxembourg law of 5 August 2005 on financial collateral arrangements.
- 12.11 <u>Delivery of Information</u>. Neither the Administrative Agent nor the Collateral Agent shall be required to deliver to any Lender originals or copies of any documents, instruments, notices, communications or other information received by the Administrative Agent or the Collateral Agent from any Credit Party, any Subsidiary thereof, the Required Lenders, any Lender or any other Person under or in connection with this Agreement or any other Credit Document except (a) as specifically provided in this Agreement or any other Credit Document and (b) as specifically requested from time to time in writing by any Lender with respect to a specific document, instrument, notice or other written communication received by and in the possession of the Administrative Agent or the Collateral Agent, as the case may be, at the time of receipt of such request and then only in accordance with such specific request.
- 12.12 Withholding. To the extent required by Applicable Law, the Administrative Agent may withhold from any payment to any Lender or other Person receiving a payment under the Credit Documents an amount equivalent to any applicable withholding tax. If the Internal Revenue Service or any other Governmental Authority asserts a claim that the Administrative Agent did not properly withhold tax from amounts paid to or for the account of any Lender or other Person receiving payment under the Credit Documents because such Lender or other Person failed to notify the Administrative Agent that withholding on payments was required, including, without limitation, because of a change in circumstances which rendered an exemption from or reduction of withholding tax ineffective or for any other reason, such Lender shall indemnify the Administrative Agent fully for all amounts paid, directly or indirectly, by the Administrative Agent as tax or otherwise, including any penalties or interest and together with all expenses incurred.

12.13 Delegation of Duties. Each of Administrative Agent and Collateral Agent may perform any and all of its duties and exercise its rights and powers under this Agreement or under any other Credit Document by or through any one or more sub-agents appointed by Administrative Agent or Collateral Agent, as applicable. Administrative Agent, Collateral Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Affiliates. The exculpatory, indemnification and other provisions of Section 12.02 and of Section 12.03 shall apply to any the Affiliates of Administrative Agent or Collateral Agent and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent and Collateral Agent. All of the rights, benefits, and privileges (including the exculpatory and indemnification provisions) of Section 12.02 and of Section 12.03 shall apply to any such sub-agent and to the Affiliates of any such sub-agent, and shall apply to their respective activities as sub-agent as if such sub-agent and Affiliates were named herein. Notwithstanding anything herein to the contrary, with respect to each sub-agent appointed by Administrative Agent or Collateral Agent, as applicable (i) such sub-agent shall be a third party beneficiary under this Agreement with respect to all such rights, benefits and privileges (including exculpatory rights and rights to indemnification) and shall have all of the rights and benefits of a third party beneficiary, including an independent right of action to enforce such rights, benefits and privileges (including exculpatory rights and rights to indemnification) directly, without the consent or joinder of any other Person, against any or all of Credit Parties and the Lenders, (ii) such rights, benefits and privileges (including exculpatory rights and rights to indemnification) shall not be modified or amended without the consent of such sub-agent, and (iii) such sub-agent shall only have obligations to Administrative Agent or Collateral Agent, as applicable, and not to any Credit Party, Lender or any other Person and no Credit Party, Lender or any other Person shall have any rights, directly or indirectly, as a third party beneficiary or otherwise, against such sub-agent.

12.14 Quebec Security. Without limiting the powers of the Collateral Agent hereunder or under any of the other Credit Documents, each Lender (for its benefit and the benefit of its Affiliates), each Issuing Lender, the Administrative Agent, the Collateral Agent and each other Agent (all such Lenders (for their benefit and the benefit of their respective Affiliates), Issuing Lender, Administrative Agent, Collateral Agent and other Agents are collectively called, for purposes of this Section 12.14, the "Quebec Secured Parties") hereby acknowledges and agrees that DBNY shall, for purposes of holding any security granted by any Borrower or by any Affiliate or Subsidiary of any Borrower on property pursuant to the laws of the Province of Quebec to secure obligations of such Borrower or such Affiliate or Subsidiary under any bond or debenture (the "Quebec Secured Obligations"), be the holder of an irrevocable power of attorney (fondé de pouvoir) (within the meaning of the Civil Code of Quebec) for all present and future Quebec Secured Parties and holders of any bond or debenture. Each of the Quebec Secured Parties, for itself and for all present and future Affiliates that are or may become Quebec Secured Parties hereby irrevocably constitutes, to the extent necessary, DBNY as the holder of an irrevocable power of attorney (fondé de pouvoir) (within the meaning of Article 2692 of the Civil Code of Quebec) in order to hold security granted by any of the Borrowers or by any of their Affiliates or Subsidiaries in the Province of Quebec to secure the Quebec Secured Obligations. Furthermore, each of the Quebec Secured Parties hereby appoints DBNY to act in the capacity of the holder and depositary of such bond or debenture on its own behalf as Collateral Agent and for and on behalf and for the benefit of all present and future Quebec Secured Parties. Each assignee (for itself and for all present and future Affiliates) of a Quebec Secured Party shall be deemed to have confirmed and ratified the constitution of the Collateral Agent as the holder of such irrevocable power of attorney (fondé de pouvoir) by execution of the relevant Assignment and Assumption Agreement or other relevant documentation relating to such assignment. Notwithstanding the provisions of Section 32 of the An Act respecting the special powers of legal persons (Quebec), DBNY may acquire and be the holder of any bond or debenture. The Borrowers hereby acknowledge that such bond or debenture constitutes a title of indebtedness, as such term is used in Article 2692 of the Civil Code of Quebec.

SECTION 13. Miscellaneous.

13.01 Payment of Expenses, etc. The Credit Parties hereby jointly and severally agree to: (a) whether or not the transactions herein contemplated are consummated, pay all reasonable out-of-pocket costs and expenses (including Expenses) of the Administrative Agent (including, without limitation, the reasonable fees and disbursements of Latham & Watkins LLP and the Administrative Agent's other counsel and consultants and the fees and expenses in connection with the appraisals and collateral examinations required pursuant to Section 9.02) in connection with the preparation, execution, delivery and administration of this Agreement and the other Credit Documents and the documents and instruments referred to herein and therein and any amendment, waiver or consent relating hereto or thereto, of the Administrative Agent and its Affiliates in connection with its or their syndication efforts with respect to this Agreement and of the Administrative Agent and, after the occurrence of an Event of Default, each of the Issuing Lenders and Lenders in connection with the enforcement of this Agreement and the other Credit Documents and the documents and instruments referred to herein and therein or in connection with any refinancing or restructuring of the credit arrangements provided under this Agreement in the nature of a "work-out" or pursuant to any insolvency or bankruptcy proceedings (including, in each case without limitation, the reasonable fees and disbursements of counsel and consultants for the Administrative Agent and, after the occurrence of an Event of Default, counsel for each of the Issuing Lenders and Lenders); (b) pay and hold the Administrative Agent, each of the Issuing Lenders and each of the Lenders harmless from and against any and all present and future stamp, excise and other similar documentary taxes (collectively, the "Other Taxes") with respect to the foregoing matters and save the Administrative Agent, each of the Issuing Lenders and each of the Lenders harmless from and against any and all liabilities with respect to or resulting from any delay or omission (other than to the extent attributable to the Administrative Agent, such Issuing Lender or such Lender) to pay such Other Taxes; and (c) indemnify the Administrative Agent, each Issuing Lender and each Lender, and each of their respective officers, directors, employees, representatives, agents, affiliates, trustees and investment advisors from and hold each of them harmless against any and all liabilities, obligations (including removal or remedial actions), losses, damages, penalties, claims, actions, judgments, suits, costs, expenses and disbursements (including reasonable attorneys' and consultants' fees and disbursements) incurred by, imposed on or assessed against any of them as a result of, or arising out of, or in any way related to, or by reason of, (i) any investigation, litigation or other proceeding (whether or not the Administrative Agent, any Issuing Lender or any Lender is a party thereto and whether or not such investigation, litigation or other proceeding is brought by or on behalf of any Credit Party) related to the entering into and/or performance of this Agreement or any other Credit Document or the use of any Letter of Credit or the proceeds of any Loans hereunder or the consummation of the Transactions or any other transactions contemplated herein or in any other Credit Document or the exercise of any of their rights or remedies provided herein or in the other Credit Documents, or (ii) the actual or alleged presence of Hazardous Materials in the air, surface water or groundwater or on the surface or subsurface of any Real Property at any time owned, leased or operated by US Company or any of its Subsidiaries, the generation, storage, transportation, handling or disposal of Hazardous Materials by US Company or any of its Subsidiaries at any location, whether or not owned, leased or operated by US Company or any of its Subsidiaries, the non-compliance by US Company or any of its Subsidiaries with any Environmental Law (including applicable permits thereunder) applicable to any Real Property, or any Environmental Claim asserted against US Company, any of its Subsidiaries or any Real Property at any time owned, leased or operated by US Company or any of its Subsidiaries, including, in each case, without limitation, the reasonable fees and disbursements of counsel and other consultants incurred in connection with any such investigation, litigation or other proceeding (but excluding any losses, liabilities, claims, damages or expenses to the extent incurred by reason of the gross negligence or willful misconduct of the Person to be indemnified (as determined by a court of competent jurisdiction in a final and nonappealable decision)). To the extent that the undertaking to indemnify, pay or hold harmless the Administrative Agent, any Issuing Lender or any Lender set forth in the preceding

sentence may be unenforceable because it is violative of any law or public policy, the Credit Parties jointly and severally shall make the maximum contribution to the payment and satisfaction of each of the indemnified liabilities which is permissible under Applicable Law. In addition, the Credit Parties jointly and severally agree to reimburse the Administrative Agent for all reasonable third party administrative, audit and monitory expenses incurred in connection with the Borrowing Base and determinations thereunder. For the avoidance of doubt, except with respect to Other Taxes, this <u>Section 13.01</u> shall not apply to Taxes which shall be governed by <u>Section 5.04</u> or <u>Section 5.05</u>, as applicable.

13.02 Right of Setoff. (a) In addition to any rights now or hereafter granted under Applicable Law or otherwise, and not by way of limitation of any such rights, upon the occurrence and during the continuance of an Event of Default, the Administrative Agent, the Collateral Agent, each Issuing Lender and each Lender is hereby authorized at any time or from time to time, without presentment, demand, protest or other notice of any kind to any Credit Party or to any other Person, any such notice being hereby expressly waived, to set off and to appropriate and apply any and all deposits (general or special) and any other Indebtedness at any time held or owing by the Administrative Agent, the Collateral Agent, such Issuing Lender or such Lender (including, without limitation, by branches and agencies of the Administrative Agent, the Collateral Agent, such Issuing Lender or such Lender wherever located) to or for the credit or the account of US Company or any of its Subsidiaries against and on account of the Obligations and liabilities of the Credit Parties to the Administrative Agent, the Collateral Agent, such Issuing Lender or such Lender under this Agreement or under any of the other Credit Documents, including, without limitation, all interests in Obligations purchased by such Lender pursuant to Section 13.04(b), and all other claims of any nature or description arising out of or connected with this Agreement or any other Credit Document, irrespective of whether or not the Administrative Agent, the Collateral Agent, such Issuing Lender or such Lender shall have made any demand hereunder and although said Obligations, liabilities or claims, or any of them, shall be contingent or unmatured; provided that notwithstanding anything to the contrary contained in this Section 13.02(a), the Administrative Agent and each Lender hereby waive all rights of setoff, whether granted hereunder, pursuant to any Applicable Law or otherwise, with respect to any LKE Proceeds.

(b) NOTWITHSTANDING THE FOREGOING SUBSECTION (a), AT ANY TIME THAT THE LOANS OR ANY OTHER OBLIGATION SHALL BE SECURED BY REAL PROPERTY LOCATED IN CALIFORNIA, NO LENDER SHALL EXERCISE A RIGHT OF SETOFF, LIEN OR COUNTERCLAIM OR TAKE ANY COURT OR ADMINISTRATIVE ACTION OR INSTITUTE ANY PROCEEDING TO ENFORCE ANY PROVISION OF THIS AGREEMENT OR ANY NOTE UNLESS IT IS TAKEN WITH THE CONSENT OF THE REQUIRED LENDERS OR APPROVED IN WRITING BY THE ADMINISTRATIVE AGENT, IF SUCH SETOFF OR ACTION OR PROCEEDING WOULD OR MIGHT (PURSUANT TO CALIFORNIA CODE OF CIVIL PROCEDURE SECTIONS 580a, 580b, 580d AND 726 OF THE CALIFORNIA CODE OF CIVIL PROCEDURE OR SECTION 2924 OF THE CALIFORNIA CIVIL CODE, IF APPLICABLE, OR OTHERWISE) AFFECT OR IMPAIR THE VALIDITY, PRIORITY OR ENFORCEABILITY OF THE LIENS GRANTED TO THE SECURITY AGENT PURSUANT TO THE SECURITY DOCUMENTS OR THE ENFORCEABILITY OF THE NOTES AND OTHER OBLIGATIONS HEREUNDER, AND ANY ATTEMPTED EXERCISE BY ANY LENDER OF ANY SUCH RIGHT WITHOUT OBTAINING SUCH CONSENT OF THE REQUIRED LENDERS OR THE ADMINISTRATIVE AGENT SHALL BE NULL AND VOID. THIS SUBSECTION (b) SHALL BE SOLELY FOR THE BENEFIT OF EACH OF THE LENDERS AND THE ADMINISTRATIVE AGENT HEREUNDER.

13.03 Notices. Except as otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (including telegraphic,

telecopier or cable communication) and mailed, telegraphed, telecopied, cabled or delivered: (a) if to any Credit Party, at the following address:

Mobile Mini, Inc. 7420 South Kyrene Road, Suite 101 Tempe, Arizona 85283 Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

With a copy to:

Squire, Sanders & Dempsey L.L.P. 40 North Central Avenue, Suite 2700 Phoenix, AZ 85004

Attention: Joseph P. Richardson

Phone: (602) 528-4801 Fax: (602) 253-8129

(b) if to any Lender, at its address specified on Schedule 13.03; and (c) if to the Administrative Agent, at the Notice Office; or, as to any Credit Party or the Administrative Agent, at such other address as shall be designated by such party in a written notice to the other parties hereto and, as to each Lender, at such other address as shall be designated by such Lender in a written notice to US Company and the Administrative Agent; provided that any notice to the Administrative Agent of the filing by a Credit Party of a document with the SEC pursuant to Section 9.01(g) or (h) may be given by email. All such notices and communications shall, when mailed, telegraphed, telecopied, or cabled or sent by overnight courier, be effective when deposited in the mails, delivered to the telegraph company, cable company or overnight courier, as the case may be, or sent by telecopier, except that notices and communications to the Administrative Agent, US Company and UK Company shall not be effective until received by the Administrative Agent, US Company or UK Company, as the case may be.

13.04 Benefit of Agreement; Assignments; Participations. (a) This Agreement shall be binding upon and inure to the benefit of and be enforceable by the respective successors and assigns of the parties hereto; provided, however, that neither US Company nor any other Credit Party may assign or transfer any of its rights, obligations or interest hereunder without the prior written consent of the Lenders and, provided further, that, although any Lender may transfer, assign or grant participations in its rights hereunder, such Lender shall remain a "Lender" for all purposes hereunder (and may not transfer or assign all or any portion of its Revolving Loan Commitment hereunder except as provided in Sections 2.13 and 13.04(b)) and the transferee, assignee or participant, as the case may be, shall not constitute a "Lender" hereunder and, provided further, that no Lender shall transfer or grant any participation under which the participant shall have rights to approve any amendment to or waiver of this Agreement or any other Credit Document except to the extent such amendment or waiver would (i) extend the final scheduled maturity of any Loan, Note or Letter of Credit (unless such Letter of Credit is not extended beyond the Final Maturity Date) in which such participant is participating, or reduce the rate or extend the time of payment of interest or Fees thereon (except in connection with a waiver of applicability of any post-default increase in interest rates) or reduce the principal amount thereof (it being understood that any amendment or modification to the financial definitions in this Agreement or to Section 13.07(a) shall not constitute a reduction in the rate of interest or Fees payable hereunder), or increase the amount of the participant's participation over the amount thereof then in effect (it being understood that a waiver of any Default or Event of Default or of a mandatory reduction in the Total Revolving Loan Commitment shall not constitute a change in the terms

of such participation, and that an increase in any Revolving Loan Commitment (or the available portion thereof) or Loan shall be permitted without the consent of any participant if the participant's participation is not increased as a result thereof), (ii) consent to the assignment or transfer by US Company or any Credit Party of any of its rights and obligations under this Agreement or (iii) release all or substantially all of the Collateral under all of the Security Documents (except as expressly provided in the Credit Documents) supporting the Loans or Letters of Credit hereunder in which such participant is participating. In the case of any such participation, the participant shall not have any rights under this Agreement or any of the other Credit Documents (the participant's rights against such Lender in respect of such participation to be those set forth in the agreement executed by such Lender in favor of the participant relating thereto) and all amounts payable by the Credit Parties hereunder shall be determined as if such Lender had not sold such participation. Notwithstanding the foregoing, a participant that holds a US Obligation shall be entitled to the benefits, and bound by the obligations, of Section 5.04 as though it were a Lender; provided, however, that such participant shall not be entitled to receive any greater payment under Section 5.04 than the applicable Lender would have been entitled to receive with respect to the participation transferred to such participant.

(b) Notwithstanding the foregoing, any Lender (or any Lender together with one or more other Lenders) may (x) assign all or a portion of its Revolving Loan Commitment and related outstanding Obligations (or, if the Revolving Loan Commitment has terminated, outstanding Obligations) hereunder to (i) (A) its parent company and/or any affiliate of such Lender which is at least 50% owned by such Lender or its parent company or (B) to one or more other Lenders or any affiliate of any such other Lender which is at least 50% owned by such other Lender or its parent company (provided that any fund that invests in loans and is managed or advised by the same investment advisor of another fund which is a Lender (or by an Affiliate of such investment advisor) shall be treated as an affiliate of such other Lender for the purposes of this sub-clause (x)(i)(B)), or (ii) in the case of any Lender that is a fund that invests in loans, any other fund that invests in loans and is managed or advised by the same investment advisor of any Lender or by an Affiliate of such investment advisor or (y) assign all, or if less than all, a portion equal to at least \$5,000,000 or the Equivalent Amount thereof in the aggregate for the assigning Lender or assigning Lenders, of such Revolving Loan Commitments and related outstanding Obligations (or, if the Revolving Loan Commitments have terminated, outstanding Obligations) hereunder to one or more Eligible Transferees (treating any fund that invests in loans and any other fund that invests in loans and is managed or advised by the same investment advisor of such fund or by an Affiliate of such investment advisor as a single Eligible Transferee), each of which assignees shall become a party to this Agreement as a Lender by execution of an Assignment and Assumption Agreement, provided that (v) at such time, Schedule 1.01(a) shall be deemed modified to reflect the Revolving Loan Commitments and/or outstanding Revolving Loans, as the case may be, of such new Lender and of the existing Lenders, (w) upon the surrender of the relevant Notes by the assigning Lender (or, upon such assigning Lender's indemnifying the Borrowers for any lost Note pursuant to a customary indemnification agreement) new Notes will be issued, at the Borrowers' joint and several expense, to such new Lender and to the assigning Lender upon the request of such new Lender or assigning Lender, such new Notes to be in conformity with the requirements of Section 2.05 (with appropriate modifications) to the extent needed to reflect the revised Revolving Loan Commitments and/or outstanding Revolving Loans, as the case may be, (x) the consent of the Administrative Agent and, so long as no Event of Default then exists and the Syndication Date shall have occurred, the US Company, shall be required in connection with any such assignment pursuant to clause (y) above (such consents, in any case, not to be unreasonably withheld, delayed or conditioned), (y) the Administrative Agent shall receive at the time of each such assignment, from the assigning or assignee Lender, the payment of a non-refundable assignment fee of \$3,500 and (z) no such transfer or assignment will be effective until recorded by the Administrative Agent on the Register pursuant to Section 13.15. To the extent of any assignment pursuant to this Section 13.04(b), the assigning Lender shall be relieved of its obligations hereunder with respect to its assigned Revolving Loan Commitment and outstanding Revolving Loans. At the time of

each assignment of a US Obligation pursuant to this <u>Section 13.04(b)</u> to a Person which is not already a Lender hereunder which is a Non-U.S. Lender, the respective assignee Lender shall, to the extent legally entitled to do so, provide to the US Company the appropriate Internal Revenue Service Forms (and, if applicable, a Section 5.04(b)(ii) Certificate) described in Section 5.04(b). To the extent that an assignment of all or any portion of a Lender's Revolving Loan Commitment and related outstanding Obligations pursuant to <u>Section 2.13</u> or this <u>Section 13.04(b)</u> would, at the time of such assignment, result in increased costs under <u>Section 2.10</u>, <u>3.06</u>, <u>5.04</u> or <u>5.05</u> from those being charged by the respective assigning Lender prior to such assignment, then the Borrowers shall not be obligated to pay such increased costs (although the Borrowers, in accordance with and pursuant to the other provisions of this Agreement, shall be obligated to pay any other increased costs of the type described above resulting from changes after the date of the respective assignment).

- (c) Nothing in this Agreement shall prevent or prohibit any Lender from pledging its Loans and Notes hereunder to a Federal Reserve Bank in support of borrowings made by such Lender from such Federal Reserve Bank and, with prior notification to the Administrative Agent (but without the consent of the Administrative Agent or US Company), any Lender which is a fund may pledge all or any portion of its Loans and Notes to its trustee or to a collateral agent providing credit or credit support to such Lender in support of its obligations to such trustee, such collateral agent or a holder of such obligations, as the case may be. No pledge pursuant to this clause (c) shall release the transferor Lender from any of its obligations hereunder.
- (d) Any Lender which assigns all of its Revolving Loan Commitment and/or Loans hereunder in accordance with <u>Section 13.04(b)</u> shall cease to constitute a "Lender" hereunder, except with respect to indemnification provisions under this Agreement (including, without limitation, <u>Sections 2.10</u>, <u>2.11</u>, <u>3.06</u>, <u>5.04</u>, <u>5.05</u>, <u>12.06</u>, <u>13.01</u> and <u>13.06</u>), which shall survive as to such assigning Lender.
- (e) In case of assignment, transfer or novation by the assigning Lender to the assignee of all or any part of its rights and obligations under any of the Revolving Loan Commitment, the assigning Lender and the assignee shall agree that, for the purposes of Article 1278 of the Luxembourg Civil Code (to the extent applicable), the Securities created under the Luxembourg Security Agreement, securing the rights assigned, transferred or novated thereby, will be preserved for the benefit of the assignee.

13.05 No Waiver; Remedies Cumulative. No failure or delay on the part of the Administrative Agent, the Collateral Agent, any Issuing Lender or any Lender in exercising any right, power or privilege hereunder or under any other Credit Document and no course of dealing between US Company or any other Credit Party and the Administrative Agent, the Collateral Agent, any Issuing Lender or any Lender shall operate as a waiver thereof; nor shall any single or partial exercise of any right, power or privilege hereunder or under any other Credit Document preclude any other or further exercise thereof or the exercise of any other right, power or privilege hereunder or thereunder. The rights, powers and remedies herein or in any other Credit Document expressly provided are cumulative and not exclusive of any rights, powers or remedies which the Administrative Agent, the Collateral Agent, any Issuing Lender or any Lender would otherwise have. No notice to or demand on any Credit Party in any case shall entitle any Credit Party to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the Administrative Agent, the Collateral Agent, any Issuing Lender or any Lender to any other or further action in any circumstances without notice or demand.

13.06 <u>Payments Pro Rata</u>. (a) Except as otherwise provided in this Agreement, the Administrative Agent agrees that promptly after its receipt of each payment from or on behalf of any Borrower in respect of any Obligations hereunder, the Administrative Agent shall distribute

such payment to the Lenders entitled thereto (other than any Lender that has consented in writing to waive its <u>pro rata</u> share of any such payment) <u>pro rata</u> based upon their respective shares, if any, of the Obligations with respect to which such payment was received.

- (b) Each of the Lenders agrees that, if it should receive any amount hereunder (whether by voluntary payment, by realization upon security, by the exercise of the right of setoff or banker's lien, by counterclaim or cross action, by the enforcement of any right under the Credit Documents, or otherwise), which is applicable to the payment of the principal of, or interest on, the Loans, Unpaid US Drawings, Unpaid UK Drawings, Commitment Commission or Letter of Credit Fees, of a sum which with respect to the related sum or sums received by other Lenders is in a greater proportion than the total of such Obligation then owed and due to such Lender bears to the total of such Obligation then owed and due to all of the Lenders immediately prior to such receipt, then such Lender receiving such excess payment shall purchase for cash without recourse or warranty from the other Lenders an interest in the Obligations of the respective Credit Party to such Lenders in such amount as shall result in a proportional participation by all the Lenders in such amount; provided that if all or any portion of such excess amount is thereafter recovered from such Lenders, such purchase shall be rescinded and the purchase price restored to the extent of such recovery, but without interest.
- (c) Notwithstanding anything to the contrary contained herein, the provisions of the preceding Sections <u>13.06(a)</u> and <u>(b)</u> shall be subject to the express provisions of this Agreement which require, or permit, differing payments to be made to Non-Defaulting Lenders as opposed to Defaulting Lenders.
- 13.07 <u>Calculations; Computations</u>. (a) The financial statements to be furnished to the Lenders pursuant hereto shall be made and prepared in accordance with GAAP consistently applied throughout the periods involved (except as set forth in the notes thereto or as otherwise disclosed in writing by US Company to the Lenders); <u>provided</u> that, (i) except as otherwise specifically provided herein, all computations shall utilize GAAP and policies in conformity with those used to prepare the audited financial statements of US Company referred to in <u>Section 8.10(a)</u> for the fiscal year ended December 31, 2007 and (ii) to the extent expressly provided herein, certain calculations shall be made on a <u>pro forma</u> basis.
- (b) All computations of interest, Commitment Commission and other Fees hereunder shall be made on the basis of a year of 360 days for the actual number of days (including the first day but excluding the last day; except that in the case of Letter of Credit Fees and Facing Fees, the last day shall be included) occurring in the period for which such interest, Commitment Commission or Fees are payable; provided, however, all computations of interest on Base Rate Loans determined by reference to the Prime Lending Rate shall be made on the basis of a year of 365/366 days (as applicable) for the actual number of days occurring in the period for which such interest is payable.
- (c) For purposes of the *Interest Act* (Canada), (i) whenever any interest or fee under this Agreement or any Note is calculated using a rate based on a year of 360 days, the rate determined pursuant to such calculation, when expressed as an annual rate, is equivalent to (x) the applicable rate based on a year of 360 days, (y) multiplied by the actual number of days in the calendar year in which the period for which such interest or fee is payable (or compounded) ends, and (z) divided by 360, (ii) the principle of deemed reinvestment of interest does not apply to any interest or fee calculation under this Agreement and any Note, and (iii) the rates of interest stipulated in this Agreement and any Note are intended to be nominal rates and not effective rates or yields.

13.08 <u>GOVERNING LAW; SUBMISSION TO JURISDICTION; VENUE; WAIVER OF JURY TRIAL</u>. (a) THIS AGREEMENT AND THE OTHER

CREDIT DOCUMENTS AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND THEREUNDER SHALL, EXCEPT AS OTHERWISE PROVIDED IN ANY MORTGAGE, BE CONSTRUED IN ACCORDANCE WITH AND BE GOVERNED BY THE LAW OF THE STATE OF NEW YORK (WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES). ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT MAY BE BROUGHT IN THE COURTS OF THE STATE OF NEW YORK OR OF THE UNITED STATES FOR THE SOUTHERN DISTRICT OF NEW YORK, IN EACH CASE WHICH ARE LOCATED IN THE COUNTY OF NEW YORK, AND, BY EXECUTION AND DELIVERY OF THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT, EACH BORROWER HEREBY IRREVOCABLY ACCEPTS FOR ITSELF AND IN RESPECT OF ITS PROPERTY, GENERALLY AND UNCONDITIONALLY, THE JURISDICTION OF THE AFORESAID COURTS. EACH BORROWER HEREBY FURTHER IRREVOCABLY WAIVES ANY CLAIM THAT ANY SUCH COURTS LACK PERSONAL JURISDICTION OVER SUCH CREDIT PARTY, AND AGREES NOT TO PLEAD OR CLAIM, IN ANY LEGAL ACTION PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT BROUGHT IN ANY OF THE AFOREMENTIONED COURTS, THAT SUCH COURTS LACK PERSONAL JURISDICTION OVER SUCH BORROWER. EACH BORROWER FURTHER IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OUT OF ANY OF THE AFOREMENTIONED COURTS IN ANY SUCH ACTION OR PROCEEDING BY THE MAILING OF COPIES THEREOF BY REGISTERED OR CERTIFIED MAIL, POSTAGE PREPAID, TO SUCH BORROWER AT ITS ADDRESS SET FORTH OPPOSITE ITS SIGNATURE BELOW, SUCH SERVICE TO BECOME EFFECTIVE 30 DAYS AFTER SUCH MAILING. EACH BORROWER HEREBY IRREVOCABLY WAIVES ANY OBJECTION TO SUCH SERVICE OF PROCESS AND FURTHER IRREVOCABLY WAIVES AND AGREES NOT TO PLEAD OR CLAIM IN ANY ACTION OR PROCEEDING COMMENCED HEREUNDER OR UNDER ANY OTHER CREDIT DOCUMENT THAT SERVICE OF PROCESS WAS IN ANY WAY INVALID OR INEFFECTIVE. NOTHING HEREIN SHALL AFFECT THE RIGHT OF THE ADMINISTRATIVE AGENT, THE COLLATERAL AGENT, ANY ISSUING LENDER ANY LENDER OR THE HOLDER OF ANY NOTE TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR TO COMMENCE LEGAL PROCEEDINGS OR OTHERWISE PROCEED AGAINST EACH BORROWER IN ANY OTHER JURISDICTION.

- (b) EACH BORROWER IRREVOCABLY WAIVES ANY OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY OF THE AFORESAID ACTIONS OR PROCEEDINGS ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT BROUGHT IN THE COURTS REFERRED TO IN CLAUSE (a) ABOVE AND HEREBY FURTHER IRREVOCABLY WAIVES AND AGREES NOT TO PLEAD OR CLAIM IN ANY SUCH COURT THAT ANY SUCH ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.
- (c) EACH OF THE PARTIES TO THIS AGREEMENT HEREBY IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE OTHER CREDIT DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY.
- 13.09 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and by the different parties hereto on separate counterparts, each of which when so executed and delivered shall be an original, but all of which shall together constitute one and the same instrument. A set of counterparts executed by all the parties hereto shall be lodged with US Company, the Borrowers and the Administrative Agent. Delivery of an executed counterpart hereof by facsimile or other electronic transmission shall be as effective as delivery of an original executed counterpart hereof.

- 13.10 Effectiveness. This Agreement shall become effective on the date (the "Effective Date") on which US Company, the Borrowers, the Administrative Agent and each of the Lenders shall have signed a counterpart hereof (whether the same or different counterparts) and shall have delivered the same to the Administrative Agent at the Notice Office or, in the case of the Lenders, shall have given to the Administrative Agent telephonic (confirmed in writing), written or telex notice (actually received) at such office that the same has been signed and mailed to it. The Administrative Agent will give US Company, US Company, the other Borrowers and each Lender prompt written notice of the occurrence of the Effective Date.
- 13.11 <u>Headings Descriptive</u>. The headings of the several sections and subsections of this Agreement are inserted for convenience only and shall not in any way affect the meaning or construction of any provision of this Agreement.
- 13.12 Amendment or Waiver; etc. (a) Neither this Agreement nor any other Credit Document nor any terms hereof or thereof may be changed, waived, discharged or terminated unless such change, waiver, discharge or termination is in writing signed by the respective Credit Parties party hereto or thereto and the Required Lenders although additional parties may be added to (and annexes may be modified to reflect such additions) and, other than as further set forth below. Subsidiaries of the US Company may be deleted from, the Guaranty and the Security Documents in accordance with the provisions hereof and thereof without the consent of the other Credit Parties party thereto or the Required Lenders, provided that no such change, waiver, discharge or termination shall, without the consent of each Lender (other than a Defaulting Lender) (with Obligations being directly affected in the case of following clause (i)), (i) extend the final scheduled maturity of any Loan or Note or extend the stated expiration date of any Letter of Credit beyond the Final Maturity Date, or reduce the rate or extend the time of payment of interest or Fees thereon (except in connection with the waiver of applicability of any post-default increase in interest rates), or reduce (or forgive) the principal amount thereof (it being understood that any amendment or modification to the financial definitions in this Agreement or to Section 13.07(a) shall not constitute a reduction in the rate of interest or Fees for the purposes of this clause (i)), (ii) release all or substantially all of the Collateral (except as expressly provided in the Credit Documents) under all the Security Documents, (iii) amend, modify or waive any provision of this Section 13.12(a) (except for technical amendments with respect to additional extensions of credit pursuant to this Agreement which afford the protections to such additional extensions of credit of the type provided to the Revolving Loan Commitments and the Loans on the Effective Date), (iv) reduce the "majority" voting threshold specified in the definition of Required Lenders (it being understood that, with the consent of the Required Lenders, additional extensions of credit pursuant to this Agreement may be included in the determination of the Required Lenders on substantially the same basis as the extensions of Revolving Loan Commitments are included on the Effective Date), (v) consent to the assignment or transfer by US Company or any Borrower of any of their rights and obligations under this Agreement, (vi) amend, change or modify this Agreement in any manner that would change the effect of Section 5.03(d), (vii) amend, change or modify the provisions of Section 2.17 or 5.02(a)(i), (iii), (iii) or (iv); provided further, that no such change, waiver, discharge or termination shall (1) increase the Revolving Loan Commitment of any Lender over the amount thereof then in effect without the consent of such Lender (it being understood that waivers or modifications of conditions precedent, covenants, Defaults or Events of Default or of a mandatory reduction in the Total Revolving Loan Commitment shall not constitute an increase of the Revolving Loan Commitment of any Lender, and that an increase in the available portion of the Revolving Loan Commitment of any Lender shall not constitute an increase of the Revolving Loan Commitment of such Lender), (2) without the consent of each Issuing Lender, amend, modify or waive any provision of Section 2 or alter its rights or obligations with respect to Letters of Credit, (3) without the consent of the Swingline Lender, alter the Swingline Lender's rights or obligations with respect to Swingline Loans, (4) without the consent of the Administrative Agent, amend, modify or waive any provision of Section 12 or any other provision of this Agreement or any other Credit Document as same

relates to the rights or obligations of the Administrative Agent, (5) without the consent of the Collateral Agent, amend, modify or waive any provision relating to the rights or obligations of the Collateral Agent, or (6) without the consent of the Supermajority Lenders, (w) releases any Guarantor with assets in the US Borrowing Base or UK Borrowing Base from any obligations arising under the Guarantees, (x) amend the definition of Supermajority Lenders (it being understood that, with the consent of the Required Lenders, additional extensions of credit pursuant to this Agreement may be included in the determination of the Supermajority Lenders on substantially the same basis as the extensions of Loans and Revolving Loan Commitments are included on the Effective Date), (y) amend the definition of Aggregate Borrowing Base, US Borrowing Base or UK Borrowing Base (or, in each case, any defined terms as used therein) as such definitions are set forth herein on the Effective Date (or as same may be amended from time to time pursuant to this clause (y)) in any manner which would have the effect of increasing availability thereunder as determined in good faith by the Administrative Agent (it being understood that the establishment, modification or elimination of Reserves and adjustment, establishment and elimination of criteria for Eligible Accounts, Eligible Container Fleet Inventory, Eligible Container Inventory Held For Sale, Eligible Goods Inventory, Eligible Machinery and Equipment, Eligible Raw Materials Inventory, Eligible Real Property, Eligible Trailer Fleet Inventory and Eligible Work-In-Process Container Inventory, in each case by the Administrative Agent or the Collateral Agent in accordance with the terms hereof, will not be deemed such an increase in advance rates) or (z) increase the percentage of the US Borrowing Base for which Agent Advances may be made pursuant to Section 2.01(h).

(b) If, in connection with any proposed change, waiver, discharge or termination of or to any of the provisions of this Agreement as contemplated by clauses (i) through (v), inclusive, of the first proviso to Section 13.12(a), the consent of the Required Lenders is obtained but the consent of one or more of such other Lenders whose consent is required is not obtained, then the Borrowers shall have the right, so long as all non-consenting Lenders whose individual consent is required are treated as described in either clause (A) or (B) below, to either (A) replace each such non-consenting Lender or Lenders with one or more Replacement Lenders pursuant to Section 2.13 so long as at the time of such replacement, each such Replacement Lender consents to the proposed change, waiver, discharge or termination or (B) terminate such non-consenting Lender's Revolving Loan Commitment and/or repay all outstanding Revolving Loans of such Lender and/or cash collateralize its applicable RL Percentage of the Letter of Credit of Outstandings in accordance with Sections 4.02(b) and/or 5.01(b), provided that, unless the Revolving Loan Commitments which are terminated and Revolving Loans which are repaid pursuant to preceding clause (B) are immediately replaced in full at such time through the addition of new Lenders or the increase of the Revolving Loan Commitments and/or outstanding Revolving Loans of existing Lenders (who in each case must specifically consent thereto), then in the case of any action pursuant to preceding clause (B), the Required Lenders (determined after giving effect to the proposed action) shall specifically consent thereto, provided further, that the Borrowers shall not have the right to replace a Lender, terminate its Revolving Loan Commitment or repay its Revolving Loans solely as a result of the exercise of such Lender's rights (and the withholding of any required consent by such Lender) pursuant to the second proviso to Section 13.12(a).

13.13 <u>Survival</u>. All indemnities set forth herein including, without limitation, in <u>Sections 2.10</u>, <u>2.11</u>, <u>3.06</u>, <u>5.04</u>, <u>5.05</u>, <u>12.06</u> and <u>13.01</u> shall survive the execution, delivery and termination of this Agreement and the Notes and the making and repayment of the Obligations.

13.14 <u>Domicile of Loans</u>. Each Lender may transfer and carry its Loans at, to or for the account of any office, Subsidiary or Affiliate of such Lender. Notwithstanding anything to the contrary contained herein, to the extent that a transfer of Loans pursuant to this <u>Section 13.14</u> would, at the time of such transfer, result in increased costs under <u>Section 2.10</u>, <u>2.11</u>, <u>3.06</u>, <u>5.04</u> or <u>5.05</u> from those being charged by the respective Lender prior to such transfer, then the Borrowers shall not be

obligated to pay such increased costs (although the Borrowers shall be jointly and severally obligated to pay any other increased costs of the type described above resulting from changes after the date of the respective transfer).

13.15 Register. The Borrowers hereby designate the Administrative Agent to serve as its agent, solely for purposes of this Section 13.15, to maintain a register (the "Register") on which it will record the Revolving Loan Commitments from time to time of each of the Lenders, the Loans made by each of the Lenders and each repayment in respect of the principal or interest amount of the Loans of each Lender. Failure to make any such recordation, or any error in such recordation, shall not affect the Borrowers' obligations in respect of such Loans. With respect to any Lender, the transfer of the Revolving Loan Commitment of such Lender and the rights to the principal of, and interest on, any Loan made pursuant to such Revolving Loan Commitment shall not be effective until such transfer is recorded on the Register maintained by the Administrative Agent with respect to ownership of such Revolving Loan Commitment and Loans and prior to such recordation all amounts owing to the transferor with respect to such Revolving Loan Commitment and Loans shall remain owing to the transferor. The registration of assignment or transfer of all or part of any Revolving Loan Commitments and Loans shall be recorded by the Administrative Agent on the Register only upon the acceptance by the Administrative Agent of a properly executed and delivered Assignment and Assumption Agreement pursuant to Section 13.04(b). Coincident with the delivery of such an Assignment and Assumption Agreement to the Administrative Agent for acceptance and registration of assignment or transfer of all or part of a Loan, or as soon thereafter as practicable, the assigning or transferor Lender shall surrender the Note (if any) evidencing such Loan, and thereupon one or more new Notes in the same aggregate principal amount shall be issued to the assigning or transferor Lender and/or the new Lender at the request of any such Lender. The Borrowers jointly and severally agree to indemnify the Administrative Agent from and against any and all losses, claims, damages and liabilities of whatsoever nature which may be imposed on, asserted against or incurred by the Administrative Agent in performing its duties under this Section 13.15. Upon the request of any Lender, the Administrative Agent shall provide such Lender with the Revolving Loan Commitment and outstanding Loan amount of such Lender as such information has been recorded in the Register.

13.16 <u>Confidentiality.</u> (a) Subject to the provisions of clause (b) of this <u>Section 13.16</u>, each Lender agrees that it will use its reasonable efforts not to disclose without the prior consent of the US Company (other than to its employees, auditors, advisors or counsel or to another Lender if such Lender or such Lender's holding or parent company in its sole discretion determines that any such party should have access to such information, <u>provided</u> such Persons shall be subject to the provisions of this <u>Section 13.16</u> to the same extent as such Lender) any information with respect to US Company or any of its Subsidiaries which is now or in the future furnished pursuant to this Agreement or any other Credit Document, <u>provided</u> that any Lender may disclose any such information (i) as has become generally available to the public other than by virtue of a breach of this <u>Section 13.16(a)</u> by the respective Lender, (ii) as may be required or appropriate in any report, statement or testimony submitted to any municipal, state or Federal regulatory body having or claiming to have jurisdiction over such Lender or to the Federal Reserve Board or the Federal Deposit Insurance Corporation or similar organizations (whether in the United States or elsewhere) or their successors, (iii) as may be required or appropriate in respect to any summons or subpoena or in connection with any litigation, (iv) in order to comply with any law, order, regulation or ruling applicable to such Lender, (v) to the Administrative Agent or the Collateral Agent, (vi) to any direct or indirect contractual counterparty in any swap, hedge or similar agreement (or to any such contractual counterparty's professional advisor), so long as such contractual counterparty (or such professional advisor) agrees to be bound by the provisions of this <u>Section 13.16</u> and (vii) to any prospective or actual transferee, pledgee or participant in connection with any contemplated transfer, pledge or participation of any of the Notes or Revolving Loan Commitments

or any interest therein by such Lender, <u>provided</u> that such prospective transferee, pledgee or participant agrees to be bound by the confidentiality provisions contained in this <u>Section 13.16</u>.

- (b) US Company and the Borrowers hereby acknowledge and agree that each Lender may share with any of its affiliates, and such affiliates may share with such Lender, any information related to US Company or any of its Subsidiaries (including, without limitation, any non-public customer information regarding the creditworthiness of US Company and its Subsidiaries), <u>provided</u> such Persons shall be subject to the provisions of this <u>Section 13.16</u> to the same extent as such Lender.
- 13.17 <u>Patriot Act</u>. Each Lender subject to the USA PATRIOT ACT (Title 111 of Pub. L. 107-56 (signed into law October 26, 2001)) (the "<u>Patriot Act</u>") hereby notifies US Company and the Borrowers that pursuant to the requirements of the Patriot Act, they are required to obtain, verify and record information that identifies US Company, the Borrowers and the other Credit Parties and other information that will allow such Lender to identify US Company, the Borrowers and the other Credit Parties in accordance with the Patriot Act.
- 13.18 <u>Release of Borrowers</u>. In the event that all of the Equity Interests of one or more Borrowers (other than the US Company) is sold or otherwise disposed of or liquidated in compliance with the requirements of <u>Section 10.03</u> (or such sale, other disposition or liquidation has been approved in writing by the Required Lenders (or all Lenders if required by <u>Section 13.12</u>)) and the proceeds of such sale, disposition or liquidation are applied in accordance with the provisions of <u>Section 5.02(c)</u>, to the extent applicable, such Borrower shall, upon consummation of such sale or other disposition (except to the extent that such sale or disposition is to US Company or any of its Wholly-Owned Subsidiaries), be released from this Agreement and each remaining Borrower agrees that, with no action on its part required, the remaining Borrowers shall remain jointly and severally liable for all Obligations.

SECTION 14. Nature of Borrower Obligations.

14.01 Nature of Borrower Obligations. Notwithstanding anything to the contrary contained elsewhere in this Agreement, it is understood and agreed by the various parties to this Agreement that all Obligations to repay principal of, interest on, and all other amounts with respect to, all Loans, Letters of Credit and all other Obligations pursuant to this Agreement and each other Credit Document (including, without limitation, all fees, indemnities, taxes and other Obligations in connection therewith or in connection with the related Revolving Loan Commitments) shall constitute the joint and several obligations of each of the Borrowers. In addition to the direct (and joint and several) obligations of the Borrowers with respect to Obligations as described above, all such Obligations shall be guaranteed pursuant to, and in accordance with the terms of, the Guaranties.

14.02 <u>Independent Obligation</u>. The obligations of each Borrower with respect to the Obligations are independent of the obligations of each other Borrower or any Guarantor under the Guaranty of such Obligations, and a separate action or actions may be brought and prosecuted against each Borrower, whether or not any other Borrower or any Guarantor is joined in any such action or actions. Each Borrower waives, to the fullest extent permitted by law, the benefit of any statute of limitations affecting its liability hereunder or the enforcement thereof. Any payment by any Borrower or other circumstance which operates to toll any statute of limitations as to any Borrower shall, to the fullest extent permitted by law, operate to toll the statute of limitations as to each Borrower.

14.03 <u>Authorization</u>. Each of the Borrowers authorizes the Administrative Agent, the Issuing Lenders and the Lenders without notice or demand (except as shall be

required by applicable statute and cannot be waived), and without affecting or impairing its liability hereunder, from time to time to:

- (a) exercise or refrain from exercising any rights against any other Borrower or any Guarantor or others or otherwise act or refrain from acting;
- (b) release or substitute any other Borrower, endorsers, Guarantors or other obligors;
- (c) settle or compromise any of the Obligations of any other Borrower or any other Credit Party, any security therefor or any liability (including any of those hereunder) incurred directly or indirectly in respect thereof or hereof, and may subordinate the payment of all or any part thereof to the payment of any liability (whether due or not) of any Borrower to its creditors other than the Lenders;
- (d) apply any sums paid by any other Borrower or any other Person, howsoever realized to any liability or liabilities of such other Borrower or other Person regardless of what liability or liabilities of such other Borrower or other Person remain unpaid; and/or
- (e) consent to or waive any breach of, or act, omission or default under, this Agreement or any of the instruments or agreements referred to herein, or otherwise, by any other Borrower or any other Person.
- 14.04 <u>Reliance</u>. It is not necessary for the Administrative Agent, the Collateral Agent, any Issuing Lender or any Lender to inquire into the capacity or powers of US Company, any Borrower or any of its Subsidiaries or the officers, directors, members, partners or agents acting or purporting to act on its behalf, and any Obligations made or created in reliance upon the professed exercise of such powers shall constitute the joint and several obligations of the Borrowers hereunder.
- 14.05 <u>Contribution</u>; <u>Subrogation</u>. No Borrower shall have any rights of contribution or subrogation with respect to any other Borrower as a result of payments made by it hereunder, in each case unless and until the Total Revolving Loan Commitment and all Letters of Credit have been terminated and all Obligations have been paid in full in cash.
- 14.06 <u>Waiver</u>. Each Borrower waives any right to require the Administrative Agent, the Collateral Agent, the Issuing Lenders or the Lenders to (i) proceed against any other Borrower, any Guarantor or any other party, (ii) proceed against or exhaust any security held from any Borrower, any Guarantor or any other party or (iii) pursue any other remedy in the Administrative Agent's, the Collateral Agent's, any Issuing Lender's or Lenders' power whatsoever. Each Borrower waives any defense based on or arising out of suretyship or any impairment of security held from any Borrower, any Guarantor or any other party or on or arising out of any defense of any other Borrower, any Guarantor or any other party other than payment in full in cash of the Obligations, including, without limitation, any defense based on or arising out of the disability of any other Borrower, any Guarantor or any other party, or the unenforceability of the Obligations or any part thereof from any cause, or the cessation from any cause of the liability of any other Borrower, in each case other than as a result of the payment in full in cash of the Obligations.

SECTION 15. Guarantee.

15.01 <u>The Guarantees</u>. (a) The US Guarantors hereby jointly and severally guarantee, as a primary obligor and not as a surety to each Secured Creditor and their respective

successors and assigns, the prompt payment in full when due (whether at stated maturity, by required prepayment, declaration, demand, by acceleration or otherwise) of the principal of and interest (including any interest, fees, costs or charges that would accrue but for the provisions of the Title 11 of the United States Code after any bankruptcy or insolvency petition under Title 11 of the United States Code) on the US Revolving Loans made by the Lenders to, and the corresponding Notes held by each Lender of, the US Borrowers, and all other US Obligations from time to time owing to the Secured Creditors by any US Credit Party (such obligations being herein collectively called the "US Guaranteed Obligations"). The US Guarantors hereby jointly and severally agree that if any US Borrower or any US Guarantor shall fail to pay in full when due (whether at stated maturity, by acceleration or otherwise) any of the US Guaranteed Obligations, the US Guarantors will promptly pay the same in cash, without any demand or notice whatsoever, and that in the case of any extension of time of payment or renewal of any of the US Guaranteed Obligations, the same will be promptly paid in full when due (whether at extended maturity, by acceleration or otherwise) in accordance with the terms of such extension or renewal.

(b) The UK Guarantors hereby jointly and severally guarantee, as a primary obligor and not as a surety to each Secured Creditor and their respective successors and assigns, the prompt payment in full when due (whether at stated maturity, by required prepayment, declaration, demand, by acceleration or otherwise) of the principal of and interest (including any interest, fees, costs or charges that would accrue but for the provisions of the Title 11 of the United States Code after any bankruptcy or insolvency petition under Title 11 of the United States Code) on the UK Revolving Loans made by the Lenders to, and the corresponding Notes held by each Lender of, each UK Borrower, and all other UK Obligations from time to time owing to the Secured Creditors by any UK Credit Party (such obligations being herein collectively called the "UK Guaranteed Obligations"). The UK Guarantors hereby jointly and severally agree that if any UK Borrower or other UK Guarantor shall fail to pay in full when due (whether at stated maturity, by acceleration or otherwise) any of the UK Guaranteed Obligations, the UK Guarantors will promptly pay the same in cash, without any demand or notice whatsoever, and that in the case of any extension of time of payment or renewal of any of the UK Guaranteed Obligations, the same will be promptly paid in full when due (whether at extended maturity, by acceleration or otherwise) in accordance with the terms of such extension or renewal.

15.02 Obligations Unconditional. (a) The obligations of the US Guarantors under Section 15.01(a) shall constitute a guaranty of payment (and not of collection) and to the fullest extent permitted by Applicable Law, are absolute, irrevocable and unconditional, joint and several, irrespective of the value, genuineness, validity, regularity or enforceability of the US Guaranteed Obligations under this Agreement, the corresponding Notes, if any, or any other agreement or instrument referred to herein or therein, or any substitution, release or exchange of any other guarantee of or security for any of the US Guaranteed Obligations, and, irrespective of any other circumstance whatsoever that might otherwise constitute a legal or equitable discharge or defense of a surety or US Guarantor (except for payment in full). Without limiting the generality of the foregoing, it is agreed that the occurrence of any one or more of the following shall not alter or impair the liability of the US Guarantors hereunder which shall remain absolute, irrevocable and unconditional under any and all circumstances as described above:

- (i) at any time or from time to time, without notice to the US Guarantors, the time for any performance of or compliance with any of the US Guaranteed Obligations shall be extended, or such performance or compliance shall be waived;
- (ii) any of the acts mentioned in any of the provisions of this Agreement or the Notes, if any, or any other agreement or instrument referred to herein or therein shall be done or omitted;

- (iii) the maturity of any of the US Guaranteed Obligations shall be accelerated, or any of the US Guaranteed Obligations shall be amended in any respect, or any right under the Credit Documents or any other agreement or instrument referred to herein or therein shall be amended or waived in any respect or any other guarantee of any of the US Guaranteed Obligations or any security therefor shall be released or exchanged in whole or in part or otherwise dealt with:
- (iv) any Lien or security interest granted to, or in favor of, any US Issuing Lender or any Lender or Collateral Agent as security for any of the US Guaranteed Obligations shall fail to be perfected; or
 - (v) the release of any other US Guarantor pursuant to Section 15.09(a).
- The US Guarantors hereby expressly waive diligence, presentment, demand of payment, protest and all notices whatsoever, and any requirement that any Secured Creditor exhaust any right, power or remedy or proceed against any US Borrower under this Agreement or the Notes, if any, or any other agreement or instrument referred to herein or therein, or against any other person under any other guarantee of, or security for, any of the US Guaranteed Obligations. The US Guarantors waive any and all notice of the creation, renewal, extension, waiver, termination or accrual of any of the US Guaranteed Obligations and notice of or proof of reliance by any Secured Creditor upon this US Guarantee or acceptance of this Guarantee, and the US Guaranteed Obligations, and any of them, shall conclusively be deemed to have been created, contracted or incurred in reliance upon this Guarantee, and all dealings between US Borrowers and the Secured Creditors shall likewise be conclusively presumed to have been had or consummated in reliance upon this Guarantee. This Guarantee shall be construed as a continuing, absolute, irrevocable and unconditional guarantee of payment (and not of collection) without regard to any right of offset with respect to the US Guaranteed Obligations at any time or from time to time held by Secured Creditors, and the obligations and liabilities of the US Guarantors hereunder shall not be conditioned or contingent upon the pursuit by the Secured Creditors or any other person at any time of any right or remedy against the US Borrowers or against any other person which may be or become liable in respect of all or any part of the US Guaranteed Obligations or against any collateral security or guarantee therefor or right of offset with respect thereto. This Guarantee shall remain in full force and effect and be binding in accordance with and to the extent of its terms upon the US Guarantors and the successors and assigns thereof, and shall inure to the benefit of the Lenders, and their respective successors and a
- (b) The obligations of the UK Guarantors under Section 15.01(b) shall constitute a guaranty of payment (and not of collection) and to the fullest extent permitted by Applicable Law, are absolute, irrevocable and unconditional, joint and several, irrespective of the value, genuineness, validity, regularity or enforceability of the UK Guaranteed Obligations under this Agreement, the corresponding Notes, if any, or any other agreement or instrument referred to herein or therein, or any substitution, release or exchange of any other guarantee of or security for any of the UK Guaranteed Obligations, and, irrespective of any other circumstance whatsoever that might otherwise constitute a legal or equitable discharge or defense of a surety or UK Guarantor (except for payment in full). Without limiting the generality of the foregoing, it is agreed that the occurrence of any one or more of the following shall not alter or impair the liability of the UK Guarantors hereunder which shall remain absolute, irrevocable and unconditional under any and all circumstances as described above:
 - (i) at any time or from time to time, without notice to the UK Guarantors, the time for any performance of or compliance with any of the UK Guaranteed Obligations shall be extended, or such performance or compliance shall be waived;

- (ii) any of the acts mentioned in any of the provisions of this Agreement or the Notes, if any, or any other agreement or instrument referred to herein or therein shall be done or omitted;
- (iii) the maturity of any of the UK Guaranteed Obligations shall be accelerated, or any of the UK Guaranteed Obligations shall be amended in any respect, or any right under the Credit Documents or any other agreement or instrument referred to herein or therein shall be amended or waived in any respect or any other guarantee of any of the UK Guaranteed Obligations or any security therefor shall be released or exchanged in whole or in part or otherwise dealt with:
- (iv) any Lien or security interest granted to, or in favor of, any UK Issuing Lender or any Lender or Collateral Agent as security for any of the UK Guaranteed Obligations shall fail to be perfected; or
 - (v) the release of any other UK Guarantor pursuant to Section 15.09(b).
- (c) The UK Guarantors hereby expressly waive diligence, presentment, demand of payment, protest and all notices whatsoever, and any requirement that any Secured Creditor exhaust any right, power or remedy or proceed against the UK Borrowers under this Agreement or the Notes, if any, or any other agreement or instrument referred to herein or therein, or against any other person under any other guarantee of, or security for, any of the UK Guaranteed Obligations. The UK Guarantors waive any and all notice of the creation, renewal, extension, waiver, termination or accrual of any of the UK Guaranteed Obligations and notice of or proof of reliance by any Secured Creditor upon this Guarantee or acceptance of this Guarantee, and the UK Guaranteed Obligations, and any of them, shall conclusively be deemed to have been created, contracted or incurred in reliance upon this Guarantee, and all dealings between any UK Borrower and the Secured Creditors shall likewise be conclusively presumed to have been had or consummated in reliance upon this Guarantee. This Guarantee shall be construed as a continuing, absolute, irrevocable and unconditional guarantee of payment (and not of collection) without regard to any right of offset with respect to the UK Guaranteed Obligations at any time or from time to time held by Secured Creditors, and the obligations and liabilities of the UK Guarantors hereunder shall not be conditioned or contingent upon the pursuit by the Secured Creditors or any other person at any time of any right or remedy against any UK Borrower or against any other person which may be or become liable in respect of all or any part of the UK Guaranteed Obligations or against any collateral security or guarantee therefor or right of offset with respect thereto. This Guarantee shall remain in full force and effect and be binding in accordance with and to the extent of its terms upon the UK Guarantors and the successors and assigns thereof, and shall inure to the benefit of the Lenders, and their respective successors a
- 15.03 <u>Reinstatement</u>. (a) The obligations of the US Guarantors under this Section 15 shall be automatically reinstated if and to the extent that for any reason any payment by or on behalf of US Borrower or any US Guarantor in respect of the US Guaranteed Obligations is rescinded or must be otherwise restored by any holder of any of the US Guaranteed Obligations, whether as a result of any proceedings in bankruptcy or reorganization or otherwise.
- (b) The obligations of the UK Guarantors under this <u>Section 15</u> shall be automatically reinstated if and to the extent that for any reason any payment by or on behalf of any of the UK Borrower or any UK Guarantor in respect of the UK Guaranteed Obligations is rescinded or must be otherwise restored by any holder of any of the UK Guaranteed Obligations, whether as a result of any proceedings in bankruptcy or reorganization or otherwise.

- 15.04 <u>Subrogation</u>; <u>Subordination</u>. (a) Each US Guarantor hereby agrees that until the indefeasible payment and satisfaction in full in cash of all US Guaranteed Obligations and the expiration and termination of the Revolving Loan Commitments of the Lenders under this Agreement it shall waive any claim and shall not exercise any right or remedy, direct or indirect, arising by reason of any performance by it of its guarantee in <u>Section 15.01(a)</u>, whether by subrogation or otherwise, against US Borrower or any other US Guarantor of any of the US Guaranteed Obligations or any security for any of the US Guaranteed Obligations. Any Indebtedness of any US Credit Party permitted pursuant to <u>Section 10.05(e)</u> shall be subordinated to such US Credit Party's US Obligations in the manner set forth in the intercompany note evidencing such Indebtedness (in a form reasonably acceptable to the Administrative Agent).
- (b) Each UK Guarantor hereby agrees that until the indefeasible payment and satisfaction in full in cash of all UK Guaranteed Obligations and the expiration and termination of the Revolving Loan Commitments of the Lenders to make UK Revolving Loans under this Agreement it shall waive any claim and shall not exercise any right or remedy, direct or indirect, arising by reason of any performance by it of its guarantee in Section 15.01(b), whether by subrogation or otherwise, against any of the UK Borrowers or any other UK Guarantor of any of the UK Guaranteed Obligations or any security for any of the UK Guaranteed Obligations. Any Indebtedness of any UK Credit Party permitted pursuant to Section 10.05(e) shall be subordinated to such UK Credit Party's UK Obligations in the manner set forth in the intercompany note evidencing such Indebtedness (in a form reasonably acceptable to the Administrative Agent).
- 15.05 <u>Remedies</u>. (a) The US Guarantors jointly and severally agree that, as between the US Guarantors and the Lenders, the obligations of a US Borrower under this Agreement and the corresponding Notes, if any, may be declared to be forthwith due and payable as provided in <u>Section 11</u> (and shall be deemed to have become automatically due and payable in the circumstances provided in <u>Section 11</u>) for purposes of <u>Section 15.01(a)</u>, notwithstanding any stay, injunction or other prohibition preventing such declaration (or such obligations from becoming automatically due and payable) as against any US Borrower and that, in the event of such declaration (or such obligations being deemed to have become automatically due and payable), such obligations (whether or not due and payable by US Borrower) shall forthwith become due and payable by the US Guarantors for purposes of <u>Section 15.01(a)</u>.
- (b) The UK Guarantors jointly and severally agree that, as between the UK Guarantors and the Lenders, the obligations of the UK Borrowers under this Agreement and the corresponding Notes, if any, may be declared to be forthwith due and payable as provided in Section 11 (and shall be deemed to have become automatically due and payable in the circumstances provided in Section 11) for purposes of Section 15.01(b), notwithstanding any stay, injunction or other prohibition preventing such declaration (or such obligations from becoming automatically due and payable) as against any UK Borrower and that, in the event of such declaration (or such obligations being deemed to have become automatically due and payable), such obligations (whether or not due and payable by a UK Borrower) shall forthwith become due and payable by the UK Guarantors for purposes of Section 15.01(b).
- 15.06 <u>Instrument for Payment of Money</u>. (a) Each US Guarantor hereby acknowledges that the guarantee in this Section 15 constitutes an instrument for the payment of money, and consents and agrees that any Lender or Administrative Agent, at its sole option, in the event of a dispute by such US Guarantor in the payment of any moneys due hereunder, shall have the right to bring a motion-action under New York CPLR Section 3213.

- (b) Each UK Guarantor hereby acknowledges that the guarantee in this Section 15 constitutes an instrument for the payment of money, and consents and agrees that any Lender or Administrative Agent, at its sole option, in the event of a dispute by such UK Guarantor in the payment of any moneys due hereunder, shall have the right to bring a motion-action under New York CPLR Section 3213.
- 15.07 <u>Continuing Guarantee</u>. (a) The guarantee in this <u>Section 15.01(a)</u> is a continuing guarantee of payment (and not of collection), and shall apply to all US Guaranteed Obligations whenever arising.
- (b) The guarantee in this <u>Section 15.01(b)</u> is a continuing guarantee of payment (and not of collection), and shall apply to all UK Guaranteed Obligations whenever arising.
- 15.08 General Limitation on Guarantee Obligations. (a) In any action or proceeding involving any state corporate limited partnership or limited liability company law, or any applicable state, federal or foreign bankruptcy, insolvency, reorganization or other law affecting the rights of creditors generally, if the obligations of any US Guarantor under Section 15.01(a) would otherwise be held or determined to be void, voidable, invalid or unenforceable, or subordinated to the claims of any other creditors, on account of the amount of its liability under Section 15.01(a), then, notwithstanding any other provision to the contrary, the amount of such liability shall, without any further action by such US Guarantor, any US Credit Party or any other person, be automatically limited and reduced to the highest amount (after giving effect to the right of contribution established in Section 15.10(a)) that is valid and enforceable and not subordinated to the claims of other creditors as determined in such action or proceeding.
- (b) In any action or proceeding involving any state corporate limited partnership or limited liability company law, or any applicable state, federal or foreign bankruptcy, insolvency, reorganization or other law affecting the rights of creditors generally, if the obligations of any UK Guarantor under Section 15.01(b) would otherwise be held or determined to be void, voidable, invalid or unenforceable, or subordinated to the claims of any other creditors, on account of the amount of its liability under Section 15.01(b), then, notwithstanding any other provision to the contrary, the amount of such liability shall, without any further action by such UK Guarantor, any UK Credit Party or any other person, be automatically limited and reduced to the highest amount (after giving effect to the right of contribution established in Section 15.10(b)) that is valid and enforceable and not subordinated to the claims of other creditors as determined in such action or proceeding.
- (c) The guarantee provided under Section 15.01(b) by any UK Guarantor organized under the laws of the Netherlands shall be limited to the extent required so that such guarantee does not constitute a breach of the financial assistance prohibitions contained in Section 2:98c or 2:207c of the Dutch Civil Code. For the avoidance of any doubt it is expressly acknowledged that such UK Guarantors will continue to provide a guarantee to the extent doing so does not constitute a violation of aforementioned Dutch financial assistance prohibitions.
- (d) The obligations of any UK Guarantor incorporated in England and Wales under this <u>Section 15</u> shall not extend beyond a point where they would cause the infringement of any of sections 151 to 158 (inclusive) of the United Kingdom Companies Act 1985 (as amended or otherwise re-enacted from time to time).
- (e) Notwithstanding any provision to the contrary contained in this agreement or in any Credit Document, the liability of the Luxembourg Subsidiary under this section 15 shall be limited at any time to a maximum aggregate amount equal to 90 per cent. of the greater of the Luxembourg Subsidiary's

own funds ("capitaux propres") as mentioned in its most recently approved financial statements, or existing as at the date of this agreement.

15.09 Release of Guarantors. (a) If, in compliance with the terms and provisions of the Credit Documents, all or substantially all of the Equity Interests or property of any US Guarantor are sold or otherwise transferred (a "US Transferred Guarantor") to a person or persons, none of which is a US Borrower or a Domestic Subsidiary of US Company, such US Transferred Guarantor shall, upon the consummation of such sale or transfer, be automatically released from its obligations under this Agreement and its obligations to pledge and grant any Collateral owned by it pursuant to any Security Document or Pledge Agreement and, in the case of a sale of all or substantially all of the Equity Interests of the US Transferred Guarantor, the pledge of such Equity Interests to the Collateral Agent pursuant to the Security Agreements or Pledge Agreements shall be automatically released, and the Collateral Agent shall take such actions as are necessary to effect each release described in this Section 15.09 in accordance with the relevant provisions of the Security Documents or Pledge Agreement, so long as US Company shall have provided the Administrative Agent such certifications or documents as Administrative Agent shall reasonably request in order to demonstrate compliance with this Agreement.

(b) If, in compliance with the terms and provisions of the Credit Documents, all or substantially all of the Equity Interests or property of any UK Guarantor are sold or otherwise transferred (a "<u>UK Transferred Guarantor</u>") to a person or persons, none of which is a UK Borrower or a UK Credit Party, such UK Transferred Guarantor shall, upon the consummation of such sale or transfer, be automatically released from its obligations under this Agreement and its obligations to pledge and grant any Collateral owned by it pursuant to any Security Document and, in the case of a sale of all or substantially all of the Equity Interests of the UK Transferred Guarantor, the pledge of such Equity Interests to the Collateral Agent pursuant to the Security Agreements or Pledge Agreement shall be automatically released, and the Collateral Agent shall take such actions as are necessary to effect each release described in this Section 15.09 in accordance with the relevant provisions of the Security Documents or Pledge Agreement, so long as the applicable Borrower shall have provided the Administrative Agent such certifications or documents as the Administrative Agent shall reasonably request in order to demonstrate compliance with this Agreement.

15.10 <u>Right of Contribution</u>. (a) Each US Guarantor hereby agrees that to the extent that a US Guarantor shall have paid more than its proportionate share of any payment made hereunder, such Guarantor shall be entitled to seek and receive contribution from and against any other US Guarantor hereunder which has not paid its proportionate share of such payment. Each US Guarantor's right of contribution shall be subject to the terms and conditions of Section 15.04(a). The provisions of this Section 15.10(a) shall in no respect limit the obligations and liabilities of any US Guarantor to the Secured Creditors, and each US Guarantor shall remain liable to the Secured Creditors for the full amount guaranteed by such US Guarantor hereunder.

(b) Each UK Guarantor hereby agrees that to the extent that a UK Guarantor shall have paid more than its proportionate share of any payment made hereunder, such Guarantor shall be entitled to seek and receive contribution from and against any other UK Guarantor hereunder which has not paid its proportionate share of such payment. Each UK Guarantor's right of contribution shall be subject to the terms and conditions of Section 15.04(b). The provisions of this Section 15.10(b) shall in no respect limit the obligations and liabilities of any UK Guarantor to the Secured Creditors, and each UK Guarantor shall remain liable to the Secured Creditors for the full amount guaranteed by such UK Guarantor hereunder.

Execution Copy

IN WITNESS WHEREOF, the parties hereto have caused their duly authorized officers to execute and deliver this Agreement as of the date first above written.

Address: **US BORROWERS**

Attention: Chief Financial Officer

Phone: (480) 894-6311

Fax: (480) 894-6433

Title: Executive Vice President **UK BORROWERS**

MOBILE MINI, INC.

Attention: Chief Financial Officer RAVENSTOCK MSG LIMITED

Phone: (480) 894-6311

Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311

Fax: (480) 894-6433

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

MOBILE MINI UK LIMITED

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg

Title: Director

US GUARANTORS

MSG WC HOLDINGS CORP.

Attention: Chief Financial Officer By: /s/ Allan A. Villegas

Phone: (480) 894-6311

Name: Allan A. Villegas
Fax: (480) 894-6433

Title: Chief Financial Officer

MSG WC INTERMEDIARY CO.

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433 MOBILE SERVICES GROUP, INC.

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

CACTUS MERGER SUB, INC.

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg Title: Executive Vice President

MOBILE STORAGE GROUP, INC.

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

MSG INVESTMENTS, INC.

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

A BETTER MOBILE STORAGE COMPANY

Attention: Chief Financial Officer

Phone: (480) 894-6311

Fax: (480) 894-6433

By: /s/ Allan A. Villegas

Name: Allan A. Villegas
Title: Chief Financial Officer

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433 MOBILE STORAGE GROUP (TEXAS), L.P.

By: MOBILE STORAGE GROUP, INC., its General

Partner

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

MOBILE MINI I, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg Title: Executive Vice President

DELIVERY DESIGN SYSTEMS, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg Title: Executive Vice President

MOBILE MINI, LLC

By: MOBILE MINI, INC., its Sole Member

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg Title: Executive Vice President

MOBILE MINI, LLC

By: MOBILE MINI, INC., its Sole Member

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg Title: Executive Vice President

MOBILE MINI OF OHIO, LLC

By: MOBILE MINI, INC., its Sole Member

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg Title: Executive Vice President

A ROYAL WOLF PORTABLE STORAGE, INC.

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433 By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg

Title: Secretary

TEMPORARY MOBILE STORAGE, INC.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg

Title: Secretary

UK GUARANTORS

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433 MOBILE STORAGE (UK) LIMITED

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

RAVENSTOCK (TAM) HIRE LIMITED

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

MOBILE STORAGE UK FINANCE, LP

By: MOBILE STORAGE GROUP, INC., its

General Partner

By: /s/ Allan A. Villegas

Name: Allan A. Villegas Title: Chief Financial Officer

LIKO LUXEMBOURG INTERNATIONAL

S.A.R.L.

By: /s/ Douglas A. Waugaman

Name: Douglas A. Waugaman Title: Category B Manager

MOBILE MINI UK HOLDINGS LIMITED

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg

Title: Director

BOX LEASE LIMITED

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg

Title: Director

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433

Attention: Chief Financial Officer

Phone: (480) 894-6311 Fax: (480) 894-6433 MOBILE MINI HOLDING B.V.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg Title: Managing Director

MOBILE MINI B.V.

By: /s/ Lawrence Trachtenberg

Name: Lawrence Trachtenberg Title: Managing Director

DEUTSCHE BANK AG NEW YORK BRANCH

Individually and as Administrative Agent

By: /s/ Marguerite Sutton
Name: Marguerite Sutton
Title: Director

By: /s/ Carin Keegan
Name: Carin Keegan Title: Director

ROYAL BANK OF CANADA

as Lender

By: /s/ Dustin Craven

Name: Dustin Craven
Title: Attorney-in-Fact

ALLIED IRISH BANKS, PLC

as Lender

By: /s/ Martin Chin

Name: Martin Chin Title: SVP

By: /s/ Mia Bolin

Name: Mia Bolin Title: AVP

HSBC BUSINESS CREDIT (USA) INC.

as Lender

By: /s/ Edward Chonko

Name: Edward Chonko Title: Vice President

SUMITOMO MITSUI BANKING CORPORATION

as Lender

By: /s/ Leo E. Pagarigan

Name: Leo E. Pagarigan Title: General Manager

BANK OF AMERICA, N.A.

as Lender

By: /s/ Jason Riley

Name: Jason Riley

Title: Senior Vice President

FIFTH THIRD BANK

as Lender

By: /s/ John T. Penny

Name: John T. Penny Title: Vice President

BURDALE CAPITAL FINANCE, INC. as Lender

By: /s/ Antimo Barbieri

Name: Antimo Barbieri Title: Senior Vice President

By: /s/ Philip R. Webb

Name: Philip R. Webb Title: Director

JPMORGAN CHASE BANK, N.A.

as Lender

By: /s/ Timothy J. Whitefoot

Name: Timothy J. Whitefoot
Title: Vice President

THE BANK OF NOVA SCOTIA

as Lender

By: /s/ Diane Emanuel

Name: Diane Emanuel Title: Director

NORTHERN TRUST BANK

as Lender

By: /s/ Morgan A. Lyons

Name: Morgan A. Lyons Title: Vice President

RBS BUSINESS CAPITAL, A DIVISION OF RBS ASSET

FINANCE, INC. as Lender

By: /s/ James H. Herzog, Jr.

Name: James H. Herzog, Jr.
Title: Senior Vice President

WELLS FARGO FOOTHILL, LLC

as Lender

By: /s/ Rohan Damani

Name: Rohan Damani Title: Vice President

SIEMENS FINANCIAL SERVICES, INC.

as Lender

By: /s/ John Finore

Name: John Finore
Title: Vice President

By: /s/ David Kantes

Name: David Kantes

Title: Senior Vice President

ING CAPITAL LLC

as Lender

By: /s/ William C. Beddingfield

Name: William C. Beddingfield Title: Managing Director

CAPITAL ONE LEVERAGE FINANCE CORP.

as Lender

By: /s/ Thomas F. Furst

Name: Thomas F. Furst Title: Vice President

ISRAEL DISCOUNT BANK OF NEW YORK

as Lender

By: /s/ Edward R. Behnen

Name: Edward R. Behnen
Title: Assistant Vice President

By: /s/ Jeffrey Ackerman

Name: Jeffrey Ackerman Title: Senior Vice President

KBC BANK N.V.

as Lender

By: /s/ Edward Eijlers

Name: Edward Eijlers Title: Vice President

By: /s/ Sandra T. Johnson

Name: Sandra T. Johnson Title: Managing Director

UPS CAPITAL CORPORATION

as Lender

By: /s/ John P. Holloway

Name: John P. Holloway

Title: Director of Portfolio Management

BANK OF ARIZONA, N.A.

as Lender

By: /s/ Christine A. Nowaczyk

Name: Christine A. Nowaczyk
Title: Senior Vice President

PNC BANK, NATIONAL ASSOCIATION

as Lender

By: /s/ Lawrence Weinstein

Name: Lawrence Weinstein Title: Vice President

BANK LEUMI USA

as Lender

By: /s/ Jacques Delvoye

Name: Jacques Delvoye

Title: FVP

BANK OF THE WEST

as Lender

By: /s/ Cecile Segovia

Name: Cecile Segovia Title: Vice President



FOR IMMEDIATE RELEASE

MOBILE MINI, INC. COMPLETES MERGER WITH

MOBILE STORAGE GROUP, INC.

Announces Appointment of Three New Directors

Tempe, AZ – June 27, 2008 — Mobile Mini, Inc. (Nasdaq GS: MINI), the world's leading provider of portable storage solutions, today announced that it has completed its merger with Mobile Storage Group, Inc. of Glendale, California, following approval at a Special Meeting of Stockholders on June 26, 2008 and securing a new \$900 million asset-based revolving credit facility to fund the cash portion of the transaction, including the assumption of Mobile Storage's outstanding debt.

Commenting on the merger, Steven Bunger, Chairman, President & Chief Executive Officer of Mobile Mini, stated, "This transaction has established Mobile Mini as the undisputed world leader of portable storage solutions with 79 branches in the US, three in Canada, 20 in the UK and one in The Netherlands. As such, we expect to continue providing our customers with leading portable storage solutions, our employees with unparalleled career opportunities, and our shareholders with industry leading growth. Financially, realizing on the consolidation synergies and capital expenditure reductions achievable through the merger will allow us to significantly increase operating free cash flow and thereby drive shareholder value through both growth and debt paydown. Since February, we have been fine tuning plans and timetables for branch consolidation and integration, and our initial expectations for cost synergies of at least \$25 million on an annualized basis appear to be fully realizable by the end of fiscal 2009. We remain confident that the combined business will be solidly accretive to our earnings in the first full year of the combination."

The merged company includes senior executives from both companies. Steven G. Bunger continues to serve as Chairman, President and Chief Executive Officer and Lawrence Trachtenberg as Mobile Mini's Chief Financial Officer and Executive Vice President. Doug Waugaman, CEO of Mobile Storage Group, joins Mobile Mini as COO of Integration, reporting directly to Mr. Bunger. Jody Miller, Bill Armstead and Ron Halchishak, senior executives at Mobile Storage Group, have all been named Senior Vice Presidents of the combined organization.

"The merger with Mobile Storage Group will benefit our customers, employees and stockholders by broadening our geographic base and allowing us to cover most major markets for portable storage in both the US and the UK. The addition of Mobile Storage Group's talented and experienced employees, as well as its significant resources, gives us a strategic advantage in servicing an expanded customer base," Mr. Bunger continued.

Upon closing, Mobile Mini assumed Mobile Storage Group's outstanding indebtedness and paid cash totaling approximately \$562 million and issued 8,555,556 shares of Mobile Mini preferred stock with a liquidation preference of \$154 million. As such, the transaction is valued at approximately \$716 million.

(more)

Mobile Mini News Release
June 27, 2008
Page 2

The preferred stock is initially convertible into 8,555,556 shares of Mobile Mini's common stock for a fully diluted ownership in Mobile Mini of approximately 20%, as of March 31, 2008. In addition, the preferred stock will be mandatorily convertible into Mobile Mini common stock if, after the first year following the issuance of the preferred stock, Mobile Mini's common stock trades above \$23.00 per share for a period of 30 consecutive days. The preferred stock will not have any cash or payment-in-kind dividends, will impose no covenants upon Mobile Mini, and will include an optional redemption feature following the tenth year after the issue date.

In connection with merger, the Company entered into a new \$900 million asset-based revolving credit facility, fully underwritten by Deutsche Bank AG, Bank of America and JP Morgan. Oppenheimer & Co. Inc. and Deutsche Bank Securities Inc. acted as financial advisors to Mobile Mini, and White & Case LLP acted as legal counsel. Lehman Brothers Inc. acted as financial advisor to Mobile Storage Group and Kirkland & Ellis LLP acted as legal counsel. Following the closing of the transaction, including repayment of Mobile Storage Group's indebtedness, Mobile Mini currently has availability of over \$280 million remaining under its new revolving credit facility.

Mobile Mini expects to issue revised 2008 pro forma guidance on or before August 7th when second quarter results are scheduled to be announced.

Appointment of New Directors

Following the consummation of the merger transaction, Mobile Mini expanded the size of its Board of Directors from six to eight by appointing Sanjay Swani and Michael E. Donovan to serve as directors, as designated by Welsh, Carson, Anderson & Stowe ("WCAS"), the major stockholder of Mobile Storage Group. Both Messrs. Donovan and Swani have served as members of the Mobile Storage Group Board of Directors since August 2006.

In addition, Frederick G. ("Rick") McNamee has been named to Mobile Mini's Board of Directors. He fills the vacancy resulting when Ronald J. Marusiak, a Mobile Mini director since 1996, stepped down to devote more of his time to his duties as the Division President of Micro-Tronics, Inc., a precision machining and tool and die company. Mr. Bunger commented, "We owe a huge debt of gratitude to Ron for all his years of dedicated service to our Company. During his tenure on the Board, Mobile Mini grew from 8 locations to 103 today and our total revenues went from \$42.2 million to just under \$320 million in 2007. We have enjoyed the benefits of Ron's wise counsel, fairness and integrity, and natural entrepreneurial instincts for more than a decade. Therefore we are all delighted that Ron, who has considerable international business experience, has agreed to become a director of our UK business unit. We also welcome Rick to the Board, as well as Sanjay and Mike, and look forward to their insights and guidance, particularly following the business combination with Mobile Storage Group."

Mr. Swani has been a General Partner of WCAS since 2001, having joined the firm as a Vice President in 1999. Earlier, Swani worked at Fox Paine & Company, L.L.C. and Morgan Stanley & Co. Incorporated in their mergers & acquisitions area and in their debt capital markets area. Mr. Swani currently serves on the Board of Directors of ITC DeltaCom, Inc. and several privately-held companies including Ozburn-Hessey Logistics, Inc., Vision Logistics Holding Corp., and Global Knowledge Networks, Inc. Mr. Swani has an undergraduate degree from Princeton University and graduate degrees from the MIT Sloan School of Management and Harvard Law School.

(more)

Mobile Mini News Release Page 3
June 27, 2008

Mr. Donovan joined WCAS in 2001 and is currently a Principal. Prior to joining WCAS, Mr. Donovan worked at Windward Capital Partners and the investment banking division of Merrill Lynch. Mr. Donovan currently serves on the Board of Directors of several privately-held companies, including Ozburn-Hessey Logistics, Inc., Vision Logistic Holding Corp. and United Surgical Partners International Inc. Mr. Donovan graduated with a B.A. from Yale University in 1998.

With 31 years experience in electronic manufacturing, Rick McNamee, has since 2000 been co-principal of Quadrus Consulting, a consulting practice primarily focused in the manufacturing operations and strategic planning domains. He has taken on a broad range of assignments utilizing his expertise in contract manufacturing, mergers and acquisitions, and business divestitures accumulated over a long career growing successful enterprises. Mr. Bunger noted, "When Ron decided to step down from our Board of Directors, I immediately thought of Rick, whom I have known and respected for many years. My fellow Board members agreed that we should fill the vacancy with someone who served as a CEO of a public company. We found the right person with Rick, who knows first hand what it takes to build successful companies and at the same time, enhance shareholder value, through a combination of internal growth and acquisitions."

Upon earning his B.S. in Industrial Engineering from Purdue University in 1979, Mr. McNamee joined IBM in Austin, Texas working in manufacturing and engineering, eventually managing the Printed Wiring Board business unit, which generated \$100 million in revenues and employed 1,000. In 1993 he joined Continental Circuits which manufactured complex, multi-layer circuit boards used in sophisticated electronic equipment for leaders in the computer, communications, instrumentation and industrial controls industries. As Chairman, President and CEO, he spearheaded Continental Circuits' initial public offering and later sale to Hadco Corporation, where he served as Hadco's interim chief technology officer, and senior vice president in charge of operations in Malaysia and Phoenix.

Mobile Mini, Inc. is the world's leading provider of portable storage solutions through its total fleet of approximately 285,000 of portable storage units and portable offices with 103 branches in the U.S., United Kingdom, Canada and The Netherlands. Mobile Mini is included on the Russell 2000® and 3000® Indexes and the S&P Small Cap Index.

This news release contains forward-looking statements, particularly statements relating to planned reductions in capital expenditures, the ability to increase operating free cash flow, expected synergies and cost savings anticipated from the acquisition of Mobile Storage Group and expectations that the business combination will be accretive to earnings, which involve risks and uncertainties that could cause actual results to differ materially from those currently anticipated, including the ability of Mobile Mini to successfully integrate Mobile Storage Group. Additional risks and uncertainties are described from time to time in the Company's SEC filings. Forward-looking statements represent the judgment of the Company, as of the date of they are made (in this instance, the date of this release), and Mobile Mini disclaims any intent or obligation to update forward-looking statements.

CONTACT

Larry Trachtenberg, Executive VP & Chief Financial Officer
Mobile Mini, Inc.
(480) 894-6311
www.mobilemini.com

-OR-

INVESTOR RELATIONS COUNSEL:

The Equity Group Inc. Linda Latman (212) 836-9609 Lena Cati (212) 836-9611 www.theequity.group.com

#####